THIS PROSPECTUS IS A PROSPECTUS FOR DISTRIBUTION IN AND FROM SWITZERLAND ONLY AND DOES NOT CONSTITUTE A PROSPECTUS FOR THE PURPOSES OF IRISH APPLICABLE LAW

THIS EXTRACT PROSPECTUS REFERS TO THE OFFERING OF THE FUNDS LISTED IN THE TABLE OF CONTENTS. OTHER FUNDS ARE AVAILABLE IN THE COMPANY WHICH ARE NOT CURRENTLY OFFERED FOR SALE TO NON-QUALIFIED INVESTORS IN SWITZERLAND.

UBS (Irl) Fund Solutions plc

(an umbrella fund with segregated liability between sub-funds)

A company incorporated with limited liability as an open-ended investment company with variable capital under the laws of Ireland with registered number 484724

PROSPECTUS FOR SWITZERLAND

This Prospectus is dated 04 June 2025

The Directors of UBS (IrI) Fund Solutions PIc whose names appear in this Prospectus accept responsibility for the information contained in this Prospectus. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure such is the case) the information contained in this document is in accordance with the facts and does not omit anything likely to affect the import of such information. This Prospectus should be read in conjunction with the Supplements dealing with the relevant Fund(s).

1 Important Information

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS PROSPECTUS YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Authorisation

The authorisation of UBS (Irl) Fund Solutions plc (the "Company") by the Central Bank of Ireland (the "Central Bank") shall not constitute a warranty as to the performance of the Company and the Central Bank shall not be liable for the performance or default of the Company.

The value of and income from Shares in the Company may go up or down and you may not get back the amount you have invested in the Company.

Information applicable to the Company generally is contained in this Prospectus. Shares constituting each Fund offered by the Company are described in the Supplements to this Prospectus.

The Company is an umbrella investment company with segregated liability between sub-funds (hereinafter referred to as "Funds") and with variable capital incorporated on 21 May 2010 and is authorised in Ireland as an undertaking for collective investment in transferable securities pursuant to the UCITS Regulations. Such authorisation is not an endorsement or guarantee of the Company or any Fund by the Central Bank, nor is the Central Bank responsible for the contents of this Prospectus.

General

The Company is structured as an umbrella investment company, in that different Funds may be established from time to time by the Directors with the prior approval of the Central Bank.

The particulars of each Fund will be set out in a separate Supplement. A separate Supplement will be issued at the time of establishment of each Fund. Each Supplement forms part of and should be read in the context of and in conjunction with this Prospectus.

Any such Supplement shall list all of the existing Funds. Shares of more than one Class may be issued in relation to a Fund. Information contained within the Supplements is selective and should be read in conjunction with this Prospectus. The creation of any new Classes of Shares must be notified to and cleared in advance by the Central Bank. On the introduction of any new Class of Shares, the Company will prepare and the Directors will issue documentation setting out the relevant details of each such Class of Shares. A separate portfolio of assets shall be maintained for each Fund and shall be invested in accordance with the investment objective applicable to such Fund.

Shares in any of the Funds may be subscribed for or redeemed in cash. Other than in respect of Unlisted Funds, Shares may also be bought or sold on the secondary market (as described below).

The Company may decline any application for Shares in whole or in part without assigning any reason therefor and will not accept an initial subscription for Shares of any amount which is less than the Minimum Subscription Amount or the Minimum Initial Investment Amount as set forth in the Supplement for the relevant Fund, unless the Minimum Subscription Amount or the Minimum Initial Investment Amount (as appropriate) is waived by the Directors.

After the initial issue, Shares will be issued and redeemed at the Net Asset Value per Share plus or minus duties and charges (as the case may be) including any Exchange Charge and any Primary Market Transaction Costs specified in the relevant Supplement. The Net Asset Value of the Shares of each Class and the Issue and Redemption Prices will be calculated in accordance with the provisions summarised under the heading "Issue and Redemption Prices/Calculation of Net Asset Value" in this Prospectus.

Details of Dealing Days in respect of each Fund are set out in the relevant Supplement.

In deciding whether to invest in the Company, investors should rely on information in this Prospectus and the relevant Fund's most recent annual and/or semi-annual reports. Prospective investors may also wish to consider the KIID for the relevant Class prior to subscribing for Shares in that Class in order to assist them in making an informed investment decision. Furthermore, before investing in the Company, you should consider the risks involved in such investment. Please see "**Risk Factors**" below and where applicable to each Fund in the relevant Supplement.

Because the Prospectus and KIID may be updated from time to time, investors should make sure they have the most recent versions.

The decision to invest in any Fund, and if so how much, should be based on a realistic analysis of the investor's own financial circumstances and tolerance for investment risk.

As with any investment, future performance may differ from past performance, and Shareholders could lose money. There is no guarantee that any Fund will meet its objectives or achieve any particular level of future performance. These are investments, not bank deposits.

No Fund in this Company is intended as a complete investment plan, nor are all Funds appropriate for all investors. Before investing in a Fund, each prospective investor should read the Prospectus and should understand the risks, costs and terms of investment in that Fund. In particular, investors should read and consider the section entitled "Risk Factors" before investing in the Company.

All Shareholders are entitled to the benefit of, are bound by and are deemed to have notice of, the provisions of the Articles, copies of which are available upon request.

Defined terms used in this Prospectus shall have the meaning attributed to them in Appendix I.

Description of Classes

Shares may be issued with one or several of the following characteristics:

dis	Shares in Classes with "dis" shall distribute their net income (e.g. from any interest or dividends received) at regular intervals.
acc	Shares in Classes with "acc" in their name shall automatically reinvest earnings on the corresponding Fund assets back into the corresponding Fund.
UKdis	For Classes with "UKdis" in their name, the Company intends to distribute a sum which corresponds to 100% of the reportable income within the meaning of the UK reporting fund rules where the Classes are subject to the reporting fund rules. Dividend payments may be made more frequently than semi-annually. The Company does not intend to prepare tax reporting in other countries in respect of these Classes which are intended for investors who are subject to UK taxation on their investment in the Class.
h	Shares in Classes with "h" in their name shall engage in currency hedging as described in section 4.14 below.

Listing and Secondary Market Trading

The Shares of each Fund (other than the Unlisted Funds) may be listed on one or more Relevant Stock Exchanges and if so listed shall be fully transferable by Shareholders. In such circumstances, it is envisaged that Shares will be bought and sold by retail and institutional investors and professional traders in the secondary market like the ordinary shares of a listed company. However, the Company cannot guarantee that a liquid secondary market will develop in relation to the Shares of any such Funds. Shares in the relevant Fund which are purchased on the secondary market (as further described in section 10 of the Prospectus) cannot usually be redeemed directly from the Company. Investors normally sell their Shares on the secondary market with the assistance of an intermediary (e.g. a stockbroker or other investment broker) and may incur fees for investing in this manner. In addition, please note that such investors may pay more than the current Net Asset Value per Share when purchasing Shares on the secondary market and may receive less than the current Net Asset Value when selling their shareholding. Where the value of the Shares quoted on the secondary market significantly differs or varies from the current Net Asset Value per Share, investors who hold their shares through a secondary market will be permitted to redeem their shareholding directly from the Company. For example, this may apply in cases of market disruption such as the absence of a market maker. In such situations, information will be communicated to the regulated market indicating that the Company is open for direct redemptions. Such secondary market investors should refer to section 9.3 of the Prospectus for details on how to process such redemption requests. Only the actual costs of providing this facility (i.e. those costs associated with liquidating any underlying positions) will be charged to such secondary market investors and in any event, the fees in respect of any such redemptions shall not be excessive.

The preceding paragraph is subject to the requirements of the Central Bank and any applicable laws.

Selling Restrictions

Distribution of this Prospectus is not authorised in any jurisdiction unless it is accompanied by a copy of the then latest annual report and audited accounts of the Company and, if published after such report, a copy of the then latest semi-annual report and unaudited accounts. Such reports and this Prospectus together form the prospectus for the issue of Shares in the Company.

This Prospectus may not be used for the purpose of an offer or solicitation in any jurisdiction or in any circumstances in which such offer or solicitation is unlawful or not authorised.

The Shares have not been approved by the United States Securities and Exchange Commission, any state securities commission or other U.S. regulatory authority, nor have any of the foregoing authorities passed upon or endorsed the merits of this offering or the accuracy or adequacy of these offering materials. Any representation to the contrary is unlawful. The Directors and/or the Manager may impose restrictions on the holding of Shares by (and consequently to compulsorily redeem Shares held by) any person who appears to be in breach of any law or requirement of any country or governmental authority by virtue of which such person is not qualified to hold such Shares; or by any person or persons in circumstances (whether directly or indirectly affecting such person or persons, and whether taken alone or in conjunction with any other persons, connected or not, or any other circumstances appearing to the Directors to be relevant) which, in the opinion of the Directors, might result in the Company or the relevant Fund incurring any liability to taxation or suffering any other pecuniary, regulatory, legal or material administrative disadvantages which the Company or the relevant Fund might not otherwise have incurred, suffered or breached; or any individual under the age of 18 (or such other age as the Directors think fit) or of unsound mind. The Articles also permit the Directors where necessary to redeem and cancel Shares (including a fraction of portion thereof) held by a person who is, or is deemed to be, or is acting on behalf of, an Irish Resident on the occurrence of a chargeable event for Irish taxation purposes as described under "Taxation" below (together "Prohibited Persons").

Potential subscribers and purchasers of Shares should consult a stockbroker, bank manager, solicitor, accountant or other financial advisor and inform themselves as to (a) the possible tax consequences, (b) the legal requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which they might encounter under the laws of the countries of their incorporation, citizenship, residence or domicile and which might be relevant to the subscription, purchase, holding or disposal of Shares.

This Prospectus may be translated into other languages. Any such translation shall only contain the same information and have the same meanings as this English language document. Where there is any inconsistency between this English language document and the document in another language, this English language document shall prevail except to the extent (but only to the extent) required by the laws of any jurisdiction where the Shares are sold so that in an action based upon disclosure in a document of a language other than English, the language of the document on which such action is based shall prevail, solely for the purposes of such action and to the extent so required.

Any information given, or representations made, by any dealer, salesman or other person not contained in this Prospectus or in any reports and accounts of the Company forming part hereof must be regarded as unauthorised and accordingly must not be relied upon. Neither the delivery of this Prospectus nor the offer, issue or sale of Shares shall under any circumstances constitute a representation that the information contained in this Prospectus is correct as of any time subsequent to the date of this Prospectus. To reflect material changes, this Prospectus may from time to time be updated and Shareholder notifications may from time to time be issued in respect of such changes. Intending subscribers should request from the Manager any such Shareholder notification or later Prospectus and should further enquire of the Administrator or the Manager as to the issue of reports and accounts of the Company.

United States

The Company does not currently issue, offer for sale (directly or indirectly) or permit the transfer of Shares to or for the benefit of U.S. Persons. The Shares may not be assigned, resold, pledged, exchanged or otherwise transferred (each, a "**Transfer**") to or for the benefit of a U.S. Person (as defined in this Prospectus).

The Shares have not been, and will not be, registered under the US Securities Act of 1933, as amended (the "**1933 Act**"), or qualified under any applicable state statutes and may not be offered, sold or transferred in the United States (including its territories and possessions) or to or for the benefit of, directly or indirectly, any U.S. Person (as that term is defined herein). The Company has not been, and will not be, registered under the U.S. Investment Company Act of 1940, as amended (the "**1940 Act**"), and investors will not be entitled to the benefits of such registration.

The Directors will reject any proposed purchase of Shares by, or Transfer of Shares to, a U.S. Person or for the benefit of a U.S. Person and will require the transfer or redemption of any Shares held by, or for the benefit of, any person who is a U.S. Person or is holding Shares for the account of, or for the benefit of, a U.S. Person.

Any Shareholder who becomes a U.S. Person after their initial investment in the Company may have their Shares compulsorily redeemed in accordance with the provisions of this Prospectus and will be prohibited from purchasing additional Shares of any Fund of the Company, or exchanging Shares of one Fund for Shares of another Fund, for so long as such Shareholder is a U.S. Person.

Each Shareholder is required to notify the Company immediately of any change in their status as a U.S. Person.

Investors may not enter into or issue any derivative or structured product (each, a "Structured Product"), the return on which is based, directly or indirectly, in whole or in part, on the value of the

Company, any Fund of the Company or any Shares without the prior consent of the Company, which may be granted, conditioned or withheld in the sole discretion of the Directors.

Notwithstanding the foregoing, the Directors shall <u>not</u> consent to the creation or issuance of any Structured Product that is entered into or issued with or to any entity (each, a "**Structured Product Investor**"), such that (i) the Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor on behalf of any underlying beneficial owner, such underlying beneficial owner) would be a beneficial owner of Shares for purposes of the 1940 Act unless such Structured Product Investor (and, where applicable, underlying beneficial owner) is a U.S. Person; and (ii) the sale of the Structured Product or the purchase of the Structured Product by any Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor (and, where the Structured Product is held by the Structured Product Investor on behalf of any underlying beneficial owner, such underlying beneficial owner) would result in any violation by the Company, any Fund of the Company and/or any investment adviser to the Company or any Fund of the Company of any laws or regulations in any jurisdiction.

Although certain affiliates of the Manager or the Investment Manager may be registered under the United States Investment Advisers Act of 1940, as amended (the **"Advisers Act**"), because the funds are non-US investment entities, the funds' investors will not have the benefit of the substantive provisions of US law, including the Advisers Act.

Australia

UBS Fund Management (Ireland) Limited is exempt from the requirement to hold an Australian Financial Services Licence under the Corporations Act 2001 (Cth.) (the "Act") in respect of financial services provided to Australian wholesale clients (within the meaning of section 761G of the Act).

MiFID II Product Governance Rules – UCITS as non-complex financial instruments

Article 25 of MiFID II sets out requirements in relation to the assessment of suitability and appropriateness of financial instruments for clients. Article 25(4) contains rules relating to the selling of financial instruments by a MiFID-authorised firm to clients in an execution only manner. Provided the financial instruments are comprised from the list contained in Article 25(4)(a) (referred to broadly as non-complex financial instruments for these purposes), a MiFID-authorised firm selling the instruments will not be required to also conduct what is referred to as an "appropriateness test" on its clients. An appropriateness test would involve requesting information on the client's knowledge and experience on the type of investment offered and, on this basis, assessing whether the investment is appropriate for the client. If the financial instruments fall outside the list contained in Article 25(4)(a) (i.e. are categorised as complex financial instruments), the MiFID-authorised firm selling the instruments will be required to also conduct an appropriateness test on its clients.

UCITS (other than structured UCITS) are specifically referenced in the list in Article 25(4)(a). Accordingly, each Fund is deemed to be a non-complex financial instrument for these purposes.

Potential for Capital Reduction

Where provided for in the relevant Supplement, (i) dividends may be declared out of the capital of the relevant Fund; and/or (ii) fees and expenses may be paid out of the capital of the relevant Fund, in each case in order to preserve cash flow to Shareholders. In any such cases, there is a greater risk that capital may be eroded and distribution will be achieved/fees will be paid in a manner that foregoes the potential for future capital growth of your investment. This cycle may continue until all capital is depleted.

Distributions out of capital may have different tax consequences to distributions of income and it is recommended that you seek appropriate advice in this regard.

Redemption Fee

Where provided for in the relevant Supplement, the Directors or the Manager may levy a Redemption Fee of up to 3 % of the Net Asset Value per Share. Details of any such charge with respect to one or more Funds will be set out in the relevant Supplement.

The difference at any one time between the subscription price (to which may be added a Subscription Fee) and the Redemption Price (from which may be deducted a Redemption Fee) means that an investment should be viewed as medium to long-term.

Table of Contents

1	Important Information	2	
2	Directory	9	
3	Management & Administration	11	
4	Investment Objective & Policies		
5	Investment Restrictions		
6	Risk Factors		
7	Conflicts of Interest	60	
8	Issue and Redemption Prices/Calculation and Publication of Net Asset Value/Valuation	on of	
	Assets		
9	Share Dealing	67	
10	Secondary Market	78	
11	Fees & Expenses	82	
12	Share Information	86	
13	Summary of Articles	89	
14	Miscellaneous	96	
Appendix I – Definitions		103	
Appen	Appendix II - Markets		
Appen	Appendix III – Taxation		
	Appendix IV - Current List of the Depositary's Sub-Delegates		
UBS (UBS CMCI COMPOSITE SF UCITS ETF		
	BBG COMMODITY INDEX SF UCITS ETF		
UBS S	UBS S&P 500 SF UCITS ETF		
	MSCI USA SF UCITS ETF		
UBS N	MSCI EM SF UCITS ETF	203	
	MSCI AC ASIA EX JAPAN SF UCITS ETF		
	MSCI ACWI SF UCITS ETF		
	CMCI EX-AGRICULTURE SF UCITS ETF		
UBS E	3BG Commodity CMCI SF UCITS ETF	277	
UBS (CMCI Commodity Carry SF UCITS ETF	298	
	MSCI China A SF UCITS ETF		
	Euro Equity Defensive Covered Call SF UCITS ETF		
UBS E	Euro Equity Defensive Put Write SF UCITS ETF	361	
UBS L	JBS US Equity Defensive Covered Call SF UCITS ETF		
UBS L	JBS US Equity Defensive Put Write SF UCITS ETF		
	CMCI Future Commodity SF UCITS ETF		
	MSCI USA SF INDEX FUND		
	JBS EUR Overnight Rate SF UCITS ETF Supplement to the Prospectus4		
INFOF	NFORMATION FOR INVESTORS IN SWITZERLAND DATED 04 JUNE 20254		

2 Directory

COMPANY

UBS (IrI) Fund Solutions plc 2nd Floor 5 Earlsfort Terrace Dublin 2 Ireland

DIRECTORS

Alan White Markus Goetschi Marie Antoinette (Nina) Petrini Naomi Daly William Kennedy

MANAGER

UBS Fund Management (Ireland) Limited 1st Floor College Park House South Fredrick Street Dublin 2 Ireland

INVESTMENT MANAGER

UBS Asset Management (UK) Limited 5 Broadgate London, EC2M 2QS England

DISTRIBUTOR

UBS Asset Management Switzerland AG Bahnhofstrasse 45

> CH-8001 Zurich Switzerland

DEPOSITARY

State Street Custodial Services (Ireland) Limited 78 Sir John Rogerson's Quay Dublin 2 Ireland

ADMINISTRATOR, REGISTRAR AND TRANSFER AGENT

State Street Fund Services (Ireland) Limited 78 Sir John Rogerson's Quay Dublin 2 Ireland

COMPANY SECRETARY

Dechert Secretarial Limited 2nd Floor 5 Earlsfort Terrace Dublin 2 Ireland

AUDITORS

Ernst & Young EY Building Harcourt Centre Harcourt Street Dublin 2 Ireland

IRISH LEGAL ADVISERS TO THE COMPANY

Dechert LLP 2nd Floor 5 Earlsfort Terrace Dublin 2 Ireland

3 Management & Administration

3.1 Directors of the Company

The "Directors" of the Company are described below:

Markus Goetschi

Markus Goetschi (Swiss resident) is head of passive products at UBS Asset Management He is responsible for the strategic development of new and the management of existing ETF and passive fund capabilities at UBS across various European fund platforms in Luxembourg, Ireland and Switzerland. Mr. Goetschi is also responsible for the development and maintenance of global vendor and index provider relationships and he chairs the ETF and indexed investments working group at SFAMA.

Mr. Goetschi started his professional careers At Zurich Cantonal Bank as a business organiser in the area of electronic sales. On completing his studies, Mr. Goetschi moved to SIX Swiss Exchange as an index analyst in 2004. From 2007 to 2009, he led the index unit at SIX before joining STOXX as part of the transfer of the operative index business, where he focused on the development of new global index concepts. Since 2011, Mr. Goetschi has been with UBS Asset Management in the passive product business area.

Mr. Goetschi completed his studies at the University of Zurich in 2004 and graduated with a Master of Science in Computer Science (commercial information technology). In 2008, he also completed further training as a financial analyst with the CIIA diploma.

Alan White

Mr. White has over 20 years' experience in banking and asset management and currently holds the position of designated person for distribution of UBS Fund Management (Ireland) Limited. Mr. White has worked with UBS since July 2002; from January 2004 to September 2006 he held the position of manager of Alternative Investment Strategies Trade Support and from September 2006 to December 2006 he worked as an Equity Derivatives Structurer in UBS Investment Bank. Mr. White has worked with the Manager since January 2007 where he has held various positions including head of product development and from 2011 to 2022 acted as an executive director of the management company board. He holds a degree in Finance and a Diploma in Computer Science from University College Cork. Mr. White is a director of several investment funds managed by the Manager.

Marie Antoinette (Nina) Petrini

Nina joined UBS Asset Management at the end of 2019 as the Head of ETF and Index Fund sales for Spain and Portugal. She was later also assigned the coverage for the ETF distribution effort for Latin America. Her appointment has thus far led to a successful relaunch of the brand in these regions, with strengthened business ties and activities.

Ms. Petrini started her career in 2001 in Lehman Brothers as a US Equity Sales in the Investment Banking Division, where she developed her business skills until 2008 as a Vice President in the sales team. She later transitioned through Nomura Securities in 2009 in the same role, but finally joined Barclays Capital as a Director to continue within her Global Equity Sales and Distribution coverage for over 9 years. The entity was new to the local market and Ms. Petrini was critical in the establishment and development of institutional client business which eventually led to strong market share.

Naomi Daly

Naomi Daly (Irish resident) serves as a full time specialist independent director to a number of Irish domiciled investment funds. Naomi Daly has over 24 years of experience in asset management, including prime brokerage, hedge fund consulting, distribution, managed account platforms and governance. Naomi serves as a Chair and independent non-executive Director to Irish regulated entities, having previously been CEO, executive director and head of business development at MPMF Fund Management. Prior to MPMF, she spent a decade at Goldman Sachs in London as an Executive Director. Naomi started as a business analyst at AIB.

William Kennedy

William Kennedy (Irish Resident) is an international finance professional with over 25 years' experience in financial services and in the wider UBS group. Mr. Kennedy joined UBS investment bank in 1993 and held a variety of positions before being appointed as head of European Derivative Sales and Structuring in 2005. He acted as Head of European Equities from 2008 to 2010 before moving to UBS wealth management division in Zurich as Head of Investment Products and Services in 2011. In 2015, he transferred to the UBS asset management division in Switzerland and held a number of positions, including Head of Distribution and latterly, Head of Products and Platforms. Mr. Kennedy holds a Bachelor of Commerce from the University of Galway and a Masters in Business Studies (MBS) from the U.C.D. Michael Smurfit Graduate School of Business. For the purposes of this Prospectus, the address of all the Directors is the registered office of the Company.

The Company has granted indemnities to the Directors in respect of any loss or damages that they may suffer, save where this results from a Director's negligence, default, breach of duty or breach of trust in relation to the Company.

The Company has delegated the day to day management and running of the Company in accordance with policies approved by the Directors to the Manager. Consequently, all Directors of the Company are non-executive.

3.2 Promoter

The "**Promoter**" of the Company is UBS AG, London Branch.

The Promoter is registered as a limited company in England and Wales and is authorised and regulated by the UK Financial Conduct Authority ("**FCA**").

3.3 Manager

The Company has appointed UBS Fund Management (Ireland) Limited (the "**Manager**") to act as manager to the Company and each Fund with power to delegate one or more of its functions subject to the overall supervision and control of the Company. The Manager is a private limited company. It was incorporated in Ireland on 1 December 2005 and is a wholly owned subsidiary of UBS AG. The authorised share capital of the Manager is \in 10,000,000. The Manager currently acts as manager or alternative investment fund manager to four other collective investment schemes. The Manager is authorised and regulated by the Central Bank.

The directors of the Manager are:

Ian Fitzpatrick

Ian Fitzpatrck is an Executive Director with UBS Asset Management, based in Dublin, and is the CEO of UBS Fund Management (Ireland) Limited. Prior to this he managed a team

of global product managers that covered the hedge fund & private market products on the UBS Asset Management alternative product shelf. He has also worked with UBS Wealth Management in Dublin and UBS Investment Bank in Sydney, Australia. Mr Fitzpatrick trained as a Chartered Accountant in the audit and advisory division of Deloitte, Dublin. He holds a Bachelor of Accounting & Finance and a Master of Accounting, both from Dublin City University, and completed the Chartered Tax Adviser qualification with the Irish Tax Institute.

Deirdre Gormley

Deirdre Gormley (Irish) is an independent investment fund director with over 30 years' experience in the asset management and investment funds industry, having held senior executive and board positions in large international organizations. In her previous executive roles Ms Gormley was responsible for a wide range of investment management, business development, governance and regulatory activities. She was involved in product management for Irish, Luxembourg and Dutch domiciled investment products. Ms Gormley is the former CEO/Head of Management Company for Northern Trust Asset Management in Dublin Ireland. In this role she was responsible for the management of UCITS and IPM business which included Money Market, Equity, Fixed Income and ETF products. She was responsible for the Northern Trust Fund Managers Ireland branches in Europe and the oversight of delegated service providers. Prior to this role Deirdre spent 12 years with Pioneer Investment Limited (now Amundi Ireland Limited) as Head of Product and Marketing Services. Prior to joining Pioneer Deirdre held various senior management posts with JPMorgan both in Dublin and New York covering a range of operational and client relationship roles. Deirdre has a Bachelor of Science degree in Finance from Marist College in Poughkeepsie, New York.

Valerie Bernard

Valérie Bernard is conducting officer and deputy CEO of UBS Fund Management (Luxembourg) S.A ("UBS FML"), a UBS management company domiciled in Luxembourg, managing UBS investment funds and white labelling business. She has joint responsibility for setting the firm's strategic direction, as well as leading day to day management and governance of the business. Ms Bernard joined UBS in January 1997 and has over 31 years' experience in the financial services industry. Before joining UBS FML in June 2013, Ms Bernard had management responsibilities within UBS Fund Services Luxembourg where she acted as Product Control Head and subsequently as Head NAV Administration. Prior to joining UBS, Ms Bernard began her professional career as an external auditor at Deloitte Luxembourg. Mrs Bernard holds a Master in Business Administration from the University of Liege (B) and a post-graduate diploma in European business from the Northampton University.

Hannah Vinci

Ms. Vinci is Head of Platform Solutions and Credit Structuring in EMEA at UBS, with responsibility for identifying and delivering credit based and other investment solutions for a wide variety of institutional clients, including banks, asset managers, insurance companies and pension funds, family offices and private wealth organisations. Ms.Vinci joined UBS in September 2014 and has over 19 years' experience obtained between working in the legal and financial services industries. Prior to joining UBS, Ms. Vinci worked at Bank of America Merrill Lynch in the Credit Trading and Correlation Trading/Structuring businesses, and previous to that, she qualified as a solicitor into the Derivatives & Structured Finance department at Allen & Overy LLP. Ms. Vinci holds a BA in Philosophy from Birmingham University, and a Postgraduate Diploma in Law from College of Law, London. She completed her Legal Practice Course at Oxford Law School.

Tom Payne

Mr Payne is a Managing Director at UBS AG. He is Head of Strategic Projects and Financial Resource Management for UBS's Non-Core and Legacy division. Prior to this, he was EMEA Head of Global Markets Structuring with responsibility for developing and structuring products linked to FX, Rates, Credit, Equities, Commodities and Funds, including UCITS funds managed by FMI, for a wide variety of institutional clients. Mr. Payne joined UBS AG in October 2005 and has over 22 years' experience in the financial services industry. Prior to joining UBS AG, Mr. Payne worked at Goldman Sachs International in the Fixed Income, Currencies and Commodities Structuring division. Mr. Payne holds a Master of Engineering (Honours) in Engineering, Economics and Management from the University of Oxford.

Naomi Daly is also a director of the Manager. The secretary of the Manager is Dechert LLP.

3.4 Investment Manager

The Investment Manager of the Company is UBS Asset Management (UK) Limited, unless otherwise specifically stated in the Supplement for the relevant Fund.

The Investment Manager is part of UBS Asset Management, a business group of UBS AG. UBS AG is one of the world's leading financial firms, which also include one of the world's largest wealth managers, a premier investment bank and securities firm, and one of the world's leading asset managers.

The Investment Manager is regulated by the UK Financial Conduct Authority.

Details of any sub-investment manager appointed by the Investment Manager will be provided to Shareholders on request and will be disclosed in the Supplement for the relevant Fund and the periodic reports issued by the Company.

3.5 Distributor

The Manager has appointed UBS Asset Management Switzerland AG the "**Distributor**" to act as distributor of shares of the Company and each Fund. The Distributor is registered as a limited company in Switzerland ("**Aktiengesellschaft**") and is authorised and regulated by the Swiss Financial Market Authorities ("**FINMA**").

3.6 Depositary

The Company has appointed State Street Custodial Services (Ireland) Limited as depositary of its assets pursuant to a depositary agreement dated 12 October 2016, as amended from time to time (the "**Depositary Agreement**"). The Depositary provides safe custody for the Company's assets.

The Depositary is a limited liability company incorporated in Ireland on 23 May 1991 and is, like the Administrator, ultimately owned by the State Street Corporation. Its authorised share capital is £5,000,000 and its issued and paid up share capital is £200,000. The principal activity of the Depositary is to act as depositary of the assets of collective investment schemes. The Depositary is regulated by the Central Bank.

The Depositary shall carry out functions in respect of the Company including but not limited to the following:

(i) the Depositary shall hold in custody all financial instruments capable of being registered or held in a financial instruments account opened in the Depositary's

books and all financial instruments capable of being physically delivered to the Depositary;

- the Depositary shall verify the Company's ownership of all other assets (other than those referred to in (i) above) and maintain and keep up-to-date a record of such assets it is satisfied are owned by the Company;
- (iii) the Depositary shall ensure effective and proper monitoring of the Company's cash flows;
- (iv) the Depositary shall be responsible for certain oversight obligations in respect of the Company see "Summary of Oversight Obligations" below.

Duties and functions in relation to (iii) and (iv) above may not be delegated by the Depositary.

Summary of Oversight Obligations:

The Depositary is obliged to ensure, among other things, that:

- the sale, issue, redemption and cancellation of Shares effected on behalf of the Company are carried out in accordance with the Companies Act, the conditions imposed by the Central Bank and the Articles;
- the value of Shares is calculated in accordance with the Companies Act and the Articles;
- in transactions involving the Company's assets, any consideration is remitted to it within time limits which are acceptable market practice in the context of a particular transaction;
- the Company and each Fund's income is applied in accordance with the Companies Act and the Articles;
- the instructions of the Company are carried out unless they conflict with the Companies Act or the Articles; and
- it has enquired into the conduct of the Company in each Accounting Period and reports thereon to the Shareholders. The Depositary's report will be delivered to the Company in good time to enable the Directors to include a copy of the report in the annual report of the Company. The Depositary's report will state whether in the Depositary's opinion each Fund has been managed in that period:

(i) in accordance with the limitations imposed on the investment and borrowing powers of the Fund imposed by the Articles and/or the Central Bank under the powers granted to the Central Bank under the Companies Act; and

(ii) otherwise in accordance with the provisions of the Companies Act and the Articles.

If the Company has not complied with (i) or (ii) above, the Depositary will state why this is the case and will outline the steps that the Depositary has taken to rectify the situation. The duties provided for above may not be delegated by the Depositary to a third party.

In discharging its role, the Depositary shall act honestly, fairly, professionally, independently and in the interests of the Company and the Shareholders.

The Depositary shall act as depositary of the Company's assets and shall be responsible for the oversight of the Company to the extent required by and in accordance with applicable law, rules and regulations. The Depositary shall exercise the supervisory duties in accordance with applicable law, rules and regulations as well as the Depositary Agreement.

The Depositary shall perform its obligations with due skill, care and diligence as determined in accordance with the standards and practices of a professional depositary for hire in the markets or jurisdictions in which the Depositary performs services under the Depositary Agreement.

The Depositary shall be liable to the Company, or to the Shareholders, for all losses suffered by them as a result of the Depositary's negligent or intentional failure to properly fulfil its obligations as set out in the Depositary Agreement and UCITS V. The Depositary shall be liable to the Company and to the Shareholders, for the loss by the Depositary or a duly appointed third party of any financial instruments held in custody (determined in accordance with UCITS V) and shall be responsible for the return of financial instruments or corresponding amount to the Company without undue delay. The Depositary Agreement contains indemnities in favour of the Depositary for certain losses incurred but excluding circumstances where the Depositary is liable for the losses incurred.

The Depositary Agreement shall continue in force unless and until terminated by either party giving not less than 90 days' prior written notice to the other, although termination may be immediate in certain circumstances, such as the insolvency of the Depositary. Upon an (envisaged) removal or resignation of the Depositary, the Company shall with due observance of the applicable requirements of the Central Bank, appoint a successor Depositary. The Depositary may not be replaced without the approval of the Central Bank.

The Depositary Agreement shall be governed by the laws of Ireland and the courts of Ireland shall have non-exclusive jurisdiction to hear any disputes or claims arising out of or in connection with the Depositary Agreement.

3.7 Administrator

The Manager has appointed State Street Fund Services (Ireland) Limited (the "Administrator") to act as administrator, registrar and transfer agent of the Company with responsibility for performing the day to day administration of the Company, including the calculation of the Net Asset Value and the Net Asset Value per Share of each Fund.

The Administrator is a limited liability company. It was incorporated in Ireland on 23 March 1992 and is ultimately a wholly owned subsidiary of the State Street Corporation. The authorised share capital of the Administrator is Stg£5,000,000 with an issued and paid-up share capital of Stg£351,000.

State Street Corporation is a leading world-wide specialist in providing sophisticated global investors with investment servicing and investment management. State Street Corporation is headquartered in Boston, Massachusetts, U.S.A., and trades on the New York Stock Exchange under the symbol "STT". The Administration Agreement provides that the appointment of the Administrator will continue in force unless and until terminated by either party giving to the other not less than ninety days' written notice although in certain circumstances (e.g., the insolvency of either party, unremedied breach after notice etc.) the Agreement may be terminated forthwith by notice in writing by either party to the other. The Administration Agreement contains indemnities in favour of the Administrator other than matters arising by reason of its failure to exercise reasonable care in the performance of its duties under the Administrator or any of its agents appointed for the performance of the Administrator's duties under the Administration Agreement and other than tax on the Administrator's overall income or profits. Losses under the indemnity do not include any

consequential losses, any loss of profit, any loss of goodwill, any loss of business opportunity or any indirect loss suffered by the Administrator.

3.8 Paying Agents/ Representatives/ Distributors

Local laws or regulations in certain EEA jurisdictions may require that the Manager appoints a local Paying Agent and/or other local representatives. The role of the Paying Agent may entail, for example maintaining accounts through which subscription and redemption proceeds and dividends are paid.

Investors who choose or are obliged under local regulations to pay/receive subscription/redemption monies via the intermediary entity rather than directly to the Administrator or the Company bear a credit risk against that entity with respect to a) subscription monies prior to the transmission of such monies to the Depositary for the account of the Company and b) redemption monies payable by such intermediate entity to the relevant investor. The appointment of a Paying Agent (including a summary of the agreement appointing such Paying Agent) may be detailed in a Country Supplement.

Fees and expenses of Paying Agents and/or other local representatives, which will be at normal commercial rates, will be borne by the relevant Fund(s).

Investors who do not themselves wish to be registered as Shareholders may use the services of a nominee. Where Shares are held through a nominee, those underlying investors who avail of the services of such nominee may be obliged to pay a fee directly to it in relation to the subscription, redemption or conversion of Shares, details of which will be provided by the nominee. Regard must be had to the anti-money laundering requirements set out in the section entitled "Share Dealing".

4 Investment Objective & Policies

4.1 Investment Objective and Policies

The Articles provide that the investment objective and policies for each Fund will be formulated by the Directors at the time of the creation of the Fund and will be set out in the Supplement for the relevant Fund.

The investment objective of each Fund will be to provide Shareholders with a return (either at the relevant Scheduled Maturity Date or on each Dealing Day) linked to a Reference Index or Reference Asset.

Pending investment of the proceeds of a placing or offer of Shares or where market or other factors so warrant, a Fund's assets may be invested in Money Market Instruments and in cash deposits.

The return that the Shareholder will receive will be dependent on the performance of the relevant transferable securities, other eligible assets and financial derivative instruments including OTC derivatives owned by the Fund, and the performance of any efficient portfolio management technique or other techniques used to link the transferable securities, other eligible assets, financial derivative instruments and OTC derivatives to the Reference Index or Reference Asset (where appropriate). Thus the return Shareholders receive may not wholly correspond to the performance of the Reference Index or Reference Asset. There is no assurance that the investment objective of any Fund whose performance is linked to the Reference Index or Reference.

While it is not the Company's intention that any Fund be leveraged (unless disclosed in the Supplement for the relevant Fund), any leverage resulting from the use of FDIs will be in accordance with the requirements of the Central Bank.

In certain circumstances, the Company may receive interest on accounts maintained to receive subscription proceeds (prior to such proceeds being invested) and/or interest on accounts where cash for trades that have not settled is held. Any such interest accrued is expected to be negligible and all such negligible amounts will be utilised to reimburse service providers to the Company (including, for the avoidance of doubt, the Investment Manager) for any costs and/or expenses which are attributable to the Company or a Fund but which have been paid by such service providers.

Any change in the investment objective or any material change to the investment policies of a Fund may only be made with the approval of an ordinary resolution of the Shareholders of the Fund. Subject and without prejudice to the preceding sentence of this paragraph, in the event of a change of investment objective and/or policies of a Fund, a reasonable notification period must be given to each Shareholder of the Fund to enable a Shareholder to have its Shares redeemed prior to the implementation of such change.

Further details of the investment objective and policies for each Fund are set out in the relevant Supplement.

4.2 Additional information relating to the Reference Index or Reference Asset

A number of circumstances may arise in respect of the replication of or delivery of the performance of a Reference Index or Reference Asset by a Fund. These include, but are not limited to, the following:

- (a) Each Fund is subject to the UCITS Regulations which include, inter alia, certain restrictions on the proportions of that Fund's Net Asset Value which may be held in individual securities. Depending on the concentration of the Reference Index or Reference Asset a Fund may also hold synthetic securities which are correlated to, or the return on which is based on, securities which form part of the Reference Index or Reference Asset within the limits set out in the Prospectus.
- (b) The constituent securities of the Reference Index or Reference Asset change from time to time. The Investment Manager may adopt a variety of strategies when managing a Fund to bring it in line with the changed Reference Index or Reference Asset. For example, where a security which forms part of the Reference Index or Reference Asset is not available or a market for such security does not exist, a Fund may instead hold depository receipts relating to such securities (e.g. ADRs and GDRs and other equity-related securities).
- (c) From time to time securities in the Reference Index or Reference Asset may be subject to corporate actions. The Investment Manager has discretion to manage these events as appropriate.
- (d) Securities included in the Reference Index or Reference Asset may, from time to time, become illiquid or otherwise unobtainable at fair value. In these circumstances, the Investment Manager may use a number of techniques, including purchasing securities whose returns, individually or collectively, are seen to be well-correlated to the constituents of the Reference Index or Reference Asset.
- (e) Where the Fund uses one or more FDIs to gain exposure to the Reference Index or Reference Asset, the relevant Calculation Agent will have the discretion to make such changes to the FDI(s) to preserve the economic effect of the transaction for both the Approved Counterparty and the Fund in accordance with the detailed provisions of such FDI.
- (f) Securities included in the Reference Index may, from time to time, be excluded from the Reference Index composition if certain thresholds as set forth by the Index Provider are exceeded. For more details on exclusion parameters including sectors and revenue thresholds, please refer to the section "Index methodology" in each relevant Supplement. The Investment Manager may adopt a variety of strategies when managing a Fund to bring it in line with the changed Reference Index or Reference Asset.
- (g) Where the Fund uses one or more FDIs to gain exposure to the Reference Index or Reference Asset, the Fund promotes the same environmental characteristics as those of the Reference Index or Reference Asset, as relevant.

The Index Provider rebalances the Reference Index on a regular basis, as described in the Relevant Supplement, and the Investment Manager tracks the Reference Index or Reference Asset in line with the limits set out in the investment policy of the relevant Fund as set out in the Relevant Supplement.

Unless otherwise set out in the Relevant Supplement, the investment objective of Funds complying with Article 8 or Article 9 of SFDR (as disclosed in the Relevant Supplement) is to track the Fund's Reference Index or Reference Asset price and

income performance and its characteristics, including ESG characteristics, in accordance with the provisions of the Prospectus.

4.3 Change or Substitution of Reference Index or Reference Asset

The Directors may decide, if they consider it to be in accordance with the Investment Restrictions and the UCITS Regulations and in the interest of the Company or any relevant Fund, to change or substitute the existing Reference Index or Reference Asset of a Fund with one or more Reference Indices or Reference Assets.

The Board of Directors may, for instance, decide to substitute such a Reference Index or Reference Asset in the following circumstances:

- (a) the transferable securities, swaps or other techniques or instruments described under "Investment Restrictions" which are necessary for the implementation of the relevant Fund's investment objective cease to be available in a manner which is regarded as acceptable by the Directors;
- (b) the quality, accuracy and availability of data of a particular Reference Index or Reference Asset has deteriorated;
- (c) the components of the Reference Index or Reference Asset would cause the Fund (if it were to follow the Reference Index or Reference Asset closely) to be in breach of the limits set out under "Investment Restrictions" and/or materially affect the taxation or fiscal treatment of the Company or any of its Shareholders;
- (d) the particular Reference Index or Reference Asset ceases to exist or, in the determination of the Directors, there is (or there is expected to be) a material change in the formula for or the method of calculating a component of the Reference Index or Reference Asset or there is (or there is expected to be) a material modification of a component of the Reference Index or Reference Asset;
- (e) the Directors become aware that there is limited liquidity in one or more component securities of the Reference Index or Reference Asset, or it becomes impractical to invest in the components of the Reference Index or Reference Asset;
- (f) the Index Provider increases its licence fees to a level which the Directors consider excessive;
- (g) a change of ownership of the relevant Index Provider and/or a change of name of the relevant Reference Index;
- (h) any successor Index Provider is not considered acceptable by the Directors; or
- (i) a new Reference Index or Reference Asset becomes available which supersedes the existing Reference Index or Reference Asset.

The above list is indicative only and cannot be understood as being exhaustive in respect of the ability of the Directors to change the Reference Index or Reference Asset in any other circumstances as they consider appropriate.

Any proposal by the Directors to change a Reference Index or Reference Asset shall be (a) subject to the prior approval of the Shareholders of the relevant Fund by ordinary resolution; or (b) notified to Shareholders where the circumstances set out in (g) above apply. The Prospectus and any of the relevant Supplements will be updated in the case of substitution or change of the existing Reference Index or Reference Asset of a Fund for another Reference Index or Reference Asset, or in the event of (g) above in accordance with the requirements of the Central Bank.

4.4 Disruption Events

Upon the occurrence of a Disruption Event (including an Index Disruption and Adjustment Event as defined below, and without limitation to the Directors' personal powers as further described herein):

- (a) to the extent that the Fund has entered into FDIs, an Approved Counterparty (whether acting as the relevant Calculation Agent or otherwise) may either (i) terminate one or more of the relevant FDIs, or (ii) adjust the terms of the relevant FDIs held by the Fund to account for such event, including adjustment to or substitution of the Reference Index, the calculation of the Reference Index level or the valuation of the FDI (and, provided that the Investment Manager (and where appropriate the Approved Counterparty) considers that it is commercially reasonable to do so, the relevant Fund may continue to operate by using such formula for and method of calculating the Reference Index level last in effect prior to the occurrence of any such event with such adjustments as the Investment Manager may deem necessary for the purpose of continuing the operation of the relevant Fund), and such adjustment(s) may have a positive or negative impact on the Net Asset Value of the relevant Fund; and/or
- (b) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions under the section 8.3 "Suspension of Calculation of Net Asset Value"; and/or
- (c) if the Directors shall have resolved that it is impracticable or inadvisable for a Fund to continue to operate having regard to the market conditions (taking into account such disruption or adjustment events and the best interests of the Shareholders), the Directors shall terminate the Fund.

Any change of a Reference Index shall be subject to (i) the prior approval of the Shareholders of the relevant Fund by ordinary resolution; or (ii) shall be notified to Shareholders in the circumstances set out in section 4.3(g) above.

Certain events ("**Index Disruption and Adjustment Events**") may occur with respect to a Reference Index or the ability of an Approved Counterparty to perform its obligations under one or more derivative contracts. These events include, but are not limited to, those items in section 4.3 above and the events below:

- (i) the Reference Index is deemed to be inaccurate or does not reflect actual market developments;
- (ii) the Reference Index is permanently cancelled by the Index Provider;
- (iii) the Index Provider fails to calculate and announce the Reference Index level;
- the Index Provider makes a material change in the formula for or the method of calculating the Reference Index (other than a modification prescribed in that formula or method to maintain the calculation of the Reference Index level in the event of changes in the constituent components and weightings and other routine events);
- (v) the licence to use and reference the Reference Index by the Company is terminated;

- (vi) it becomes impossible or commercially unreasonable, in the determination of the Investment Manager, for the Approved Counterparty to continue to perform its obligations under the derivatives;
- (vii) to the extent the Fund has entered into FDIs, and / or options or futures contracts on the Reference Index where (a) the costs associated with the Approved Counterparty hedging its liability and obligations under the relevant FDIs and / or options or futures contracts on the Reference Index increase; or (b) the ability of the Approved Counterparty to hedge its liability becomes impaired or commercially unreasonable or impracticable; or
- (viii) if any law shall be passed or change in law is implemented which renders it illegal, impracticable or inadvisable to (a) continue to reference or replicate the relevant Reference Index; or (b) for the Approved Counterparty to continue to perform its obligations under one or more derivative contracts.

The provisions in this section 4.4 apply to Reference Assets in the same way as they apply to a Reference Index.

4.5 Use of Financial Derivative Instruments

Subject to the UCITS Regulations and to the conditions within the limits laid down by the Central Bank, the Company, on behalf of a Fund may invest in FDIs dealt on a regulated market and/or "over the counter" derivatives ("**OTCs**") which will be used for investment purposes, hedging and/or efficient portfolio management purposes.

The FDIs in which a Fund may invest include, without limitation, options and futures transactions, swaps, forward contracts, credit derivatives, spot foreign exchange transactions, caps and floors, contracts for differences or other derivative transactions, further details of which will be set out in the relevant Supplement.

A Fund may invest in OTC derivatives in accordance with the Central Bank Rules and provided that the counterparties to the OTC derivatives are Approved Counterparties

The Company must employ through its service providers a risk management process which enables it to monitor, measure and manage at any time the risks attached to a Fund's FDI positions and their contribution to the overall risk profile of the portfolio of assets of a Fund. It must employ a process for accurate and independent assessment of the value of OTC derivative. The Company must provide the Central Bank with details of its FDI activity and risk assessment methodology and, in accordance with particular requirements of the Central Bank shall specify, for that purpose, the permitted types of FDI, the underlying risks, the guantitative limits and how these will be monitored and enforced and the methods which are chosen in order to estimate the risks associated with transactions in any FDI applicable to a Fund. A Fund may only employ FDIs that have been specified in the risk management process that the Company has submitted to the Central Bank. Where a Fund uses the "value-at-risk" (VaR) approach to measure market risk, the Company will ensure that a Fund's market risk relating to the use of FDIs is in accordance with the Central Bank Regulations and the limits contained therein and that counterparty risk exposure to any OTC derivative transactions shall never exceed the limits permitted under the UCITS Regulations. Relevant information in respect of the use of "value-at risk" or VaR shall be set out in the Supplement for the relevant Fund. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments in respect of the relevant Fund.

For example, such use may include using FDIs such as swaps to exchange the performance of the securities held by a Fund for the performance of the Reference Index or Reference Asset.

4.6 Efficient Portfolio Management

A Fund may also employ techniques and instruments relating to transferable securities and/or other financial instruments in which it invests for efficient portfolio management purposes, a list of which are set out in the relevant Supplement. Use of such techniques and instruments should be in line with the best interests of shareholders and will generally be made for one or more of the following reasons:

- (a) the reduction of risk;
- (b) the reduction of cost; or
- (c) the generation of additional capital or income for the relevant Fund with an appropriate level of risk, taking into account the risk profile of the Fund as described in this Prospectus and the relevant Supplement and the risk diversification limits set out in the Central Bank Regulations.

In addition, the use of such techniques and instruments must be realised in a cost-effective way and must not result in a change to the investment objective of the Fund or add supplementary risks not covered in this Prospectus. Please refer to the section entitled "Risk Factors" for more details. The risks arising from the use of such techniques and instruments shall be adequately captured in the Company's risk management process.

Such techniques and instruments may also include foreign exchange transactions which alter the currency characteristics of assets held by the relevant Fund.

Assets of a Fund may be denominated in a currency other than the Base Currency of the Fund and changes in the exchange rate between the Base Currency and the currency of the asset may lead to a depreciation of the value of the Fund's assets as expressed in the Base Currency. The Investment Manager may seek to mitigate this exchange rate risk by using FDI.

4.7 Mitigation of Counterparty Risk Exposure

When applying the investment restrictions specified in section 5.5 in respect of an OTC derivative, reference must be made to the net counterparty risk exposure as determined pursuant to the UCITS Regulations. In order to reduce its net counterparty risk exposure, the Company may in relation to any of its Funds avail itself of all mitigation techniques such as netting and financial collateral techniques which are or would become authorised by the UCITS Regulations.

In particular, the Company may reduce the overall counterparty risk of each Fund's OTC derivative by causing the counterparty to deliver Collateral (which will comply with the requirements of the Central Bank) to the Depositary (or as otherwise permitted by the Central Bank). Such Collateral will be enforceable by the Company at all times and will be marked to market on a daily basis. The amount of Collateral to be delivered will be at least equal to the value by which the overall exposure limit as determined pursuant to the UCITS Regulations has been exceeded.

The Company may also reduce the overall counterparty risk of the Fund's OTC derivative by resetting the OTC derivative. The effect of resetting the OTC derivative is to reduce the mark to market value of the OTC derivative and, thereby, reduce the net counterparty exposure accordingly.

4.8 Borrowing and Lending Powers

The Company may borrow up to 10% of a Fund's net assets at any time for the account of any Fund and the Depositary may charge the assets of such Fund as security for any such borrowing, provided that such borrowing is only for temporary purposes. In accordance with the provisions of the UCITS Regulations, the Company may charge the assets of a Fund as security for borrowings of that Fund.

Any particular borrowing restrictions for a Fund will appear in the Supplement for the relevant Fund. Without prejudice to the powers of the Fund to invest in transferable securities or enter into Repurchase Transactions, the Company may not lend cash to, or act as guarantor on behalf of, third parties. A Fund may acquire debt securities and securities which are not fully paid.

4.9 Securities Lending, Repurchase Transactions and Securities Financing Transactions

A Fund may use Securities Financing Transactions in accordance with normal market practice and subject to the conditions and limits of the SFTR and the Central Bank Rules where provided for in the relevant Supplement. Such Securities Financing Transactions may only be entered into for the purposes of efficient portfolio management.

Any type of assets that may be held by each Fund in accordance with its investment objective and policies may be subject to such Securities Financing Transactions. Where provided for in the relevant Supplement, the Fund may also use Total Return Swaps for investment purposes.

Subject to each Fund's investment objective and polices, there is no limit on the proportion of assets that may be subject to Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of a Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the relevant Fund. The expected proportion of a Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%, unless otherwise specified in the relevant Fund's Supplement. In any case the most recent semi-annual and annual accounts of each Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Securities lending means transactions by which one party transfers securities to the other party subject to a commitment that the other party will return equivalent securities on a future date or when requested to do so by the party transferring the securities, that transaction being considered as securities lending for the party transferring the securities. The Company has the power to lend on an exclusive basis or otherwise any or all of the securities of each Fund to one or several financial institutions experienced in such transactions, or via a central counterparty or standardised lending system operated by a securities clearing institution, and may do so from time to time in accordance with applicable laws, regulations and market practice for the purposes of efficient portfolio management. In particular, the Company may lend some or all of the securities of each Fund to a member of the UBS AG Group ("UBS") as the sole securities borrowing counterparty and, for that purpose it may enter into an exclusive arrangement with UBS. Any such securities lending arrangements, whether entered into on an exclusive basis or not, will be in compliance with the requirements of the Central Bank, will be with counterparties that are institutions of appropriate financial standing which engage in these types of arrangements and approved by the Manager and will be on normal commercial terms negotiated at arm's length. In accordance with normal market practice, counterparties to such securities lending arrangements will be required to provide Collateral to the Company on behalf of the relevant Fund(s) (or as otherwise permitted by the Central Bank) of a value of at least equal to the market value of any securities loaned to the counterparty. The Collateral will comply with the Collateral Policy as detailed below. Under a securities lending arrangement, the requisite eligible Collateral may be transferred to and held in one or several accounts maintained in the name of the relevant Fund, or the Company with the securities borrowing counterparty, Affiliate(s) of the securities borrowing counterparty or a depositary bank, clearing institution or provider of collateral managements services ("Company Collateral Account"). The title of any Company Collateral Account(s) will make it clear that Collateral credited to such accounts belongs to the Company and such accounts will be separate from any accounts in which the securities borrowing counterparty, its Affiliate(s), depositary bank, clearing institution or provider of collateral managements services holds any of its own assets, and at all times in accordance with the requirements of the Central Bank.

Any Fund that seeks to engage in securities lending shall ensure that it is able at any time to recall any security that has been lent out or terminate any securities lending agreement into which it has entered.

Repurchase agreements are a type of securities lending transaction in which one party sells a security to the other party with a simultaneous agreement to repurchase the security at a fixed future date at a stipulated price reflecting a market rate of interest unrelated to the coupon rate of the securities. The Company has the power to enter into Repurchase Transactions, acting as either buyer or seller, on an exclusive basis or otherwise, with one or more counterparties that are financial institutions experienced in such transactions, which transactions may involve a central counterparty or other settlement system, and may do so from time to time in accordance with applicable laws, regulations and market practice for the purposes of efficient portfolio management. In particular, the Company may enter into such transactions with UBS as sole counterparty and, for that purpose it may enter into an exclusive arrangement with UBS. Any such Repurchase Transactions, whether entered into on an exclusive basis or not, will be in compliance with the requirements of the Central Bank, will be with counterparties that are institutions of appropriate financial standing which engage in these types of arrangements and approved by the Manager and will be on normal commercial terms negotiated at arm's length. In accordance with normal market practice, counterparties to such Repurchase Transactions will be required to provide Collateral to the Company (or as otherwise permitted by the Central Bank) of a value of at least equal to the cash leg of the Repurchase Transaction. The Collateral will comply with the Collateral Policy as detailed below. Under a Repurchase Transaction, if any, the requisite eligible Collateral may be transferred to one or several Company Collateral Accounts. The title of any Company Collateral Account(s) will make it clear that Collateral credited to such accounts belongs to the Company and such accounts will be separate from any accounts in which the counterparty, its Affiliate(s), depositary bank, clearing institution or provider of collateral managements services holds any of its own assets, and at all times in accordance with the requirements of the Central Bank.

A Fund that enters into a repurchase agreement should ensure that it is able at any time to recall any securities subject to the repurchase agreement or to terminate the repurchase agreement into which it has entered. Fixed-term repurchase and reverse repurchase agreements that do not exceed seven days should be considered as arrangements on terms that allow the assets to be recalled at any time by the Fund.

A reverse repurchase agreement is a transaction whereby a Fund purchases securities from a counterparty and simultaneously commits to resell the securities to the counterparty at an agreed upon date and price.

Any Fund that enters into a reverse repurchase agreement should ensure that it is able at any time to recall the full amount of cash or to terminate the reverse repurchase agreement on either an accrued basis or a mark-to-market basis. When the cash is recallable at any time on a mark-to-market basis, the mark-to-market value of the reverse repurchase agreement should be used for the calculation of the Net Asset Value of the Fund.

All the revenues arising from Securities Financing Transactions and any other efficient portfolio management techniques shall be returned to the relevant Fund following the deduction of any direct and indirect operational costs and fees arising. Such direct and

indirect operational costs and fees (which are all fully transparent), which shall not include hidden revenue, shall include fees and expenses payable to repurchase/reverse repurchase agreements counterparties and/or securities lending agents engaged by the Company from time to time . Such fees and expenses of any repurchase/reverse repurchase agreements counterparties and/or securities lending agents engaged by the Company, which will be at normal commercial rates together with VAT, if any, thereon, will be borne by the Company or the Fund in respect of which the relevant party has been engaged. Details of Fund revenues arising and attendant direct and indirect operational costs and fees as well as the identity of any specific repurchase/reverse repurchase agreements counterparties and/ or securities lending agents engaged by the Company from time to time (including whether they are related to the Manager or the Depositary) shall be included in the relevant Fund's semi-annual and annual reports.

While the Company will conduct appropriate due diligence in the selection of counterparties, including consideration of the legal status, country of origin, credit rating and minimum credit rating (where relevant), it is noted that the Central Bank Rules do not prescribe any pre trade eligibility criteria for counterparties to a Fund's Securities Financing Transactions. Counterparties to such transactions shall: (1) be entities regulated, approved, registered or supervised in their home jurisdiction; and (2) be located in an OECD Member State, which together will constitute the Company's criteria to select counterparties. Counterparties need not have a minimum credit rating.

From time to time, a Fund may engage repurchase/reverse repurchase agreements counterparties and/or securities lending agents that are related parties to the Depositary or other service providers of the Company. Such engagement may on occasion cause a conflict of interest with the role of the Depositary or other service provider in respect of the Company. Please refer to the section entitled "Conflicts of Interest" for further details on the conditions applicable to any such related party transactions. The identity of any such related parties will be specifically identified in the relevant Fund's semi-annual and annual reports.

Repurchase/reverse repurchase agreements or securities lending do not constitute borrowing or lending for the purposes of Regulation 103 and Regulation 111 of the UCITS Regulations respectively.

The Manager will, at least annually, review and/or confirm the arrangements for securities lending and Repurchase Transactions and associated fees invoiced to the relevant Fund, if any.

4.10 Risk Management Process

The Company, or its delegates, on behalf of each Fund has filed with the Central Bank its risk management process which enables it to accurately measure, monitor and manage the various risks associated with the use of FDI and Securities Financing Transactions where appropriate. Any FDI not included in the risk management process will not be utilised until such time as the risk management process has been updated, in accordance with the Central Bank requirements. The Company, or its delegates, will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

4.11 References to Ratings

The European Union (Alternative Investment Fund Managers) (Amendment) Regulations 2014 (S.I. No. 379 of 2014) (the "**Amending Regulations**") transpose the requirements of the Credit Ratings Agencies Directive (2013/14/EU) ("**CRAD**") into Irish law. CRAD aims to restrict the reliance on ratings provided by credit rating agencies and to clarify the obligations for risk management. In accordance with the Amending Regulations and the CRAD (which amended the UCITS Regulations), notwithstanding anything else in this

Prospectus, the Manager or the Investment Manager shall not solely or mechanistically rely on credit ratings in determining the credit quality of an issuer or counterparty.

4.12 Collateral Policy

In the context of efficient portfolio management techniques, Securities Financing Transactions and/or the use of FDI for hedging or investment purposes, Collateral may be received from a counterparty for the benefit of a Fund or posted to a counterparty by or on behalf of a Fund. Any receipt or posting of Collateral by a Fund will be conducted in accordance with the requirements of the Central Bank and the terms of the Collateral Policy outlined below.

The types of collateral acceptable for a Fund shall include but not be limited to: (i) cash; (ii) government or other public securities; (iii) certificates of deposit issued by relevant institutions; (iv) bonds/commercial paper issued by relevant institutions or by non-bank issuers; and (v) equity securities traded on certain stock exchanges.

(a) Collateral - received by the Company

Collateral posted by the counterparty for the benefit of a Fund may be taken into account as reducing the exposure to such counterparty. Each Fund will require receipt of the necessary level of Collateral so as to ensure counterparty exposure limits are not breached. Counterparty risk may be reduced by an amount equivalent to the value of the collateral received after taking into account appropriate discounts.

Risks linked to the management of Collateral, such as operational and legal risks, shall be identified, managed and mitigated by the Manager's risk management process. A Fund receiving Collateral for at least 30% of its Net Asset Value will put in place an appropriate stress testing policy to ensure regular stress tests are carried out under normal and exceptional liquidity conditions to enable the relevant Fund to assess the liquidity risk attached to the Collateral. The liquidity stress testing policy will at least prescribe the components set out in Regulation 24 paragraph 8 of the Central Bank Regulations.

For the purpose of providing margin or Collateral in respect of transactions in techniques and instruments, the Fund may transfer, mortgage, pledge, charge or encumber any assets or cash forming part of the Fund in accordance with normal market practice and the requirements outlined in the Central Bank's Rules.

All assets received by a Fund in the context of Repurchase Transactions and securities lending shall be considered as Collateral and must comply with the terms of the Collateral Policy.

(i) Non-cash Collateral

Collateral received must, at all times, meet with the following criteria:

- (A) Liquidity: Collateral received other than cash should be highly liquid and traded on a regulated market or multilateral trading facility with transparent pricing in order that it can be sold quickly at a price that is close to pre-sale valuation. Collateral received should also comply with the provisions of the Central Bank Regulations. There is no restriction on maturity provided the collateral is sufficiently liquid.
- (B) Valuation: Collateral received should be valued on at least a daily basis and assets that exhibit high price volatility should not be accepted as Collateral unless suitably conservative haircuts are in place. Collateral may be marked to market daily by the counterparty using its procedures, subject to

any agreed haircuts, reflecting market values and liquidity risk and may be subject to variation margin requirements.

- (C) Issuer credit quality: Collateral received should be of high quality.
- (D) Correlation: Collateral received should be issued by an entity that is independent from the counterparty and is not expected to display a high correlation with the performance of the counterparty.
- (E) Diversification (asset concentration): Collateral should be sufficiently diversified in terms of country, markets and issuers with a maximum exposure to a given issuer of 20% of the Fund's Net Asset Value. When the Fund is exposed to different counterparties, the different baskets of Collateral should be aggregated to calculate the 20% limit of exposure to a single issuer. To the extent that a Fund avails of the increased issuer exposure facility in section 5(ii) of Schedule 3 of the Central Bank Regulations, such increased issuer exposure may be to any of the issuers listed in paragraph 5.2(I) of the "Investment Restrictions" section of the Prospectus.
- (F) Immediately available: Collateral received should be capable of being fully enforced by the Fund at any time without reference to or approval from the counterparty.
- (G) Safe-keeping: Collateral received on a title transfer basis should be safekept by the Depositary or its agent. For other types of Collateral arrangement, the Collateral can be held by a third party depositary which is subject to prudential supervision, and which is unrelated and unconnected to the provider of the Collateral.
- (H) Haircuts: The Company (or its delegate), on behalf of each Fund, shall apply suitably conservative haircuts or discounts to the market value of the assets being received as Collateral where appropriate on the basis of an assessment of the characteristics of the assets such as the credit standing or the price volatility, as well as the outcome of any stress tests performed in accordance with the requirements of EMIR. EMIR does not require the application of a haircut for cash variation margin. Accordingly any haircut applied to cover currency risk will be as agreed with the relevant counterparty. The Company (or its delegate) has determined that generally if issuer or issue credit quality of the Collateral is not of a very high quality or the Collateral carries a significant level of price volatility, a conservative haircut must be applied in accordance with the relevant haircut policy. However, the application of such a haircut will be determined on a case by case basis. The Company (or its delegate), in its discretion, may accept certain Collateral with more conservative, less conservative or no haircuts applied in accordance with its haircut policy. To the extent that a Fund avails of the increased issuer exposure facility in section 5(ii) of Schedule 3 of the Central Bank Regulations, such increased issuer exposure may be to any of the issuers listed in section 5.2(I) of the Prospectus.

Non-cash Collateral cannot be sold, pledged or re-invested.

(ii) Cash Collateral

Cash Collateral received by the Company for a Fund may not be invested other than in the following:

- (A) deposits with Relevant Institutions;
- (B) high-quality government bonds;
- (C) reverse repurchase agreements provided the transactions are with credit institutions subject to prudential supervision and the Fund is able to recall at any time the full amount of cash on an accrued basis;
- (D) short-term money market funds as defined in the ESMA Guidelines on a Common Definition of European Money Market Funds (ref CESR/10-049).

Re-invested cash Collateral should be diversified in accordance with the diversification requirements applicable to non-cash Collateral outlined in 4.12(a)(i)(E). Invested cash Collateral may not be placed on deposit with the counterparty or a related entity. Exposure created through the reinvestment of Collateral must be taken into account in determining risk exposures to a counterparty. Re-investment of cash Collateral in accordance with the provisions above can still present additional risk for the Fund. Please refer to the section of this Prospectus entitled "Risk Factors; Reinvestment of Cash Collateral Risk" for more details.

(b) Collateral – posted by the Company

Collateral posted to a counterparty by or on behalf of the Fund must be taken into account when calculating counterparty risk exposure other than when it is protected by client money rules or similar arrangements. Collateral posted to a counterparty and Collateral received from such counterparty may be taken into account on a net basis provided the Fund is able to legally enforce netting arrangements with the counterparty.

4.13 Common Investment Pools

To reduce operational and administrative charges and to facilitate diversification of investments the Manager may authorise the assets of any Fund to be managed in conjunction with other funds established by the Manager or other funds promoted or managed by the Manager or any company affiliated to the Manager. This will be done by establishing a pool of assets ("**Pool**") comprising cash and investments contributed by all funds which participate in the Pool ("**Participating Funds**"). This technique is known as pooling.

Opportunities to establish pooling arrangements arise where the investment objectives and policies of Participating Funds are sufficiently similar so as to enable the assets contributed by a Participating Fund to be managed in a manner identical to that of all other Participating Funds in the Pool. However, it is not essential that the investment objectives and policies of each Participating Fund in the Pool be identical. It is sufficient that the Investment Manager be in a position to manage the Pool as one portfolio of assets whilst complying with the investment objectives, policies and restrictions applicable to each Participating Fund.

A Pool is not a separate legal entity and an investor may not invest directly in a Pool. The Investment Manager shall not be permitted to manage the assets of any Fund on a pooled basis without the prior consent of the Manager. The Manager shall be notified in respect of the admission of any fund as a Participating Fund in a Pool in which a Fund participates.

Operational Issues relating to Pooling

Assets may be contributed to and withdrawn from the Pool by a Participating Fund at any time. A record shall be maintained of all of the assets contributed to the Pool by a Participating Fund and the percentage allocation of each of the pooled assets within the Pool that is attributable to each Participating Fund, which shall be allocated on a pro rata basis. This percentage allocation shall be applied to all assets held in the Pool. When additional cash or securities are contributed to or withdrawn from the Pool by a Participating Fund the allocation percentage of each Participating Fund will be adjusted to reflect the change. Where a contribution is made in cash, a deduction may be made where the Investment Manager considers this necessary to discharge transactions costs and fiscal charges incurred in investing the cash. Similarly, in the case of a cash withdrawal, a deduction may be made to reflect transaction costs in disposing of securities. Dividends and any other distribution of income received in respect of assets will be allocated pro-rata to the Participating Fund's holding of assets. All assets comprising a Pool will be valued in accordance with the provisions of the section "Valuation of Assets" below.

Investors should note that the pooling arrangement may cause the composition of the assets of a Fund to be altered as a result of subscriptions and redemptions in another Participating Fund which would cause the Investment Manager to dispose of or acquire assets for the Pool or may cause the Investment Manager to increase the amount of ancillary liquid assets held by the Investment Manager.

Custody of Assets

A Fund will participate in pooling arrangements only with Participating Fund who have appointed the Depositary as depositary and the Administrator as administrator. The Depositary shall, by relying on a set of records produced by the Administrator's accounting systems, at all times ensure that it is in a position to identify the assets of the Fund even though the sub-custodian's records may identify the assets as being held in a Pool.

Termination of Pooling Arrangement

The Manager may elect at any time to terminate a Fund's participation in the pooling arrangements on notice to the Investment Manager, the Administrator and the Depositary. In such event that portion of the assets in the pool representing each Fund's percentage allocation of assets shall be withdrawn.

4.14 Hedged Classes

The Company may (but is not obliged to) enter into certain currency-related transactions in order to hedge the currency exposure of the assets of a Fund attributable to a particular Class into the currency of denomination of the relevant Class for the purposes of efficient portfolio management and in order to hedge the currency exposure of a Fund where the Fund invests in assets denominated in currencies other than the Base Currency. In addition, a Class designated in a currency other than the Base Currency shall, where disclosed in the relevant Supplement, be hedged against exchange rate fluctuation risks between the designated currency of the Class and the Base Currency. Any financial instruments used to implement such strategies with respect to one or more Classes shall be assets/liabilities of the Fund as a whole but will be attributable to the relevant Class(es) and the gains/losses on, and the costs of, the relevant financial instruments will accrue solely to the relevant Class. However, investors should note that there is no segregation of liability between Share Classes.

Where a Class of Shares is to be currency hedged, this will be disclosed in the Supplement for the Fund in which such Class is issued. Any currency exposure of a Class may not be combined with or offset against that of any other Class of a Fund. The currency exposure of the assets attributable to a Class must be allocated to the Class being hedged and may

not be allocated to other Classes. Where the Investment Manager seeks to hedge against currency fluctuations, while not intended, this could result in over-hedged or under-hedged positions due to external factors outside the control of the Company. Any such hedging will hedge no less than 95% of the portion of the net assets of the relevant Class of Shares which is to be hedged against currency risk. Due to matters outside the control of the Company, currency exposure may be over or under hedged but over hedged positions executed will not be permitted to exceed 105% of the net assets of the relevant Class of Shares. Hedged positions will be kept under review on an ongoing basis, at least at the same valuation frequency of the relevant Fund, to ensure that over hedged or under hedged positions do not exceed/fall short of the permitted levels disclosed above. Such review will incorporate a procedure to rebalance the hedging arrangements on a regular basis to ensure that any such position stays within the permitted position levels disclosed above and that positions materially in excess of 100% of the net assets of the relevant Class of Shares are not carried forward from month to month. To the extent that hedging is successful for a particular Class, the performance of the Class is likely to move in line with the performance of the underlying assets (before taking account of any relevant hedging fees or costs), with the result that investors in that Class will not gain/ loose if the Class currency falls/ rises against the Base Currency. Given that there is no segregation of liability between Classes of Shares, there is a risk that, under certain circumstances, currency hedging transactions in relation to Classes of Shares which have "h" in their name could result in liabilities which might affect the Net Asset Value of other Classes of Shares of the relevant Fund.

4.15 References to Benchmarks

Certain Funds may refer to indices within the Supplement of the relevant Funds. These indices may be referenced for various purposes including, but not limited to operating as a reference benchmark and relative VaR measurement. Where an index is used for such purposes the relevant index will not be used to measure the performance of the Fund for the purposes of defining asset allocation in accordance with Article 3 (1)(7)(e) of the Benchmark Regulation. The particular purpose of the relevant index shall be clearly disclosed in the relevant Supplement. Shareholders should note that the Company and/or its distributors may from time to time refer to other indices in marketing literature or other communications purely for financial or risk comparison purposes. However, unless such indices are referred to as such in the Supplement of the Fund they are not formal benchmarks against which the Fund is managed.

Where relevant the Manager shall put in place written plans, in accordance with Article 28(2) of the Benchmark Regulation, detailing the actions it will take in the event that any index it uses for any Fund in accordance with Article 3 (1)(7)(e) of the Benchmark Regulation materially changes or ceases to be provided. These written plans shall detail the steps the Manager will take to nominate a suitable alternative index. Copies of these written plans may be obtained on request from the Manager.

Any index used by a Fund in accordance with Article 3 (1)(7)(e) of the Benchmark Regulation shall be provided by an administrator and either the index or the administrator of such index shall be included in the register referred to in Article 36 of the Benchmark Regulation or availing of the transitional arrangements pursuant to Article 51 of the Benchmark Regulation.

4.16 References to Securitisations

Regulation EU 2017/2402 (the "Securitisation Regulation") applies to UCITS such as the Company from 1 January 2019. Accordingly, where a Fund is exposed to securitisations, the Investment Manager of the Fund will carry out a due diligence process before becoming exposed to a securitisation and on an ongoing basis as long as they remain exposed to a securitisation. The Investment Manager will ensure that the securitisation is risk retention compliant with the originator retaining a material net economic interest of not less than 5%

in the securitisation and on an ongoing basis, the originator of the securitisation will make available to holders of a securitisation position certain information on the transaction and underlying exposures in accordance with the Securitisation Regulation.

4.17 Miscellaneous

Where the Directors consider it to be in the best interests of Shareholders, the Company on behalf of a Fund may grant a security interest over some or all of the assets of the Fund in connection with any borrowing, securities lending activity or OTC derivatives entered into by the Company on behalf of a Fund.

5 Investment Restrictions

The particular investment restrictions for each Fund will be formulated by the Directors at the time of the creation of the Fund and will appear in the Supplement for the relevant Fund.

Details of the investment restrictions laid down in accordance with the UCITS Regulations in respect of each Fund are set out below:

5.1 Permitted Investments

Investments of each Fund are confined to:

- (a) Transferable securities and money market instruments which are either admitted to official listing on a stock exchange in an EU Member State or non-EU Member State or which are dealt on a market which is regulated, operates regularly, is recognised and open to the public in an EU Member State or non-EU Member State (and which in each case is listed in Appendix II).
- (b) Recently issued transferable securities which will be admitted to official listing on a stock exchange or other market (as described above) within a year.
- (c) Money market instruments other than those dealt on a regulated market.
- (d) Units/shares of UCITS.
- (e) Units/shares of AIFs.
- (f) Deposits with credit institutions.
- (g) FDI.
- 5.2 Investment Restrictions
 - (a) Each Fund may invest no more than 10% of net assets in transferable securities and money market instruments other than those referred to in paragraph 5.1.
 - (b) Each Fund may invest no more than 10% of net assets in recently issued transferable securities which will be admitted to official listing on a stock exchange or other market (as described in paragraph 5.1) within a year. This restriction will not apply in relation to investment by each Fund in certain US securities known as Rule 144A securities provided that:
 - (i) the securities are issued with an undertaking to register with the US Securities and Exchanges Commission within one year of issue; and
 - (ii) the securities are not illiquid securities i.e. they may be realised by each Fund within seven days at the price, or approximately at the price, at which they are valued by the Fund.
 - (c) Each Fund may invest no more than 10% of net assets in transferable securities or money market instruments issued by the same body provided that the total value of transferable securities and money market instruments held in the issuing bodies in each of which it invests more than 5% is less than 40%.

- (d) Subject to the prior approval of the Central Bank, the limit of 10% in (c) is raised to 25% in the case of bonds that are issued by a credit institution which has its registered office in a Member State and is subject by law to special public supervision designed to protect bond-holders. If a Fund invests more than 5% of its net assets in these bonds issued by one issuer, the total value of these investments may not exceed 80% of the net asset value of the Fund.
- (e) The limit of 10% in (c) is raised to 35% if the transferable securities or money market instruments are issued or guaranteed by a Member State or its local authorities or by a non-Member State or public international body of which one or more Member States are members.
- (f) The transferable securities and money market instruments referred to in (d) and (e) shall not be taken into account for the purpose of applying the limit of 40% referred to in (c).
- (g) Each Fund may not invest more than 20% of net assets in deposits made with the same credit institution.

Deposits with any one credit institution, other than Relevant Institutions held as ancillary liquidity, must not exceed 10% of net assets.

This limit may be raised to 20% in the case of deposits made with the Depositary.

(h) The risk exposure of each Fund to a counterparty in an "over the counter" ("OTC") derivative transaction may not exceed 5% of net assets.

This limit is raised to 10% in the case of Relevant Institutions.

- Notwithstanding paragraphs (c), (g) and (h) above, a combination of two or more of the following issued by, or made or undertaken with, the same body may not exceed 20% of net assets:
 - (i) investments in transferable securities or money market instruments;
 - (ii) deposits, and/or
 - (iii) counterparty risk exposures arising from OTC derivatives transactions.
- (j) The limits referred to in (c), (d), (e), (g), (h) and (i) above may not be combined, so that exposure to a single body shall not exceed 35% of net assets.
- (k) Group Companies are regarded as a single issuer for the purposes of (c), (d), (e), (g), (h) and (i). However, a limit of 20% of net assets may be applied to investment in transferable securities and money market instruments within the same group.
- (I) Each Fund may invest up to 100% of its net assets in different transferable securities and money market instruments issued or guaranteed by any Member State, or any local authority of a Member State, or non-Member States or public international bodies of which one or more Member States are members or by Australia, Canada, Hong Kong, Japan, New Zealand, Switzerland, United States or any of the following: or by OECD Governments (provided the relevant issues are investment grade), Government of Brazil (provided the issues are of investment grade), Government of the People's Republic of China, Government of India (provided the issues are of investment grade), Government of Singapore, European Investment Bank, European Bank for Reconstruction and Development, International Finance Corporation, International Monetary Fund, Euratom, The Asian Development Bank, International Central Bank, Council of Europe, Eurofima, African Development Bank, International

Bank for Reconstruction and Development (The World Bank), The Inter American Development Bank, European Union, Federal National Mortgage Association (Fannie Mae), Federal Home Loan Mortgage Corporation (Freddie Mac), Government National Mortgage Association (Ginnie Mae), Student Loan Marketing Association (Sallie Mae), Federal Home Loan Bank, Federal Farm Credit Bank, Tennessee Valley Authority or Straight-A Funding LLC.

- (m) Each Fund must hold securities from at least 6 different issues, with securities from any one issue not exceeding 30% of net assets.
- 5.3 Investment in Collective Investment Schemes
 - (a) A Fund may not invest more than 20% of net assets in any one CIS.
 - (b) Investments in AIFs may not in aggregate exceed 30% of net assets.
 - (c) A Fund may not invest more than 10% of net assets in total in other CIS. Such CIS must themselves be prohibited from investing more than 10% of net assets in total in other CIS.
 - (d) When a Fund invests in the units of other CIS that are managed, directly or by delegation by the Fund's investment manager or by any other company with which the investment manager is linked by common management or control, or by a substantial direct or indirect holding, neither the investment manager nor that other company may charge subscription, conversion or redemption fees on account of that Fund's investment in the units of such other CIS. Maximum management fees or TER invoiced by such other CIS will be set out in the Supplement for each Fund that elected for investing in such other CIS.
 - (e) Where a commission (including a rebated commission) is received by a Fund's investment manager by virtue of an investment in the units of another CIS, this commission must be paid into the property of the Fund.
- 5.4 General Provisions
 - (a) The Company may not acquire any shares carrying voting rights which would enable it to exercise significant influence over the management of an issuing body.
 - (b) Each Fund may acquire no more than:
 - (i) 10% of the non-voting shares of any single issuing body;
 - (ii) 10% of the debt securities of any single issuing body;
 - (iii) 25% of the units/shares of any single CIS;
 - (iv) 10% of the money market instruments of any single issuing body.

NOTE: The limits laid down in (ii), (iii) and (iv) above may be disregarded at the time of acquisition if at that time the gross amount of the debt securities or of the money market instruments or the net amount of the securities in issue cannot be calculated.

- (c) 5.4(a) and (b) shall not be applicable to:
 - (i) transferable securities and money market instruments issued or guaranteed by a Member State or its local authorities;

- (ii) transferable securities and money market instruments issued or guaranteed by a non-Member State;
- (iii) transferable securities and money market instruments issued by public international bodies of which one or more Member States are members;
- (iv) shares held by each Fund in the capital of a company incorporated in a nonmember State which invests its assets mainly in the securities of issuing bodies having their registered offices in that State, where under the legislation of that State such a holding represents the only way in which each Fund can invest in the securities of issuing bodies of that State. This waiver is applicable only if in its investment policies the company from the non-Member State complies with the limits laid down in 5.2(c) to 5.2(k), 5.3(a), 5.3(b), 5.4(a), (b), (d), (e) and (f) and provided that where these limits are exceeded, 5.4(e) and (f) are observed;
- (v) shares held by the Fund in the capital of subsidiary companies carrying on only the business of management, advice or marketing in the country where the subsidiary is located, in regard to the redemption of shares at Shareholders' request exclusively on their behalf.
- (d) A Fund need not comply with the investment restrictions herein when exercising subscription rights attaching to transferable securities or money market instruments which form part of their assets.
- (e) The Central Bank may allow recently authorised Funds to derogate from the provisions of 5.2(c) to 5.2(l), 5.3(a), 5.3(b), 5.6(a) and 5.6(b) for six months following the date of their authorisation, provided they observe the principle of risk spreading.
- (f) If the limits laid down herein are exceeded for reasons beyond the control of a Fund, or as a result of the exercise of subscription rights, the Fund must adopt as a priority objective for its sales transactions the remedying of that situation, taking due account of the interests of its shareholders.
- (g) A Fund may not carry out uncovered sales of:
 - (i) transferable securities;
 - (ii) money market instruments;
 - (iii) units/shares of CIS; or
 - (iv) FDI.
- (h) A Fund may hold ancillary liquid assets.
- 5.5 Financial Derivative Instruments
 - (a) A Fund's global exposure relating to FDI must not exceed its total Net Asset Value.
 - (b) A Fund may invest in FDIs dealt in "over the counter" provided that the counterparties to over-the-counter transactions are institutions subject to prudential supervision and belonging to the categories approved by the Central Bank.
 - (c) Position exposure to the underlying assets of FDI, including embedded FDI in transferable securities or money market instruments, when combined where relevant with positions resulting from direct investments, may not exceed the investment limits set out in the Central Bank Rules. (This provision does not apply in the case of index

based FDI provided the underlying index is one which meets with the criteria set out in the Central Bank Rules.)

- (d) A transaction in FDI which gives rise to a future commitment on behalf of each Fund must be covered as follows:
 - (i) in the case of FDI which requires the Fund to physically deliver the underlying asset, the asset must be held at all times by the Fund;
 - (ii) in the case of FDI which automatically, or at the discretion of the Fund, are cash settled, each Fund must hold, at all times, liquid assets which are sufficient to cover the exposure.

It is intended that each Fund should have the power to avail of any change in applicable laws, UCITS Regulations or Central Bank Rules which would permit investment in assets and securities on a wider basis.

The Company will not amend such investment restrictions except in accordance with the requirements of the Central Bank.

5.6 Index Tracking Funds

- (a) Notwithstanding the provisions of paragraph 5.2(c), a Fund may, in accordance with the Articles, invest up to 20 per cent. of its net assets in shares and/or debt securities issued by the same body where the investment policy of the Fund is to replicate or reference an index. Where a Fund gains exposure to an index via an FDI, no component of the index should comprise more than 20% of the total of the index. The index must be recognised by the Central Bank in accordance with the Central Bank Rules.
- (b) The limit in paragraph (a) may be raised to 35%, and applied to a single issuer or component, where this is justified by exceptional market conditions (which justification, where relevant, shall be disclosed in the relevant Supplement).
- (c) The reference in paragraph (a) to replication of the composition of a shares or debt securities index shall be understood as replication of the composition of the underlying assets of the index, including the use of derivatives or other techniques as referred to in Regulation 69(1), (2) and (3) of the UCITS Regulations.

In addition to the above investment restrictions contained in the UCITS Regulations, the following investment restrictions shall apply with respect to each relevant Fund:

United Nations Conventions on Cluster Munitions and Anti-Personnel Landmines: the Investment Manager shall ensure that no investments are made in companies that are directly involved in the development, brokerage, acquisition, import, export, execution, production, storage or trading of nuclear, biological or chemical weapons (NBC weapons), cluster munitions and/or anti-personnel mines (prohibited war material), or that provide related services. In addition, these restrictions also exclude such Funds from securities of companies that are involved in the manufacture or sale of controversial weapons. Such exclusions are based on UBS's methodology.

Cannabis Exclusion List

In respect of all Funds which replicate their Reference Index, the Investment Manager will use its proprietary tools to ensure that the portfolio of transferable securities held by the Fund will be screened to exclude investments in certain companies that are involved in the commercial production, distribution or sale of cannabis ("cannabis securities").

This exclusion is based on the UBS methodology.

6.1 General

The discussion below is of general nature and is intended to describe various risk factors which may be associated with an investment in the Shares of a Fund to which the attention of investors is drawn. See also the section of the relevant Supplement for a discussion of any additional risks particular to Shares of that Fund. However, these are not intended to be exhaustive and there may be other considerations that should be taken into account in relation to an investment. Investors should consult their own advisors before considering an investment in the Shares of a particular Fund. What factors will be of relevance to the Shares of a particular Fund will depend upon a number of interrelated matters including, but not limited to, the nature of the Shares, the Reference Index or Reference Asset (if applicable), the investments and assets of the Fund to the Reference Index or Reference Asset (if applicable).

No investment should be made in the Shares of a particular Fund until careful consideration of all those factors has been made.

The investments of the Company in securities are subject to normal market fluctuations and other risks inherent in investing in securities. The value of investments and the income from them, and therefore the value of and income from Shares relating to each Fund, can go down as well as up and an investor may not get back the amount they invest. Changes in exchange rates between currencies or the conversion from one currency to another may also cause the value of the investments to diminish or increase. Due to any Redemption Fee and/or Primary Market Transaction Costs which may be payable on the Shares, an investment in Shares (where such charges are levied, as further disclosed in the relevant Supplement) should be viewed as a medium to long term investment. An investment in a Fund should not constitute a substantial proportion of an investment portfolio and may not be appropriate for all investors.

Risk factors may occur simultaneously and/or may compound each other resulting in an unpredictable effect on the value of the Shares. No assurance can be given as to the effect that any combination of risk factors may have on the value of the Shares.

Achievement of Investment Objective: There is no assurance that any Fund will achieve its investment objective. The following are some, but not necessarily all, of the factors which may result in the value of the Shares varying from the value of the Reference Index or Reference Asset: investments in assets other than the constituents of the Reference Index or Reference Asset may give rise to delays or additional costs and taxes compared to an investment in the constituents of the Reference Index or regulatory constraints may affect the Company but not the constituents of the Reference Index or Reference Asset; the fluctuation in value of Fund's assets; and the existence of a cash position held by a Fund.

Segregation of Liability: Under the provisions of the Companies Act, the Directors shall maintain for each Fund a separate portfolio of assets. As between Shareholders, each portfolio of assets shall be invested for the exclusive benefit of the relevant Fund. The Shareholders shall only be entitled to the assets and profits of that Fund in which they participate. The Company shall be considered one single legal entity. With regard to third parties, in particular towards the Company's creditors, the Company shall be responsible for all liabilities incurred by a Fund exclusively based on the assets of this relevant Fund.

Among the Shareholders, the liabilities of each Fund shall only be incurred to the respective Fund. While the provisions of the Companies Act provide for segregated liability between Funds, these provisions have yet to be tested in foreign courts, in particular, in satisfying local creditors' claims. Accordingly, it is not free from doubt that the assets of any Fund of the Company may not be exposed to the liabilities of other Funds of the Company.

Legal and Regulatory: The Company must comply with regulatory constraints or changes in the laws affecting it, the Shares, or the Investment Restrictions, which might require a change in the investment policy and objectives followed by a Fund. A Fund's assets may also be subject to change in laws or regulations and/or regulatory action which may affect their value.

Listing: There can be no certainty that a listing on any stock exchange applied for by the Company will be achieved and/or maintained or that the conditions of listing will not change. Further, trading in Shares of such relevant Funds on a Relevant Stock Exchange may be halted pursuant to that Relevant Stock Exchange's rules due to market conditions and investors may not be able to sell their Shares of such relevant Funds until trading resumes.

Nominee Arrangements: Where an investor holds Shares via a nominee or intermediary, or holds interests in Shares through a Clearing Agent, such Shareholder will typically not appear on the Register of the Company and may not therefore be able to exercise voting or other rights available to those persons appearing on the Register.

Political Factors, Emerging Market and Non-OECD Member State Assets: The performance of the Shares and/or the possibility to purchase, sell, or redeem the Shares may be affected by changes in general economic conditions and uncertainties such as political developments, changes in government policies, the imposition of restrictions on the transfer of capital and changes in regulatory requirements. Such risks can be heightened in investments in, or relating to, emerging markets or non-OECD member states. The currencies in which investments are denominated may be unstable, may be subject to significant depreciation and may not be freely convertible. Settlement of transactions may be subject to delay and administrative uncertainties. In addition, local custody services remain underdeveloped in many non-OECD and emerging market countries and there is a transaction and custody risk involved in dealing in such markets. In certain circumstances, a Fund may not be able to recover or may encounter delays in the recovery of some of its assets. Furthermore, the legal infrastructure and accounting. auditing and reporting standards in emerging markets or non-OECD member states, may not provide the same degree of investor information or protection as would generally apply to major markets.

Changes in the UK Political Environment: On 23 June 2016, the UK held a referendum to decide on its membership of the EU. The resulting vote was to leave the EU. The UK withdrew from the EU as of midnight CET on 31 January 2020. Following a transition period which ended on 31 December 2020, the UK generally ceased to have access to the EU single market.

While the UK and the EU have concluded an agreement on trade and cooperation (the "**Trade and Cooperation Agreement**") which has been provisionally applied from 1 January 2021, this agreement does not necessarily create a permanent set of rules, but is a basis for an evolving relationship, with scope for increasing divergence or closer cooperation which may vary between different areas.

Accordingly, there remain a number of uncertainties in connection with the future of the UK and its relationship with the EU. This introduces significant uncertainty in the business, legal and political environment, and risks ("**Brexit Risks**") including increased potential for short and long-term market volatility and currency volatility, macroeconomic risk to the UK and European economies, internal political instability within the UK including impetus for the break-up of the UK and with respect to the border between Ireland and Northern Ireland

and related political and economic stresses, impetus for further disintegration of the EU and related political stresses (including those related to sentiment against cross-border capital movements), the potential for exacerbation of supply bottlenecks and labour shortages resulting from disruption to international trade and restrictions of the movement of people, legal uncertainty regarding achievement of compliance with applicable financial and commercial laws and regulations, and the effects of future divergence between the legal, regulatory and tax regimes in the EU and the UK.

This uncertainty surrounding the UK's relationship with the EU, together with the impact of the changes that have already occurred as a result of the end of the transition period and the termination of the UK's access to the EU single market, may adversely impact the Company and its investments (in particular those that relate to companies or assets based in, doing business in, or having services or other significant relationships in or with, the UK).

There can be no assurance that the Brexit Risks will not alter significantly the attractiveness of an investment in the Company including as a result of the potential for capital losses, delays, legal and regulatory risk and general uncertainty. Brexit Risks also include the potential for prejudice to financial services businesses that are conducting business in the EU and which are based in the UK, disruption to regulatory regimes related to the operations of the Company or its service providers. As such, it may be necessary for the Manager, the Investment Manager or a Fund's advisers or service providers to restructure their arrangements with the Company.

Eurozone Crisis: As a result of the crisis of confidence in the markets which has caused bond yield spreads (the cost of borrowing in the debt capital markets) and credit default spreads (the cost of purchasing credit protection) to increase, most notably in relation to certain Eurozone countries, certain countries in the EU have had to accept "bailouts" from banks and lines of credit from supra-governmental agencies such as the International Monetary Fund (the "**IMF**") and the recently created European Financial Service Facility (the "**EFSF**"). The European Central Bank (the "**ECB**") has also been intervening to purchase Eurozone debt in an attempt to stabilise markets and reduce borrowing costs. In December 2011, leaders of the countries in the Eurozone, as well as the leaders of certain other countries in the EU, met in Brussels and agreed a "fiscal compact" which includes a commitment to a new fiscal rule, to be introduced into the legal systems of the relevant countries, as well as acceleration of the entry into force of the European Stability Mechanism treaty.

Notwithstanding the measures described above, and future measures which may be introduced, it is possible that a country may leave the Eurozone and return to a national currency, and as a result may leave the EU and/or that the Euro, the European single currency, will cease to exist in its current form and/or lose its legal status in one or more countries in which it currently has such status. The effect of such potential events on the Funds which are denominated in Euro or which invest in instruments predominantly tied to Europe is impossible to predict.

External Factors may impact on Performance: Each Fund's performance is influenced by a variety of external factors which are beyond the control of the Company, or its delegates, including: changing supply and demand relationships; trade, fiscal, monetary and exchange control programs and policies of governments; political and economic events, policies and political unrest; changes in interest rates and rates of inflation; currency devaluations and re-evaluations; market sentiment; and force majeure events, including natural disasters (such as hurricanes, earthquakes or floods), pandemics or any other serious public health concern, war or terrorism or the threat of or perceived potential for these events, each of which or a combination of which may have a negative impact on the performance of each Fund. These events could adversely affect levels of business activity and precipitate sudden significant changes in regional and global economic conditions and cycles. These events also pose significant risks to people and physical facilities and operations around the world.

Concerns about the spread of the novel coronavirus disease that emerged in late December 2019 ("**Covid-19**"), and other outbreaks of health epidemics and contagious diseases in the past, have caused governments at various times to take measures to prevent the spread of such diseases, including restrictions on travel and public transport and prolonged closures of businesses, workplaces, schools and other public amenities. The outbreak of Covid-19, and measures imposed by governments in response, have materially restricted, and may continue to materially restrict, the level of business activity globally, which may adversely affect the Fund and its Investments, or its ability to source new investments. Any other such widespread health crisis could have a similar effect in the future.

In particular, the outbreak of Covid-19 spread rapidly around the globe, and was characterised as a pandemic by the World Health Organisation in March 2020. To date, local and national governments worldwide have imposed guarantines, significant domestic and international travel restrictions and "lockdown" measures restricting public gatherings, restricting the purposes for which individuals may leave their homes and requiring the temporary closure of workplaces and businesses. Although such restrictions may be relaxed from time to time, they could be re-imposed in response to further outbreaks of Covid-19 or the emergence of new Covid-19 variants. The deterioration of political, socioeconomic and financial conditions globally has resulted in widespread disruption to certain sectors, including both the real estate and the financial sectors. Solvency issues may arise for key market participants if the Covid-19 pandemic and the disruption it has caused results in working capital lines being blocked, financial covenants being breached, events of default occurring and/or the triggering of termination payments or other contingent liabilities for non-performance, whilst any continuation or intensification of the slow-down in business activity associated with the pandemic may also negatively impact the liquidity of the Company's investments. The Company has invested, and may invest in the future, in investments that have exposure to sectors that have been adversely affected by the Covid-19 pandemic, and which could be affected by other widespread health crises in the future.

In addition, although the Investment Manager remains operational amidst the Covid-19 outbreak, travel and transportation restrictions, quarantine requirements, mandatory working from home arrangements and/or absence due to illness have affected or may affect all or a significant number of staff, and the employees of other service providers to the Company. If these conditions are further prolonged or reintroduced, they have the potential to disrupt the ability of the Investment Manager and such other service providers to perform their functions on behalf of the Company.

While the Investment Manager believes that it can pursue its investment strategy during this pandemic, any of the foregoing could materially and adversely affect the Company's ability to source, manage and divest its investments and its ability to fulfil its investment objectives. Similar consequences could arise with respect to other comparable infectious diseases (including any potential further Covid-19 outbreak that may occur in the future or the emergence of new Covid-19 variants). The full scope of the Covid-19 outbreak, its duration, intensity and consequences are uncertain and the resultant economic slowdown and/or negative business sentiment across markets and/or any long-term changes to working patterns and human behaviour that may arise from the pandemic could have a negative and long-lasting impact on the business operations and financial condition of the Company and its investments and its ability to source new investments which satisfy the Funds' investment criteria. The longer the pandemic and related adverse business environment persist, the greater the risk of materially adverse impact on the Company.

Operational Risks (including Cyber Security and Identity Theft): An investment in a Fund, like any fund, can involve operational risks arising from factors such as processing errors, human errors, inadequate or failed internal or external processes, failure in systems and technology, changes in personnel, infiltration by unauthorised persons and errors caused by service providers such as the Investment Manager or the Administrator. While the Funds seek to minimise such events through controls and oversight, the inherent risks

of IT are continuously increasing and as a result, there may still be failures that could cause losses to a Fund.

The Investment Manager, Administrator and Depositary (and their respective groups) each maintain appropriate information technology systems and seek to ensure that there is a well-defined and comprehensive IT and cybersecurity risk management framework in place that provides effective oversight of IT related risks. However, like any other system, these systems could be subject to cyber security attacks or similar threats resulting in data security breaches, theft, a disruption in the Investment Manager's, Administrator's and/or Depositary's service or ability to close out positions and the disclosure or corruption of sensitive and confidential information. Notwithstanding the existence of policies and procedures designed to identify, manage and prevent such breaches and ensure the security, integrity and confidentiality of such information as well as the existence of business continuity and disaster recovery measures designed to mitigate any such breach or disruption at the level of the Company and its delegates, such security breaches may potentially also result in loss of assets and could create significant financial and or legal exposure for the Company.

Fund Expenses: Returns on Shares will be net of all fees and expenses incurred in the establishment and ongoing running of the relevant Fund and may not be directly comparable to the yields which could be earned if any investment were instead made directly in the assets of the relevant Fund or the constituents of the Reference Index or the Reference Asset.

Delivery Failure: In some securities markets, deliveries of securities and other Fund Assets and payments therefor may not be or are not customarily made simultaneously. Further due to the nature of the investment policy and structuring of transactions involving the Fund Assets the deliveries of securities and payments may not be made simultaneously. The Depositary or a sub-custodian may make or accept payment for or delivery of Fund Assets in such form and manner and shall not be contrary to the customs prevailing in the relevant market or among securities dealers or in accordance with the terms of the Depositary Agreement. The Company shall bear the risk that:-(i) the recipient of Fund Assets delivered by the Depositary or any sub-custodian may fail to make payment for or return such Fund Assets or hold such Fund Assets or the proceeds of their sale in trust for the Depositary or the Company; and (ii) the recipient of payment for Fund Assets made by the Depositary or any sub-custodian including without limitation amounts paid as premium or margin on derivatives contracts may fail to deliver the Fund Assets (such failure to include, without limitation, delivery of forged or stolen Fund Assets) or to return such payment, or hold such payment in trust for the Depositary or the Company in each case whether such failure is total or partial or merely a failure to perform on a timely basis. Neither the Depositary nor any sub-custodian shall be liable to the Company for any loss resulting from any of the foregoing events or from the liquidation, insolvency or bankruptcy of such recipient.

Failure to Reproduce the Composition of a Reference Index or Reference Asset: The intention is that each Fund will seek to achieve a return by tracking or replicating the relevant Reference Index or Reference Asset. Investors should, however, be aware that the Fund incurs expenses and fees which reduce returns accordingly, and that if and when dividends are paid by the issuers of the securities the Fund's income may be reduced by withholding taxes on such dividend, such taxes may not be reflected in the return on the Reference Index or Reference Asset concerned.

While the Company will, in the event of changes to the composition of the Reference Index or Reference Asset, seek to adjust the portfolio composition of the Fund, it should also be noted that a period of time generally elapses between any change in the composition of the Reference Index or Reference Asset and the corresponding adjustment being made to the composition of the Fund's portfolio.

To assist in the achievement of the investment objective, the Fund may also employ FDIs. However, prices of FDIs develop differently from the underlying securities. No guarantee can therefore be given that the return on the Fund concerned will at all times be identical to that of the Reference Index or Reference Asset.

Interest and Exchange Rate Fluctuations: An investment in the Shares may directly or indirectly involve exchange rate risk. Because the Net Asset Value of the Fund will be calculated in its Base Currency, the performance of any of its constituents denominated in another currency other than the Base Currency will also depend on the strength of such currency against the Base Currency. Equally, the currency denomination of any Fund Asset in another currency than the Base Currency will involve exchange rate risk for the Fund. Furthermore, an investor will be subject to exchange rate risk where they invest in a Fund whose Base Currency is different to the functional currency of the investor.

Each Fund may, but is not obliged to, enter into hedging transactions on currencies to protect against a decline in the value of investments denominated in currencies other than the Base Currency, and against any increase in the cost of investments denominated in currencies other than the Base Currency; however there is no assurance that any such hedging transactions will be successful.

Hedging activity at Share Class level may expose the Fund to cross contamination risk as it may not be possible to ensure (contractually or otherwise) that a counterparty's recourse in any such arrangements is limited to the assets of the relevant Share Class. Although the costs, gains and losses of the currency hedging transactions will accrue solely to the relevant Share Class, investors are nonetheless exposed to the risk that currency hedging transactions undertaken in one Share Class may impact negatively on another Share Class, particularly where (pursuant to EMIR) such currency hedging transactions require the Fund to post collateral (i.e. initial or variation margin). Any such collateral is posted by a Fund and at the Fund's risk (rather than by the Share Class and at the risk of the Share Class only because the Share Class does not represent a segregated portion of the Fund's assets) thus exposing investors in other Share Classes to a proportion of this risk.

Fluctuations in interest rates of the country or region in whose currency or currencies the Shares, the Fund's Assets and/or the Reference Index or Reference Asset are denominated may affect financing costs and the real value of the Shares. Interest rates vary over time, and the Net Asset Value of a Fund invested in fixed-interest securities will change in response to fluctuations in interest rates (and credit spreads). For example, when interest rates decline the value of fixed-income securities can generally be expected to rise, and conversely, when interest rates rise, the value of fixed-income securities can generally be expected to decline.

Securities Lending Arrangements and Repurchase Transactions: The Company may engage in securities lending or Repurchase Transactions over a period of time with one or more counterparties, which could include an exclusive arrangement with a member of the UBS AG Group as the exclusive counterparty. Collateral which meets the requirements of the Collateral Policy will be posted by the relevant counterparty. A default by the counterparty to such a securities lending arrangement or Repurchase Transaction, or a fall in the value of the Collateral posted in connection with such transactions below that of the value of the securities lent or the cash leg of the Repurchase Transaction may result in a reduction in the value of the relevant Fund and the Fund may suffer loss as a result. The Company will use reasonable endeavours to ensure that any Collateral transferred to it in connection with such transactions will be segregated from the bankruptcy estate of the counterparty and not available to the creditors of the counterparty. Shareholders are advised, however, that third parties may seek to challenge such segregation which, if successful, would result in a total loss of both the Collateral and the assets of the Fund that were lent or otherwise transferred. In the case of cash Collateral, as a matter of applicable law, such cash Collateral might not be held in a segregated manner in favour of the Company, which may result in a total loss of cash Collateral upon insolvency of the relevant counterparty.

Reinvestment of Cash Collateral: As a Fund may reinvest cash Collateral received, subject to the conditions and within the limits laid down by the Central Bank, a Fund reinvesting cash Collateral will be exposed to the risk associated with such investments, such as failure or default of the issuer of the relevant security.

Collateral Risk: Collateral or margin may be passed by the Fund to a counterparty or broker in respect of OTC FDI transactions. Assets deposited as collateral or margin with brokers may not be held in segregated accounts by the brokers and may therefore become available to the creditors of such brokers in the event of their insolvency or bankruptcy. Where collateral is posted to a counterparty or broker by way of title transfer, the collateral may be re-used by such counterparty or broker for their own purpose, thus, exposing the Fund to additional risk.

SFTs, TRSs and Collateral Management Risk: SFTs create several risks for the Company and its investors, including counterparty risk, if the counterparty to an SFT or TRS defaults on its obligation to return assets equivalent to the ones provided to it by the Company and liquidity risk if the Company is unable to liquidate collateral provided to it to cover a counterparty default (investors may be unable to redeem their interest in the Company if the Company is unable to liquidate the portfolio as a result of its exposure to illiquid assets).

Risks related to the counterparty's right of re-use of any collateral include that upon the exercise of such right of re-use, such assets will no longer belong to the Company and the Company will only have a contractual claim for the return of equivalent assets. In the event of insolvency of a counterparty the Company shall rank as an unsecured creditor and may not recover its assets from the counterparty. More broadly, assets subject to a right of re-use by a counterparty may form part of a complex chain of transactions over which the Company or its delegates will not have any visibility or control.

Margin lending transactions are defined in SFTR as being transactions in which a counterparty extends credit in connection with the purchase, sale, carrying or trading of securities, but not including other loans that are secured by collateral in the form of securities (for example, credit lines in connection with futures trading). In the context of prime brokerage and other credit facilities that the Fund may utilise, it may be difficult to identify whether a particular transactions falls within the definition of SFT or not.

In respect of TRSs, if the volatility or expectation of volatility of the reference asset(s) varies, the market value of the financial instruments may be adversely affected. A Fund will be subject to the credit risk of the counterparty to the swap, as well as that of the issuer of the reference obligation, and the documentation risk associated with these instruments. The Company generally will have no right to directly enforce compliance by the issuer of the reference obligation with the terms of that obligation, and will not have any rights of set-off against the issuer.

Counterparty Rating Downgrade Risk: The Company will enter into OTC derivative transactions and Securities Financing Transactions only with those counterparties that it believes to be sufficiently creditworthy.

If a counterparty (which is not a Relevant Institution) engaged by the Company, in respect of a Fund, is subject to a credit rating downgrade, this could potentially have significant implications for the relevant Fund both from a commercial perspective and a regulatory perspective. Pursuant to the Central Bank Rules, a rating downgrade for a counterparty to an OTC derivative transaction or a Securities Financing Transaction to A-2 or below (or a comparable rating) shall require the relevant Fund without delay to conduct a new credit assessment of that counterparty. Regardless of the measures the Company, in respect of a Fund, may implement to reduce counterparty credit risk, there can be no assurance that a counterparty will not default or that the relevant Fund will not sustain losses on the transactions as a result.

Investments in Small and Mid-Cap Companies: The equity securities of smaller and midcap companies tend to be more volatile and less liquid than the equity securities of large companies. As smaller companies may experience more market price volatility than equity securities of larger companies, the Net Asset Value of any Funds which invest in such companies may reflect this volatility. Smaller companies, as compared with larger companies, may have a shorter history of operations, may not have as great an ability to raise additional capital, may have a less diversified product line making them susceptible to market pressure and may have a smaller public market for their shares.

Investment in such companies may involve relatively higher investment costs and accordingly investment in a Fund which invests in smaller and mid-cap companies should be viewed as a long-term investment. Such a Fund may however dispose of an investment made by it within a relatively short period of time, for example, to meet requests for redemption of shares.

Corporate Bonds: A corporate bond-focused Fund may invest in, or otherwise gain exposure to, corporate bonds from companies with a range of credit worthiness. A default by the issuer of a bond may result in a reduction in the value of that Fund.

Although a Fund will invest in bonds that invest and trade in the secondary market, the secondary market for corporate bonds can often be illiquid and therefore it may be difficult to achieve fair value on purchase and sale transactions.

Covered Bonds: Where a Fund invests in, or otherwise gains exposure to, covered bonds, such covered bonds will generally be of a higher quality than comparable uncovered bonds, however there is no guarantee that such covered bonds will be free from counterparty default.

Government Bonds: Although a government bond-focused Fund will invest in, or otherwise gain exposure to, government bonds that invest and trade in the secondary market, the secondary market for government bonds or government-issued inflation linked bonds can become illiquid and therefore it may be more difficult to achieve fair value on purchase and sale transactions.

Structured Finance and Other Securities: A Fund may be exposed directly or indirectly to Structured Finance Securities and other assets which involve substantial financial risk, including distressed debt and low quality credit securities, asset-backed securities and credit-linked securities. These securities may entail a higher liquidity risk than exposure to sovereign or corporate bonds. The Fund's primary credit risk would be to the issuer of the Structured Finance Security.

Concentration Risk: If the Reference Index or Reference Asset of a Fund concentrates in a particular industry, group of industries or sector, that Fund may be adversely affected by the performance of those securities and may be subject to price volatility. In addition, a Fund that concentrates in a single industry or group of industries may be more susceptible to any single economic, market, political or regulatory occurrence affecting that industry or group of industries.

Leverage Risk: The Fund Assets, Reference Index or Reference Asset and the derivative techniques used to link the two may comprise elements of leverage (or borrowings) which may potentially magnify losses and may result in losses greater than the amount borrowed or invested.

Capital Erosion Risk: Certain Funds and Share Classes may have as the priority objective the generation of income rather than capital. Investors should be noted that the focus on income and the charging of investment management fees and any other fees to capital may erode capital and diminish the Fund's ability to sustain future capital growth. In this regard, distributions made during the life of the Fund or an applicable Share Class should be understood as a type of capital reimbursement.

Share Subscriptions and Redemptions: Provisions relating to the subscription and redemption of Shares grant the Company discretion to limit the amount of Shares available for subscription or redemption on any Dealing Day and, in conjunction with such limitations, to defer or pro rata such subscription or redemption. In addition, where requests for subscription or redemption are received late, there will be a delay between the time of submission of the request and the actual date of subscription or redemption. Such deferrals or delays may operate to decrease the number of Shares or the redemption amount to be received.

Subscriptions/Redemptions Account: The Company operates а Subscriptions/Redemptions Account for all of Funds. Monies the the in Subscriptions/Redemptions Account are deemed assets of the respective Funds and shall not have the protection of the Investor Money Regulations. There is a risk for investors to the extent that monies are held by the Company in the Subscriptions/Redemptions Account for the account of a Fund at a point where such Fund (or another Fund of the Company) becomes insolvent. In respect of any claim by an investor in relation to monies held in the Subscriptions/Redemptions Account, the investor shall rank as an unsecured creditor of the Company.

Counterparty Recovery Risk: The Company has appointed a number of service providers to the Company. Where a service provider's error results in a loss to a Fund, the compensation which may be payable to the relevant Fund by that service provider will be limited to the amount which can be recovered from the service provider.

Index Licences: Investors should note that there may be instances where, under the terms of an index licence agreement between the Manager and an Index Provider (relating to a Reference Index), the Index Provider has limited or excluded its liability under such index licence agreement. In such circumstances, and for example where an Index Provider miscalculates the value or level of a Reference Index, the Fund and/or Shareholders may have no recourse against the relevant Index Provider and as a result Shareholders in the relevant Fund may incur a loss.

Index calculation risk: the values of Reference Indices are calculated and published on a regular basis by the respective Index Provider or their delegates, and the published values can form an integral part of the value of the Swap Agreement and consequently of the Net Asset Value of the Fund. Investors are therefore subject to the risk that the levels of the Reference Indices are calculated or published incorrectly or inaccurately. In situations where, subsequent to the initial publication of the levels for any day and subsequent to the release of the Net Asset Value of the Fund for that day, the levels of the Reference Indices are revised, the Net Asset Value of the Fund for that day will not be amended to reflect the new levels of the Reference Indices.

Index Tracking Risk: there is no assurance that the Index Provider will compile the Reference Index accurately, or that the Reference Index will be determined, composed or calculated accurately. While the Index Provider does provide descriptions of what the Reference Index is designed to achieve, the Index Provider does not provide any warranty or accept any liability in relation to the quality, accuracy or completeness of data in respect of the Reference Index, and does not guarantee that the Reference Index will be in line with the described index methodology.

The Company and affiliates do not provide any warranty or guarantee for Index Provider errors. Errors in respect of the quality, accuracy and completeness of the data may occur from time to time and may not be identified and corrected for a period of time, particularly where the indices are less commonly used. Therefore gains, losses or costs associated with Index Provider errors will be borne by the Funds and their investors. For example, during a period where the Index contains incorrect constituents, a Fund tracking such published Reference Index would have market exposure to such constituents and would be underexposed to the constituents that should have been included in the Reference Index. As such, errors may result in a negative or positive performance impact to the Funds and their investors. Investors should understand that any gains from Index Provider errors will be kept by the Funds and their investors and any losses resulting from Index Provider errors will be borne by the Funds and their investors.

Short Selling Risk: Although the UCITS Regulations prohibit the short selling of physical securities, UCITS are permitted to create synthetic short positions through the use of FDIs. A short sale means any sale of a security which the seller does not own at the time of entering into the agreement to sell including such a sale where at the time of entering into the agreement to sell the borrowed or agreed to borrow the security for delivery at settlement. The seller sells the borrowed or agreed to be borrowed securities in anticipation of a decline in price of the relevant security. The benefit to the seller where the value of the security declines is the difference between the price at which the security is sold and the cost of repurchasing the borrowed security in order to return it to the person from whom it was borrowed. A synthetic short position allows a fund to achieve a similar economic outcome without short selling the physical securities.

Synthetic short selling may be achieved through the use of a variety of FDIs including contracts for differences, futures and options. Please refer to the section 'Use of Financial Derivative Instruments' for further details in relation to the risks attached to trading each of these FDIs.

Short Selling Regulations

Pursuant to the EU Short Selling Regulation 236/2012 (the "SSR"), information on net short positions, in shares admitted to trading on a trading venue in the EU (except where the principal trading venue of that instrument is outside the EU) or sovereign debt issued by a Member State or the EU, is required to be notified to the relevant competent authority as prescribed in the SSR and the delegated regulations adopted by the European Commission to supplement the SSR. In brief, under the SSR, a short position may be generated either by the short selling of physical shares or sovereign debt or by entering into a transaction relating to a financial instrument, other than shares or sovereign debt, where the effect is to confer a financial advantage on the person entering in to the transaction in the event of a decrease in the price or value of the relevant share or sovereign debt instrument. The term 'financial instrument' is defined by reference to Annex I, section C of the Markets in Financial Instruments Directive (MiFID) and includes transferable securities, money market instruments, units in collective investment schemes and a broad range of derivatives referencing various underlying investments. Accordingly, the SSR notification requirements cover net short positions created by the use of FDIs such as options, futures, index-related instruments, contracts for differences and spread bets relating to shares or sovereign debt.

The SSR and the delegated regulations set out the deadlines by which notifications of net short positions must be made to the relevant competent authority and the thresholds at which a notification requirement is triggered. The thresholds, in the case of shares, are set by reference to the value of the short position relative to the issued share capital of the issuer and, in the case of sovereign debt, by reference to the total amount of outstanding issued sovereign debt. Depending on the value of the short position, notifications may constitute private notifications to the relevant competent authority or public disclosure where information on net short positions notified will be available to the public. In order to comply with the SSR, where a Fund is engaging in synthetic shorting of shares or sovereign debt, the Company must be aware of the notification and disclosure obligations under the SSR. Failure to adhere to the notification and disclosure requirements under the SSR could result in losses to the Company.

Compliance with the SSR and the delegated regulations may represent a significant increase in the administrative burden on the Company in respect of Funds impacted by the SSR with inevitable adverse cost implications.

Hong Kong Political Risk

Hong Kong reverted to Chinese sovereignty on July 1, 1997 as a Special Administrative Region (SAR) of the People's Republic of China ("**PRC**") under the principle of "one country, two systems." Although PRC is obligated to maintain the current capitalist economic and social system of Hong Kong through June 30, 2047, the continuation of economic and social freedoms enjoyed in Hong Kong is dependent on the government of PRC. Any attempt by PRC to tighten its control over Hong Kong's political, economic, legal or social policies may result in an adverse effect on Hong Kong's markets. In addition, the Hong Kong dollar trades at a fixed exchange rate in relation to (or, is "pegged" to) the U.S. dollar, which has contributed to the growth and stability of the Hong Kong economy. However, it is uncertain how long the currency peg will continue or what effect the establishment of an alternative exchange rate system would have on the Hong Kong economy. Because a Fund's Net Asset Value may be denominated in U.S. dollars, the establishment of an alternative exchange rate system could result in a decline in the relevant Fund's Net Asset Value.

Risk of Investing in Hong Kong

A Fund's investment in Hong Kong issuers may subject the Fund to legal, regulatory, political, currency, security, and economic risk specific to Hong Kong. PRC is Hong Kong's largest trading partner, both in terms of exports and imports. Any changes in the Chinese economy, trade regulations or currency exchange rates may have an adverse impact on Hong Kong's economy.

The economy of Hong Kong is closely tied to the economy of PRC. The Chinese economy has grown rapidly during the past several years and there is no assurance that this growth rate will be maintained. PRC may experience substantial rates of inflation or economic recessions, causing a negative effect on the economy and securities market. Delays in enterprise restructuring, slow development of well-functioning financial markets and widespread corruption have also hindered performance of the Chinese economy, and PRC continues to receive substantial pressure from trading partners to liberalise official currency exchange rates.

Investment in the PRC

PRC Economic Risks

The economy in the PRC, which has been in a state of transition from a planned economy to a more market orientated economy, differs from the economies of most developed countries and investing in the PRC may be subject to greater risk of loss than investments in developed markets, due, among other factors, to greater market volatility, lower trading volume, political and economic instability, greater risk of market shut down, greater control of foreign exchange and greater limitations on foreign investment policy than those typically found in a developed market. There may be substantial government intervention in the PRC economy, including restrictions on investment in companies or industries deemed to be sensitive to relevant national interests. The PRC government and regulators may also intervene in the financial markets, such as by the imposition of trading restrictions, which may affect the trading of Chinese securities and ultimately may impact on the performance of a Fund. The Chinese companies in which the relevant Reference Index invests may be held to lower disclosure, corporate governance, accounting and reporting standards than companies in more developed markets. In addition, some of the securities held with the Reference Index may be subject to higher transaction and other costs, foreign ownership limits, the imposition of withholding or other taxes, or may have liquidity issues which make such securities more difficult to sell at reasonable prices. These factors ultimately may have an unpredictable impact on a Fund's investments and increase the volatility. Furthermore, market interventions may have a negative impact on market sentiment which may in turn affect the performance of the relevant Reference Index and therefore the performance of the Fund.

The PRC economy has experienced significant and rapid growth in the past 20 years. However, such growth may or may not continue, and may not apply evenly across different geographic locations and sectors of the PRC economy. Economic growth has also been accompanied by periods of high inflation. The PRC government has implemented various measures from time to time to control inflation and restrain the rate of economic growth of the PRC economy. Furthermore, the PRC government has carried out economic reforms to achieve decentralisation and utilisation of market forces to develop the economy of the PRC. These reforms have resulted in significant economic growth and social progress. There can, however, be no assurance that the PRC government will continue to pursue such economic policies or, if it does, that those policies will continue to be successful. Any such adjustment and modification of those economic policies may have an adverse impact on the securities markets in PRC and therefore ultimately on the performance of a Fund.

These factors may increase the volatility of a Fund and hence the risk of loss to the value of your investment.

PRC Political Risks

Any political changes, social instability and adverse diplomatic developments which may take place in or in relation to the PRC could result in the imposition of additional governmental restrictions including expropriation of assets, confiscatory taxes or nationalisation of some or all of the constituents of the relevant Reference Index. Investors should also note that any change in the policies of the government and relevant authorities of the PRC may adversely impact the securities markets in the PRC.

Legal System of the PRC

The PRC legal system is based on written statutes and their interpretation by the Supreme People's Court. Prior court decisions may be cited for reference but do not have precedent value. Since 1979, the PRC government has been developing a comprehensive system of commercial law, and progress has been made in introducing laws and regulations dealing with economic matters such as foreign investment, corporate organisation and governance, commerce, taxation and trade. However, because of the limited volume of published cases and judicial interpretation and their non-binding nature, the interpretation and enforcement of these regulations involves significant uncertainties.

Potential market volatility risk

Investors should note that the Shanghai Stock Exchange and Shenzhen Stock Exchange on which China A Shares are traded are undergoing development. Market volatility may result in significant fluctuation in the prices of securities traded on such markets, which could therefore ultimately impact upon the Net Asset Value of a Fund.

Accounting and Reporting Standards

PRC companies are required to follow PRC accounting standards and practices which follow international accounting standards to a certain extent. However, the accounting,

auditing and financial reporting standards and practices applicable to PRC companies may be less rigorous, and there may be significant differences between financial statements prepared in accordance with the PRC accounting standards and practice and those prepared in accordance with international accounting standards. For example, there are differences in the valuation methods of properties and assets and in the requirements for disclosure of information to investors.

Taxation in the PRC

A 10% PRC withholding tax is applicable to the payment of dividends and interest to foreign investors by PRC listed companies. From a technical perspective, withholding tax is also legally applicable to capital gains realised by foreign investors on the disposal of China A Shares under the Corporate Income Tax. However, before the release of its Caishui ("Circular") 79, the PRC tax authorities had not been reported to have sought in practice to collect such withholding tax on capital gains realised by foreign investors on the disposal of China A Shares.

Circulars 79 and 81 were released by the PRC tax authorities on 14 November 2014 in anticipation of the launch of the Stock Connect and in response to market concerns on the capital gains tax treatment of trading activity under this new programme and its comparability to the existing QFII/ RQFII regime. Under Circulars 79 and 81, capital gains realised on or after 17 November 2014 by foreign investors including QFIIs and RQFIIs from the disposal of China A Shares, including PRC "land-rich" companies (i.e. companies that have derived more than 50% of their value from immovable property located in the PRC at any time in the 3 years prior to the disposal), are temporarily exempt from Chinese taxation. The duration of this period of temporary exemption is not stated in either Circular 79 or 81 and is subject to termination by the PRC tax authorities with or without notice.

To the extent that a Fund uses swaps to achieve its investment objective, such instruments may include a provision whereby the Fund undertakes to keep the relevant counterparty fully or partially indemnified for any prospective or retrospective tax liability that it may suffer either directly or indirectly as a result of the relevant counterparty's direct or indirect holding of, interest in, or dealing in China A Shares for the purpose of hedging such swaps and/or other FDI.

If the relevant calculation agent determines, in its sole and absolute discretion, that the risk of the relevant counterparty suffering from such tax liability becomes material, it may at any time elect to make a downward adjustment to the valuation of the swap to reflect the expected amount of such tax liability. Any such adjustment of the valuation of the swap will remain in effect until such tax position of the relevant counterparty can be ascertained, at which point the calculation agent may make further upward or downward adjustments of the valuation of the swap accordingly. As a result, the attention of investors is drawn to the fact that the Net Asset Value of the Fund may be adversely impacted by any such adjustments to the valuation of the swap.

Other risks associated with the PRC

Risks of trading liquidity and other disruptions

China A Shares are traded on the Shanghai and/or Shenzhen Stock Exchanges.

Liquidity for China securities will be impacted by any temporary or permanent suspensions of particular securities imposed from time to time by the Shanghai and/or Shenzhen Stock Exchanges or as a result of any regulatory or governmental intervention in relation to particular investments or the markets. Any such suspension may impact on the Net Asset Value of a Fund and may increase the tracking error of a Fund and may expose the Fund to losses.

If a significant portion of the constituents of the relevant Reference Index are restricted or suspended, a Fund may, in the sole discretion of the Directors, suspend the determination of the Net Asset Value and the issue and redemption of Shares of the Fund in accordance with the section entitled "Suspension of Calculation of Net Asset Value" on page 65 of this Prospectus. In addition, where the last the closing or bid prices, as applicable, for suspended securities do not, in the opinion of the Manager, reflect their fair value or if prices are unavailable, the value will be calculated with care and in good faith by the Directors, or a competent person or firm appointed by the Directors and approved for that purpose by the Depositary, on the basis of the probable realisation value for such securities in accordance with the section titled "Valuation of Assets" at section 8.2 of this Prospectus. Any temporary suspension of the issue and redemption of Shares in the Fund may cause the Shares of a Fund to trade at a significant premium or discount to the Net Asset Value on any stock exchange on which they are admitted for trading.

Use of Financial Derivative Instruments:

General: The Fund may use derivative techniques to track the performance of the Reference Index. While the prudent use of such derivatives can be beneficial, derivatives also involve risks which, in certain cases, can be greater than the risks presented by investing directly in the constituents of the Reference Index or Reference Asset. There may be transaction costs associated with the use of derivatives.

Control and Monitoring: Derivative products are highly specialised instruments that require investment techniques and risk analysis different from those associated with equity and fixed income securities. The use of derivative techniques requires an understanding not only of the relevant Reference Index or Reference Asset but also of the derivative itself. In particular, the use and complexity of derivatives require the maintenance of adequate controls to monitor the transactions entered into, the ability to assess the risk that a derivative adds to a Fund and the ability to forecast the relative price, interest rate or currency rate movements correctly.

Liquidity Risk: Liquidity risk exists when a particular instrument is difficult to purchase or sell. If a derivative transaction is particularly large or if the relevant market is illiquid as is the case with many privately negotiated derivatives, it may not be possible to initiate a transaction or liquidate a position at an advantageous price, or at all.

Credit Risk and Counterparty Risk: The Company on behalf of a Fund may enter into transactions in "over-the-counter" markets, which will expose the Fund to the credit risk of its counterparties with whom they transact or place margins or Collateral in respect of such transactions and their ability to satisfy the terms of such contracts, and could result in the relevant Fund suffering a loss thereon. For example, the Company on behalf of the Fund may enter into securities lending arrangements and Repurchase Transactions, forward contracts, options and swap arrangements or other derivative techniques, each of which exposes the Fund to the risk that the counterparty may default on its obligations to perform under the relevant contract. In the event of the bankruptcy or insolvency of a counterparty, the Fund could experience delays in liquidating the position and significant losses, including declines in the value of its investment during the period in which the Company seeks to enforce its rights, inability to realise any gains on its investment during such period and fees and expenses incurred in enforcing its rights. There is also a possibility that the above agreements and derivative techniques are terminated due, for instance, to bankruptcy, supervening illegality or change in the tax or accounting laws relative to those at the time the agreement was originated. In such circumstances, investors may be unable to recover any losses incurred.

Other Risks: Other risks in using derivatives include the risk that there are differing valuations for such derivatives arising out of different permitted valuation methods. Many derivatives, in particular OTC derivatives, are complex and often valued subjectively and the valuation can only be provided by a limited number of market professionals which often

are acting as counterparties to the transaction to be valued. Inaccurate valuations can result in increased cash payment requirements to counterparties or a loss of value to a Fund. The value of any OTC derivatives shall be the value obtained from the Approved Counterparty or the Administrator and shall be valued daily. Such valuations will be approved or verified at least weekly by a party independent of the Approved Counterparty as determined by the Board of Directors and approved by the Depositary. Alternatively, the valuation methodology explained in section 8.2 could be used. Derivatives do not always perfectly or even highly correlate or replicate the value of the securities, rates or indices they are designed to replicate. Consequently, a Fund's use of derivative techniques may not always be an effective means of, and sometimes could be counterproductive to, implementing such Fund's investment objective.

Investors should note that where an OTC derivative is terminated and the Directors and the Investment Manager decide to invest in a new OTC derivative in its place, the terms of the new OTC derivative may be different and in some cases less favourable than the terms of the previous OTC derivative.

EPM Risk: The Company on behalf of a Fund may employ techniques and instruments relating to Transferable Securities, Money Market Instruments and/or other financial instruments in which it invests for efficient portfolio management purposes. Many of the risks attendant in utilising derivatives, as disclosed in the section entitled "Use of Financial Derivative Instruments" above, will be equally relevant when employing such efficient portfolio management techniques. In addition to the sub-section entitled "General", particular attention is drawn to the sub-sections entitled "Credit Risk and Counterparty Risk" and "Securities Lending Arrangements and Repurchase Transactions". Investors should also be aware that from time to time, a Fund may engage with Repurchase Transaction and other financial derivative contracts' counterparties and/or securities lending agents that are related parties to the Depositary or other service providers of the Company. Such engagement may on occasion cause a conflict of interest with the role of the Depositary or other service provider in respect of the Company. Please refer to section "Conflicts of Interest" for further details on the conditions applicable to any such related party transactions. The identity of any such related parties will be specifically identified in the Company's semi-annual and annual reports.

Absence of Regulation; Counterparty Risk: In general, there is less government regulation and supervision of transactions in the OTC markets (in which currencies, spot and option contracts, certain options on currencies and swaps are generally traded) than of transactions entered into on recognised exchanges (as referred to in the Prospectus). OTC derivatives lack transparency as they are privately negotiated contracts and any information concerning them is usually only available to the contracting parties. While measures are being introduced under EMIR that aim to mitigate risks involved in investing in OTC derivatives and improve transparency, these types of investments continue to present challenges in clearly understanding the nature and level of risks involved. In addition, many of the protections afforded to participants on some recognised exchanges, such as the performance guarantee of an exchange clearing house, might not be available in connection with OTC transactions.

OTC derivatives are generally not regulated. OTC derivatives are non-exchange traded option agreements, which are specifically tailored to the needs of an individual investor. These options enable the user to structure precisely the date, market level and amount of a given position. The counterparty for these derivatives will be the specific firm involved in the transaction rather than an exchange, and accordingly the bankruptcy or default of a counterparty with which the Company on behalf of a Fund trades OTC derivatives could result in substantial losses to that Fund. In addition, a counterparty may refrain from settling a transaction in accordance with its terms and conditions because the contract is not legally enforceable or because it does not accurately reflect the intention of the parties or because of a dispute over the terms of the contract (whether or not bona fide) or because of a credit or liquidity problem, thus causing that Fund to suffer a loss. To the extent that a

counterparty defaults on its obligation and that Fund is delayed or prevented from exercising its rights with respect to the investments in its portfolio, it may experience a decline in the value of its position, lose income and incur costs associated with asserting its rights. Regardless of the measures that Fund may implement to reduce counterparty credit risk, however, there can be no assurance that a counterparty will not default or that the Fund will not sustain losses on the transactions as a result. Counterparty exposure will be in accordance with that Fund's investment restrictions.

Risks related to SFTs, TRSs and collateral management

Any assets received by a Fund from a counterparty on a title transfer basis (whether in respect of an OTC derivatives transaction, SFT, TRS or otherwise) shall be held by the Depositary or a sub-custodian. For other types of collateral arrangements, the collateral may be held with a third party depositary which is unrelated to the collateral provider.

Assets provided by a Fund on a title transfer basis no longer belong to the Fund and pass outside the custodial network. Such counterparty may use those assets at its absolute discretion. Assets provided to a counterparty other than on a title transfer basis shall be held by the Depositary or a sub-custodian of the Depositary.

Pursuant to Regulation 24(4) of the Central Bank Regulations, the Company may not reuse any non-cash collateral it receives. Any cash collateral received by the Company may only be re-invested in accordance with the Central Bank Regulations (briefly in highly liquid government bonds, short term money market funds, deposits with qualifying credit institutions or reverse repurchase agreements with qualifying credit institutions, provided the Company is able to recall at any time the full amount of cash on an accrued basis.

Any profits and losses under SFTs and TRS will be for the account of the relevant Fund. SFTs and TRS may be subject to costs, which shall be at normal commercial rates, including fees and spreads payable to third parties unaffiliated to the Investment Manager or which may include entities affiliated to the Company/Investment Manager, and any such expenses will be borne by the relevant Fund.

6.2 Depositary Risk

If a Fund invests in assets that are financial instruments that can be held in custody ("Custody Assets"), the Depositary is required to perform full safekeeping functions and will be liable for any loss of such assets held in custody unless it can prove that the loss has arisen as a result of an external event beyond its reasonable control, the consequences of which would have been unavoidable despite all reasonable efforts to the contrary. In the event of such a loss (and the absence of proof of the loss being caused by such an external event), the Depositary is required to return identical assets to those lost or a corresponding amount to the Fund without undue delay.

If a Fund invests in assets that are not financial instruments that can be held in custody ("Non-Custody Assets"), the Depositary is only required to verify the Fund's ownership of such assets and to maintain a record of those assets which the Depositary is satisfied that the Fund holds ownership of. In the event of any loss of such assets, the Depositary will only be liable to the extent the loss has occurred due to its negligent or intentional failure to properly fulfil its obligations pursuant to the Depositary Agreement.

As it is likely that the Funds may each invest in both Custody Assets and Non-Custody Assets, it should be noted that the safekeeping functions of the Depositary in relation to the respective categories of assets and the corresponding standard of liability of the Depositary applicable to such functions differs significantly.

The Funds enjoy a strong level of protection in terms of Depositary liability for the safekeeping of Custody Assets. However, the level of protection for Non-Custody Assets

is significantly lower. Accordingly, the greater the proportion of a Fund invested in categories of Non-Custody Assets, the greater the risk that any loss of such assets that may occur may not be recoverable. While it will be determined on a case-by-case whether a specific investment by the Fund is a Custody Asset or a Non-Custody Asset, generally it should be noted that derivatives traded by a Fund over-the-counter will be Non-Custody Assets. There may also be other asset types that a Fund invests in from time to time that would be treated similarly. Given the framework of Depositary liability under UCITS V, these Non-Custody Assets, from a safekeeping perspective, expose the Fund to a greater degree of risk than Custody Assets, such as publicly traded equities and bonds.

6.3 Inaction by the CSDs and/or an ICSD

Investors that settle or clear through an ICSD will not be a registered Shareholder in the Company, they will hold an indirect beneficial interest in such Shares and the rights of such investors, where such person is an ICSD Participant, shall be governed by the terms and conditions applicable to the arrangement between such ICSD Participant and their ICSD and where the holder of the indirect beneficial interests in the Shares is not an ICSD Participant, shall be governed by their arrangement with their respective nominee, broker or CSD (as appropriate) which may be an ICSD Participant or have an arrangement with an ICSD Participant. The Company will issue any notices and associated documentation to the registered holder of the Shares i.e. the ICSD, with such notice as is given by the Company in the ordinary course when convening general meetings. The ICSD has a contractual obligation to relay any such notices to its ICSD Participants in accordance with its rules and procedures. The ICSD is contractually bound to collate all votes received from the ICSD Participants and is obligated to vote in accordance with such instructions. The Company has no power to ensure the ICSD relays notices of votes in accordance with the ICSD Participant's instructions. The Company cannot accept voting instructions from any persons other than the ICSD.

6.4 Payments

Any dividends declared and any liquidation and mandatory redemption proceeds are paid by the Company or its authorised agent (for example, the Paying Agent) to the ICSD. Investors, where they are ICSD Participants, must look solely to the ICSD for their share of each dividend payment or any liquidation or mandatory redemption proceeds paid by the Company or, where they are not ICSD Participants, they must look to their respective nominee, broker or CSD (as appropriate, which may be an ICSD Participant or have an arrangement with an ICSD Participant) for any share of each dividend payment or any liquidation or mandatory redemption proceeds paid by the Company that relates to their investment.

Investors shall have no claim directly against the Company in respect of dividend payments and any liquidation and mandatory redemption proceeds due on Shares and the obligations of the Company will be discharged by payment to the ICSD.

6.5 Secondary Market Trading Risk

Even though the Shares of certain Funds (not to include the Unlisted Funds) are to be listed on one or more Relevant Stock Exchanges, there can be no certainty that there will be liquidity in the Shares on any Relevant Stock Exchange or that the market price at which such Shares may be traded on a Relevant Stock Exchange will be the same as or approximately equal to the Net Asset Value per Share. However, as such Shares may be dealt in by means of subscription and redemption, the Directors consider that large discounts or premiums in the Net Asset Value of such a Fund would not be sustainable. There can be no guarantee that once such Shares are listed on a Relevant Stock Exchange they will remain listed or that the conditions of listing will not change. Trading in such Shares on a Relevant Stock Exchange, or the requirement for market makers to make two way prices on a Relevant Stock Exchange, may be halted or suspended due to market conditions, or because the Relevant Stock Exchange considers that trading in such Shares is inadvisable, or the discontinuance in the calculation or publication of the Reference Index or Reference Asset or a component thereof, or otherwise pursuant to the Relevant Stock Exchange's rules. If trading on a Relevant Stock Exchange is halted or suspended, investors in such Shares may not be able to sell their Shares until trading resumes however such investors should be able to apply to the Company to redeem such Shares in accordance with the provisions set out below.

6.6 Taxation

Investors in the Shares should be aware that they may suffer income tax, withholding tax, capital gains tax, wealth tax, stamp taxes or any other kind of tax on distributions or deemed distributions of the Fund, capital gains within the Fund whether realised or unrealised, income received or accrued or deemed received within the Fund etc., subject to the laws and practices of the country where the Shares are purchased, sold, held or redeemed and subject to the country of tax residence or nationality of the Shareholder.

Investors should be aware of the fact that taxes may be calculated based on income received and/or deemed to be received and/or accrued in the Fund in relation to the assets of a Fund, whereas the performance of the Fund, and subsequently the return investors receive after redemption of the Shares, might partially or fully depend on the performance of the Reference Index or Reference Asset.

Investors who are in any doubt as to their tax position should consult their own independent tax advisors. In addition, investors should be aware that tax regulations and their application or interpretation by the relevant taxation authorities' change from time to time. Accordingly, it is not possible to predict the precise tax treatment, which will apply at any given time.

6.7 FATCA

The United States and Ireland have entered into an inter-governmental agreement to implement FATCA (the "**IGA**"). Under the IGA, an entity classified as a Foreign Financial Institution (an "**FFI**") that is treated as resident in Ireland is expected to provide the Revenue Commissioners with certain information in respect of its "account holders" (i.e. Shareholders). The IGA further provides for the automatic reporting and exchange of information in relation to accounts held in Irish FFIs by U.S. Persons, and the reciprocal exchange of information regarding U.S. financial accounts held by Irish residents. Provided the Company complies with the requirements of the IGA and the Irish legislation, it should not be subject to FATCA withholding on any payments it receives and may not be required to impose FATCA withholding on payments which it makes.

Although the Company will attempt to satisfy any obligations imposed on it to avoid the imposition of the FATCA withholding tax, no assurance can be given that the Company will be able to satisfy these obligations. In order to satisfy its FATCA obligations, the Company will require certain information from investors in respect of their FATCA status. If the Company becomes subject to a withholding tax as a result of the FATCA regime, the value of the Shares held by all Shareholders may be materially affected.

All prospective investors and Shareholders should consult with their own tax advisors regarding the possible implications of FATCA on an investment in the Company and the Funds.

6.8 CRS

Ireland has provided for the implementation of CRS through Section 891F of the TCA and the enactment of the Returns of Certain Information by Reporting Financial Institutions Regulations 2015 (the "**CRS Regulations**").

The CRS is a global OECD tax information exchange initiative which is aimed at encouraging a coordinated approach to disclosure of income earned by individuals and organisations.

The Company is a Reporting Financial Institution for CRS purposes and will be required to comply with the Irish CRS obligations. In order to satisfy its CRS obligations, the Company will require its investors to provide certain information in respect of their tax residence and may, in some cases, require information in relation to the tax residence of the beneficial owners of the investor. The Company, or a person appointed by the Company, will report the information required to the Revenue Commissioners by 30 June in the year following the year of assessment for which a return is due. The Revenue Commissioners will share the appropriate information with the relevant tax authorities in participating jurisdictions.

All prospective investors / shareholders should consult with their own tax advisors regarding the possible CRS implications of an investment in the Company.

6.9 DAC6 – Disclosure requirements for reportable cross-border tax arrangements

On 25 June 2018, Council Directive (EU) 2018/822 ("**DAC6**") introduced rules regarding the mandatory automatic exchange of information in the field of taxation in relation to reportable cross-border arrangements ("**RCBAs**"). DAC6 is intended to provide the tax authorities of EU member states with information about potentially aggressive tax-planning arrangements. One of the aims of DAC 6 is that such information will enable the authorities to react promptly against harmful tax practices and close loopholes by enacting legislation or by undertaking adequate risk assessments and carrying out tax audits. DAC 6 is capable of applying to arrangements that do not necessarily constitute aggressive tax planning.

The DAC6 obligations apply from 1 July 2020, but also require reporting of arrangements implemented between 25 June 2018 and 30 June 2020. DAC6 generally requires EU intermediaries to report to their local tax authorities information about RCBAs, including details of the arrangement as well as identification information about the involved intermediaries and relevant taxpayers, i.e. the persons to whom the RCBA is made available. Subsequently, the local tax authorities exchange the information with the tax authorities of other EU member states. As such, intermediaries engaged by the Company, or in certain cases the Company itself, may be legally required to report information on any RCBA entered into by the Company to the relevant tax authorities.

6.10 Section 871(m)

Section 871(m) of the US Tax Code requires withholding (up to 30%, depending on whether a treaty applies) on certain financial instruments to the extent that the payments or deemed payments on the financial instruments are contingent upon or determined by reference to U.S.-source dividends. Under U.S. Treasury Department regulations, certain payments or deemed payments to the Company with respect to certain equity-linked instruments that reference U.S. stocks may be treated as dividend equivalents that are subject to U.S. withholding tax at a rate of 30% (or lower treaty rate). Under these regulations, withholding may be required even in the absence of any actual dividend-related payment or adjustment made pursuant to the terms of the instrument. If the Company becomes subject to a withholding tax as a result of 871(m), the value of the Shares held by the Shareholders may be materially affected. All prospective investors/Shareholders should consult with their own

tax advisors regarding the possible implications of 871(m) on an investment in the Company.

6.11 Consequences of winding-up proceedings

If the Company fails for any reason to meet its obligations or liabilities, or is unable to pay its debts, a creditor may be entitled to make an application for the winding-up of the Company. The commencement of such proceedings may entitle creditors (including Approved Counterparties) to terminate contracts with the Company (including FDIs) and claim damages for any loss arising from such early termination. The commencement of such proceedings may result in the Company being dissolved at a time and its assets (including the assets of all Funds) being realised and applied to pay the fees and expenses of the appointed liquidator or other insolvency officer, then in satisfaction of debts preferred by law and then in payment of the Company's liabilities, before any surplus is distributed to the Shareholders of the Company. In the event of proceedings being commenced, the Company may not be able to pay the full amounts anticipated by the Supplement in respect of any Fund.

6.12 Potential Conflicts of Interest

The Promoter, the Directors, the Manager, the Investment Manager, the Depositary, the Administrator, the Registrar, the Index Provider, the Distributor, any Approved Counterparty, the Calculation Agent under any FDI, any counterparty to securities lending arrangements and Repurchase Transactions, any Shareholder, any Authorised Participant or market maker which has been appointed to offer prices for the Shares on any Relevant Stock Exchange on which the Classes to which the Shares belong are listed (for the purposes hereof, a "**Market Maker**") and any of their respective subsidiaries, Affiliates, associates, agents or delegates (for the purposes hereof, "**Connected Persons**" and each a "**Connected Person**") may undertake activities which may give rise to potential conflicts of interest, including but not limited to:

- (a) enter into any financial, banking or other transactions or arrangements with one another or with the Company including, without limitation, investment by the Company in securities of any Connected Person, or investment by any Connected Person in any company or body any of whose investments form part of the assets of the Company, or be interested in any such contracts or transactions;
- (b) invest in and deal in Shares relating to any Fund, and shares, securities, assets or any property of the kind included in the property of the Company for their respective individual accounts or for the account of a third party; and
- (c) deal as agent, principal or counterparty in the sale or purchase of securities and other investments (including foreign exchange, FDI and securities lending and Repurchase Transactions) to or from the Company.

Any assets of the Fund in the form of cash or securities may be deposited, subject to the provisions of the Central Bank Acts, 1942 to 1998, of Ireland as amended by the Central Bank and Financial Services Regulatory Authority of Ireland Acts, 2003 to 2004, with any Connected Person. Any assets of the Fund in the form of cash may be invested in certificates of deposit or banking investments issued by any Connected Person. Banking or similar transactions may also be undertaken with or through a Connected Person.

Further details of any risk factors which are applicable to a particular Fund are set out in the relevant Supplement. The risk factors set out in this Prospectus do not purport to be an exhaustive or complete explanation of all the risks. Investors should seek professional advice before investing.

6.13 Sustainability Risks

Sustainability risks means an environmental, social or governance event or condition that, if it occurs, could cause an actual or a potential material negative impact on the value of the investment. If a sustainability risk associated with an investment materialises, it could lead to the loss in value of an investment.

Sustainability risks may vary from investment to investment and could include, but are not limited to, risks of environmental damage, social risks (including public health, safety and human rights violations and exploitation), governance risks (inadequate oversight and internal governance of the companies, including management and board structure, compensation and approach to anti-bribery and anti-corruption), litigation risks linked to ESG issues, as well as the risk of political and regulatory changes on investments related to each of the foregoing.

In addition, sustainability risks can either represent a risk of their own or have an impact on other risks and may contribute significantly to risks such as market risks, operational risks, liquidity risks or counterparty risks.

7 Conflicts of Interest

Any Connected Person may also deal as agent, principal or counterparty in the sale or purchase of securities and other investments (including foreign exchange, FDI and securities lending arrangements and Repurchase Transactions) to or from the relevant Fund. There will be no obligation on the part of any Connected Person to account to the relevant Fund or to Shareholders of that Fund for any benefits so arising, and any such benefits may be retained by the relevant party, provided that such transactions are carried out as if effected on normal commercial terms negotiated at arm's length, are consistent with the best interests of the Shareholders of that Fund and:

- (a) a certified valuation of such transaction by a person approved by the Depositary (or in the case of any such transaction entered into by the Depositary, the Directors) as independent and competent has been obtained; or
- (b) such transaction has been executed on best terms reasonably obtainable on an organised investment exchange under its rules; or
- (c) where (a) and (b) are not reasonably practical, such transaction has been executed on terms which the Depositary is (or in the case of any such transaction entered into by the Depositary, the Directors are) satisfied conform with the principle that such transactions be carried out as if effected on normal commercial terms negotiated at arm's length and consistent with the best interests of the Shareholders.

In addition, in many cases each Approved Counterparty may be required to provide valuations of financial derivative instruments entered into between the relevant Fund and the Approved Counterparty, whether as Calculation Agent or otherwise. These daily valuations will form the basis upon which the value of certain assets of a Fund is calculated. The Directors acknowledge that each Approved Counterparty or the relevant Affiliates may have a potential conflict of interest by virtue of acting as the Approved Counterparty and/or providing such valuations (including in the case of an Affiliate providing an independent verification of the valuation provided by an Approved Counterparty under 8.2 in respect of OTC instruments). However, the Directors believe that such conflicts can be adequately managed, and expect that each Approved Counterparty or one of its Affiliates will be suitable and competent to provide such valuations and will do so at no further cost to the relevant Fund than would be the case if the services of a third party were engaged to provide valuations. In addition the valuation provided by the Approved Counterparty will be verified by an entity independent of the Approved Counterparty will be verified by an entity independent of the Approved Counterparty as described above. Alternatively, the valuation methodology explained in 8.2 could be used.

Members of the UBS AG Group may act as Approved Counterparty and Calculation Agent to the swaps and other FDIs to which the Company is a party, Authorised Participant, Index Provider, counterparty to any securities lending arrangements or Repurchase Transactions, Market Maker and/or the Depositary to the Company all in accordance with the relevant agreements which are in place. The Directors acknowledge that, by virtue of carrying-out these functions in connection with the Company, potential conflicts of interest are likely to arise. In such circumstances, each of the foregoing will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that such members of the UBS AG Group are suitable and competent to perform such functions.

The Depositary (or in the case of a transaction involving the Depositary, the Directors) shall document how it complied with paragraphs (a), (b) and (c) above and where transactions are conducted in accordance with paragraph (c), the Depositary (or in the case of a transaction involving the Depositary, the Directors), must document the rationale for being satisfied that the transaction conformed to the principles outlined above.

Potential conflicts of interest may arise from time to time from the provision by the Depositary and/or its affiliates of other services to the Company and/or other parties. For example, the Depositary and/or its affiliates may act as the depositary, trustee, depositary and/or administrator of other funds. It is therefore possible that the Depositary (or any of its affiliates) may in the course of its business have conflicts or potential conflicts of interest with those of the Company and/or other funds for which the Depositary (or any of its affiliates) act.

Where a conflict or potential conflict of interest arises, the Depositary will have regard to its obligations to the Company and will treat the Company and the other funds for which it acts fairly and such that, so far as is practicable, any transactions are effected on terms which are not materially less favourable to the Company than if the conflict or potential conflict had not existed.

Unless otherwise provided for in the Supplement for a Fund, an Approved Counterparty shall not have any discretion over the composition or management of the relevant Fund's investment portfolio or over the underlyings of the FDIs.

As the fees of the Manager are based on the Net Asset Value of a Fund, if the Net Asset Value of the Fund increases so do the fees payable to the Manager and accordingly there is a conflict of interest for the Manager in cases where the Manager or its delegate is responsible for determining the valuation price of a Fund's investments.

The Directors may act as directors of other collective investment vehicles. Where any potential conflicts of interest arise between their duties to the Company and to third parties, the Directors will endeavour to ensure that any such conflicts will not unfairly prejudice the Company.

Each Connected Party will provide the Company with relevant details of each transaction (including the name of the party involved and where relevant, fees paid to that party in connection with the transaction) in order to facilitate the Company discharging its obligation to provide the Central Bank with a statement within the relevant Fund's annual and semi-annual reports in respect of all Connected Party transactions.

The preceding list of potential conflicts of interest does not purport to be a complete enumeration or explanation of all of the conflicts of interest that may be involved in an investment in the Company.

8 Issue and Redemption Prices/Calculation and Publication of Net Asset Value/Valuation of Assets

8.1 Issue and Redemption Prices/Calculation of Net Asset Value

The Initial Issue Price for Shares of each Fund shall be the amount(s) set out in the Supplement for the relevant Fund.

The price at which Shares of any Fund will be issued on a Dealing Day, after the initial issue, is calculated by ascertaining the Net Asset Value of the relevant Fund (i.e. the value of the assets of the Fund having deducted the liabilities of the Fund therefrom) as at the Valuation Point for that Fund for the relevant Dealing Day. The Net Asset Value per Share of the relevant Fund is calculated by dividing the Net Asset Value of the relevant Fund by the total number of Shares in issue in the Fund at the relevant Valuation Point and rounding the result to up to four decimal places. Where applicable, the Net Asset Value per Share of each Class in a Fund is calculated by determining that portion of the Net Asset Value of the Fund which is attributable to the relevant Class and by dividing this sum by the total number of Shares of the relevant Class in issue at the relevant Valuation Point and rounding the resulting amount to up to four decimal places. If a Fund has more than one Class of Share, additional fees may be charged against certain Classes, and details of such fees will be set forth in the Supplement for the relevant Fund. This may result in the Net Asset Value per Share of each Class being different. The Valuation Point for each Fund is set out in the Supplement for the relevant Fund.

The price at which Shares will be issued on a Dealing Day is, subject as hereinafter provided, the Net Asset Value per Share of the relevant Class which is calculated in the manner described above. The Company may, in calculating such price, add to the Net Asset Value per Share of the relevant Class, for its own account, a charge ("**Subscription Fee**") which is sufficient to cover stamp duties, Transfer Taxes (if applicable) or other taxation (if any), fiscal and purchase charges (including but not limited to spreads, transaction costs and any transaction-driven custody charges) in respect of the issue of Shares in the relevant Fund (the "**Issue Price**"). Applicants may also be charged Primary Market Transaction Costs as may be specified in the relevant Supplement.

The price at which Shares will be redeemed on a Dealing Day is, subject as hereinafter provided, the Net Asset Value per Share of the relevant Class which is calculated in the manner described above. The Company may, in calculating such price, deduct from the Net Asset Value per Share of the relevant Class, for its own account, a charge ("**Redemption Fee**") in respect of any fiscal and sales charges (including but not limited to spreads, transaction costs and any transaction-driven custody charges) or Transfer Taxes (if applicable) in respect of the redemption of Shares in the relevant Fund (the "**Redemption Price**"). Applicants may also be charged Primary Market Transaction Costs as may be specified in the relevant Supplement.

In respect of the Primary Market Transaction Costs, the Directors or the Investment Manager as approved by the Directors may adjust the Issue Price and/or the Redemption Price by adding to or deducting from (as appropriate) the Issue Price per Share and/or the Redemption Price per Share such costs, for retention as part of the assets of the Fund. The Primary Market Transaction Costs will include dealing costs to preserve the value of the assets of the Fund.

8.2 Valuation of Assets

The Articles provide for the method of valuation of the assets and liabilities of each Fund which will be valued at the Valuation Point as follows.

- 8.2.1 Assets listed or traded on a recognised exchange for which market quotations are readily available shall be valued at the closing or last known market price which for the purposes of the Company shall be understood to mean the closing mid-market price. Where a security is listed or dealt in on more than one recognised exchange, the relevant exchange or market shall be the principal or main stock exchange or market on which the security is listed or dealt on or the exchange or market which the Directors and/or the Manager determine provides the fairest criteria in determining a value for the relevant investment.
- 8.2.2 The value of any investment which is not quoted, listed or dealt in on a Market, or of any investment which is normally quoted, listed or dealt in on a Market but in respect of which the closing price is currently unavailable or the current price of which does not in the opinion of the Directors and/or the Manager represent fair market value, shall be the probable realisation value thereof estimated with care and in good faith by (i) the Directors and/or the Manager or (ii) a competent person, firm or corporation (including the Investment Manager) selected by the Directors and/or the Manager and approved for such purpose by the Depositary or (iii) any other means provided that the value is approved by the Depositary. In determining the probable realisation value of any such investment, a certified valuation thereof provided by a competent independent person or in the absence of any independent person, the Investment Manager, who in each case shall have been approved for such purposes by the Depositary, shall be sufficient. Where reliable market quotations are not available for fixed income securities, the value of such securities may be determined using matrix methodology compiled by the Directors and/or the Manager or competent person (as approved by the Depositary) whereby such securities are valued by reference to the valuation of other securities which are comparable in rating, yield, due date and other characteristics.
- 8.2.3 The value of any investment listed on a Market but acquired or traded at a premium or at a discount outside the relevant Market may be valued taking into account the level of premium or discount as at the date of valuation of the investment and the Depositary must ensure the adoption of such a procedure is justifiable in the context of establishing the probable realisation value for the security.
- 8.2.4 Cash (in hand or deposit) and other liquid assets will be valued at their face value plus interest accrued, where applicable as at the Valuation Point.
- 8.2.5 Exchange-traded derivative instruments will be valued based on the settlement price as determined by the market where the instrument is traded. If such settlement price is not available, such value shall be calculated in accordance with 8.2.2 above.
- 8.2.6 The value of any prepaid expenses, cash dividends and interest declared or accrued and not yet received as at a Valuation Point shall be deemed to be the face value thereof, unless in any case the Directors and/or the Manager are of the opinion that the same is unlikely to be paid or received in full in which case the value thereof shall be arrived at after making such discount as the Directors and/or the Manager may consider appropriate in such case to reflect the true value thereof as at any Valuation Point.
- 8.2.7 The value of any demand notes, promissory notes and accounts receivable shall be deemed to be the face value or full amount thereof after making such discount

as the Directors and/or the Manager may consider appropriate to reflect the true current value thereof as at any Valuation Point.

- 8.2.8 Certificates of deposit, treasury bills, bank acceptances, trade bills and other negotiable instruments shall be valued at the closing price, or if unavailable, the last known market price for such certificates of deposit, treasury bills, bank acceptances, trade bills and other negotiable instruments.
- 8.2.9 Forward foreign exchange contracts shall be valued by reference to freely available market quotations as at the Valuation Point or if unavailable in accordance with the provisions for off-exchange derivative contracts below.
- 8.2.10 OTC FDI will be valued either using the counterparty's valuation or an alternative valuation provided by the Company or by an independent pricing vendor appointed by the Directors and/or the Manager and approved for this purpose by the Depositary. OTC FDI shall be valued at least daily. If using the counterparty's valuation, such valuation must be approved or verified by a party independent of the counterparty and approved by the Depositary (which may include the Company or a party related to the OTC counterparty provided that it is an independent unit within the same group and which does not rely on the same pricing models employed by the counterparty) on a weekly basis. In the event that the Company opts to use an alternative valuation, the Company will use a competent person appointed by the Directors and/or the Manager, approved for this purpose by the Depositary, or will use a valuation by any other means provided that the value is approved by the Depositary. All alternative valuations will be reconciled with the counterparty's valuation on at least a monthly basis. Any significant differences to the counterparty valuation will be promptly investigated and explained.
- 8.2.11 The valuation of units or shares or other similar participations in any collective investment scheme which provides for the units or shares or other similar participations therein to be redeemed at the option of the holder out of the assets of that undertaking shall be the last available net asset value per unit or share or other relevant participation as published by the scheme as at the relevant Valuation Point, or (if bid and offer prices are published), at the last bid price.

Notwithstanding the provisions of paragraphs above:-

- (i) The Directors and/or the Manager or their delegate shall, at their discretion in relation to any particular Fund which is a short-term money market fund, have in place an escalation procedure to ensure that any material discrepancy between the market value and the amortised cost value of a money market instrument is brought to the attention of the Investment Manager or a review of the amortised cost valuation vis-à-vis market valuation will be carried out in accordance with the requirements of the Central Bank.
- (ii) Where it is not the intention or objective of the Directors and/or the Manager to apply amortised cost valuation to the portfolio of the Fund as a whole, a money market instrument within such a portfolio shall only be valued on an amortised basis if the money market instrument has a residual maturity of less than 3 months and does not have any specific sensitivity to market parameters, including credit risk.

If the Directors and/or the Manager deem it necessary, a specific investment may be valued under an alternative method of valuation approved by the Depositary and the rationale/methodologies used must be clearly documented.

The foregoing valuation principles are subject to any prevailing rules that may apply to how the Company is required to value particular instruments as may be contained in EMIR.

Any value expressed otherwise than in the Base Currency of the relevant Fund (whether of any investment or cash) and any non-Base Currency borrowing shall be converted into the Base Currency at the rate which the Administrator deems appropriate in the circumstances.

Notwithstanding the generality of the foregoing, the Directors and/or the Manager may with the approval of the Depositary adjust the value of any investment if taking into account currency, marketability and/or such other considerations as they may deem relevant, such as, applicable rate of interest, anticipated rate of dividend, maturity or liquidity, they consider that such adjustment is required to reflect the fair value thereof. The rationale for adjusting the value must be clearly documented.

8.3 Suspension of Calculation of Net Asset Value

The Directors may at any time temporarily suspend the calculation of the Net Asset Value of any Fund and the subscription, redemption or exchange of Shares and the payment of redemption proceeds of any Class during (i) any period when any of the principal Markets on which any of the direct investments, or indirect investments (e.g. the components of a Reference Index or Reference Asset), of the relevant Fund are quoted is closed, otherwise than for ordinary holidays, or during which dealings therein are restricted or suspended; (ii) any period when, as a result of political, economic, military or monetary events or any circumstances outside the control, responsibility and power of the Directors, disposal or valuation of the investments of the relevant Fund is not reasonably practicable without this being seriously detrimental to the interests of Shareholders of the relevant Fund or if, in the opinion of the Directors, the Net Asset Value of the Fund cannot fairly be calculated; (iii) any breakdown in the means of communication normally employed in determining the price of any of the relevant Fund's investments and other assets, or when, for any other reason the current prices on any Market of any assets of the relevant Fund cannot be promptly and accurately ascertained; (iv) any period during which the Company is unable to repatriate funds required for the purpose of making payments due on redemption of Shares of any Class in the relevant Fund or during which the transfer of funds involved in the realisation or acquisition of investments or payments due on redemption of Shares cannot, in the opinion of the Directors, be effected at normal prices or rates of exchange; (v) any period where in the opinion of the Directors such suspension is justified having regard to the interests of the Company and/or the relevant Fund; (vi) following the circulation to the relevant Shareholders of a notice of a general meeting at which a resolution proposing to wind-up the Company or terminate the relevant Fund (or relevant Class of Shares) is to be considered; or (vii) upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (in accordance with section 4.4).

The Company will, whenever possible, take all reasonable steps to bring any period of suspension to an end as soon as possible.

Shareholders who have requested the issue or redemption of Shares of any Class or exchanges of Shares of one Fund to another will be notified of any such suspension in such manner as may be directed by the Directors and their requests will be dealt with on the first Dealing Day after the suspension is lifted. Any such suspension shall be notified immediately, and in any event within the same Business Day to the Central Bank, to the competent authorities in any jurisdiction where the Company (and its relevant Funds) are registered for sale and to the Relevant Stock Exchanges (if any) where the Shares of the relevant Fund are listed. Details of any such suspension will also be notified to all Shareholders and will be published in a newspaper circulating in an appropriate jurisdiction (or such other publications as the Directors may determine) if the Directors consider it is likely to exceed 14 days.

Where a Fund temporarily suspends the redemption of Shares the Company shall notify the Central Bank immediately upon the lifting of the temporary suspension. In circumstances where the temporary suspension has not been lifted within 21 Business Days, the Company shall provide the Central Bank with an update on the temporary suspension upon the expiration of the 21 Business Day period and each subsequent 21 Business Day period where the temporary suspension continues to apply.

8.4 Publication of the Net Asset Value

The Net Asset Value per Share of each Class of Shares within each Fund expressed in the Base Currency and, as the case may be, translated into other currencies as specified in the relevant Supplement, and any dividend declaration, will be available on the Website each Business Day and at the office of the Administrator. The Company will notify the Relevant Stock Exchanges where the Shares are listed, as appropriate. The Company will not accept any responsibility for any error or delay in publication or for non-publication of prices which are beyond its control. Such Net Asset Values per Share (and translations into other currencies as appropriate) may also be available on the Website. The access to such publication on the Website may be restricted and is not to be considered as an invitation to subscribe for, purchase, convert, sell or redeem Shares.

9 Share Dealing

9.1 General Information in respect of Applications for Shares

The Directors may restrict or prevent the ownership of Shares by (i) any person, firm or corporate body, if in the opinion of the Directors such holding may be detrimental to the Company, if it may result in the Company incurring any liability to taxation or suffering other pecuniary, regulatory, legal or material administrative disadvantages or being in breach of any law or regulation which the Company might not otherwise have incurred, suffered or breached; or (ii) any person who appears to be in breach of any law or requirement of any country or governmental authority or by virtue of which such person is not qualified to hold such Shares; or (iii) any United States Person; or (iv) any individual under the age of 18 (or such other age as the Directors think fit) or of unsound mind (such persons, firms or corporate bodies to be determined by the Directors being herein referred to as "**Prohibited Persons**").

As with other Irish companies limited by shares, the Company is required to maintain a register of Shareholders. The Directors have resolved that Shares in the Funds will be issued in dematerialised (or uncertificated) form and that the Funds will apply for admission for clearing and settlement through a clearing system. As the Company is an Irish company, the operation of a clearing system in respect of these Shares is governed by the Companies Act.

- 9.2 Subscription for Shares
- 9.2.1 Application for Shares

Save as set out in the relevant Supplement, the following provisions apply in respect of Subscription for Shares of each of the Funds, other than the Unlisted Funds:

Investors can subscribe for their Shares for cash on the relevant Dealing Day, except during any period in which the calculation of the Net Asset Value is suspended. It is also possible for investors to buy their Shares on the Secondary Market, as set out in section 10 below. The details on the specific cash subscription procedures are set out below under the heading "Cash Subscriptions and Redemptions". The amount of any Primary Market Transaction Costs payable by the applicant will be set out in the relevant Supplement.

It is envisaged by the Directors and by the Manager that investors will generally buy and sell their Shares through the Secondary Market (as set out in section 10 below, and other than for the Unlisted Funds), given the nature of the Funds of the Company and the terms and conditions relating to the subscription and redemption of Shares other than on the Secondary Market.

Initial applications for Shares must be made by applicants in writing to the Company care of the Administrator using an Application Form. For the avoidance of doubt, the completed Application Form must first be provided to the Company care of the Administrator in electronic form or by fax in order for the Administrator to open an account in the name of the applicant(s), with the original of same to be provided as soon as reasonably practicable, where applicable. Measures aimed at the prevention of money laundering and terrorist financing require an applicant to provide verification of identity to the Company and/or the Administrator. Details of these requirements are set out in the Application Form and, for the Avoidance of doubt, may require that certain original documentation, as will be set out in the Application Form, is provided to the Administrator as soon as reasonably practicable in respect of an account being opened in the name of the applicant(s).

Application Forms may be obtained from the Company or the Administrator. Joint applicants must each sign the Application Form unless an acceptable power of attorney or other written authority is provided. Shares will not be issued until such time as the Administrator has opened an account in the name of the applicant(s); i.e. it has received and is satisfied with all the information and documentation required to verify the identity of the applicant(s). This may result in Shares being issued on a Subscription Date subsequent to the Subscription Date on which an applicant initially wished to have Shares issued to the applicant(s). It is further acknowledged that the Administrator shall be held harmless by the applicant against any loss arising as a result of the failure to process the subscription if such information as has been requested by the Administrator may refuse to process a redemption request until relevant information has been provided including but not limited to any relevant anti-money laundering / counter-terrorist financing documentation.

Existing Shareholders may subscribe for further Shares by submitting an Application Form either in writing, using an Application Form obtainable from the Company or the Administrator, or by fax, provided that all ongoing anti-money laundering/counter-terrorist financing and client identification requirements are complete.

The Directors have also decided that initial and subsequent subscription applications may be made by electronic or other means (provided that (i) a duly completed Application Form is received by the Administrator for account opening purposes and (ii) such electronic or other means are in accordance with the requirements of the Central Bank and previously agreed with the Administrator).

Applications for initial subscriptions will be accepted at the Initial Issue Price plus Primary Market Transaction Costs and/or Subscription Fee (as applicable). Subsequent subscriptions will be accepted at a price corresponding to the Net Asset Value per Share as determined as at the Valuation Point for the relevant Dealing Day, plus Primary Market Transaction Costs and/or Subscription Fee (as applicable).

Any subscription orders received after the Dealing Deadline for a particular Dealing Day will be processed on the following Dealing Day unless the Directors in their absolute discretion (and such discretion only to be exercised in exceptional circumstances) otherwise determine to accept one or more orders received after the Dealing Deadline for processing on that Dealing Day provided that such application(s) have been received prior to the Valuation Point for that particular Dealing Day.

The Net Asset Value per Share of each Class in each Fund will be published in its respective Base Currency and, as the case may be, translated into other currencies as specified in the relevant Supplement, as set out in section 8.4.

9.2.2 Fractions

Unless otherwise set out in the Supplement, fractions of Shares shall not be issued. Subscription monies representing fractions of Shares will not be returned to the applicant but will be retained as part of the assets of the relevant Fund and accordingly available to Shareholders of the Fund on a pro rata basis on each of the Shareholders holding of Shares.

9.2.3 Restrictions on Subscriptions

The Directors retain the right to offer only one Class of Shares for purchase by investors in any particular jurisdiction in order to conform to local law, custom or business practice. The Directors also reserve the right to adopt standards applicable to classes of investors or transactions that permit or require the purchase of a particular Class of Shares.

The Directors reserve the right to reject, in whole or in part, any application or subsequent subscription for Shares. In particular, if the Directors determine that it would be detrimental to the existing Shareholders to accept applications for Shares of any Fund which represents more than 10% of the Net Asset Value of such Fund, the Directors may decide that all or part of the application for Shares in excess of 10% be deferred until the next Dealing Day. If the Directors decide to defer all or part of the application in excess of 10%, the applicants shall be informed prior to the deferral taking place.

9.2.4 Minimum Subscription Amount, Minimum Initial Investment Amount, Minimum Additional Investment Amount and Minimum Holding

Details of the Minimum Subscription Amount, Minimum Initial Investment Amount, Minimum Additional Investment Amount and Minimum Holding for each Fund will be set out in the relevant Supplement. The Directors reserve the right from time to time to waive any requirements relating to a Minimum Subscription Amount, the Minimum Initial Investment Amount and/or the Minimum Additional Investment Amount as and when it determines in its absolute discretion.

9.2.5 Subscriptions/Redemptions Account

Where the subscription monies are received into the Subscriptions/Redemptions Account from an investor in advance of Shares being issued (as will be the case in the context of a Fund which operates on a cleared funds basis), such subscription monies will be the property of the relevant Fund and accordingly an investor will be treated as a general unsecured creditor of the Company during the period between receipt of subscription monies into the Subscriptions/Redemptions Account and the issue of Shares.

9.2.6 Data Protection

Prospective investors should note that, by virtue of making an investment in the Company and the associated interactions with the Company and its affiliates and delegates (including completing the Application Form, and including the recording of electronic communications or phone calls where applicable), or by virtue of providing the Company with personal information on individuals connected with the investor (for example directors, trustees, employees, representatives, shareholders, investors, clients, beneficial owners or agents) such individuals will be providing the Company and its affiliates and delegates with certain personal information which constitutes personal data within the meaning of the Data Protection Legislation. The Company shall act as a data controller in respect of this personal data and its affiliates and delegates, such as the Manager, the Administrator, the Investment Manager and the Distributor, may act as data processors (or joint data controllers in some circumstances).

The Company has prepared a document outlining the Company's data protection obligations and the data protection rights of individuals under the Data Protection Legislation (the **"Privacy Notice"**).

All new investors shall receive a copy of the Privacy Notice as part of the process to subscribe for Shares in the Company. The Privacy Notice contains information on the following matters in relation to data protection:

- that investors will provide the Company with certain personal information which constitutes personal data within the meaning of the Data Protection Legislation;
- a description of the purposes and legal basis for which the personal data may be used;
- details on the transmission of personal data, including (if applicable) to entities located outside the EEA;
- details of data protection measures taken by the Company;

- an outline of the various data protection rights of individuals as data subjects under the Data Protection Legislation;
- information on the Company's policy for retention of personal data;
- contact details for further information on data protection matters.

Given the specific purposes for which the Company and its affiliates and delegates envisage using personal data, under the provisions of the Data Protection Legislation, it is not anticipated that individual consent will be required for such use. However, as outlined in the Privacy Notice, individuals have the right to object to the processing of their data where the Company has considered this to be necessary for the purposes of its or a third party's legitimate interests.

- 9.3 Redemption of Shares
- 9.3.1 Applications for Redemptions

Save as set out in the relevant Supplement, the following provisions apply in respect of Redemption of Shares of each of the Funds other than the Unlisted Funds:

Investors can redeem their Shares for cash on the relevant Dealing Day except during any period in which the calculation of the Net Asset Value is suspended. It is also possible for investors to sell their Shares on the Secondary Market, as set out in section 10 below. The details on the specific cash procedures are set out below under the heading "Cash Subscriptions and Redemptions". The amount of any Primary Market Transaction Costs and/or Redemption Fee (as applicable) payable by the Shareholder will be set out in the relevant Supplement.

Investors should note that any redemption proceeds being paid out by a Fund and held for any time in the Subscriptions/Redemptions Account shall remain an asset of the relevant Fund until such time as the proceeds are released to the investor. This would include, for example, cases where redemption proceeds are temporarily withheld pending the receipt of any outstanding identity verification documents as may be required by the Company or the Administrator – enhancing the need to address these issues promptly so that the proceeds may be released. It should also be noted that the investor shall have ceased being considered a Shareholder and instead will rank as a general unsecured creditor of the Company.

It is envisaged by the Directors and by the Manager that investors will generally buy and sell their Shares through the Secondary Market (as set out under section 10 of this Prospectus, and other than for the Unlisted Funds), given the nature of the Funds of the Company and the terms and conditions relating to the subscription and redemption of Shares other than on the Secondary Market.

Shareholders wishing to have all or some of their Shares redeemed by the Company may make an application for redemption using a Redemption Form obtainable from the Company or the Administrator (a) in writing, or (b) by fax provided that (i) payment shall be made to the account on record (any changes to the account on record may only be made upon receipt of original written instructions), (ii) an original Application Form has been received and all anti-money laundering/counter-terrorist financing and client identification requirements are complete, and (iii) that the Shareholder has requested this facility, or (c) to the extent that the Directors have so decided, by electronic or other means. Applications must include details of the name of the Fund, Class of Share, the number of Shares or the amount the Shareholder wishes to have redeemed, the Shareholder's details, the Shareholder's account number and any other information required by the Redemption Form. In no event shall applications for redemption be processed until the original Application Form has been received by the Administrator from the investor and all of the necessary anti-money laundering/counter-terrorist financing requirements have been

completed. Any delay or failure by a Shareholder to provide any of this information may result in a delay in the processing of (or failure to process) the application for redemption whilst verification (which may be requested in writing) is sought from the Shareholder, and the Company and the Administrator shall not be liable for any such delay in the processing of (or failure to process) the application for redemption resulting from a Shareholder delaying or failing to provide the verification information required.

Any redemption requests received after the Dealing Deadline for a particular Dealing Day will be processed on the following Dealing Day unless the Directors in their absolute discretion (and such discretion only to be exercised in exceptional circumstances) otherwise determine to accept one or more orders received after the Dealing Deadline for processing on that Dealing Day provided that such application(s) have been received prior to the Valuation Point for that particular Dealing Day.

Written confirmations may be required by the Company and must be duly signed by all registered holders, unless in the case of joint registered holders, each such holder has sole signing authority.

The redemption proceeds will correspond to the Net Asset Value per Share as determined as at the Valuation Point for the relevant Dealing Day, less the Primary Market Transaction Costs and/or Redemption Fee (as applicable).

The Net Asset Value per Share of each Class in each Fund will be published in its respective Base Currency and, as the case may be, translated into other currencies as specified in the relevant Supplement, as set out in section 8.4.

Details of the Minimum Redemption Amount and Minimum Holding for each Fund will be set out in the relevant Supplement. The Directors reserve the right from time to time to waive any requirements relating to a Minimum Redemption Amount and Minimum Holding as and when it determines in its absolute discretion.

9.3.2 Restrictions on Redemptions

The Company is entitled to limit the number of Shares of any Fund redeemed on any Dealing Day to Shares representing not more than 10% of the Net Asset Value of that Fund on that Dealing Day. The Directors will advise the Administrator when to apply this based on reporting provided by the Administrator. In this event, the limitation will apply *pro rata*, so that all Shareholders wishing to redeem their shareholding in that Fund on the relevant Dealing Day will realise the same proportion of their redemption request. Shares not redeemed but which would otherwise have been redeemed, will be carried forward for redemption on the next Dealing Day and will be dealt with in priority (on a *pro rata* basis as detailed above) to redemption requests received subsequently. If requests for redemptions are so carried forward, the Administrator will inform the Shareholders affected.

The Articles contain special provisions with respect to a redemption request received from a Shareholder which would result in Shares representing more than 5% of the Net Asset Value of any Fund being redeemed by the Company on any Dealing Day. The Directors will advise the Administrator when to apply this based on reporting provided by the Administrator. In such a case the Company, at the discretion of the Directors and with the consent of the relevant Shareholder, may satisfy the redemption request in whole or in part by a distribution of investments of the relevant Fund *in specie*, provided that such a distribution would not be prejudicial to the interests of the remaining Shareholders of that Fund. The assets to be transferred shall be selected at the discretion of the Investment Manager, subject to the approval of the Depositary, and taken at their value used in determining the redemption price of the Shares being redeemed. Where a Shareholder requesting such redemption receives notice of the Company's intention to elect to satisfy the redemption request by such a distribution of assets, the Shareholder may require that

the Company, instead of transferring those assets, arrange for their sale and the payment of the net proceeds of sale to that Shareholder.

9.4 Cash Subscriptions and Redemptions

An investor may subscribe for or redeem Shares for cash on each Dealing Day (except during any period in which the calculation of the Net Asset Value is suspended) by notifying the Administrator as described below.

Applications for cash subscriptions or redemptions received by the Administrator on any Business Day before the relevant Dealing Deadline will be processed by the Administrator on that Business Day at a price corresponding to the Net Asset Value per Share as at the Valuation Point for the relevant Dealing Day. Applications for cash subscriptions or redemptions received after the Dealing Deadline for the relevant Dealing Day shall, unless the Manager or its delegate shall otherwise agree, be deemed to have been received by the next Dealing Deadline. For the avoidance of doubt, where relevant, the Minimum Subscription Amount, the Minimum Initial Investment Amount, the Minimum Additional Investment and the Minimum Redemption Amount shall be set out in the Supplement for the relevant Fund. Applications for subscriptions and redemptions will be irrevocable unless the Directors, or a delegate, otherwise agree. If requested, the Directors may, in their absolute discretion and subject to the prior approval of the Depositary, agree to designate additional Dealing Days and Valuation Points in respect of those Dealing Days for the purchase of Shares relating to any Fund which will be open to all Shareholders. Shareholders will be notified in advance of any such additional Dealing Days.

If the full payment in cleared funds relating to an application for subscription (including any Primary Market Transaction Costs and/or Subscription Fee, as applicable) has not been received by or on behalf of the Company by the relevant Settlement Date, the allotment of Shares made in respect of such application may, at the discretion of the Directors, be cancelled, or, alternatively, the Directors may treat the application as an application for such number of Shares as may be purchased with such payment on the Dealing Day next following receipt of payment in full of cleared funds. In such cases the Company may charge the applicant for any resulting bank charges or market losses incurred by the relevant Fund.

Shareholders wishing to make a cash redemption must also make arrangements for the transfer of their Shares into the Company's account at a clearing system. Delivery instructions are available from the Administrator upon written request. Payment for Shares redeemed will be effected by the Settlement Date as specified in the Supplement for the relevant Fund (provided that the Shares have been transferred into the Company's account at a clearing system). Redemption proceeds (net of any Primary Market Transaction Costs and/or Redemption Fee, as applicable) in the Base Currency of the Fund will be paid by electronic transfer to the account on record (any changes to the account on record may only be made upon receipt of original written instructions). On a redemption, the Depositary will release cash at the instruction of the Administrator. The cost of any transfer of proceeds by electronic transfer will be deducted from such proceeds.

Investors should note that although the Directors intend to pay all redemption proceeds by the relevant Settlement Date, such payment may be delayed if the Directors consider such delay to be in the best interests of the Shareholders. In all instances the payment of the redemption proceeds shall be within 10 Business Days of the relevant Dealing Deadline.

9.5 Use of a Subscriptions/Redemptions Account

The Company operates a single, omnibus Subscriptions/Redemptions Account for all of the Funds, in accordance with the Central Bank's guidance relating to umbrella fund cash accounts. Accordingly, monies in the Subscriptions/Redemptions Account are deemed

assets of the respective Funds and shall not have the protection of the Investor Money Regulations. It should be noted however that the Depositary will monitor the Subscriptions/Redemptions Account in performing its cash monitoring obligations and ensuring effective and proper monitoring of the Company's cash flows in accordance with its obligations as prescribed under UCITS V. There nonetheless remains a risk for investors to the extent that monies are held by the Company in the Subscriptions/Redemptions Account of a Fund at a point where such Fund (or another Fund of the Company) becomes insolvent. In respect of any claim by an investor in relation to monies held in the Subscriptions/Redemptions Account, the investor shall rank as an unsecured creditor of the Company.

The Company in conjunction with Depositary shall establish a policy to govern the operation of the Subscriptions/Redemptions Account, in accordance with the Central Bank's guidance in this area. This policy shall be reviewed by the Company and the Depositary at least annually.

9.6 Form of the Shares and Register

The Shares shall be issued in the form of Registered Shares.

Save in respect of the Unlisted Funds, Registered Shares may be represented by a Global Share Certificate.

The Administrator will maintain the register for the Company and will be responsible for the issue of Shares to investors (which may include Market Makers and other shareholders that choose to subscribe for Shares directly to the Company).

9.7 Registered Shares/ Registered Shares represented by Global Share Certificates

The Shares can be issued in registered form and the Shareholders' register is conclusive evidence of the ownership of such Shares.

Other than in the case of Unlisted Shares, Registered Shares shall be issued without share certificates. The uncertified form enables the Company to effect redemption instructions without undue delay.

Global Share Certificates will be issued in the name of the Company and deposited with the Clearing Agents. Global Share Certificates will be transferable in accordance with applicable laws and any rules and procedures issued by any Clearing Agent concerned with such transfer. Such Registered Shares represented by a Global Share Certificate are freely transferable subject to and in accordance with the rules of the relevant Clearing Agent. Shareholders who are not participants in such systems will only be able to transfer such Registered Shares represented by a Global Share Certificate through a financial intermediary who is a participant in the settlement system of the relevant Clearing Agent.

9.8 General Provisions

The Directors reserve the right to reject any application for subscription or exchange in whole or in part. Furthermore, the Directors reserve the right at any time, without notice, to discontinue the issue and sale of Shares of any Fund of the Company. The Directors also reserve the right to authorise at any time and without notice the issue and sale of Shares for Funds that were previously closed for further subscriptions.

The Directors may in its discretion decide, prior to the Launch Date, to cancel the offering of a Fund. The Directors may also decide to cancel the offering of a new Class of Shares. In such case, investors having made an application for subscription will be duly informed and any subscription monies already paid will be returned. For the avoidance of doubt, no interest will be payable on such amount prior to their return to the investors.

No Shares will be issued, redeemed or exchanged during any period when the calculation of the Net Asset Value per Share of the relevant Fund is suspended pursuant to the Articles and as discussed herein under "**Suspension of Calculation of Net Asset Value**". Notice of any such suspension will be given to applicants for subscription, redemption or exchange of Shares (in addition to the notifications to be made pursuant to section 8.3). Applications will be considered on the first Dealing Day following the end of the suspension period.

The Company is regulated by the Central Bank, and must comply with the measures provided for in the CJA which are aimed towards the prevention and detection of money laundering and terrorist financing.

The CJA requires a detailed verification of the investor's identity including any persons purporting to act on the investor's behalf. This may include obtaining proof of address, source of funds, source of wealth or other additional information which may be requested from time to time, monitoring the business relationship on an on-going basis and where applicable, identifying and verifying the identity of the beneficial owner on a risk sensitive basis in order to comply with the obligations set out in the CJA. Politically exposed persons ("**PEPs**"), an individual who is or has been entrusted with prominent public functions, their immediate family members and/or persons known to be close associates of such persons, must also be identified and will be subject to enhanced due diligence measures in accordance with the CJA.

By way of example an individual may be required to produce a duly certified copy of their passport or identification card together with evidence of their address such as utility bills or bank statements (not more than six months old). Date of birth and tax residence details may also need to be provided and verified.

In the case of corporate investors, such measures may require production of a certified copy of the certificate of incorporation (and any change of name), memorandum and articles of association (or equivalent), the identification of the directors of the company, verification of two directors (or equivalent) of the company and identification of the beneficial ownership of the corporate applicant including identification of the ultimate beneficial owner and/or a certified copy of the corporate investor's authorised signatory list.

The level of customer due diligence/verification documentation required will depend on the circumstances of each application following a risk based assessment of the applicant. For example, a detailed verification might not be required where the application is deemed low risk after consideration of a number of risk variables including jurisdiction, customer type and distribution channels. The Company will have regard to the relevant business risk assessment when determining the level of customer due diligence required under Sections 33 and 35 of the CJA.

Pursuant to Section 35 of the CJA, prior to establishing a business relationship with an applicant to which the European Union (Anti-Money Laundering: Beneficial Ownership Of Corporate Entities) Regulations 2019 apply, the Company is required to confirm that information concerning the beneficial ownership of any Irish domiciled or registered applicant has been entered in the relevant central beneficial ownership register that applies to such applicant.

The Company and the Administrator reserve the right to request such information and documentation as is necessary to verify the identity of an applicant in line with the requirements of the CJA. In the event of delay or failure by the applicant to produce any information required for verification purposes, the Administrator, on behalf of the Company, may refuse to accept the application and return all subscription money or compulsorily redeem such Shareholder's Shares and/or payment of redemption proceeds may be delayed and none of the Company, the Directors, the Manager, the Investment Manager or the Administrator shall be liable to the subscriber or Shareholder where an application for Shares is not processed or Shares are compulsorily redeemed in such circumstances.

The Administrator, on behalf of the Company, may refuse to pay redemption proceeds or accept further subscription money where the requisite information for verification purposes has not been produced by a Shareholder.

Appropriate measures to verify an applicant's identity are required to take place before the establishment of the business relationship or as soon as practicable after initial contact is made with an applicant. For the avoidance of doubt, no transactions will be facilitated on non-verified accounts.

Each applicant agrees in the Application Form to hold harmless and indemnify the Company, the Manager, the Administrator and their delegates ("**Indemnified Persons**"), and agrees to keep them indemnified, against any direct or indirect loss, damages, liabilities or reasonable costs of any nature whatsoever arising to any of them, arising as a result of:

- (i) any of the Indemnified Persons acting upon or pursuant to facsimile instructions reasonably believed in good faith to be genuine and to be signed by properly authorised persons. The Indemnified Persons may rely conclusively upon and shall incur no liability in respect of any action taken upon any notice, consent, request, instruction or other instrument reasonably believed in good faith to be genuine and to be signed by such properly authorised persons;
- (ii) a failure to process any application for Shares due to the applicant failing to provide the necessary identification requirements under the CJA.
- (iii) any breach of any representation, statement, warranty, covenant or confirmation by the applicant or any failure by the applicant to disclose any relevant details or to provide any information reasonably requested of them, other than where such failure results from legal or regulatory prohibitions imposed on the applicant or where such loss has been caused by the negligence, wilful default or fraud of the Company, the Manager or its delegates.
- (iv) any non-delivery or late delivery of cash or the securities in respect of the applicant's subscription for Shares.

Activities which may adversely affect the interests of the Shareholders (for example, activities that disrupt the Company's investment strategies or impact expenses for the Company) are not permitted. The Directors may, in their discretion, if they deem such activities adversely affect the interests of the Shareholders, take action as appropriate to prevent such activities.

All subscription, redemption and exchange requests shall be dealt with promptly, however investors should note that neither the Company nor the Administrator shall be liable for any delays in processing subscriptions, redemptions and/or exchange requests.

9.9 Compulsory Redemption

The Company may compulsorily redeem all of the Shares of any Fund if the Net Asset Value of the relevant Fund is less than the Minimum Fund Size (if any) specified in the relevant Supplement and in accordance with the terms of the Supplement for the relevant Fund.

The Company reserves the right to redeem any Shares which are or become owned, directly or indirectly by a U.S. Person or a Prohibited Person as set out under the heading "General Information in respect of Applications for Shares".

The Board of Directors may, at any time, redeem all Shares from Shareholders whose holding is less than the Minimum Holding. In such case the Shareholder concerned will

receive prior notice so as to be able to increase their holding above such amounts during a period of 10 Business Days following the receipt of such notice.

In the event the Shares cease to be held via a Recognised Clearing and Settlement System and such Shares are held by Irish Residents, the Company shall, where necessary for the collection of Irish tax, redeem and cancel a sufficient portion of the Shares held by a person who is or is deemed to be an Irish Resident or is acting on behalf of an Irish Resident on the occurrence of a chargeable event for taxation purposes and to pay the proceeds thereof to the Revenue Commissioners.

9.10 Exchange of Shares

At the sole discretion of the Directors, Shareholders will be able to apply to exchange on any Dealing Day all or part of their holding of Shares of any Class of one Fund (the **"Original Class**") for Shares of another Class in the same Fund or a separate Fund which is being offered at that time (the **"New Class**") provided that all the criteria for applying for Shares in the New Class have been met, by giving notice to the Administrator on behalf of the Company on or prior to the Dealing Deadline for the relevant Valuation Point. The Manager however may, at its sole discretion, agree to accept requests for exchange received after that time provided they are received prior to the relevant Valuation Point. The Manager may not be able to exercise this discretion in all circumstances, for example where requests for exchanges of Shares are made via dealing platforms or other electronic means. Shareholders making requests for exchanges via dealing platforms or other electronic means are reminded that they must refer to the provider of the dealing platform or electronic means for the procedures that apply to such trading arrangements.

The general provisions and procedures relating to subscriptions and redemptions will apply equally to exchanges. All exchanges will be treated as a redemption of the Shares of the Original Class and application of the net proceeds to the purchase of Shares of the New Class, based upon the then current Issue and Redemption Prices of Shares in each Fund. The Articles allow for an Exchange Charge of up to 3% of the total Redemption Price of the Shares of the Original Class redeemed to be charged, and the Directors, in their sole discretion, reserve the right to impose such fee within this limit together with any Primary Market Transaction Costs as shall be set out in the Supplement in respect of each Fund. Furthermore, where exchanges of Shares are not permitted between Classes of a certain Fund, this will be set out in the Supplement for the relevant Fund. Unless otherwise set out in the Supplement, fractions of shares shall not be issued.

The number of Shares of the New Class to be issued will be calculated in accordance with the following formula:

$$S = \frac{\left[Rx(RPxER)\right] - F}{SP}$$

where:

- R = the number of Shares of the Original Class to be exchanged;
- S = the number of Shares of the New Class to be issued;
- RP = the Redemption Price per Share of the Original Class as at the Valuation Point for the relevant Dealing Day;
- ER = in the case of an exchange of Shares designated in the same Base Currency is 1. In any other case, it is the currency conversion factor determined by the Directors on or about the Valuation Point for the relevant Dealing Day as representing the effective rate of exchange applicable to the

transfer of assets relating to the Original and New Classes of Shares after adjusting such rate as may be necessary to reflect the effective costs of making such transfer;

- SP = the issue price per Share of the New Class as at the Valuation Point for the relevant Dealing Day; and
- F = the Exchange Charge, if any payable to the Company, or as it may direct, on the exchange of Shares.

Where there is an exchange of Shares, Shares of the New Class will be allotted and issued in respect of and in proportion to the Shares of the Original Class in the proportion S to R.

Shares may not be exchanged for Shares in a different Fund and exchanges between the Classes of Shares of the same Fund shall not be permitted during any period when the calculation of the Net Asset Value of the relevant Fund is suspended in the manner described under "**Suspension of Calculation of Net Asset Value**" above. Applicants for exchange of Shares will be notified of such postponement and their applications will be considered as at the next Dealing Day following the ending of such suspension.

When requesting the exchange of Shares as an initial investment in a Fund, Shareholders should ensure that the value of the Shares exchanged is equal to or exceeds the Minimum Subscription Amount or the Minimum Initial Investment Amount (as appropriate) for the relevant New Class specified in the relevant Supplement. In the case of an exchange of a partial holding only, the value of the remaining holding must also be at least equal to the Minimum Holding for the Original Class.

The Administrator will on request arrange for any necessary currency transaction required if there is an exchange of Shares of any Class of a Fund for Shares of the same Class in another Fund. Any such currency transaction may be effected with the Depositary or an Authorised Participant and will be at the applicant's cost. Currency exchange transactions may delay any dealing in Shares as the Administrator may choose at its option to delay executing any foreign exchange transaction until cleared funds have been received.

9.11 ESG Focus/Impact Funds

The Investment Manager may categorise certain Funds as ESG focus/impact funds. ESG focus/impact funds promote ESG characteristics or have a specific sustainability or impact objective which is defined in the investment policy.

Unless otherwise indicated in the relevant Supplement for a Fund, investments underlying a particular Fund do not take into account the EU criteria for environmentally sustainable economic activities.

10 Secondary Market

This section applies to all Funds of the Company other than the Unlisted Funds.

10.1 Listing on a Stock Exchange

It is the intention of the Company for each of its Funds (other than for the Unlisted Funds), through having its Shares listed on one or more Relevant Stock Exchanges, to qualify as an exchange traded fund ("ETF"). As part of those listings there is an obligation on one of more members of the Relevant Stock Exchange to act as market makers offering prices at which the Shares can be purchased or sold by investors. The spread between those purchase and sale prices may be monitored and regulated by the relevant stock exchange authority.

Unless otherwise stated in the Supplement for the relevant Fund, it is contemplated that application will be made to list the Shares of each Fund on one or more Relevant Stock Exchange(s). If the Directors decide to create additional Funds or Classes they may in their discretion apply for the Shares of such Funds to be listed on the Relevant Stock Exchange(s). For so long as the Shares of any Fund are listed on any Relevant Stock Exchange, the Fund shall endeavour to comply with the requirements of the Relevant Stock Exchange relating to those Shares. For the purposes of compliance with the national laws and regulations concerning the offering and/or listing of the Shares outside Ireland this document may have attached to it one or more documents setting out information relevant for the jurisdictions in which the Shares are offered for subscription.

The Company does not charge any transfer fee for purchases of Shares on the secondary market. Orders to buy Shares through the Relevant Stock Exchanges can be placed via a member firm or stockbroker. Such orders to buy Shares may incur costs over which the Company has no control.

The approval of any listing particulars pursuant to the listing requirements of the Relevant Stock Exchange does not constitute a warranty or representation by such Relevant Stock Exchange as to the competence of the service providers or as to the adequacy of information contained in the listing particulars or the suitability of the Shares for investment or for any other purpose.

Certain Authorised Participants who subscribe for Shares may act as market makers; other Authorised Participants are expected to subscribe for Shares in order to be able to offer to buy Shares from or sell Shares to their customers as part of their broker/dealer business. Through such Authorised Participants being able to subscribe for or redeem Shares, a liquid and efficient secondary market may develop over time on one or more Relevant Stock Exchanges as they meet secondary market demand for such Shares. Through the operation of such a secondary market, persons will be able to buy Shares from or sell Shares to other secondary market investors or market makers, broker/dealers, or other Authorised Participants at prices which should approximate, after currency conversion, the Net Asset Value of the Shares. Investors should be aware that on days other than Business Days or Dealing Days of a Fund when one or more markets are trading Shares but the underlying Market(s) on which the Reference Index or Reference Asset of the Fund are traded are closed, the spread between the quoted bid and offer prices in the Shares may widen and the difference between the market price of a Share and the last calculated Net Asset Value per Share may, after currency conversion, increase. Investors should also be aware that on such days the Reference Index or Reference Asset value would not necessarily be calculated and available for investors in making their investment decisions because prices of Reference Index or Reference Asset securities in the underlying Market(s) would not be available on such days. Nonetheless, one or more Relevant Stock

Exchanges may provide a calculation of such Reference Index or Reference Asset based upon trading, if any, of such Reference Index or Reference Asset securities on marketplaces other than the underlying Market(s). The settlement of trades in Shares on Relevant Stock Exchanges will be through the facilities of one or more Recognised Clearing and Settlement Systems following applicable procedures which are available from the Relevant Stock Exchanges. Further details of the Relevant Stock Exchanges for each Fund are set out in the relevant Supplement.

10.2 Intra-Day Portfolio Value ("**iNAV**")

The Investment Manager may at its discretion make available on the Website, or may designate other persons to make available on its behalf, on each Business Day, an intraday portfolio value or "**iNAV**" for one or more Funds. If the Investment Manager makes such information available on any Business Day, the iNAV will be calculated at least once per day based upon information available during the trading day or any portion of the trading day, and will ordinarily be based upon the current value of the assets/exposures of the Fund in effect on such Business Day, together with any cash amount in the Fund as at the previous Business Day. The Investment Manager will make available an iNAV if this is required by any Relevant Stock Exchange.

An iNAV is not, and should not be taken to be or relied on as being, the value of a Share or the price at which Shares may be subscribed for or redeemed or purchased or sold on any Relevant Stock Exchange. In particular, any iNAV provided for any Fund where the constituents of the Reference Index or Reference Asset are not actively traded during the time of publication of such iNAV may not reflect the true value of a Share, may be misleading and should not be relied on. The inability of the Investment Manager or its designee to provide an iNAV, on a real-time basis, or for any period of time, will not in itself result in a halt in the trading of the Shares on a Relevant Stock Exchange, which will be determined by the rules of the Relevant Stock Exchange in the circumstances. Investors should be aware that the calculation and reporting of any iNAV may reflect time delays in the receipt of the relevant constituent securities prices in comparison to other calculated values based upon the same constituent securities including, for example, the Reference Index or Reference Asset itself or the iNAV of other exchange traded funds based on the same Reference Index or Reference Asset. Investors interested in subscribing for or redeeming Shares on a Relevant Stock Exchange should not rely solely on iNAV in making investment decisions, but should also consider other market information and relevant economic and other factors (including, where relevant, information regarding the Reference Index or Reference Asset, the relevant constituent securities and financial instruments based on the Reference Index or Reference Asset corresponding to the relevant Fund). None of the Company, the Directors, the Manager, the Investment Manager, the Depositary, the Administrator, any Authorised Participant and the other service providers shall be liable to any person who relies on the iNAV.

10.3 Clearing and Settlement

The settlement of trading in Shares of the Funds is centralised in the ICSD settlement structure operated by Clearstream which provides centralised issuance in Clearstream and allows for centralised settlement in the ICSD structure jointly operated by Clearstream and Euroclear (these ICSDs being the Recognised Clearing and Settlement Systems through which the Funds' Shares will be settled). Shares in the Funds will generally be issued in Dematerialised Form and no temporary documents of title or share certificates will be issued in respect of Shares of the Funds. The Funds will apply for admission for clearing and settlement through the ICSD.

Under the ICSD settlement model, all Shares in the Funds will ultimately settle in an ICSD but investors may have their holdings within CSDs which will be participants. Accordingly, an Investor will either hold its beneficial interests in Funds Shares within the ICSD (as an ICSD Participant) or within other CSDs which are ICSD Participants.

A purchaser of interests in Shares in the Funds will not be a registered Shareholder in the Company, but will hold an indirect beneficial interest in such Shares. Legal title to the Shares of the Funds will be held by the relevant ICSD as the registered holder of the Funds Shares. The rights of the holder of the indirect beneficial interests in the Shares, where such person is a participant in the ICSD, shall be governed by the terms and conditions applicable to the arrangement between such participant and their ICSD and where the holder of the indirect beneficial interests in the Shares is not an ICSD Participant, shall be governed by their arrangement with their respective nominee, broker or CSD (as appropriate) which may be an ICSD Participant or have an arrangement with an ICSD Participant. The extent to which, and the manner in which, participants may exercise any rights arising under the Shares will be determined by the respective rules and procedures of their ICSD. All references herein to actions by holders of the Funds Shares will refer to actions taken by the ICSD as registered Shareholder following instructions from the ICSD Participants. All distributions, notices, reports, and statements issued to such Shareholder by the Company shall be distributed to the ICSD Participants in accordance with such ICSD's procedures.

Interests in the Shares represented by the Global Share Certificate will be transferable in accordance with applicable laws, any rules and procedures issued by the ICSDs and this Prospectus. Beneficial interests in such Shares will only be transferable in accordance with the rules and procedures for the time being of the relevant ICSD and this Prospectus.

International Central Securities Depositary

Each ICSD Participant must look solely to its ICSD for documentary evidence of the amount of such ICSD Participant's interests in any Shares. Any certificate or other document issued by the relevant ICSD, as to the interest in such Shares standing to the account of any person shall be conclusive and binding as accurately representing such records. Each ICSD Participant must look solely to its ICSD for such ICSD Participant's (and therefore any person with an interest in the Shares) portion of each payment or distribution made by the Funds to or on the instructions of the ICSD and in relation to all other rights arising under the Shares.

ICSD Participants shall have no claim directly against the Company, the Funds, any Paying Agent or any other person (other than their ICSD) relating to payments or distributions due in respect of the Shares which are made by the Company or the Funds to or on the instructions of the ICSD and such obligations of the Company shall be discharged thereby.

The Company or its duly authorised agent may from time to time require the holder of the indirect beneficial interest in the Shares to provide them with information relating to: (i) the capacity in which they hold an interest in Shares; (ii) the identity of any other person or persons then or previously interested in such Shares; (iii) the nature of any such interests; and (iv) any other matter where disclosure of such matter is required to enable compliance by the Company with applicable laws or the constitutional documents of the Company.

The Company or its duly authorised agent may from time to time request the ICSD to provide the Company with certain details in relation to ICSD Participants that hold interests in Shares in each Fund including (but not limited to): ISIN, ICSD Participant name, ICSD Participant type (e.g. fund/bank/individual), residence of ICSD Participants, number of ETFs and holdings of the ICSD Participant within the ICSD, as appropriate including which Funds, types of Shares and the number of such interests in the Shares held by each such ICSD Participant, and details of any voting instructions given and the number of such interests in the Shares held by each such ICSD Participant. ICSD Participants that are holders of interests in Shares or intermediaries acting on behalf of such account holders will provide such information upon request of the ICSD or its duly authorised agent and have been authorised pursuant to the respective rules and procedures of each ICSD to disclose such information to the Company of the interest in Shares or to its duly authorised agent. Similarly, the Company or its duly authorised agent may from time to time request

any CSD to provide the Company with details in relation to Shares in each Fund or interests in Shares in each Fund held in each CSD and details in relation to the holders of those Shares or interests in Shares, including (without limitation) holder types, residence, number and types of holdings and details of any voting instructions given by each holder. Holders of Shares and interests in Shares in a CSD or intermediaries acting on behalf of such holders agree to the CSD, pursuant to the respective rules and procedures of the relevant CSD, disclosing such information to the Company or its duly authorised agent.

The holder of the indirect beneficial interest in the Shares may be required to agree to the ICSD providing the identity of an ICSD Participant or investor to the Company upon their request.

Notices of Meetings and the Exercise of Voting Rights through the International Central Securities Depositaries

Notices of general meetings and associated documentation will be issued by the Company to the registered holder of the Shares i.e. the ICSD. Each ICSD Participant must look solely to its ICSD and the rules and procedures for the time being of the relevant ICSD governing onward delivery of such notices to the ICSD Participants and the ICSD Participant's right to exercise voting rights. Investors who are not ICSD Participants in the relevant ICSD would need to rely on their broker, nominee, custodian bank or other intermediary which is an ICSD Participant, or which has an arrangement with an ICSD Participant, in the relevant ICSD to receive any notices of Shareholder meetings of the Company and to relay their voting instructions to the relevant ICSD.

The ICSD has a contractual obligation to promptly notify the ICSD Participants of any Shareholder meetings of the Company and to relay any associated documentation issued by the Company to the ICSD Participants in accordance with its rules and procedures. In accordance with their respective rules and procedures, each ICSD is contractually bound to collate and transfer all votes received from its ICSD Participants to the Company and is obligated to vote in accordance with the ICSD Participant's voting instructions.

Settlement of Subscriptions and Redemptions

Shares may be issued in Dematerialised Form (or uncertificated) and the Funds may apply for admission for clearing and settlement through a clearing system. To facilitate this arrangement, the Depositary (or its delegate) will maintain a Depot Account at the relevant clearing and settlement system. Settlement of subscriptions for shares by an Authorised Participant will take place on a delivery versus payment ("**DVP**") basis at the relevant clearing and settlement system. An Authorised Participant will arrange for delivery of the subscription monies to the Depot Account maintained by the Depositary (or its delegate) who, in turn, will arrange for the simultaneous delivery to the Authorised Participant of the shares for which it has subscribed.

Upon a redemption of Shares by an Authorised Participant, such transaction will also take place on a DVP basis at the relevant clearing and settlement system. The Authorised Participant will arrange for the delivery of Shares to the Depositary's (or its delegate's) Depot Account who, in turn, will arrange for the simultaneous credit of the redeeming investor's Depot Account with the redemption proceeds.

11 Fees & Expenses

11.1 Fees and Expenses Payable by the Company

The Company may pay the fees and expenses of each Fund, which may include the Management Fee and any Administrative Expenses incurred by the Fund as set out below. Please see the section "Flat Fee" below for further information in respect of fees and expenses.

11.1.1 Management Fee

In accordance with and subject to the terms of the Management Agreement, the annual Management Fee payable to the Manager will be a percentage of the net assets of each Fund or Class of Shares or the Initial Issue Price (as will be indicated in the Supplement). The Manager is also entitled to receive reimbursement for any reasonable disbursements and out-of-pocket expenses. The Management Fee will be calculated upon each Dealing Day and will be accrued daily and payable monthly in arrears.

(i) Investment Management Fee

The annual Investment Management Fee payable to the Investment Manager will be a percentage of the net assets of each Fund or Class of Shares or the Initial Issue Price.

11.1.2 Administrative Expenses

(i) Depositary Fee

The Depositary is entitled to receive from each Fund a depositary fee. This fee will be paid by the Company to the Depositary for and on behalf of the Funds. The Company will also reimburse the Depositary out of the assets of the Funds for reasonable and approved by the Company out-of-pocket expenses incurred by the Depositary and for fees (which will not exceed normal commercial rates) and reasonable out-of-pocket expenses of any subcustodian appointed by the Depositary and will be liable for transaction costs, according to section 11.1.2(vi). The fees and expenses of the Depositary accrue on each Dealing Day and are payable monthly in arrears.

(ii) Fees payable to the Directors and to the Administrator

The Directors are entitled to such annual fee that is consistent with market rates and as may be agreed between the Company and the Promoter, or such other amount as may be approved by a resolution of the Shareholders in a general meeting and may be paid all reasonable and properly vouched travelling, hotel and other out-of-pocket expenses properly incurred by them in connection with their attendance at meetings of the Directors.

According to the Administration Agreement, the Company shall pay to the Administrator a fee for its services as Administrator (including all vouched out-of-pocket costs and expenses reasonably incurred by it). The fees and expenses of the Administrator accrue on each Dealing Day and are payable monthly in arrears.

(iii) Exceptional Expenses

The Company shall be liable for Exceptional Expenses including, without limitation, legal fees and expenses incurred in prosecuting or defending a claim or allegation by or against the Company, and any tax, levy, duty or similar charge imposed on the Company or its

assets that would otherwise not qualify as ordinary expenses. Exceptional Expenses are allocated across each Class of Shares, according to their respective assets.

(iv) Setting Up Costs

The cost of establishing the Company and the Funds (including fees in connection with the incorporation and registration of the Company, legal, regulatory and consultancy fees, listing the Funds on the Relevant Stock Exchanges and registering the Funds for sale in other jurisdictions) may be paid by the Company and/or Promoter and/or Manager. The cost of establishing subsequent Funds may also be paid by the Company and/or Promoter and/or Prom

(v) Miscellaneous Expenses

Miscellaneous Expenses include but are not limited to the following, which may also be paid out of the assets of the Company: ongoing organisation and registration costs; licence fees payable to licence holders of an index; expenses for legal and auditing services; expenses in relation to the calculation and publication of any stamp duties, all taxes and VAT; company secretarial fees; any costs incurred in respect of meetings of Shareholders; marketing costs; investment transaction charges; costs incurred in respect of the distribution of income to Shareholders; any fees or expenses of a counterparty to any securities lending arrangements or Repurchase Transactions with the Company; the fees and expenses of any paying agent, clearing agent, settlement agent or representative appointed in compliance with the requirements of another jurisdiction; the fees and expenses associated with any liquidation, schemes of amalgamation or merger (other than where the scheme of amalgamation or merger is as a result of a commercial decision on the part of the Promoter and/or Manager), delisting or deregistration, the fees and expenses of any consultant appointed to provide services to the Company or the Manager; any amount payable under indemnity provisions contained in the Articles or any agreement with any appointee of the Company; cost of any proposed listings and maintaining such listings; all reasonable out-of-pocket expenses of the Board of Directors; foreign registration fees and fees relating to the maintenance of such registrations including translation costs and local legal costs and other expenses due to supervisory authorities in various jurisdictions and local representatives' remunerations in foreign jurisdictions; insurance; interest; the costs of printing and distributing this Prospectus and any costs incurred as a result of periodic updates of this Prospectus or the relevant Supplements: reports, accounts and any explanatory memoranda; any necessary translation fees; any fees in respect of circulating details of the Net Asset Value; and any fees in respect of circulating such other information which is required to be published in the different jurisdictions, or of a change in law or the introduction of any new law (including any costs incurred as a result of compliance with any applicable code, whether or not having the force of law).

(vi) Transaction Fees

Transaction Fees are any fees and expenses incurred in buying and selling securities or other investments held by a Fund, e.g., Transfer Taxes, brokerage costs and commissions and correspondence fees for transferring securities or investments or other interests, costs incurred relating to execution of swap transactions, including relevant collateral costs and custody transaction fees unless otherwise specified in the relevant Supplement.

For the avoidance of doubt, each of the service providers may waive all or a portion of the fee it receives in respect of any investor.

11.2 Flat Fee

The Company may also have a fee structure where, in respect of each Fund, a single fee amount (known as the "**Flat Fee**") is accrued by the relevant Fund. This Flat Fee reflects each Fund's estimated operating costs, fees and expenses (except for portfolio Transaction

Fees or charges and swap related costs, taxes or duty charges for portfolio re-balancing as further disclosed below; and any extraordinary or exceptional costs, all of which will be paid separately out of the assets of the relevant Fund) as well as an allocation of operating costs, fees and expenses of the Company. The Flat Fee rate, as a percentage of the Fund's Net Asset Value, is disclosed in the Fund's Supplement. The Flat Fee is accrued daily in the Fund's Net Asset Value and is payable monthly in arrears by the Fund to the Manager or its affiliates. The Manager or its affiliates will pay all operating costs, fees and expenses of each Fund (except for portfolio Transaction Fees or charges and swap related costs, taxes or duty charges for portfolio re-balancing, and any extraordinary or exceptional costs, all of which will be paid separately out of the assets of the relevant Fund), together with the operating costs, fees and expenses of the Company, including the fees of the Investment Manager, the Depositary, the Administrator and any out-of-pocket expenses incurred by them.

Other than: a) the Flat Fee; b) those costs relating to its investment activity (including portfolio Transaction Fees or charges and swap related costs, taxes or duty charges for portfolio re-balancing as further disclosed below, all of which will be paid separately out of the assets of the relevant Fund); and c) extraordinary or exceptional costs and expenses (if any) as may arise from time to time, such as material litigation in relation to the Company or the Fund, all of which will be paid separately out of the assets of the relevant Fund, the Fund will not incur any costs from Fund service providers or other related Fund or Company costs, such as director fees, index license fees, costs of listing or registering the Fund for public distribution in various countries, fees and expenses of professional advisers such as the statutory auditor fees, legal advisers, registration and paying agents, regulatory fees, risk management systems, collateral manager fees, etc. - such costs will be borne by the Manager or its affiliates. Consequently, the Fund will not directly bear any fees, costs or expenses in relation to the Investment Manager, Administrator or Depositary.

In circumstances where the Flat Fee accrued by the Fund is less than the fees and costs of Fund service providers (such as the fees of the Investment Manager, Depositary and Administrator) and other related Fund or Company costs payable by the Manager or its affiliates, the Manager or its affiliates will bear the shortfall and the Fund (or the investors in the Fund) will not bear or otherwise incur this shortfall. The Manager reserves the right to recharge some or all of these other costs to the Promoter or to other affiliates of UBS Group AG.

As noted above, the Manager or its affiliates will pay various Company and Fund costs, including fees due to the Company's Directors. Such fees will not exceed the sum of €40,000 per annum per Director without the approval of the Board of Directors (with each Director abstaining on any resolution relating to that Director's remuneration).

As noted above, the Flat Fee does not reflect costs relating to the Fund's investment activity (including but not limited to portfolio Transaction Fees and charges, stamp duty or other taxes on the investments of the Company or Fund including duty charges for portfolio rebalancing, withholding taxes, commissions and brokerage fees incurred with respect to the Company's investments, fees, costs and expenses payable to the Approved Counterparty under the terms of the Swap Agreements, interest on borrowings and bank charges incurred in negotiating, effecting or varying the terms of such borrowings, any commissions charged by intermediaries in relation to an investment in the Fund, and foreign currency hedging costs at the level of the portfolio or share class, all of which will be paid separately out of the assets of the relevant Fund or class).

The Flat Fee may be changed periodically. Prior to any change in Flat Fee taking effect, the Company and Fund will be subject to the approval and notification requirements of the Central Bank of Ireland.

11.3 Soft Commissions

It is not currently intended that any soft commission arrangements will be made in respect of the Company. In the event that the Investment Manager or any of its subsidiaries, Affiliates, associates, agents or delegates do enter into soft commission arrangement(s) they shall ensure that (i) the broker or counterparty to the arrangement will agree to provide best execution to the Company; (ii) the benefits under the arrangement(s) shall be those which assist in the provision of investment services to the relevant Fund and (iii) brokerage rates will not be in excess of customary institutional full service brokerage rates. Details of any such arrangements will be contained in the next following report of the Company. In the event that this is the unaudited semi-annual report, details shall also be included in the following annual report. Where appropriate, any such arrangements will comply with the requirements of Article 11 of the MiFID II Delegated Directive.

11.4 Entry/Exit Charges

11.4.1 Subscription Fee

Shareholders may be subject to a Subscription Fee of up to a maximum of 5% of subscription monies, as specified in the relevant Supplement.

11.4.2 Redemption Fee

Shareholders may be subject to a Redemption Fee up to a maximum of 3% of redemption proceeds, as specified in the relevant Supplement.

11.4.3 Exchange Charge

Shareholders may be subject to an Exchange Charge on the exchange of any Shares up to a maximum of 3% of the Net Asset Value of the Shares in the original Fund, as specified in the relevant Supplement.

11.4.4 Primary Market Transaction Costs

Shareholders may be subject to Primary Market Transaction Costs of up to a maximum of 1% of subscription monies or redemption proceeds, as specified in the relevant Supplement.

11.4.5 Anti-Dilution Levy

The Directors reserve the right to impose an Anti-Dilution Levy in the case of net subscriptions and/or net redemptions on a transaction basis, as set out in the supplement for the Fund.

12 Share Information

12.1 Dividend Policy

The dividend arrangements relating to each Fund will be decided by the Directors at the time of the creation of the relevant Fund and details are set out where applicable in the relevant Supplement.

Under the Articles, the Directors are entitled to declare such dividends on any Class of Shares at such times as they think appropriate and as appear to be justified out of the profits of the relevant Fund, being (i) the net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses and/or (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the relevant Fund and (iii) in accordance with the Central Bank Rules, partially or fully out of the capital of the relevant Fund.

To the extent Shares are not held via a Recognised Clearing and Settlement System the Company will be obliged and entitled to deduct an amount in respect of Irish tax from any dividend payable to any investor who is, or is deemed to be, or is acting on behalf of, an Irish Resident and to pay such amount to the Revenue Commissioners in Ireland.

Dividends not claimed within six years from their due date will lapse and revert to the relevant Fund. Dividends payable in cash to Shareholders will be paid by electronic transfer at the expense of the payee and will be paid within 4 months of the date the Directors declared the dividend.

Investors should note that any dividend income being paid out by a Fund and held in the Subscriptions/Redemptions Account shall remain an asset of the relevant Fund until such time as the income is released to the investor and that during this time the investor will rank as a general unsecured creditor of the Company.

12.2 Reports and Accounts

The Company's year-end is 30 June of each year. The annual report and audited accounts of the Company will be sent to Shareholders at least 21 days before the general meeting of the Company at which they are to be submitted for approval. The annual report and audited accounts of the Company will be published and filed with the Central Bank within four months after the conclusion of each accounting year.

The Company will also publish the semi-annual report and unaudited accounts on <u>www.ubs.com/ETF</u> and on <u>www.ubs.com/funds</u> and file same with the Central Bank within two months after the end of each semi-annual period which will be 31 December in each year. Such accounts will also be available to Shareholders free of charge upon request and will be available at the office of the Administrator.

Such reports and accounts will contain a statement of the Net Asset Value of each Fund and of the investments comprised therein as at the year-end or the end of such semi-annual period.

12.3 Transfer of Shares

Shares in each Fund will be transferable by instrument in writing signed by (or, in the case of a transfer by a body corporate, signed on behalf of or sealed by) the transferor provided always that the transferee completes an Application Form to the satisfaction of the Administrator and furnishes the Administrator with any documents required by it. In the case

of the death of one of joint Shareholders, the survivor or survivors will be the only person or persons recognised by the Company as having any title to or interest in the Shares registered in the names of such joint Shareholders. Shares may also be transferred in accordance with the rules of a clearing system as the Articles permit the transfer of Shares in Dematerialised Form.

Shares may not be transferred to a United States Person. Registration of any transfer may be refused by the Directors if (i) following the transfer either the transferor or the transferee would hold Shares having a value less than the Minimum Holding for the relevant Fund (if any) specified in the relevant Supplement hereto (ii) the payment of taxation remains outstanding (iii) the person to whom the transfer is to be made does not satisfy such antimoney laundering/counter-terrorist financing requirements as the Directors determine and (iv) the transfer is to be a prohibited person.

Persons dealing through a clearing system may be required to provide a representation that any transferee is not a Prohibited Person.

To the extent Shares are not held via a Recognised Clearing and Settlement System and the transferor is, or is deemed to be, or is acting on behalf of, an Irish Resident the Company may redeem and cancel a sufficient portion of the transferor's Shares as will enable the Company to pay the tax payable in respect of the transfer to the Revenue Commissioners in Ireland.

12.4 Communications with Shareholders

Communications with Shareholders may be effected by electronic mail or by any other means of communication provided that the Shareholder has consented to such method of communication. Copies of any documents sent to Shareholders will be available for inspection at the office of the Manager. Communications with Shareholders will also be published on the Website. Investors should regularly visit the Website, or request that their stockbrokers or other financial agents or advisers do so on their behalf, to ensure that they obtain such information on a timely basis. Where the Company is required to make certain information publicly available pursuant to the CBDF Directive or CBDF Regulation such information may be made available on the Website, and where relevant will be in translated form.

Unless otherwise disclosed to investors, where a Fund is marketed in another EEA Member State, the Manager shall make available facilities to perform the following tasks directly or through one or more third parties:

- process subscription, repurchase and redemption orders and make other payments to Shareholders relating to the Shares of the Fund, in accordance with the conditions set out in the Prospectus;
- (b) provide Shareholders with information on how orders, referred to in point (a) above can be made; and how repurchase and redemption proceeds are paid;
- (c) facilitate complaints handling and ensure there are procedures and arrangements relating to the Shareholders' exercise of their rights arising from their investment in the Fund in the EEA Member State where the Fund is marketed;
- (d) as further set out in the section "Documents for Inspection" below, make all required documents available for inspection by Shareholders on the Website and at the offices of the Company Secretary; and
- (e) act as a contact point for communicating with the competent authorities.

Incorporation and Share Capital

The Company was incorporated and registered in Ireland as an investment company with variable capital on 21 May 2010 with registered number 484724.

At the date hereof the authorised share capital of the Company is 2 subscriber shares ("**subscriber shares**") of €1 each and 1,000,000,000 shares of no par value initially designated as unclassified shares and available for issue as Shares.

There are no rights of pre-emption attaching to the Shares.

13 Summary of Articles

Clause 2 of the Articles provides that the sole object of the Company is the collective investment in transferable securities and/or other liquid financial assets of capital raised from the public operating on the principle of risk-spreading in accordance with the UCITS Regulations.

The Articles contain provisions to the following effect:

- (a) Directors' Authority to Allot Shares. The Directors are generally and unconditionally authorised to exercise all powers of the Company to allot relevant securities, including fractions thereof, up to an amount equal to the authorised but as yet unissued share capital of the Company.
- (b) Variation of rights. The rights attached to any Class may be varied or abrogated with the consent in writing of the holders of three-fourths in number of the issued shares of that Class, or with the sanction of a special resolution passed at a separate general meeting of the holders of the shares of the Class, and may be so varied or abrogated either whilst the Company is a going concern or during or in contemplation of a winding-up but such consent or sanction will not be required in the case of a variation, amendment or abrogation of the rights attached to any shares of any Class if, in the view of the Directors, such variation, amendment or abrogation does not materially prejudice the interests of the relevant Shareholders or any of them. Any such variation, amendment or abrogation will be set out in a supplement to (or re-statement of) the relevant Supplement originally issued in connection with the relevant Shares, a copy of which will be sent to the relevant Shareholders entered on the register on the date of issue of such document and will be binding on the relevant Shareholders. The quorum at any such separate general meeting, other than an adjourned meeting, shall be two persons holding or representing by proxy at least one third of the issued Shares of the Class in question and the quorum at an adjourned meeting shall be one person holding Shares of the Class in guestion or their proxy.
- (c) Voting Rights. The Company may issue Voting Shares and Non-Voting Shares. The Non-Voting Shares carrying no right to notice of, attend or vote at general meetings of the Company or any Fund. In respect of the Voting Shares, subject to any rights or restrictions for the time being attached to any Class or Classes of Voting Shares, on a show of hands every holder who is present in person or by proxy shall have one vote and the holder(s) of subscriber shares present in person or by proxy shall have one vote in respect of all the subscriber shares in issue and on a poll every holder present in person or by proxy shall have one vote for every Voting Share of which they are the holder and every holder of a subscriber share present in person or by proxy shall have one vote in respect of their holding of subscriber shares. On a poll of all the holders of Shares in a Fund, where there is more than one Class of Shares in existence in that Fund, the voting rights of such holders may at the discretion of the Directors be adjusted in such manner, determined by the Directors, so as to reflect the most recently calculated price at which the Shares of each of the Classes in question may be redeemed by the Company. Holders who hold a fraction of a Voting Share may not exercise any voting rights, whether on a show of hands or on a poll, in respect of such fraction of a Voting Share. In accordance with the requirements of the Central Bank, the decision to subscribe for any Class of Shares in respect of which the voting rights are restricted shall be made solely by the investor and any Shareholder of Non-Voting Shares shall have the right to switch their holding to Voting Shares without incurring any fee or charge on such exchange;
- (d) Change in Share Capital. The Company may from time to time by ordinary resolution increase the share capital by such amount and/or number as the resolution may prescribe. The Company may also by ordinary resolution, consolidate and divide all or any of its share capital into shares of larger amount, subdivide its Shares, or any of them, into Shares of

smaller amount or value or cancel any Shares which, at the date of the passing of the resolution, have not been taken or agreed to be taken by any person and reduce the amount of its authorised share capital by the amount of the shares so cancelled or redenominate the currency of any Class of Shares.

(e) Directors' Interests. Subject to the provisions of the Companies Act and provided that the nature and extent of any material interest shall be disclosed as set out below, no Director or intending Director shall be disqualified by their office from contracting with the Company nor shall any such contract or arrangement entered into by or on behalf of any other company in which any Director shall be in any way interested be avoided nor shall any Director so contracting or being so interested be liable to account to the Company for any profit realised by any such contract or arrangement by reason of such Director holding that office or of the fiduciary relationship thereby established.

The nature of a Director's interest must be declared by him at the meeting of the Directors at which the question of entering into the contract or arrangement is first taken into consideration, or if the Director was not at the date of that meeting interested in the proposed contract or arrangement at the next meeting of the Directors held after they became so interested, and in a case where the Director becomes interested in a contract or arrangement after it is made, at the first meeting of the Directors held after they become so interested.

A Director shall not vote at a meeting of the Directors or a committee established by the Directors on any resolution concerning a matter in which they have, directly or indirectly an interest which is material (other than an interest arising by virtue of their interest in shares or debentures or other securities or otherwise in or through the Company) or a duty which conflicts or may conflict with the interest of the Company. A Director shall not be counted in the quorum present at a meeting in relation to such resolution on which they are not entitled to vote.

A Director shall be entitled to vote and be counted in the quorum in respect of any resolutions concerning the following matters, namely:

- (i) the giving of any security, guarantee or indemnity to him in respect of money lent by him to the Company or any of its subsidiary or associated companies or obligations incurred by him at the request of or for the benefit of the Company or any of its subsidiary or associated companies;
- (ii) the giving of any security, guarantee or indemnity to a third party in respect of a debt or obligation of the Company or any of its subsidiary or associated companies for which the Director has assumed responsibility in whole or in part and whether alone or jointly with others under a guarantee or indemnity or by the giving of security;
- (iii) any proposal concerning any offer of shares or debentures or other securities of or by the Company or any of its subsidiary or associated companies for subscription, purchase or exchange in which offer they are or are to be interested as a participant in the underwriting or sub-underwriting thereof;
- (iv) any proposal concerning any other company in which they are interested, directly or indirectly and whether as an officer, shareholder or otherwise howsoever.

The Company by ordinary resolution may suspend or relax the provisions described above to any extent or ratify any transaction not duly authorised by reason of a contravention thereof.

(f) **Borrowing Powers.** Subject to the UCITS Regulations, the Directors may exercise all the powers of the Company to borrow or raise money and to mortgage or charge its

undertaking, property and assets (both present and future) and uncalled capital or any part as security for any debt liability or obligation of the Company thereof provided that all such borrowings shall be within the limits and conditions laid down by the Central Bank.

- (g) **Delegation to Committee.** The Directors may delegate any of their powers to any committee comprising at least one Director. Any such delegation may be made subject to any conditions the Directors may impose, and either collaterally with or to the exclusion of their own powers and may be revoked. Subject to any such conditions, the proceedings of a committee with two or more members shall be governed by the provisions of the Articles regulating the proceedings of Directors so far as they are capable of applying.
- (h) **Retirement of Directors.** The Directors shall not be required to retire by rotation or by virtue of their attaining a certain age.
- (i) Directors' Remuneration. Unless otherwise determined from time to time by the Company in general meeting, the ordinary remuneration of each Director shall be determined from time to time by resolution of the Shareholders in general meeting. Any Director who is appointed as an executive director (including for this purpose the office of chairperson or deputy chairperson) or who serves on any committee, or who otherwise performs services which in the opinion of the Directors are outside the scope of the ordinary duties of a Director, may be paid such extra remuneration by way of fees, commission or otherwise as the Directors may determine. The Directors may be paid all travelling, hotel and other outof-pocket expenses properly incurred by them in connection with their attendance at meetings of the Directors or committees established by the Directors or general meetings or separate meetings of the holders of any Class of Shares of the Company or otherwise in connection with the discharge of their duties.
- (j) **Transfer of Shares.** Subject to the restrictions set out below, the Shares of any holder may be transferred by instrument in writing in any usual or common form or any other form, which the Directors may approve.

The Directors in their absolute discretion and without assigning any reason therefore may decline to register any transfer of a Share to (i) a Prohibited Person or; (ii) an individual under the age of 18 (or such other age as the Directors may think fit) or of unsound mind; or (iii) any person unless the transferee of such Shares would, following such transfer, be the holder of Shares equal to or greater than the Minimum Subscription Amount or the Minimum Initial Investment Amount (as appropriate); or (iv) any person in circumstances where as a result of such transfer the transferor or transferee would hold less than the Minimum Holding; or (v) any person where in respect of such transfer any payment of taxation remains outstanding; or (vi) any person who does not satisfy such anti-money laundering/counter-terrorist financing requirements as the Directors may determine.

The Directors may decline to recognise any instrument of transfer unless it is accompanied by the certificate (if issued) for the Shares to which it relates in respect of one class of Share only, is in favour of not more than four transferees and is lodged at the registered office or at such other place as the Directors may appoint.

- (k) **Right of Redemption.** Shareholders have the right to request the Company to redeem their Shares in accordance with the provisions of the Articles.
- (I) Dividends. The Articles permit the Directors to declare such dividends on any Class of Shares as appears to the Directors to be justified by the profits of the relevant Fund. The Directors may, satisfy any dividend due to holders of Shares in whole or in part by distributing to them in specie any of the assets of the relevant Fund, and in particular any investments to which the relevant Fund is entitled. A shareholder may require the Directors instead of transferring any assets in specie to him, to arrange for a sale of the assets and for payment to the shareholder of the net proceeds of same. Any dividend unclaimed after

a period of six years from the date of declaration of such dividend shall be forfeited and shall revert to the relevant Fund.

- (m) **Funds.** The Directors are required to establish a separate portfolio of assets for each Fund created by the Company from time to time, to which the following shall apply:-
 - (i) the proceeds from the allotment and issue of Shares of each Class in the Fund shall be applied to the Fund established for that purpose, and the investments and the liabilities and income and expenditure attributable thereto shall be applied to such Fund subject to the provisions of the Articles;
 - (ii) any asset derived from any other asset(s) (whether cash or otherwise) comprised in any Fund, shall be applied in the books and records of the Company to the same Fund as the asset from which it was derived and any increase or diminution in the value of such an asset shall be applied to the relevant Fund;
 - (iii) in the event that there are any assets of the Company which the Directors do not consider are attributable to a particular Fund or Funds, the Directors shall, with the approval of the Depositary, allocate such assets to and among any one or more of the Funds in such manner and on such basis as they, in their discretion, deem fair and equitable; and the Directors shall have the power to and may from time to time, with the approval of the Depositary vary the basis in relation to assets previously allocated;
 - (iv) no Shares will be issued on terms that entitle the holder of any Fund to participate in the assets of the Company other than the assets (if any) of the Fund relating to such Shares. If the proceeds of the assets of the relevant Fund are not sufficient to fund the full redemption proceeds payable to each holder for the relevant Fund, the proceeds of the relevant Fund will, subject to the terms for the relevant Fund, be distributed equally among each holder of the relevant Fund pro rata to the amount paid upon the Shares held by each holder. If the realised net assets of any Fund are insufficient to pay any amounts due on the relevant Shares in full in accordance with the terms of the relevant Fund, the relevant holders of that Fund will have no further right of payment in respect of such Shares or any claim against the Company, any other Fund or any assets of the Company in respect of any shortfall;
 - (v) each Fund shall be charged with the liabilities, expenses, costs, charges or reserves of the Company in respect of or attributable to that Fund; and
 - (vi) in the event that any asset attributable to a Fund is taken in execution of a liability not attributable to that Fund, the provisions of Section 1406(6) of the Companies Act shall apply.

(n) Fund Exchanges

Subject to the provisions of the Companies Act, the UCITS Regulations, the Articles and the Supplement for each relevant Fund, a holder holding Shares in any Class in a Fund on any Dealing Day shall have the right from time to time to exchange, subject to an Exchange Charge being applied (as described in this Prospectus), all or any of such Shares for Shares of another Class of that Fund or a separate Fund (such Class being an existing Class or a Class agreed by the Directors to be brought into existence with effect from that Dealing Day).

(o) Termination of a Fund

(i) any Fund may be terminated by the Directors, in their sole and absolute discretion, by notice in writing to the Depositary in any of the following events:-

- (A) if at any time the Net Asset Value of the relevant Fund shall be less than the Minimum Fund Size as may be determined by the Directors in respect of that Fund and disclosed in the relevant Supplement; or
- (B) if any Fund shall cease to be authorised or otherwise officially approved; or
- (C) if any law shall be passed which renders it illegal or in the opinion of the Directors impracticable or inadvisable to continue the relevant Fund; or
- (D) if there is a change in material aspects of the business, in the economic or political situation relating to a Fund which the Directors consider would have material adverse consequences on the investments of the Fund; or
- (E) if the Directors shall have resolved that it is impracticable or inadvisable for a Fund to continue to operate having regard to prevailing market conditions and the best interests of the Shareholders.
- (ii) the Directors shall give notice of termination of a Fund to the Shareholders in the relevant Fund and by such notice fix the date at which such termination is to take effect, which date shall be for such period after the service of such notice as the Directors shall in their sole and absolute discretion determine;
- (iii) with effect on and from the date as at which any Fund is to terminate or in the case of (A) below such other date as the Directors may determine:-
 - (A) No Shares of the relevant Fund may be issued or sold by the Company;
 - (B) The Investment Manager shall, on the instructions of the Directors, realise all the assets then comprised in the relevant Fund (which realisation shall be carried out and completed in such manner and within such period after the termination of the relevant Fund as the Directors think advisable);
 - (C) The Depositary shall, on the instructions of the Directors from time to time, distribute to the Shareholders in proportion to their respective interests in the relevant Fund all net cash proceeds derived from the realisation of the relevant Fund and available for the purpose of such distribution, provided that the Depositary shall not be bound (except in the case of the final distribution) to distribute any of the monies for the time being in its hands the amount of which is insufficient to pay €1 or its equivalent amount in the relevant currency in respect of each Share of the relevant Fund and provided also that the Depositary shall be entitled to retain out of any monies in its hands as part of the relevant Fund full provision for all costs, charges, expenses, claims and demands incurred, made or apprehended by the Depositary or the Directors in connection with or arising out of the termination of the relevant Fund and out of the monies so retained to be indemnified and saved harmless against any such costs, charges, expenses, claims and demands; and
 - (D) Every such distribution referred to at (C) above shall be made in such manner as the Directors shall, in their sole and absolute discretion, determine but shall be made only against production of the certificates or warrants relating to the Shares of the relevant Fund if issued in respect of which the same is made and upon delivery to the Depositary of such form of request for payment as the Depositary shall in its absolute discretion require. All certificated shall in the case of an interim distribution be enfaced by the Depositary with a memorandum of payments made and in the case of the final distribution shall be surrendered to the Depositary. Any unclaimed proceeds or other cash held by the Depositary may at the

expiration of twelve months from the date upon which the same were payable be paid into court subject to the right of the Depositary to deduct therefrom any expenses it may incur in making such payment;

Notwithstanding anything contained in the Articles, and subject to the requirements of the Competent Authority, the Directors, in their discretion, may resolve at any time to terminate and close any class of Shares which has been created.

- (iv) the Directors shall have the power to propose and implement a reconstruction and/or amalgamation of the Company or any Fund(s) on such terms and conditions as are approved by the Directors subject to the following conditions namely:
 - (A) that the prior approval of the Central Bank has been obtained; and
 - (B) that the Shareholders in the relevant Fund or Funds have been circulated with particulars of the scheme of reconstruction and/or amalgamation in a form approved by the Directors and a special resolution of the Shareholders in the relevant Fund or Funds has been passed approving the said scheme.

The relevant scheme of reconstruction and/or amalgamation shall take effect upon such conditions being satisfied or upon such later date as the scheme may provide or as the Directors may determine whereupon the terms of such scheme shall be binding upon all the Shareholders and the Directors shall have the power to and shall do all such acts and things as may be necessary for the implementation thereof.

(p) Winding up. The Articles contain provisions to the following effect:

- (i) If the Company shall be wound up the liquidator shall, subject to the provisions of the Companies Act, apply the assets of each Fund in such manner and order as they think fit in satisfaction of creditors' claims relating to that Fund.
- (ii) The assets available for distribution amongst the holders shall be applied as follows: first the proportion of the assets in a Fund attributable to each Class of share shall be distributed to the holders of Shares in the relevant Class in the proportion that the number of shares held by each holder bears to the total number of shares relating to each such Class of Shares in issue as at the date of commencement to wind up, secondly, in the payment to the holder(s) of the Subscriber Shares of sums up to the nominal amount paid thereon out of the assets of the Company not attributable to any Class of Share. In the event that there are insufficient assets to enable such payment in full to be made, no recourse shall be had to the assets of the Company attributable to each Class of Shares; and thirdly, any balance then remaining and not attributable to any of the Classes of Shares shall be apportioned pro-rata as between the Classes of shares based on the Net Asset Value attributable to each Class of Shares as at the date of commencement to wind up and the amount so apportioned to a Class shall be distributed to holders pro-rata to the number of Shares in that Class of Shares held by them.
- (iii) A Fund may be wound up pursuant to Section 1407 of the Companies Act and in such event the winding up provisions of the Articles shall apply mutatis mutandis in respect of that Fund.
- (iv) If the Company shall be wound up (whether the liquidation is voluntary, under supervision or by the court) the liquidator may, with the authority of a special resolution of the relevant Shareholders and any other sanction required by the Companies Act, divide among the holders of Shares of any Class or Classes within a Fund in specie the whole or any part of the assets of the Company relating to that Fund, and whether or not the assets shall consist of property of a single kind, and

may for such purposes set such value as they deems fair upon any one or more Class or Classes of property, and may determine how such division shall be carried out as between all the Shareholders or the holders of different Classes of Shares as the case may be. The liquidator may, with the like authority, vest any part of the assets in trustees upon such trusts for the benefit of Shareholders as the liquidator, with the like authority, shall think fit, and the liquidation of the Company may be closed and the Company dissolved, but so that no Shareholder shall be compelled to accept any assets in respect of which there is a liability. A Shareholder may request the liquidator, instead of transferring the assets in specie to it, to arrange for a sale of them and to pay the Shareholder the net sales proceeds of same instead.

- (q) Share Qualification. The Articles do not contain a share qualification for Directors.
- (r) Change of Name. In the event that UBS Limited ceases to act as the Promoter of the Company, and a company within the Promoter's group is not appointed in its place, then, prior to or immediately following such termination becoming effective, the Directors will arrange to convene an extraordinary general meeting to propose that the name of the Company be changed to a name which will not reflect any involvement on the part of UBS Limited (or any of its Affiliates) with the Company. At any such extraordinary general meeting called to change the name, those Shareholders who (being individuals) are present in person or by proxy or (being corporations) are present by proxy or by a duly authorised representative and entitled to vote and who vote on a poll in favour of the resolution proposed to change the name of the Company shall collectively have such total number of votes as is one or more than the number of votes which are required to be cast on such a poll for the said special resolution to be carried. Such a change of name shall take place in accordance with the provisions of the Companies Act and the requirements of the Central Bank.

14 Miscellaneous

14.1 Remuneration Policy, Complaint Handling Policy, Whistleblower Policy , Voting Policy and Cyber Security Policy

The Manager has a remuneration policy in place to ensure compliance with UCITS V. This remuneration policy imposes remuneration rules on staff and senior management within the Company whose activities have a material impact on the risk profile of the Funds. The Manager will ensure that its remuneration policies and practices are consistent with sound and effective risk management, will not encourage risk-taking which is inconsistent with the risk profile of the Funds and the Articles, and will be consistent with UCITS V. The Manager will ensure that the remuneration policy is at all times consistent with the business strategy, objectives, values and interests of the Company, the Funds and Shareholders, and includes measures to ensure that all relevant conflicts of interest may be managed appropriately at all times.

The Manager has a best execution, conflict of interest, complaint handling, remuneration, engagement and voting policy in place.

Further details with regard to the remuneration, complaint handling and engagement policies, including, but not limited; (i) a description of how remuneration and benefits are calculated; (ii) the identities of persons responsible for awarding the remuneration and benefits; and (iii) the composition of the remuneration committee, where such a committee exists are available at https://www.ubs.com/ch/en/assetmanagement/fund-management/fund-management/fund-management/fund-management-companies/fmie.html. A paper copy of the remuneration policy, may be obtained free of charge on request from the Manager.

The Manager also has a whistle-blower policy in place, which is available on www.ubs.com/global/en/contact/ubs-staff-misconduct.html.

14.2 The Company has a cyber security policy in place. Engagement Program

The engagement program aims to prioritize/select companies where UBS Asset Management has identified concerns or thematic topics on particular ESG factors. These companies are selected from across the universe of companies in which UBS Asset Management invests using a top-down approach in accordance with our principles, as outlined in the Global Stewardship Policy. The prioritization process determines if and when engagement with a company is required. If a company is selected for the Engagement Program, engagement dialogue will generally last for at least two years. This is not an indication that sustainability related engagement has taken place with respect to companies in this portfolio during any given time period or that the companies in this portfolio were chosen with the goal to actively engage. Information on UBS Asset Management's selection of companies, engagement activities, prioritization process and understanding of concerns can be found in the UBS Asset Management Stewardship Annual Report and Stewardship Policy available at the following website:

https://www.ubs.com/global/en/assetmanagement/capabilities/sustainable-investing/stewardshipengagement.html

14.3 Voting

UBS will actively exercise voting rights based on the principles outlined in the UBS Asset Management Proxy Voting policy and UBS Asset Management Stewardship policy, with two fundamental objectives: 1. To act in the best financial interests of our clients to enhance the long-term value of their investments. 2. To promote best practice in the boardroom and encourage strong sustainability practices. This is not an indication that voting on sustainability related topics has taken place with respect to companies held by a sub-fund during any given time period. For information about voting activities with specific companies please refer to the UBS Asset Management Stewardship Annual Report available at the following website:

https://www.ubs.com/global/en/assetmanagement/capabilities/sustainable-investing/stewardship-engagement.html

14.4 Litigation and Arbitration

The Company is not involved in any litigation or arbitration nor are the Directors aware of any pending or threatened litigation or arbitration.

- 14.5 Directors' Interests
 - 14.5.1 At the date of this Prospectus, no Director has any interest, direct or indirect, in any assets which have been or are proposed to be acquired or disposed of by, or issued to, the Company and save as disclosed below no Director is materially interested in any contract or arrangement subsisting at the date hereof which is unusual in its nature and conditions or significant in relation to the business of the Company.
 - 14.5.2 At the date of this Prospectus none of the Directors nor any Associated Person have any beneficial interest in the share capital of the Company or any options in respect of such capital.
 - 14.5.3 Markus Goetschi, Alan White, Marie Antoinette (Nina) Petrini, Naomi Daly and William Kennedy are Directors of the Company. Naomi Daly is also a director of the Manager.
 - 14.5.4 Markus Goetschi, Marie Antoinette (Nina) Petrini and Alan White are employees of the UBS AG Group or affiliate companies. Directors who are employees of the UBS AG Group or affiliate companies shall not receive a fee for their role as Directors of the Company, but may be paid all travelling, hotel and other out-of-pocket expenses properly incurred by them in connection with their attendance at meetings of the Directors or site meetings with any of the Company's delegates or otherwise in connection with the discharge of their duties.
- 14.6 Anti-Money Laundering and Counter Terrorist Financing Measures

In order to comply with the anti-money laundering and counter-terrorism financing regulations, the Administrator, on the Company's behalf, will require from any subscriber or Shareholder a detailed verification of the identity of such subscriber or Shareholder, the identity of the beneficial owners of such subscriber or Shareholder, the source of funds used to subscribe for Shares, or other additional information which may be requested from any subscriber or Shareholder for such purposes from time to time. The Company and the Administrator each reserve the right to request such information as is necessary to verify the identity of an applicant and where applicable, the beneficial owner pursuant to the Beneficial Ownership Regulations 2019 (SI 110 of 2019) or as otherwise required.

Any natural person who ultimately owns or controls 25% plus one share or an ownership interest in the entity shall inform the Manager and the Company thereof. The percentage of

25% plus one share or an ownership interest of more than 25% are considered as an indication of direct or indirect ownership or control, which means that a natural person may also be considered as a beneficial owner of a corporate entity even if the 25% threshold of ownership or control in that corporate entity is not met by them.

14.7 Material Contracts

The following contracts have been entered into otherwise than in the ordinary course of the business intended to be carried on by the Company and are or may be material:

14.7.1 The **Management Agreement** dated 25 June 2010, as amended from time to time, between the Company and the Manager. The Management Agreement provides that the appointment of the Manager will continue unless and until terminated by either the Manager or the Company giving to the other party not less than 90 days' written notice although in certain circumstances the Management Agreement may be terminated forthwith by notice in writing by either the Manager or the Company to the other party. The Management Agreement contains certain indemnities payable out of the assets of the relevant Fund in favour of the Manager which are restricted to exclude matters resulting from the fraud, bad faith, wilful default or negligence of the Manager in the performance or non-performance of its obligations and duties.

The Management Agreement contains limited recourse provisions under which the recourse against the Company of the Manager in respect of any claims arising under or in relation to the Management Agreement is expressed to be limited to the Fund established in respect of the Shares to which such claims relate, and the Manager will have no recourse to any other assets of the Company or any other Fund in respect of any such claims. If following the realisation of all of the assets of the relevant Fund and the application of such realisation proceeds in payment of all claims relating to the relevant Fund and all other liabilities (if any) of the Company ranking pari passu with or senior to such claims which have recourse to the relevant Fund, the claims are not paid in full, (a) the amount outstanding in respect of such claims will be automatically extinguished, (b) the Manager will have no further right of payment in respect thereof and (c) the Manager will not be able to petition for the winding-up of the Company or the termination of any other Fund as a consequence of any such shortfall provided however that (a) and (b) above shall not apply to any assets of the Fund that may be subsequently held or recouped by the Fund.

14.7.2 The **Investment Management Agreement** dated 25 June 2010, as amended from time to time, between the Manager and the Investment Manager. The Investment Management Agreement provides that the appointment of the Investment Manager will continue unless and until terminated by either party giving to the other not less than 90 days' written notice although in certain circumstances the Investment Management Agreement agreement may be terminated forthwith by notice in writing by either party to the other. The Investment Management Agreement contains certain indemnities in favour of the Investment Manager which are restricted to exclude matters resulting from the fraud, bad faith, wilful default or negligence of the Investment Manager in the performance or non-performance of its obligations and duties.

The Investment Management Agreement contains limited recourse provisions under which the recourse against the Manager of the Investment Manager in respect of any claims arising under or in relation to the Investment Management Agreement is expressed to be limited to the assets of the Manager and any claims the Manager has in relation to the relevant Fund established in respect of the Shares to which such claims relate, and the Investment Manager will have no recourse to any other assets of the Company or any other Fund in respect of any such claims. If following the realisation of all of the assets of the Manager and any claims the Manager has in relation to the relevant Fund and the application of such realisation proceeds in payment of all claims relating to the relevant Fund (if any) and all other liabilities (if any) of the Company ranking pari passu with or senior to such claims which have recourse to the assets of the Manager, the claims are not paid in full, (a) the amount outstanding in respect of such claims will be automatically extinguished, (b) the Investment Manager will have no further right of payment in respect thereof and (c) the Investment Manager will not be able to petition for the winding-up of the Manager, the Company or the termination of any other Fund as a consequence of any such shortfall provided however that (a) and (b) above shall not apply to any assets of the Fund payable to the Manager that may be subsequently held or recouped by the Fund.

14.7.3 The **Depositary Agreement** dated 12 October 2016, as amended from time to time, between the Company and the Depositary. The Depositary Agreement provides that the appointment of the Depositary will continue unless and until terminated by either party giving to the other not less than 90 days' written notice although in certain circumstances the Depositary Agreement may be terminated forthwith by notice in writing by either party to the other.

The Depositary Agreement shall continue in force unless and until terminated by either party giving not less than 90 days' prior written notice to the other, although termination may be immediate in certain circumstances, such as the insolvency of the Depositary. Upon an (envisaged) removal or resignation of the Depositary, the Company shall with due observance of the applicable requirements of the Central Bank, appoint a successor Depositary. The Depositary may not be replaced without the approval of the Central Bank.

The Depositary Agreement shall be governed by the laws of Ireland and the courts of Ireland shall have non-exclusive jurisdiction to hear any disputes or claims arising out of or in connection with the Depositary Agreement.

14.7.4 The Administration Agreement dated 25 June 2010, as amended from time to time, between the Manager, the Company and the Administrator. The Administration Agreement provides that the appointment of the Administrator will continue unless and until terminated by either party giving to the other not less than 90 days' written notice although in certain circumstances the Administration Agreement may be terminated forthwith by notice in writing by either party to the other. The Administration Agreement contains certain indemnities payable by the Manager or out of the assets of the relevant Fund in favour of the Administrator which are restricted to exclude matters arising by reason of the recklessness, bad faith, negligence, fraud, or wilful default of the Administrator, its directors, officers, servants, employees, or agents in the performance of its or their obligations.

The Administration Agreement contains limited recourse provisions under which the recourse against the Company of the Administrator, its permitted delegates, servants or agents in respect of any claims arising under or in relation to the Administration Agreement is expressed to be limited to the Fund established in respect of the Shares to which such claims relate, and the Administrator will have no recourse to any other assets of the Company or any other Fund in respect of any such claims. If following the realisation of the assets of the relevant Fund and the application of such realisation proceeds in payment of all claims relating to the relevant Fund (if any) and all other liabilities (if any) of the Company ranking pari passu with or senior to the claims which have recourse to the relevant Fund, such claims are not paid in full, (a) the amount outstanding in respect of such claims will be automatically extinguished, (b) the Administrator will have no further right of payment in respect thereof and (c) the Administrator will not be able to petition for the winding-up of the Company or the termination of any other Fund as a consequence of any such shortfall provided however that (a) and (b) above shall not apply to any assets of the Fund that may be subsequently held or recouped by the Fund.

14.7.5 The **Distribution Agreement** dated 25 June 2010, as amended from time to time, between the Manager and Distributor. The Distribution Agreement provides appointment of the Distributor will continue unless and until terminated by either party giving to the other not less than 180 days' written notice although in certain circumstances the Distributor Agreement may be terminated forthwith by notice in writing by either party to the other.

The Distribution Agreement contains limited recourse provisions under which the recourse against the Company of the Distributor in respect of any claims arising under or in relation to the Distribution Agreement is expressed to be limited to the Fund established in respect of the Shares to which such claims relate, and the Distribution will have no recourse to any other assets of the Company or any other Fund in respect of any such claims. If following the realisation of the assets of the relevant Fund (if any) and subject to the application of such realisation proceeds in payment of all claims relating to the relevant Fund (if any) and all other liabilities (if any) of the Company ranking pari passu with or senior to such claims which have recourse to the relevant Fund, the claims are not paid in full, (a) the amount outstanding in respect of such claims will be automatically extinguished. (b) the Distributor will have no further right of payment in respect thereof and (c) the Distributor will not be able to petition for the winding-up of the Company or the termination of any other Fund as a consequence of any such shortfall provided however that (a) and (b) above shall not apply to any assets of the Fund that may be subsequently held or recouped by the Fund.

14.7.6 Additional Contracts. In addition to the above, the Company may enter into additional contracts relating to the provision of paying agent, facilities agent, correspondent bank or other similar services as may be required in connection with an offer of Shares into a particular jurisdiction from time to time. The provision of such services shall be on arm's length commercial terms for the Company for which fees shall be charged at normal commercial rates and expenses are to be reimbursed.

14.8 General

As of the date of this Prospectus, the Company does not have any loan capital (including term loans) outstanding or created but unissued or any outstanding mortgages, charges, debentures or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts, liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantee or other contingent liabilities.

Save as disclosed under the heading "Directors' Interests" above, no Director has any interest in the promotion of or in any property acquired or proposed to be acquired by the Company.

Save as may result from the entry by the Company into the agreements listed under the heading "Material Contracts" above or any other fees, commissions or expenses discharged, no amount or benefit has been paid or given or is intended to be paid or given to any promoter of the Company.

No commissions, discounts, brokerages or other special terms have been paid or granted by the Company, or are payable by the Company for subscribing or agreeing to subscribe, or procuring or agreeing to procure subscriptions, for any Shares or loan capital of the Company. To facilitate the marketing and distribution of a Fund, the Distributor or any of its affiliates, may purchase Shares in such Fund, at the relevant Fund's Launch Date (as defined in the Supplement for the relevant Fund) or at any time thereafter.

The Manager may pay a portion of its fee to distributors, dealers or other entities that assist it in the performance of its duties or provide services, directly or indirectly, to the Funds or their Shareholders and may enter into private arrangements on a negotiated basis with a holder or prospective holder of Shares. The selection of holders or prospective holders of Shares with whom such private arrangements may be made and the terms on which the Manager or their respective affiliates, designees or placement agents may enter into such private arrangements are a matter for the relevant entity, except that as a condition of any such arrangements, the Company will not thereby incur any obligation or liability whatsoever.

14.9 Documents for Inspection

Copies of the following documents may be inspected at the offices of the Administrator at its address as set out in this Prospectus, during usual business hours on weekdays, except Saturdays, Sundays and public holidays:

- (a) the Articles;
- (b) up to date information regarding the Depositary's duties and conflicts of interest; and
- (c) a description of any safe-keeping functions delegated by the Depositary, a list of delegates and sub-delegates and any conflicts of interest that may arise from such delegation.

A list of the Depositary's sub-delegates, as at the date of this Prospectus, is set out in Appendix IV.

Copies in writing of the Articles, this Prospectus, the KIID for each Share Class of each Fund, and, after publication thereof, the periodic reports and accounts may be obtained from the Administrator on request, free of charge.

Copies of a list of past and current directorships and partnerships held by each Director over the last five years are available for inspection upon request to the Company.

14.10 Seeding Arrangements

In case of Funds with low assets under management, the sponsor of the Company, UBS Asset Management Switzerland AG (or one of its affiliates) may pay a seeding fee to investors and market participants who are investing a minimum amount as set forth in a separate agreement. The seeder represents and warrants to hold such investment for an agreed time period as set forth in a separate agreement.

Any seeding fees paid by UBS Asset Management Switzerland AG (or one of its affiliates) will solely be borne by UBS Asset Management Switzerland AG (or one of its affiliates) and will not be charged to the relevant Fund or to the Company.

UBS Asset Management Switzerland AG and the Manager believe that such arrangement will help grow small-sized sub-funds and will benefit other investors in such Funds. As many of the expenses of the Funds are fixed, a higher amount of assets under management may reduce a Fund's expenses per share and a lower amount of assets under management may increase a Fund's expenses per share.

14.11 Rebates

The Manager or its agents may pay rebates directly to investors in Funds share class.

Rebates are permitted provided that they:

- are paid out of fees of the Manager or its agents and thus do not additionally impair the assets of the Fund:
- are granted on the basis of objective criteria;
- are granted to the same extent to all investors who meet the objective criteria equally and demand rebates;
- increase the quality of the service for which the rebate is granted (e.g. by contributing to higher assets of the sub-fund that can lead to a more efficient management of the assets and a reduced liquidation probability of the sub-fund and / or a reduction of the fixed costs pro rate for all investors) and all investors bear their fair share of the sub-fund's fees and costs.

The objective criterion for granting rebates is:

- the total assets held by the investor in the Class of the Fund that qualifies for rebates; Additional criteria may be:
- the total assets in UBS collective investment schemes held by the investor and / or; the region where the investor is domiciled.

Upon request of any investor, the Manager or its agents shall disclose the further details of the rebates free of charge.

14.12 Retrocessions

The Manager may pay retrocessions to cover the distribution activities of the Company.

Appendix I – Definitions

"Accounting Period"	means a period ending on 30 June of each year.
"Administration Agreement"	means the administration agreement dated 25 June 2010 between the Company, the Manager and the Administrator as amended, supplemented or otherwise modified from time to time in accordance with the requirements of the Central Bank.
"Administrative Expenses"	means the administrative expenses defined in section 11.1.2.
"Administrator"	means State Street Fund Services (Ireland) Limited or any other person or persons for the time being duly appointed administrator in succession to the said State Street Fund Services (Ireland) Limited in accordance with the requirements of the Central Bank.
"Affiliate"	means any legal or natural person which in relation to the person concerned is (i) a holding company, (ii) a subsidiary of any such holding company; (iii) a subsidiary or (iv) controlled directly or indirectly by the person concerned and "control" of an entity for this purpose means the power, direct or indirect, to direct or cause the direction of the management and policies of such entity whether by contract or otherwise and, in any event and without limitation of the foregoing, any entity owning more than 50% of the voting securities of a second entity shall be deemed to control that second entity.
"AIF"	means an alternative investment fund as defined in regulation 5(1) of the European Union (Alternative Investment Fund Managers) Regulations 2013 (S.I. No. 257 of 2013) and/or any other collective investment undertaking meeting the criteria outlined in Regulation 68(1)(e) of the UCITS Regulations including, where relevant and in the event of the United Kingdom becoming a third country, UCITS authorised by the Financial Conduct Authority in the United Kingdom in accordance with Directive 2009/65/EC of the European Parliament and of the Council, as amended, supplemented, consolidated or otherwise modified from time to time.
"Application Form"	means the form which must be submitted, initially by electronic means or by fax, with the original to follow as soon as possible thereafter. It must be submitted upon an initial application for the purposes of account opening or for subsequent applications if the investors' details or circumstances have changed from when this form was initially submitted.

"Approved Counterparty"	means such entity or entities selected by the Manager to act as the counterparty to FDIs to which the Fund is a party, provided always that the relevant entity is, in relation to OTC derivatives, one falling within a category permitted by the Central Bank Rules. The Approved Counterparty will be a member of the UBS AG Group.
"Articles"	means the Memorandum and Articles of Association of the Company.
"Associated Person"	a person is associated with a Director if, and only if, they are:
	(a) that Director's spouse, parent, brother, sister or child;
	(b) a person acting in their capacity as the trustee of any trust, the principal beneficiaries of which are the Director, their spouse or any of their children or any body corporate which they control;
	(c) a partner of that Director.
	A company will be deemed to be connected with a Director if it is controlled by that Director.
"Authorised Participant"	means an entity or person authorised by the Company for the purposes of subscribing for and redeeming Shares in a Fund.
"Base Currency"	means in relation to any Fund such currency as is specified in the Supplement for the relevant Fund.
"Benchmark Regulation"	means Regulation (EU) 2016/1011 of the European Parliament and of the Council of 8 June 2016 on indices used as benchmarks in financial instruments and financial contracts or to measure the performance of investment funds.
"Business Day"	means a day defined as a Business Day on the Website or such other day or days as may be set out in the Supplement for the relevant Fund or any such additional day as the Directors may in their discretion determine as a Business Day and notify in advance to Shareholders.
"Calculation Agent"	means the calculation agent of any FDI to which the Fund is a party. The Calculation Agent will be a member of the UBS AG Group, unless otherwise specified in the relevant Supplement.
"CBDF Directive"	means Directive (EU) 2019/160 of the European Parliament and of the Council of 20 June 2019 amending Directives 2009/65/EC and 2011/61/EU with regard to cross-border distribution of collective investment undertakings as may be amended, supplemented, consolidated, substituted in any form or otherwise modified from time to time.

"Central Bank" means the Central Bank of Ireland or any successor authority. "Central Bank Regulations" means the Central Bank (Supervision and Enforcement) Act 2013 (Section 48(1)) (Undertakings For Collective Investment in Transferable Securities) Regulations 2019 as may be amended, supplemented, consolidated, substituted in any form or otherwise modified from time to time. "Central Bank Rules" means the Central Bank Regulations and any other statutory regulations, rules, conditions, instrument, notices. requirements or guidance of the Central Bank issued from time to time applicable to the Company pursuant to the UCITS Regulations. "CJA" means the Criminal Justice (Money Laundering and Terrorist Financing) Acts 2010 to 2021. "Class(-es)" means the class or classes of Shares relating to a Fund where specific features with respect to exchange charge, minimum subscription amount, dividend policy, hedged/ unhedged, investor eligibility criteria, voting rights or other specific features may be applicable. The details applicable to each Class will be pre-determined and as described in the Supplement for the relevant Fund. means any entity affiliated with one or more Relevant Stock "Clearing Agent" Exchanges and which facilitates the validation, delivery and settlement of transactions in the Company's Shares. "Clearstream" means Clearstream Banking S.A., Luxembourg. "Collateral" means assets delivered as defined under the relevant credit support annex, securities lending arrangement or Repurchase Transaction for a Fund and which are acceptable collateral in each case in accordance with the Collateral Policy. "Collateral Policy" means the Manager's collateral policy as it relates to the Company as summarised in the Prospectus. "Company" means UBS (Irl) Fund Solutions plc. "Companies Act" means the Companies Act 2014 including any regulations issued pursuant thereto, insofar as they apply to open-ended investment companies with variable capital. "Connected Person" means the persons defined as such in the section headed "Potential Conflicts of Interest". "Courts' Service" means the entity that is responsible for the administration of moneys under the control or subject to the orders of the courts.

means Regulation (EU) 2019/1156 as may be amended, supplemented, consolidated, substituted in any form or

otherwise modified from time to time.

"CBDF Regulations"

"CRS"	means the Standard for Automatic Exchange of Financial Account Information approved on 15 July 2014 by the Council of the Organisation for Economic Cooperation and Development, also known as the Common Reporting Standard, and any bilateral or multilateral competent authority agreements, intergovernmental agreements and treaties, laws, regulations, official guidance or other instrument facilitating the implementation thereof and any law implementing the Common Reporting Standard.

- "CSDs" (and each a "CSD") means a recognised clearing system which is a national settlement systems for individual national markets. For Funds that issue Shares through an ICSD, CSDs would be participants in an ICSD.
- "Data Protection Legislation" means, from 25 May 2018 onwards, the EU data protection regime introduced by the General Data Protection Regulation (Regulation 2016/679).
- "Dealing Day" means in relation to each Fund such day or days as is specified in the relevant Supplement or such other day(s) as the Directors may with the approval of the Depositary determine and notify in advance to Shareholders provided always that there shall be at least one Dealing Day per fortnight during each calendar month.
- "Dealing Deadline" means in relation to applications for subscription, exchange or redemption of Shares in a Fund, the dates and times specified in the Supplement for the relevant Fund.
- "Dematerialised Form" means in relation to Shares, means Shares, title to which is recorded as being in uncertificated form and which may be transferred by means of a computer based settlement system in accordance with the Companies Act.
- "Depositary" means State Street Custodial Services (Ireland) Limited or such other company as may from time to time be appointed to provide depositary services to the Company in accordance with the requirements of the Central Bank.
- "Depositary Agreement" means the agreement made between the Company and the Depositary dated 12 October 2016 as amended, supplemented or otherwise modified from time to time in accordance with the requirements of the Central Bank.
- **Depot Account** means a depot account maintained by the Depositary (or its delegate) in the relevant clearing and settlement system through which subscription monies and redemption proceeds for the relevant Fund are channelled.
- "Directors" means the directors of the Company.
- "Distributor" means UBS Asset Management Switzerland AG.
- "Disruption Events" means a Market Disruption Event, a Force Majeure Event or an Index Disruption and Adjustment Event.

"EEA"	means the European Economic Area (Member States, Iceland, Norway, and Liechtenstein).
"EEA Member State"	means the member states of the European Economic Area, the current members at the date of this Prospectus being the EU Member States, Iceland, Liechtenstein and Norway.
"EMIR"	means Regulation (EU) No 648/2012 on OTC derivatives, central counterparties and trade repositories.
"ETF"	means exchange traded fund(s).
"EU"	means the European Union.
"Euroclear"	means Euroclear Bank S.A./N.V.
"Exceptional Expenses"	means the exceptional expenses defined as such in the section headed "Fees & Expenses".
"Exchange Charge"	means the charge, if any, payable on the exchange of Shares as is specified in the Supplement for the relevant Fund.
"Exempt Irish Shareholder"	means:
	(a) a qualifying management company within the meaning of Section 739B(1) TCA;
	(b) an investment undertaking within the meaning of Section 739B(1) TCA;
	(c) an investment limited partnership within the meaning of Section 739J TCA;
	 (d) a pension scheme which is an exempt approved scheme within the meaning of Section 774 TCA, or a retirement annuity contract or a trust scheme to which Section 784 or 785 TCA applies;
	(e) a company carrying on life business within the meaning of Section 706 TCA;
	(f) a special investment scheme within the meaning of Section 737 TCA;
	(g) a unit trust to which Section 731(5)(a) TCA applies;
	(h) a charity being a person referred to in Section 739D(6)(f)(i) TCA;
	 (i) a person who is entitled to exemption from income tax and capital gains tax by virtue of Section 784A(2) TCA or Section 848B TCA and the Shares held are assets of an approved retirement fund or an approved

minimum retirement fund;

- (j) a person who is entitled to exemption from income tax and capital gains tax by virtue of Section 787I TCA and the Shares held are assets of a personal retirement savings account as defined in Section 787A TCA;
- (k) the National Asset Management Agency;
- (I) the Courts Service;
- (m) a credit union within the meaning of Section 2 of the Credit Union Act 1997;
- (n) an Irish resident company, within the charge to corporation tax under Section 739G(2) TCA, but only where the fund is a money market fund;
- (o) a company which is within the charge to corporation tax in accordance with Section 110(2) TCA in respect of payments made to it by the Company;
- (p) any other person as may be approved by the Directors from time to time provided the holding of Shares by such person does not result in a potential liability to tax arising to the Company in respect of that Shareholder under Part 27, Chapter 1A TCA; and
- (q) the National Treasury Management Agency of Ireland, or a fund investment vehicle within the meaning of Section 739D(6)(kb) TCA;

and where necessary the Company is in possession of a Relevant Declaration in respect of that Shareholder.

means:

- a) Sections 1471 to 1474 of the US Internal Revenue Code of 1986 or any associated regulations or other official guidance;
- b) any intergovernmental agreement, treaty, regulation, guidance or other agreement between the Government of Ireland (or any Irish government body) and the US, UK or any other jurisdiction (including any government bodies in such jurisdiction), entered into in order to comply with, facilitate, supplement, implement or give effect to the legislation, regulations or guidance described in paragraph (a) above; and
- c) any legislation, regulations or guidance in Ireland that give effect to the matters outlined in the preceding paragraphs.

"FATCA"

"FDI"	means a financial derivative instrument (including an OTC derivative) permitted by the UCITS Regulations.
"Flat Fee"	means the flat fee defined in section 11.2.
"Force Majeure Event"	means an event or circumstance (including, without limitation, a systems failure, natural or man-made disaster, act of God, armed conflict, act of terrorism, riot or labour disruption or any similar intervening circumstance) that is beyond the reasonable control of the Investment Manager and that the Investment Manager determines affects the Fund Assets.
"Fund"	means a separate portfolio of assets which is invested in accordance with the investment objective and policies set out in the relevant Supplement and to which all liabilities, income and expenditure attributable or allocated to such portfolio shall be applied and charged and Funds means all or some of the Funds as the context requires or any other portfolios as may be established by the Company from time to time with the prior approval of the Central Bank.
"Fund Assets"	means the transferable securities, financial derivative instruments, other financial instruments and eligible assets invested in by a Fund and cash held by the Fund in accordance with the UCITS Regulations, as further described in the relevant Supplement.
"Global Share Certificate"	means the certificates issued in the name of the Company (as described in further detail under "Form of the Shares and Register".
"Group Companies"	means companies which are included in the same group for the purposes of consolidated accounts, as defined in accordance with Directive 83/349/EEC or in accordance with international accounting rules.
"ICSD"	means International Central Securities Depositary.
"ICSD Participant"	means an accountholder in an ICSD which may include Authorised Participants, their nominees or agents, and who hold their interest in Shares of the Funds settled and/or cleared through the applicable International Central Securities Depositary.
"Index Disruption and Adjustment Events"	means in respect of a Reference Index or a Reference Asset, an event which impacts the ability of the counterparty to perform its obligations under one or more derivative contracts, further information in respect of which is set out at section 4.4.
"Index Provider"	means in relation to a Fund, the entity or person acting by itself or through a designated agent which compiles, calculates and publishes information on the Index corresponding to a Fund and who has licensed the Index to the Company, as specified in the relevant Supplement.

- "Initial Issue Price" means the price per Share (excluding any entry/exit charges as outlined in 11.4 above) at which Shares are initially offered in a Fund for such period as is specified in the Supplement for the relevant Fund.
- "Intermediary" means a person who carries on a business which consists of, or includes, the receipt of payments from an investment undertaking on behalf of other persons, or holds units in an investment undertaking on behalf of other persons.

```
"International Central means Clearstream and Euroclear.
Securities Depositories"
```

- "Investment Manager" means UBS Asset Management (UK) Limited or any other person or persons for the time being duly appointed investment manager of the Company in addition or in succession to UBS Asset Management (UK) Limited and where the Investment Manager has delegated responsibility for the management of the assets of a Fund, the term Investment Manager shall also refer to the sub-investment manager of that particular Fund.
- "Investor Money Regulations" means the Central Bank (Supervision and Enforcement) Act 2013 (Section 48(1)) Investor Money Regulations 2015 for Fund Service Providers, as may be amended from time to time.
- "Irish Resident" means any person resident in Ireland or ordinarily resident in Ireland (as described in the "Ireland Taxation" section of this Prospectus) other than an Exempt Irish Shareholder.
- "ISIN" means the International Securities Identification Number.
- "Issue Price" means the price at which Shares are issued, as further described in the section headed "Issue and Redemption Prices/Calculation of Net Asset Value".
- "KIID" means the key investor information document.
- "Launch Date" means the date on which the Company issues Shares relating to a Fund in exchange for subscription proceeds as set out in the Supplement for each Fund.
- "Management Fee" means the management fee defined in section 11.1.1.
- "Market" means a stock exchange or regulated market which is provided for in the Articles and listed in Appendix II.
- "Market Disruption Event" means the occurrence or existence of one or more of the following events, which occur in relation to any Fund Asset (or to a component of such Fund Asset, the Reference Index or Reference Asset, or any derivative contract related thereto ("Affected Instrument") and this definition is to be construed accordingly):
 - (i) it is not possible to obtain a prompt or accurate price or value (or an element of such price or value) of any

Affected Instrument according to the rules or normal accepted procedures for the determination of such price or value (whether due to the non-publication of such price or value or otherwise);

- the calculation of the price or value of any Affected Instrument is, at the relevant time, in the opinion of the Manager and/or Investment Manager, impractical or impossible to make;
- (iii) there is a reduction in liquidity in any Affected Instrument in the determination of the Manager and/or the Investment Manager;
- any suspension of or limitation is imposed on trading (iv) on any exchanges, quotation systems or "over-thecounter" market where any Affected Instrument is traded; and/or there exists an event or circumstance that prevents or materially limits transactions in any Affected Instrument. For the purpose of this definition, a limitation on the hours and number of days of trading will not constitute a Market Disruption Event if it results from an announced change in the regular business hours of the relevant exchange, provided however that where a limitation on trading imposed during the course of the day by reason of movements in price otherwise exceeding levels permitted by the relevant exchange may, if so determined by the Manager and/or Investment Manager, constitute a Market Disruption Event;
- (v) where the Affected Instrument is not traded on any exchange, quotation system or other similar system, the Manager and/or the Investment Manager is unable to obtain (a) from dealers in the Affected Instrument firm quotations in respect thereof or (b) a subscription or a redemption price of any Affected Instrument according to the rules or normal accepted procedures for such Affected Instrument;
- (vi) the occurrence of any event that generally makes it impossible or impractical to convert any currency which was, immediately prior to the occurrence of such event, a foreign exchange currency, as determined by the Manager and/or Investment Manager;
- (vii) the occurrence of any event that generally makes it impossible or impractical to convert between the currency of the country of issue and/or country of payment of any Affected Instrument and the Base Currency through customary legal channels, as determined by the Manager and/or the Investment Manager;
- (viii) the occurrence of any event that generally makes it impossible or impractical to deliver or transfer (a) the currency from accounts inside the country of issue

		and/or country of payment of any Affected Instrument to accounts outside such country of issue and/or country of payment or (b) the currency of the country of issue and/or country of payment of any Affected Instrument between accounts inside such country of issue and/or country of payment, or to a party that is a non-resident of the country of issue and/or country of payment, as determined by the Manager and/or Investment Manager;
	(ix)	a general moratorium is declared in respect of banking activities in London, Dublin, New York, or TARGET; and/or
	(x)	further Market Disruption Events may apply in respect of a specific Fund and in such instance, additional details shall be included in the Supplement for the relevant Fund.
"Market Makers"	Stock E with th	financial institutions that are member of the Relevant Exchanges and have signed a market making contract e Company or that are registered as such with the nt Stock Exchanges.
"Member State"	being:- Denma Hunga Malta,	a member state of the EU (the current member states Austria, Belgium, Bulgaria, Cyprus, Czech Republic, Irk, Estonia, Finland, France, Germany, Greece, ry, Ireland, Italy, Latvia, Lithuania, Luxembourg, Poland, Portugal, Romania, Slovakia, Slovenia, Sweden, the Netherlands and the United Kingdom).
"MiFID II"		the Markets in Financial Instruments Directive) (Directive 2014/65/EU).
"MiFID II Delegated Directive"	suppler Parlian financia govern provisio	Commission Delegated Directive (EU) of 7 April 2016 menting Directive 2014/65/EU of the European nent and of the Council with regard to safeguarding of al instruments and funds belonging to clients, product ance obligations and the rules applicable to the on or reception of fees, commissions or any monetary monetary benefits.
"Minimum Fund Size"	for eac	such amount (if any) as the Directors may consider th Fund and as set out in the Supplement for the at Fund.
"Minimum Additional Investment Amount"		such amount (whether the investment be in cash or ie) as is specified in the Supplement for the relevant
"Minimum Initial Investment Amount"		such amount (whether the investment be in cash or ie) as is specified in the Supplement for the relevant
"Minimum Holding"		such number of Shares or Shares having such value as is specified in the Supplement for the relevant

"Minimum Amount"	Redemption		such amount (whether the redemption be in cash or <i>cie</i>) as specified in the Supplement for the relevant
"Minimum Amount"	Subscription		s such amount (whether the subscription be in cash or cie) as is specified in the Supplement for the relevant
"Money Market lı	nstruments"		money market instruments permitted by the UCITS ations and as further described in the relevant ement.
"Month"		means	calendar month.
"Net Asset Value Value per Share	e or Net Asset	Share with th headin Asset	in respect of the assets of a Fund or in respect of a of any Class, the amount determined in accordance ne principles set out in this Prospectus under the og "Issue and Redemption Prices/Calculation of Net Value" as the Net Asset Value of a Fund or the Net Value per Share.
"Non-Voting Sha	res"	to noti	a particular Class of Shares that do not carry the right ce of or to attend or vote at general meetings of the any of the relevant Fund.
"OECD"			the Organisation for Economic Co-operation and opment.
"OTC derivative"		means	an FDI which is dealt in an "over-the-counter" market.
"Personal Portfo Undertaking"	lio Investment	some o selecte	an investment undertaking, under the terms of which or all of the property of the undertaking, may be or was, ed by, or the selection or some or all of the property e or was, influenced by:
		(i)	the investor;
		(ii)	a person acting on behalf of the investor;
		(iii)	a person connected with the investor;
		(iv)	a person connected with a person acting on behalf of the investor;
		(v)	the investor and a person connected with the investor; or
		(vi)	a person acting on behalf of both the investor and a person connected with the investor.
		investr been s proper identifi promo	vestment undertaking is not a personal portfolio ment undertaking if the only property which may or has selected was available to the public at the time that the ty is available for selection by an investor and is clearly ed in the investment undertaking's marketing or other tional material. The investment undertaking must also ith all investors on a non-discriminatory basis. In the

case of investments deriving 50% or more of their value from land, any investment made by an individual is limited to 1% of the total capital required.

- "Primary Market Transaction Costs" means the fee amount payable in connection with a subscription or redemption, where there are net subscriptions or net redemptions as the case may be, (including as a result of an exchange from one Class to another Class of the same or a separate Fund), in addition to the value of the Shares subscribed for, or deducted from the value of the Shares redeemed, to cover any market spreads (the difference between the prices at which assets are valued and/or bought or sold), anti-dilution levies, duties and charges and other dealing costs (including Transfer Taxes) relating to the acquisition or disposal of Fund Assets in connection with such subscription, redemption or exchange.
- "Prospectus" means the prospectus and the supplements issued from time to time by the Company as amended, supplemented, consolidated or otherwise modified from time to time.

"Recognised Clearing **and** means any clearing system for the settlement of transactions Settlement System" in relation to the securities designated by the Revenue Commissioners of Ireland as a recognised clearing system for the purposes of Chapter 1(a) of Part 27 of the Taxes Consolidation Act, 1997 which at the date hereof comprise Clearstream Banking SA, Clearstream Banking AG, Euroclear, Crest, National Securities Clearing System, Sicovam SA. SIS Sega Intersettle AG. NECIGEF(Nederlands Centraal Instituut voor Giraal Effectenverkeer B.V.-the Dutch central institute for giro transferred securities), Deutsche Bank AG, Depositary & Clearing System, Central Moneymarkets Office, Depositary Trust Company of New York, Japan Securities Depositary Centre (JASDEC), Monte Titoli SpA, The Canadian Depositary for Securities Ltd. and VPC AB (Sweden).

- "Redemption Fee" means the fee, if any, to be paid out of the Redemption Price which Shares may be subject to, as described under "Important Information" and specified in the relevant supplement.
- "Redemption Form" means the form which may be submitted to make an application to redeem Shares.
- "Redemption Price" means the price at which Shares are redeemed less any Redemption Fee, as further described in the section headed "Issue and Redemption Prices/Calculation of Net Asset Value".

"Reference Asset"	means the basket of securities or other eligible assets whose performance a Fund will aim to replicate, or track the performance of, pursuant to its investment objective and in accordance with its investment policies, as specified in the relevant Supplement. The "Reference Asset" could comprise multiple baskets of securities or other eligible assets, and references to "Reference Asset" shall be read accordingly.
"Reference Index"	means the index of securities or other eligible assets whose performance a Fund will aim to replicate, or track the performance of, pursuant to its investment objective and in accordance with its investment policies, as specified in the relevant Supplement. The Reference Index could comprise several indices, and references to "Reference Index" shall be read accordingly.
"Register"	means the register of Shareholders of the Company.
"Registrar"	means State Street Fund Services (Ireland) Limited.
"Registered Shares"	means Shares which are issued in registered form of which the ownership is registered and documented in the Company's Register.
"Relevant Declaration"	means the declaration relevant to the Shareholders as set out in Schedule 2B TCA.
"Relevant Institution"	means credit institutions authorised in an EEA Member State or credit institutions authorised within a signatory state (other than an EEA Member State) to the Basle Capital Convergence Agreement of July 1988 (which includes the United Kingdom), or credit institutions authorised in a third country deemed equivalent pursuant to Article 107(4) of the Regulation (EU) No 575/2013 of the European Parliament and of the Council of 26 June 2013 on prudential requirements for credit institutions and investment firms and amending EMIR.
"Relevant Stock Exchanges"	means markets on which the Shares of the Funds will be listed such as the SIX Swiss Exchange and/or such other stock exchanges as the Directors may determine from time to time.
"Repurchase Transactions"	means repurchase, reverse repurchase or buy and sell back transactions.
"Revenue Commissioners"	means the Irish Revenue Commissioners.
"Scheduled Maturity Date"	means, with respect to a Fund, the date indicated in the relevant Supplement on which the outstanding Shares will be redeemed, the Fund being thereafter closed. Unless a

"Securities Financing Transactions" or "SFTs"	means repurchase agreements, reverse repurchase agreements, securities lending agreements and any other transactions within the scope of SFTR that a Fund is permitted to engage in.
"Setting Up Costs"	means the costs defined as such in the section headed "Fees & Expenses".
"Settlement Date"	means in respect of receipt of monies for payment of subscription monies or dispatch of monies for the redemption of Shares the dates specified in the Supplement for each Fund.
"SF"	means synthetic fund.
"SFDR"	means Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector, as may be amended, supplemented, consolidated, substituted in any form or otherwise modified from time to time.
"SFTR"	means Regulation (EU) 2015/2365 of the European Parliament and of the Council on transparency of securities financing transactions and of reuse and amending Regulation (EU) No 648/2012, as may be amended, supplemented or replaced from time to time.
"Shares"	means participating shares in the Company and includes, where the context so permits or requires, the Shares in a Fund which may be divided into different Classes, such Shares may be Voting Shares or Non-Voting Shares.
"Shareholders"	means holders of Shares, and each a "Shareholder".
"Structured Finance Securities"	means eligible debt or equity securities or other financial instruments, including asset-backed securities and credit- linked securities, which may be issued by a member of the UBS AG Group.
"STOXX"	means STOXX Ltd, a division of Deutsche Börse AG.
"Subscription Fee"	means the fee, if any, payable to the Distributor (or any other appropriate party at the direction of the Directors) on subscription for Shares as described in the relevant Supplement.
"Subscriptions/Redemptions Accounts"	means the account in the name of the Company through which subscription monies and redemption proceeds and dividend income (if any) for each Fund are channelled in accordance with the Articles, the details of which are specified in the Application Form.
"Supplements"	means the supplements to this Prospectus (each a " Supplement ") and any Supplement issued by the Company in relation to the creation of new Funds and/or share Classes.

"TARGET"	means the Trans-European Automated Real-time Gross settlement Express Transfer system.
"Taxonomy Regulation"	Regulation EU/2020/852 of the European Parliament and of the Council of 18 June 2020 on the establishment of a framework to facilitate sustainable investment, and amending SFDR, as may be supplemented, consolidated, substituted in any form or otherwise modified from time to time.
"TCA"	means the Taxes Consolidation Act, 1997, as amended.
"Total Return Swap" or "TRS"	means a derivative (and a transaction within the scope of SFTR) whereby the total economic performance of a reference obligation is transferred from one counterparty to another counterparty.
"Transaction Fees"	means the fees defined as such under the section headed "Fees & Expenses".
"Transferable Securities"	means transferable securities permitted by the UCITS Regulations and as further described in the relevant Supplement.
"Transfer Taxes"	means all stamp, transfer and other duties and taxes for which the Company may be liable in relation to a Fund for receiving the requisite securities on a subscription for Shares of delivering the requisite securities on redemption of one or more Shares.
"UBS AG Group"	means UBS AG and its Affiliates.
"UCITS"	means an undertaking for collective investment in transferable securities pursuant to the UCITS Directive.
"UCITS Directive"	means the EC Council Directive 85/0111 EEC of 20 December 1985 (OJ No. L375/3 of 31 December 1985) on the Co-ordination of Laws, Regulations and Administrative Provisions relating to UCITS, as amended, supplemented or replaced from time to time.
"UCITS Regulations"	means the European Communities (Undertakings for Collective Investment in Transferable Securities) Regulations 2011 (S.I. 352 of 2011) as amended by the European Union (Undertakings for Collective Investment in Transferable Securities) (Amendment) Regulations 2016, as may be modified, amended, supplemented, consolidated or re-enacted from time to time.
"UCITS V"	means Directive 2014/91/EU of the European Parliament and of the Council of 23 July 2014 amending Directive 2009/65/EC on the coordination of laws, regulations and administrative provisions relating to undertakings for collective investment in transferable securities as regards depositary functions, remuneration and sanctions as amended from time to time and including any supplementing

European Commission dele	gated regulations in force from
time to time.	

- "United Kingdom" or "UK" means the United Kingdom of Great Britain and Northern Ireland.
- "United States" or "U.S. or US" means the United States of America, its territories, possessions and all areas subject to its jurisdiction (including the Commonwealth of Puerto Rico).
- "United States Person" means a person who: (i) is a United States person within the or "U.S. Person" meaning of Section 7701(a)(30) of the US Internal Revenue Code of 1986, as amended, and the Treasury Regulations promulgated thereunder; (ii) is a US person within the meaning of Regulation S under the US Securities Act of 1933 (17 CFR § 230.902(k)); (iii) is a United States person within the meaning of Executive Order 13959; (iv) is not a Non-United States person within the meaning of Rule 4.7 of the US Commodity Futures Trading Commission Regulations (17 CFR § 4.7(a)(1)(iv)); (v) is in the United States within the meaning of Rule 202(a)(30)-1 under the US Investment Advisers Act of 1940, as amended; or (vi) any trust, entity or other structure formed for the purpose of allowing US Persons to invest in the Fund.
- "Unlisted Fund" means, for the purposes of the Company, any Fund which is not an ETF.
- "Valuation Point" means the point in time by reference to which the Net Asset Value of a Fund is calculated, as is specified in the Supplement for the relevant Fund.
- "Voting Shares" means the Shares of a particular Class that carry the right to vote at general meetings of the Company and the relevant Fund.
- "Website" means <u>www.ubs.com/ETF</u> (in the case of the Funds of the Company which are ETFs) or <u>www.ubs.com/funds</u> (in the case of the Funds of the Company which are Unlisted Funds) or such other website for each Fund as set out in the relevant Supplement, on which the Net Asset Value per Share and the capitalisation of the relevant Fund in its Base Currency will be published and on which this Prospectus, the Supplements and any other information in respect of the Company or any of the Funds, including various shareholder communications may be published.

In this Prospectus references to "EUR", "Euro" and "€" are references to the lawful currency of Ireland, references to "GBP", "Sterling" or "£" are to the lawful currency of the United Kingdom, references to "US\$" or "US Dollars" are to the currency of the United States, references to "CHF" or "Swiss Franc" are to the lawful currency of the Switzerland, all references to "Yen", "JPY" or "¥" are to the lawful currency of Japan, all references to "SGD" or "Singapore Dollars" are to the lawful currency of Singapore. All references to the foregoing currencies shall include any successor currency.

Appendix II - Markets

With the exception of permitted investments in unlisted securities and derivative instruments, investments will be restricted to the following stock exchanges and markets listed below in accordance with the regulatory criteria as defined in the Central Bank Regulations.

For the purposes of this Appendix II, reference to "unlisted securities" may include securities that are listed on a market or exchange where such exchange is not set out in the below list in accordance with Regulation 68(1)(c) and 68(2)(a) of the UCITS Regulations. The Central Bank does not issue a list of approved stock exchanges or markets.

1.

- (a) any stock exchange which is:
 - (i) located in any Member State of the European Union (except Malta);
 - (ii) located in a Member State of the EEA (except Liechtenstein);
 - (iii) located in Australia, Canada, Hong Kong, Japan, New Zealand, Switzerland, United States of America or the United Kingdom (at any time it is not an EU Member State); or
- (b) any stock exchange included in the following list:

Bermuda	-	Bermuda Stock Exchange;
Brazil	-	BM&F Bovespa S.A.;
Chile	-	Bolsa de Comercio de Santiago, Bolsa Electronica de Chile and Bolsa de Valparaiso;
Peoples' Republic of China	-	Shanghai Stock Exchange and Shenzhen Stock Exchange;
Egypt	-	Egyptian Stock Exchange;
India	-	Bombay Stock Exchange and National Stock Exchange of India;
Indonesia	-	Indonesia Stock Exchange;
Israel	-	Tel Aviv Stock Exchange;
Republic of Korea	-	Korean Stock Exchange;
Malaysia	-	Bursa Malaysia Securities Berhad and Bursa Malaysia Derivatives Berhad;
Mexico	-	Bolsa Mexicana de Valores and Mercado Mexicano de Derivados;
Pakistan	-	Pakistan Stock Exchange;
Peru	-	Bolsa de Valores de Lima;
Philippines	-	Philippines Stock Exchange;

Russia	-	Moscow Stock Exchange (RTS) and MICEX;
Singapore	-	Singapore Exchange Limited and CATALIST;
South Africa	-	Johannesburg Stock Exchange and the South African Futures Exchange;
Taiwan	-	Taiwan Stock Exchange, Taiwan Futures Exchange and Gre- Tai Securities Market;
Thailand	-	The Stock Exchange of Thailand, Market for Alternative Investments, Bond Electronic Exchange and the Thailand Futures Exchange;
Turkey	-	Istanbul Stock Exchange;
United Arab Emirates (UAE)	-	UAE Nasdaq, Dubai Financial Market and Abu Dhabi Securities Exchange.

(c) any of the following "over the counter" markets:

The market organised by the International Capital Market Association;

The (i) market conducted by banks and other institutions regulated by the Financial Conduct Authority (FCA) and subject to the Inter-Professional Conduct provisions of the FCA's Market Conduct Sourcebook and (ii) market in non-investment products which is subject to the guidance contained in the Non Investment Products Code drawn up by the participants in the London market, including the FCA and the Bank of England;

The market in US government securities conducted by primary dealers regulated by the Federal Reserve Bank of New York and the US Securities and Exchange Commission;

The "over-the-counter" market in the United States conducted by primary and second dealers regulated by the Securities and Exchanges Commission and by the Financial Industry Regulatory Authority (and by banking institutions regulated by the US Comptroller of the Currency, the Federal Reserve System or Federal Deposit Insurance Corporation);

The "over-the-counter" market in Canadian Government Bonds as regulated by the Investment Industry Regulatory Organisation of Canada;

The French market for Titres de Creance Negotiable (**"over-the-counter**" market in negotiable debt instruments)

(d) any of the following electronic exchanges:

NASDAQ.

2 In relation to any exchange traded financial derivative contract, any stock exchange on which such contract may be acquired or sold and which is regulated, operates regularly, is recognised and open to the public and which is (i) located in an EEA Member State, (ii) located in Australia, Canada, Hong Kong, Japan, New Zealand, Switzerland, United States (iii) the Channel Islands Stock Exchange (iv) listed at (c) or (d) above or (v) any of the following:

Sydney Futures Exchange;

BM&F Bovespa;

Toronto Stock Exchange;

Montreal Stock Exchange;

China Financial Futures Exchange;

Hong Kong Futures Exchange;

Osaka Securities Exchange;

Tokyo Stock Exchange;

Tokyo Financial Exchange;

KRX – Korea Futures Market;

Bursa Malaysia Derivatives Market;

New Zealand Futures and Options Exchange;

Singapore International Monetary Exchange;

Singapore Commodity Exchange;

Central Depository (Pte) Limited;

EUREX;

American Stock Exchange;

Chicago Board Option Exchange;

Chicago Board of Trade;

Chicago Mercantile Exchange;

NYSE Euronext;

Philadelphia Stock Exchange;

International Securities Exchange;

ICE Futures US;

New York Mercantile Exchange; and

UAE Nasdaq.

Appendix III – Taxation

Ireland - Taxation

General

The following non-exhaustive statements are by way of a general guide to potential investors and Shareholders only and do not constitute tax or legal advice. Prospective Shareholders should familiarise themselves with and, where appropriate, take advice on the laws and regulations (such as those relating to taxation and exchange controls) applicable to the subscription for, and the holding and redemption of, Shares in the places of their citizenship, residence and domicile. Shareholders and potential investors are therefore advised by the Directors to consult their professional advisers concerning possible taxation or other consequences of purchasing, holding, switching, selling or otherwise disposing of the Shares under the laws of their country of incorporation, establishment, citizenship, residence or domicile.

Shareholders and potential investors should note that the following statements on taxation are based on advice received by the Directors regarding the law and practice in force in the relevant jurisdiction at the date of this Prospectus and proposed regulations and legislation in draft form. As is the case with any investment, there can be no guarantee that the tax position or proposed tax position prevailing at the time an investment is made in the Company will endure indefinitely, as the basis for and rates of taxation can fluctuate.

The following is a brief summary of certain aspects of Irish taxation law and practice relevant to the transactions contemplated in this Prospectus. It is based on the law and practice and official interpretations in effect at the date of this Prospectus, all of which are subject to change.

Irish Taxation

The Directors have been advised that on the basis that the Company is resident in Ireland for taxation purposes the taxation position of the Company and the Shareholders is as set out below.

The Company

The Company will be regarded as resident in Ireland for tax purposes if its central management and control is exercised in Ireland and the Company is not regarded as resident elsewhere. It is the intention of the Directors that the business of the Company will be conducted in such a manner as to ensure that it is Irish resident for tax purposes.

The Directors have been advised that the Company qualifies as an investment undertaking as defined in Section 739B TCA. Under current Irish law and practice, the Company is therefore not chargeable to Irish tax on its relevant income and relevant gains for so long as the Company is resident for tax purposes in Ireland.

Notwithstanding the above, a charge to tax can arise to the Company in respect of the Shareholders on the happening of a "Chargeable Event" in the Company.

A Chargeable Event includes:

- (i) any payment to a Shareholder by the Company in respect of their Shares;
- (ii) any transfer, cancellation, redemption or repurchase of Shares; and
- (iii) any deemed disposal by a Shareholder of their Shares at the end of a "relevant period" (a "**Deemed Disposal**").

A "relevant period" is a period of 8 years beginning with the acquisition of Shares by a Shareholder and each subsequent period of 8 years beginning immediately after the preceding relevant period.

A Chargeable Event does not include:

- (i) any transaction in relation to Shares held in a recognised clearing system;
- (ii) any exchange by a Shareholder effected by way of a bargain made at arm's length by the Company, of Shares in the Company for other Shares in the Company;
- (iii) certain transfers of Shares between spouses or civil partners and former spouses or former civil partners;
- (iv) an exchange of Shares arising on a qualifying amalgamation or reconstruction of the Company with another Irish investment undertaking; or
- (v) the cancellation of Shares in the Company arising from an exchange in relation to a scheme of amalgamation (as defined in Section 739HA TCA).

On the happening of a Chargeable Event, the Company shall be entitled to deduct the appropriate amount of tax on any payment made to a Shareholder in respect of the Chargeable Event. On the occurrence of a Chargeable Event where no payment is made by the Company to the Shareholder, the Company may appropriate or cancel the required number of Shares to meet the tax liability.

Where the Chargeable Event is a Deemed Disposal and the value of Shares held by Irish Resident Shareholders in the Company is less than 10% of the total value of Shares in the Company (or a Fund) and the Company has made an election to the Revenue Commissioners to report annually certain details for each Irish Resident Shareholder, the Company will not be required to deduct the appropriate tax and the Irish Resident Shareholder (and not the Company) must pay the tax on the Deemed Disposal on a self-assessment basis. Credit is available against appropriate tax relating to the Chargeable Event for appropriate tax paid by the Company or the Shareholder on any previous Deemed Disposal. On the eventual disposal by the Shareholder of the Shares, a refund of any unutilised credit will be payable.

Shareholders

The Irish taxation treatment applicable to Shareholders in the Company is set out below and is dependent on which of the following categories into which they fall:

Shareholders whose Shares are held in a Recognised Clearing and Settlement System

Any payments to a Shareholder or any encashment, repurchase, cancellation or transfer of Shares held in a Recognised Clearing and Settlement System will not give rise to a chargeable event in the Company (there is however ambiguity in the legislation as to whether the rules outlined in this paragraph with regard to Shares held in a Recognised Clearing and Settlement System, apply in the case of chargeable events arising on a Deemed Disposal, therefore, as previously advised, Shareholders should seek their own tax advice in this regard). Thus the Company will not have to deduct any Irish taxes on the occurrence of such events.

However, Shareholders who are Irish Resident or Ordinarily Resident in Ireland or who are not Irish Resident or Ordinarily Resident in Ireland but whose Shares are attributable to a branch or agency in Ireland may still have a liability to account for Irish tax on such events, including on a Deemed Disposal.

Where Shares are held in a Recognised Clearing and Settlement System, the obligation falls on the Shareholder (rather than the Company) to self-account for any tax arising on a chargeable event if the Shareholder is Irish Resident, Ordinary Resident and/or a non-Exempted Irish Shareholder. In the case of an individual, tax currently at the rate of 41% should be accounted for by the Shareholder in respect of any distribution or gain arising to the individual Shareholder on an encashment, redemption, cancellation or transfer of Shares by a Shareholder.

Where the investment constitutes a "PPIU", tax at a rate of 60% should be accounted for by the Shareholder. These rates apply where the individual Shareholder has correctly included details of the income in a timely tax return (tax at a rate of 80% applies where details of the payment/disposal are not correctly included in the individual's tax returns).

Where the Shareholder is a company, any payment will be treated as income which is chargeable to tax under Case IV of Schedule D of the TCA.

A Relevant Declaration or approval in relation to appropriate equivalent measures is not required to be made where the Shares are held in a Recognised Clearing and Settlement System. It is the current intention of the Directors that all of the Shares will be held in a Recognised Clearing and Settlement System. If, in the future, the Directors permit Shares to be held in certificated form outside a Recognised Clearing and Settlement System, prospective investors for Shares on subscription and proposed transferees of Shares will be required to complete a Relevant Declaration as a pre-requisite to being issued Shares in the Company or being registered as a transferee of the Shares (as the case may be). A Relevant Declaration will not be required to be completed in this regard where the Company has received approval from the Revenue Commissioners where appropriate equivalent measures have been put in place.

Shareholders who are neither Irish Residents nor Irish Ordinary Residents and their Shares are not held in a Recognised Clearing and Settlement System

Non-Irish Resident Shareholders whose shares are not held in a Recognised Clearing and Settlement System will not be chargeable to Irish tax on the happening of a Chargeable Event provided that either:

- (i) the Company is in possession of a completed Relevant Declaration to the effect that the Shareholder is not an Irish Resident, or
- (ii) the Company is in possession of written notice of approval from the Revenue Commissioners to the effect that the requirement to provide a Relevant Declaration is deemed to have been complied with in respect of that Shareholder (an "Equivalent Measures Waiver") and the written notice of approval has not been withdrawn by the Revenue Commissioners.

If, in these circumstances, the Company is not in possession of a Relevant Declaration or Equivalent Measures Waiver, or the Company is in possession of information which would reasonably suggest that the Relevant Declaration is not or is no longer materially correct, the Company must deduct tax on the happening of a Chargeable Event in relation to such Shareholder. The tax deducted will generally not be refunded.

Intermediaries acting on behalf of non-Irish Resident Shareholders can claim the same exemption on behalf of the Shareholders for whom they are acting. The intermediary must complete a Relevant Declaration that it is acting on behalf of a non-Irish Resident Shareholder.

A non-Irish Resident corporate Shareholder which holds Shares directly or indirectly by or for a trading branch or agency of the Shareholder in Ireland, will be liable for Irish corporation tax on income from the Shares or gains made on the disposal of the Shares in the event that the shares are not held in a Recognised Clearing and Settlement System.

Shareholders who are Irish Resident or Irish Ordinary Resident and their Shares are not held in a Recognised Clearing and Settlement System

(i) Exempt Irish Shareholders

The Company is not required to deduct tax in respect of an Exempt Irish Shareholder whose shares are not held in a Recognised Clearing and Settlement System so long as the Company is in possession of a completed Relevant Declaration from those persons and the Company has no reason to believe that the Relevant Declaration is materially incorrect. The Exempt Irish Shareholder must notify the Company if it ceases to be an Exempt Irish Shareholder. Exempt Irish Shareholders in respect of whom the Company is not in possession of a Relevant Declaration will be treated by the Company as if they are not Exempt Irish Shareholders.

While the Company is not required to deduct tax in respect of such Exempt Irish Shareholders, those Shareholders may themselves be liable to Irish tax on their income, profits and gains in relation to any sale, transfer, redemption, repurchase or cancellation of Shares or dividends or distributions or other payments in respect of their Shares depending on their circumstances. It is the obligation of the Exempt Irish Shareholder to account for such tax to the Revenue Commissioners.

(ii) Irish-Resident Shareholders

Irish Resident Shareholders (who are not Exempt Irish Shareholders) will be liable to tax on the happening of a Chargeable Event. Tax at the rate of 41% will be deducted by the Company on payments made to the Shareholder in relation to the Shares or on the sale, transfer, Deemed Disposal (subject to the 10% threshold outlined above), cancellation, redemption or repurchase of Shares or the making of any other payment in respect of the Shares.

An Irish Resident Shareholder who is not a company and is not an Exempt Irish Shareholder will not be liable to any further income or capital gains tax in respect of any sale, transfer, Deemed Disposal, cancellation, redemption or repurchase, of Shares or the making of any other payment in respect of their Shares.

Where the Irish Resident Shareholder is a company which is not an Exempt Irish Shareholder, and the payment is not taxable as trading income under Schedule D Case I, the amount received will be treated as the net amount of an annual payment chargeable to tax under Schedule D Case IV from the gross amount of which income tax has been deducted. The rate of tax applicable to a Chargeable Event in respect of any Irish tax resident corporate investor in this instance is 25% provided the corporate investor has made a declaration to the Company including its Irish tax reference number.

Where the Irish Resident Shareholder is a company which is not an Exempt Irish Shareholder, and the payment is taxable as trading income under Schedule D Case I, the following provisions apply:

- the amount received by the Shareholder is increased by any amount of tax deducted by the Company and will be treated as income of the Shareholder for the chargeable period in which the payment is made;
- (ii) where the payment is made on the sale, transfer, Deemed Disposal, cancellation, redemption or repurchase of Shares, such income will be reduced by the amount of consideration in money or money's worth given by the Shareholder for the acquisition of those Shares; and

(iii) the amount of tax deducted by the Company will be set off against the Irish corporation tax assessable on the Shareholder in respect of the chargeable period in which the payment is made.

Personal Portfolio Investment Undertaking

An investment undertaking will be considered to be a personal portfolio investment undertaking ("**PPIU**") in relation to a specific Irish Resident Shareholder where that Irish Resident Shareholder can influence the selection of some or all of the property of the undertaking. The undertaking will only be a PPIU in respect of those Irish Resident Shareholders who can influence the selection. A gain arising on a chargeable event in relation to a PPIU will be taxed at the rate of 60%. An undertaking will not be considered to be a PPIU where certain conditions are complied with as set out in Section 739BA TCA.

Currency Gains

Where a currency gain is made by an Irish Resident Shareholder on the disposal of Shares, that Shareholder may be liable to capital gains tax in respect of any chargeable gain made on the disposal.

Stamp Duty

On the basis that the Company qualifies as an investment undertaking within the meaning of Section 739B TCA, no Irish stamp duty will be payable on the subscription, transfer or redemption of Shares. The stamp duty implications for subscriptions for Shares or transfer or redemption of Shares in specie should be considered on a case by case basis.

Capital Acquisitions Tax

No Irish gift tax or inheritance tax (capital acquisitions tax) liability will arise on a gift or inheritance of Shares provided that:

- (i) at the date of the disposition the transferor of the Shares is neither domiciled nor ordinarily resident in Ireland, and, at the date of the gift or inheritance the transferee of the Shares is neither domiciled nor ordinarily resident in Ireland; and
- (ii) the Shares are comprised in the gift or inheritance at the date of the gift or inheritance and at the valuation date.

Other Tax Matters

The income and capital gains received by the Company from securities issued in countries other than Ireland or assets located in countries other than Ireland may be subject to taxes including withholding tax in the countries where such income and gains arise. The Company may not be able to benefit from reduced rates of withholding tax by virtue of the double taxation treaties in operation between Ireland and other countries. The Directors will have sole discretion as to whether the Company will apply for such benefits and may decide not to apply for such benefits if they determine that it may be administratively burdensome, cost prohibitive or otherwise impractical.

In the event that the Company receives any repayment of withholding tax suffered, the Net Asset Value of the Company will not be restated and the benefit of any repayment will be allocated to the then existing Shareholders rateably at the time of repayment.

Special considerations for German investors

It is the intention of the Company to seek German tax reporting status for certain Classes of certain equity Funds. On this basis, in addition to the investment restrictions set out in the Prospectus, the relevant Fund will also adhere to the restriction that a minimum percentage of the Net Asset Value

of the relevant Fund shall be invested in equity participations (the "**Equity Participation Ratio**") as set out in the Relevant Supplement. For the purpose of this investment restriction, reference to "equity participations" includes:

- (1) shares in a company (which may not include depositary receipts) that are admitted to official trading on a stock exchange or admitted to, or included in another organized market which fulfils the criteria of a "regulated market" as defined in Directive 2004/39/EC of the European Parliament and of the Council of 21 April 2004 on markets in financial instruments; and/or
- (2) shares in a company other than a real estate company which is (i) resident in a Member State or in a member state of the European Economic Area, and where it is subject to, and not exempt from corporate income tax; or (ii) is a resident in any other state and subject to corporate income tax of at least 15%; and/or
- (3) units of a UCITS and/or of an AIF that is not a partnership, which as disclosed in their respective investment terms - are permanently invested with a minimum of more than 50% of their values in equity participations (an "Equity Fund") with more than 50% of the units of Equity Funds held by the relevant Fund being taken into account as equity participations; and/or
- (4) units of a UCITS and/or of an AIF that is not a partnership, which as disclosed in their respective investment terms - are permanently invested with a minimum of at least 25% of their values in equity participations (a "**Mixed Fund**") with 25% of the units of Mixed Funds held by the relevant Fund being taken into account as equity participations; and/or
- (5) units of Equity Funds or Mixed Funds that disclose their equity participation ratio in their respective investment terms; and/or
- (6) units of Equity Funds or Mixed Funds that report their equity participation ratio on a weekly basis.

With the exception of the cases described above in paragraphs (3), (4), (5) and (6), units of a UCITS and/or of an AIF which is not a partnership are not considered equity participations.

For purposes of this section, the Equity Participation Ratio does not include equity participations, which are lent out via securities lending program as set out in the Prospectus.

Above targets will not be categorized as investment limits and it cannot be guaranteed that such target will always be achieved.

Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

German investors should consult their tax advisor regarding the tax consequences of an investment in an "equity fund", a "mixed fund" or an "other fund" under the German Investment Tax Act.

Automatic Exchange of Information

The Company is obliged, pursuant to the IGA, Council Directive 2011/16/EU, Section 891E, Section 891F and Section 891G of the TCA and regulations made pursuant to those Sections, to collect certain information about its investors.

The Company will be required to provide certain information to the Revenue Commissioners in relation to the investors (including information in respect of the investor's tax residence status) and also in relation to accounts held by investors. For further information on FATCA or CRS please refer to the website of the Revenue Commissioners at www.revenue.ie/en/business/aeoi/index.html.

Further detail in respect of FATCA and CRS is set out below.

FATCA Implementation in Ireland

On 21 December 2012, the governments of Ireland and the U.S. signed the IGA.

The IGA provides for the automatic reporting and exchange of information in relation to accounts held in Irish "financial institutions" by U.S. persons and the reciprocal exchange of information regarding U.S. financial accounts held by Irish Residents. The Company is subject to these rules. Complying with such requirements will require the Company to request and obtain certain information and documentation from its Shareholders, other account holders and (where applicable) the beneficial owners of its Shareholders and to provide any information and documentation indicating direct or indirect ownership by U.S. Persons to the competent authorities in Ireland. Shareholders and other account holders will be required to comply with these requirements, and non-complying Shareholders may be subject to compulsory redemption and/ or U.S withholding tax of 30% on withholdable payments and/or other monetary penalties.

The IGA provides that Irish financial institutions will report to the Revenue Commissioners in respect of U.S. account-holders and, in exchange, U.S. financial institutions will be required to report to the IRS in respect of any Irish-resident account-holders. The two tax authorities will then automatically exchange this information on an annual basis.

The Company (and/or any of its duly appointed agents) shall be entitled to require Shareholders to provide any information regarding their tax status, identity or residency in order to satisfy any reporting requirements which the Company may have as a result of the IGA or any legislation promulgated in connection with the IGA and Shareholders will be deemed, by their subscription for or holding of Shares to have authorised the automatic disclosure of such information by the Company or any other person to the relevant tax authorities.

OECD Common Reporting Standard

Ireland has provided for the implementation of CRS through Section 891F of the TCA and the enactment of the CRS Regulations.

The CRS is a global OECD tax information exchange initiative which is aimed at encouraging a coordinated approach to disclosure of income earned by individuals and organisations.

Ireland and a number of other jurisdictions have entered or will enter into multilateral arrangements modelled on the Common Reporting Standard for Automatic Exchange of Financial Account Information published by the OECD. The Company is required to provide certain information to the Revenue Commissioners about investors resident or established in jurisdictions which are party to CRS arrangements.

The Company, or a person appointed by the Company, will request and obtain certain information in relation to the tax residence of its shareholders or "account holders" for CRS purposes and (where applicable) will request information in relation to the beneficial owners of any such account holders. The Company, or a person appointed by the Company, will report the information required to the Revenue Commissioners by 30 June in the year following the year of assessment for which a return is due. The Revenue Commissioners will share the appropriate information with the relevant tax authorities in participating jurisdictions.

There can be no assurance as to the timing or impact of any future guidance on the mechanics and scope of FATCA and the CRS on future operations of the Company. Prospective investors should consult their own tax advisor with regard to FATCA and the CRS and the potential consequences of such automatic exchange of information regimes.

Irish Resident and Irish Ordinary Resident for Irish tax purposes

(a) Irish Resident - Company

A company which has its central management and control in Ireland is resident in Ireland irrespective of where it is incorporated. A company which does not have its central management and control in Ireland but which is incorporated in Ireland is resident in Ireland except where the company is regarded as not resident in Ireland under a double taxation treaty between Ireland and another country.

(b) Irish Resident - Individual

An individual will be regarded as being resident in Ireland for a tax year if s/he:

(i) Is present for 183 days or more in the State in that tax year;

or

(ii) has a combined presence of 280 days in the State in a 2 year period, taking into account the number of days spent in the State in that tax year together with the number of days spent in the State in the preceding year.

Presence in a tax year by an individual of not more than 30 days in the State will not be reckoned for the purpose of applying the two year test. Presence in the State for a day means the personal presence of an individual at any stage during the particular day in question.

(c) Irish Ordinarily Resident - Individual

The term "**Irish Ordinarily Resident**" as distinct from "**Irish Resident**", relates to a person's normal pattern of life and denotes residence in a place with some degree of continuity.

An individual who has been resident in the State for three consecutive tax years becomes ordinarily resident with effect from the commencement of the fourth tax year.

An individual who has been ordinarily resident in the State ceases to be ordinarily resident at the end of the third consecutive tax year in which s/he is not resident. Thus, an individual who is resident and ordinarily resident in the State in 2022 and departs from the State in that tax year will remain ordinarily resident up to the end of the tax year ending 31 December 2025.

(d) Irish Resident - Trust

A Trust will generally be Irish resident where all of the trustees are resident in Ireland.

(e) Intermediary

Intermediary means a person who:

- (i) carries on a business which consists of, or includes, the receipt of payments from an investment undertaking on behalf of other persons; or
- (ii) holds shares in an investment undertaking on behalf of other persons.

	- Our ent List of the Depositary 3 Oub-Delegates
MARKET	SUBCUSTODIAN
Albania	Raiffeisen Bank sh.a.
Argentina	Citibank, N.A.
Australia	The Hongkong and Shanghai Banking Corporation Limited
Austria	UniCredit Bank Austria AG
Bahrain	First Abu Dhabi Bank P.J.S.C.
Bangladesh	Standard Chartered Bank
Belgium	BNP Paribas, France
Benin	via Standard Chartered Bank Côte d'Ivoire S.A., Abidjan, Ivory Coast
Bermuda	HSBC Bank Bermuda Limited
Federation of Bosnia and Herzegovina	UniCredit Bank d.d.
Botswana	Standard Chartered Bank Botswana Limited
Brazil	Citibank, N.A.
Bulgaria	Citibank Europe plc, Bulgaria Branch UniCredit Bulbank AD
Burkina Faso	via Standard Chartered Bank Côte d'Ivoire S.A., Abidjan, Ivory Coast
Canada	State Street Trust Company Canada
Chile	Banco De Chile
People's Republic of China	HSBC Bank (China) Company Limited China Construction Bank Corporation
China Connect	Standard Chartered Bank (Hong Kong) Limited
Colombia	Cititrust Colombia S.A. Sociedad Fiduciaria
Costa Rica	Banco BCT S.A.
Croatia	Privredna Banka Zagreb d.d. Zagrebacka Banka d.d
Cyprus	BNP Paribas Securities Services, S.A., Greece (operating through its Athens branch)
Czech Republic	Československá obchodní banka, a.s. UniCredit Bank Czech Republic and Slovakia, a.s.
Denmark	Skandinaviska Enskilda Banken AB (publ), Sweden (operating through its Copenhagen branch)
Egypt	Citibank, N.A.
Estonia	AS SEB Pank
Finland	Skandinaviska Enskilda Banken AB (publ), Sweden (operating through its Helsinki branch)
France	BNP Paribas, S.A.
Germany	State Street Bank International GmbH Deutsche Bank AG
Ghana	Standard Chartered Bank Ghana Plc
Greece	BNP Paribas, S.A.
Guinea-Bissau	via Standard Chartered Bank Côte d'Ivoire S.A., Abidjan, Ivory Coast
Hong Kong	The Hongkong and Shanghai Banking Corporation Limited

MARKET	SUBCUSTODIAN
Hungary	Citibank Europe plc Magyarországi Fióktelepe UniCredit Bank Hungary Zrt.
Iceland	Landsbankinn hf.
India	Deutsche Bank AG Citibank, N.A. The Hongkong and Shanghai Banking Corporation Limited
Indonesia	Standard Chartered Bank Deutsche Bank AG
Israel	Bank Hapoalim B.M.
Italy	Intesa Sanpaolo S.p.A.
Ivory Coast	Standard Chartered Bank Côte d'Ivoire S.A.
Japan	Mizuho Bank, Limited The Hongkong and Shanghai Banking Corporation Limited
Jordan	Standard Chartered Bank
Kazakhstan	JSC Citibank Kazakhstan
Kenya	Standard Chartered Bank Kenya Limited
Republic of Korea	The Hongkong and Shanghai Banking Corporation Limited Deutsche Bank AG
Kuwait	First Abu Dhabi Bank P.J.S.C.
Latvia	AS SEB banka
Lebanon	HSBC Bank Middle East Limited (as delegate of The Hongkong and Shanghai Banking Corporation Limited)
Lithuania	AB SEB bankas
Malawi	Standard Bank PLC
Malaysia	Standard Chartered Bank Malaysia Berhad
Mali	via Standard Chartered Bank Côte d'Ivoire S.A., Abidjan, Ivory Coast
Mauritius	The Hongkong and Shanghai Banking Corporation Limited
Mexico	Banco Nacional de México, S.A.
Morocco	Citibank Maghreb S.A.
Namibia	Standard Bank Namibia Limited
Netherlands	BNP Paribas S.A., France
New Zealand	The Hongkong and Shanghai Banking Corporation Limited
Niger	via Standard Chartered Bank Côte d'Ivoire S.A., Abidjan, Ivory Coast
Nigeria	Stanbic IBTC Bank Plc.
Norway	Skandinaviska Enskilda Banken AB (publ), Sweden (operating through its Oslo branch)
Oman	First Abu Dhabi Bank P.J.S.C.
Pakistan	Deutsche Bank AG Citibank, N.A.
Panama	Citibank, N.A.
Peru	Citibank del Perú, S.A.
Philippines	Standard Chartered Bank
Poland	Bank Handlowy w Warszawie S.A.
Portugal	Citibank Europe plc, Dublin, Ireland

MARKET	SUBCUSTODIAN		
Puerto Rico	Citibank N.A.		
Qatar	HSBC Bank Middle East Limited (as delegate of The Hongkong and Shanghai Banking Corporation Limited)		
Republic of Srpska	UniCredit Bank d.d.		
Romania	Citibank Europe plc, Dublin – Romania Branch		
Russia	AO Citibank		
Saudi Arabia	FAB Capital J.S.C.		
Senegal	via Standard Chartered Bank Côte d'Ivoire S.A., Abidjan, Ivory Coast		
Serbia	UniCredit Bank Serbia JSC		
Singapore	Citibank N.A.		
Slovak Republic	UniCredit Bank Czech Republic and Slovakia, a.s.		
Slovenia	UniCredit Banka Slovenija d.d.		
South Africa	FirstRand Bank Limited Standard Bank of South Africa Limited		
Spain	Citibank Europe plc, Dublin, Ireland		
Sri Lanka	The Hongkong and Shanghai Banking Corporation Limited		
Republic of Srpska	UniCredit Bank d.d.		
Swaziland	Standard Bank Swaziland Limited		
Sweden	Skandinaviska Enskilda Banken AB (publ)		
Switzerland	Credit Suisse (Switzerland) Limited UBS Switzerland AG		
Taiwan - R.O.C.	Standard Chartered Bank (Taiwan) Limited		
Tanzania	Standard Chartered Bank (Tanzania) Limited		
Thailand	Standard Chartered Bank (Thai) Public Company Limited		
Togo	via Standard Chartered Bank Côte d'Ivoire S.A., Abidjan, Ivory Coast		
Tunisia	Union Internationale de Banques		
Turkey	Citibank, A.Ş.		
Uganda	Standard Chartered Bank Uganda Limited		
Ukraine	JSC Citibank		
United Arab Emirates Dubai Financial Market	First Abu Dhabi Bank P.J.S.C.		
United Arab Emirates Dubai International Financial Center	First Abu Dhabi Bank P.J.S.C.		
United Arab Emirates Abu Dhabi	First Abu Dhabi Bank P.J.S.C.		
United Kingdom	State Street Bank and Trust Company, United Kingdom branch		
United States	State Street Bank and Trust Company		
Uruguay	Banco Itaú Uruguay S.A.		

MARKET	SUBCUSTODIAN
	HSBC Bank (Vietnam) Limited
Vietnam	(as delegate of The Hongkong and Shanghai Banking Corporation
	Limited)
Zambia	Standard Chartered Bank Zambia Plc.
Zimbabwe	Stanbic Bank Zimbabwe Limited
ZIIIDabwe	(as delegate of Standard Bank of South Africa Limited)

UBS CMCI COMPOSITE SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS CMCI COMPOSITE SF UCITS ETF** (the "**Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the "**Company**") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "**Central Bank**").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being UBS Constant Maturity Commodity Index Total Return).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes for the performance of the Approved to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anticorruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and

biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index"** and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.04% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- (a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- (b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- (c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- (d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund's will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions"** in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets In securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 51% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax

reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

(i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or

 (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public
	and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), (B) banks are open for business in London and New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).
	The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.

Description of the Shares

Share Classes

Share	"USD	**"hEUR	**"hCHF	**"hGBP	**"hHKD	**"hSGD	**"hJPY
Class*	acc"	acc"	acc"	acc"	acc"	acc"	acc"
Initial Issue	N/A	N/A	N/A	N/A	HKD 1,000	SGD 100	JPY 1,000

Share	**"hSEK	**hPLN	**hNOK	***USD	***hEUR	***hCHF	***hGBP
Class*	acc"	acc	acc	UKdis	UKdis	UKdis	UKdis
Initial Issue	SEK 1,000	PLN 1,000	NOK 1,000	USD 100	EUR 100	CHF 100	GBP 100

Share	***hHKD	***hSGD	***hJPY	***hSEK	***hPLN	***hNOK
Class*	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000	PLN 1,000	NOK 1,000

* Non-Base Currency share classes will have a currency hedged exposure to the Reference Index

Together the "acc**" Share Class

*** Together the "**dis**" Share Class

Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. Such Share Class's performance may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Initial Offer Period

The hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class and the USD acc Share Class have launched and Shares in these Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hHKD acc, Share Class, the hSGD acc Share Class, the hJPY acc Share Class, the hSEK acc Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a

multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat-Fees	0.34% per annum	0.34% per annum

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index was developed to offer investors a commodity index that recognised the various maturities of commodity futures contracts that are available in commodity markets. Previously, commodity indices have been limited to those offering an index based on short term commodity futures contracts, whereas the Reference Index enables the construction of commodity exposures across maturities and across commodity categories of energy, industrial metals, precious metals, agriculture and livestock. The Reference Index is made up of 27 components across commodity categories.

For an index providing exposure to commodities such as the Reference Index, where categories of components making up the relevant index are highly correlated, such highly correlated categories must be considered as subcategories of the same commodity (in other words, components that are highly correlated must be considered as representing one component of the relevant index for the purposes of calculating the Fund's exposures). The Reference Index includes one component with a weighting of more than 20% and up to 35% (taking into account high correlation between certain components of the Reference Index such that they are considered together to constitute one component). Pursuant to the Regulations, the Fund is permitted to gain exposure to one component that represents more than 20% and up to 35% of the Reference Index where justified by exceptional market conditions.

Whilst the Reference Index is designed to provide broad-based exposure to commodities as an asset class across five main groups (agriculture, energy, industrial metals, precious metals and livestock), the oil and gasoline components within the energy group are highly correlated, and on a combined basis, they have a typical weighting of more than 20% and up to 35% of the Reference Index. Their high weighting in the Reference Index is partly due to the economic significance of the energy group and of these components as measured by their relative share of global consumption. In order to properly reflect the economic significance of the oil and gasoline components within the energy group and within the commodities sector as a whole, it is necessary for the Reference Index, and consequently the Fund, to have a typical weighting of more than 20% and up to 35% to these highly correlated components.

The Reference Index is calculated in USD, and includes a notional exposure to overnight interest rates. The Reference Index has different Currency Indices, including Currency Indices denominated in Euro, GBP, CHF, HKD, SGD, JPY and SEK. Such Currency Indices employ forward FX contracts to provide the relevant currency return. The Currency Indices are designed to represent the returns of the Reference Index while reducing currency risk but not the underlying commodity market risk. By employing such a strategy, the Currency Indices seek to reduce the risk of downward currency fluctuations although potential currency gains may also be sacrificed.

The Reference Index is rebalanced on a monthly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is governed by MerQube, Inc (the **"Index Administrator**") via its internal processes. The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index. More details on the Reference Index, including its underlying components, can be found at http://www.ubs.com/global/en/investment-bank/cmci/universe/composite-index.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (i) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or

unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.

- (j) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in commodity markets.
- (k) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (I) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (m) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (n) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (o) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF

UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

UBS Index Disclaimer

No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the Index Administrator. The Index is subject to the terms set out in the Index Manual, as such Index Manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to UBS.

CMCI is a trademark of UBS AG.

Index Disclaimer for MerQube

All information provided by MerQube in this Prospectus is impersonal and not tailored to the needs of any person, entity or group of persons or entities. MerQube receives remuneration in connection with the licensing of its indices

to third parties and the provision of customised calculation services.

Past performance of an index is not an indication or guarantee of future results.

It is not possible to invest directly in an index. Exposure to an asset class represented by an index is only possible through investable instruments based on that index. MerQube does not sponsor, endorse, sell, promote or manage investment funds or other investment vehicles offered by third parties whose objective is to achieve an investment return based on the performance of an index. MerQube makes no representation that investment products based on the Index will accurately track the performance of the Index or provide positive investment returns. MerQube, Inc. is not an investment adviser and MerQube makes no representation as to the advisability of investing in any such mutual fund or other investment vehicle. Any decision to invest in such mutual fund or other investment vehicle. Any decision to invest in such mutual fund or other investment in such a fund or other investment vehicle only after careful consideration of the risks associated with an investment in such funds, which are set out in detail in a prospectus or similar document prepared by or on behalf of the issuer of the fund or other investment product or vehicle.

The closing prices of the indices are calculated by MerQube on the basis of the closing prices of the individual components of the index as determined by the respective main stock exchange. MerQube receives the closing prices from one of its third-party providers and verifies them by comparing them with the prices of another provider. The providers receive the closing prices from the primary exchanges.

MerQube and its data suppliers (collectively, the "MerQube Parties") do not warrant the accuracy, completeness, timeliness or availability of this Prospectus. The MerQube Parties are not responsible for any errors or omissions, regardless of cause, in the results obtained from the use of the Content. THE MERQUBE PARTIES DISCLAIM ALL WARRANTIES, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE, FREE OF BUGS, SOFTWARE ERRORS OR DEFECTS, THAT THE OPERATION OF THE CONTENT WILL NOT BE UN-INTERRUPTED, OR THAT THE CONTENT WILL WORK WITH ANY SOFTWARE OR HARDWARE CONFIGURATION. In no event shall the MerQube Parties be liable for any direct, incidental, exemplary, compensatory, punitive, special or consequential damages, costs, expenses, attorneys' fees or losses (including, without limitation, lost revenues or lost profits and opportunity costs) in connection with the use of the Content, even if advised of the possibility of such damages.

Publication of performance

Past performance is not an indicator or guarantee of future results. All information presented prior to the launch date has been retrospectively tested. The back-tested performance is not an actual performance, but a hypothetical one. Backtest calculations are based on the same methodology in effect at the time of the official launch of the index(es). However, back-tested data may reflect the application of the index methodology in retrospect and historical calculations may change due to revisions in the economic data used for index calculation. Full details of the index methodology can be found at www.merqube.com. It is not possible to invest directly in an index.

The first value date is the first day for which a calculated value (either live or backtested) is available for a particular index. The inception date is the date on which the values of a particular index are first considered live: Index values provided for any date or time period prior to the launch date of the index are considered backtested. MerQube defines the Launch Date as the date on which the values of an index are known to have been made available to the public.

The index returns shown do not represent the results of actual securities trading. MerQube, Inc. does not manage actual assets. The index returns do not reflect the payment of any sales charges or fees that an investor may be required to pay for the purchase of the securities underlying the index(es) or for mutual funds designed to track the performance of the index(es). The imposition of these fees and costs would result in the actual and retrospectively tested performance of the fund/investment vehicle being lower than the performance of the index that the vehicle/fund tracks.

UBS BBG COMMODITY INDEX SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS BBG COMMODITY INDEX SF UCITS ETF** (the **"Fund"**), a sub-fund of UBS (Irl) Fund Solutions plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions Plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being Bloomberg Commodity Index Total Return).

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association), and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index / Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes as listed at (i) and (ii) above and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets**" for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anticorruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.03% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of Financial Derivative Instruments. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions**" in the Prospectus apply to the Fund. In addition, the following investment restrictions shall apply to the Fund:

- (i) the Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets In securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 51% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- IV. adjust the terms of the Swaps; and/or
- V. delay payment to the Fund under the terms of the Swaps; and/or
- VI. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a)

terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), (B) banks are open for business in London and New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).
	The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	USD 100 million.
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share.
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket size.
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.

* Non-Base Currency Share Classes performance may differ from that of the Base Currency Share Classes due to currency hedging costs.

Description of the Shares

Share Classes

Share	*USD	*hEUR	*hCHF	*hGBP	*hHKD	*hSGD	*hJPY	*hSEK
Class*	acc	acc	acc	acc	acc	acc	acc	acc
Initial Issue	N/A	N/A	N/A	N/A	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000

Share	* hPLN	*hNOK	**USD	**hEUR	**hCHF	**hGBP	**hHKD	**hSGD
Class	acc	acc	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue	PLN 1,000	NOK 1,000	USD 100	EUR 100	CHF 100	GBP 100	HKD 1,000	SGD 100

Share	**hJPY	**hSEK	**hPLN	**hNOK
Class	UKdis	UKdis	UKdis	UKdis
Initial	JPY 1,000	SEK	PLN	NOK
Issue		1,000	1,000	1,000

* Together the "acc" Share Class

** Together the "dis" Share Class

‡ Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Initial Offer Period

The hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class and the USD acc Share Class have launched and Shares in these Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hHKD acc, Share Class, the hSGD acc Share Class, the hJPY acc Share Class, the h SEK Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

*** The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat Fees	0.19% per annum	0.19% per annum

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index is designed to be a highly liquid and diversified benchmark for commodities investments composed of futures contracts on physical commodities. The Reference Index provides broad-based exposure to commodities as an asset class and is designed to minimize concentration in any one commodity or sector where no single commodity or commodity sector dominates the Reference Index. Rather than being driven by micro-economic events affecting one commodity market or sector, the diversified commodity exposure of the Reference Index potentially reduces volatility in comparison to non-diversified commodity baskets. It currently includes 22 commodities in six groups (grains, energy, industrial metals, precious metals, livestock & softs).

For an index providing exposure to commodities such as the Reference Index, where categories of components making up the relevant index are highly correlated, such highly correlated categories must be considered as sub-categories of the same commodity (in other words, components that are highly correlated must be considered as representing one component of the relevant index for the purposes of calculating the Fund's exposures). The Reference Index includes one component with a weighting of more than 20% and up to 35% (taking into account high correlation between certain components of the Reference Index such that they are considered together to constitute one component). Pursuant to the Regulations, the Fund is permitted to gain exposure to one component that represents more than 20% and up to 35% of the Reference Index where justified by exceptional market conditions.

Whilst the Reference Index is designed to provide broad-based exposure to commodities as an asset class across six main groups (grains, energy, industrial metals, precious metals, livestock and softs), the oil and gasoline components within the energy group are highly correlated, and on a combined basis, they have a typical weighting of more than 20% and up to 35% of the Reference Index. Their high weighting in the Reference Index is partly due to the economic significance of the energy group and of these components as measured by their relative share of global production. In order to properly reflect the economic significance of the oil and gasoline components within the energy group and within the commodities sector as a whole, it is necessary for the Reference Index, and consequently the Fund, to have a typical weighting of more than 20% and up to 35% to these highly correlated components.

The Reference Index is calculated in USD, and includes a notional exposure to the USD 3 month treasury bill interest rate. The Reference Index has different Currency Indices, including Currency Indices denominated in Euro, GBP, CHF, HKD, SGD, JPY and SEK. Such Currency Indices employ forward FX contracts to provide the relevant currency return. The Currency Indices are designed to represent the returns of the Reference Index while reducing currency risk but not the underlying commodity market risk. By employing such a strategy, the Currency Indices seek to reduce the risk of downward currency fluctuations although potential currency gains may also be sacrificed.

The Reference Index is rebalanced on a monthly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is sponsored by Bloomberg Finance LP. More details on the Reference Index, can be found at <u>www.bloombergindexes.com</u>.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns

in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.

- (i) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (j) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in commodity markets.
- (k) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchangebased" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (I) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (m) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (n) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (a) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

"Bloomberg®", "Bloomberg Commodity IndexSM" are service marks of Bloomberg Finance L.P. and its affiliates (collectively, "Bloomberg") and have been licensed for use for certain purposes by the Fund.

The Fund is not sponsored, endorsed, sold or promoted by Bloomberg, UBS AG, UBS Securities LLC ("UBS Securities") or any of their subsidiaries or affiliates. None of Bloomberg, UBS AG, UBS Securities or any of their subsidiaries or affiliates makes any representation or warranty, express or implied, to the owners of or counterparties to the Fund or any member of the public regarding the advisability of investing in securities or commodities generally or in the Fund particularly. The only relationship of Bloomberg, UBS AG, UBS Securities or any of their subsidiaries or affiliates to the Licensee is the licensing of certain trademarks, trade names and service marks and of the Bloomberg Commodity IndexSM, which is determined, composed and calculated by Bloomberg in conjunction with UBS Securities without regard to the Fund. Bloomberg and UBS Securities have no obligation to take the needs of the Fund or the owners of Shares in the Fund into consideration in determining, composing or calculating Bloomberg Commodity IndexSM. None of Bloomberg, UBS AG, UBS Securities or any of their respective subsidiaries or affiliates is responsible for or has participated in the determination of the timing of, prices at, or quantities of Shares in the Fund to be issued or in the determination or calculation of the equation by which Shares in the Fund are to be converted into cash. None of Bloomberg, UBS AG, UBS Securities or any of their subsidiaries or affiliates shall have any obligation or liability, including, without limitation, to Fund customers, in connection with the administration, marketing or trading of Shares in the Fund. Notwithstanding the foregoing, UBS AG, UBS Securities and their respective subsidiaries and affiliates may independently issue and/or sponsor financial products unrelated to the Shares currently being issued by the Fund, but which may be similar to and competitive with such Shares in the Fund. In addition, UBS AG, UBS Securities and their subsidiaries and affiliates actively trade commodities, commodity indexes and commodity futures (including the Bloomberg Commodity IndexSM and Bloomberg Commodity Index Total ReturnSM), as well as swaps, options and derivatives which are linked to the performance of such commodities, commodity indexes and commodity futures. It is possible that this trading activity will affect the value of the Bloomberg Commodity IndexSM and Shares in the Fund.

This Supplement relates only to the Fund and does not relate to the exchange-traded physical commodities underlying any of the Bloomberg Commodity IndexSM components. Purchasers of Shares in the Fund should not conclude that the inclusion of a futures contract in the Bloomberg Commodity IndexSM is any form of investment recommendation of the futures contract or the underlying exchange-traded physical commodity by Bloomberg, UBS AG, UBS Securities or any of their subsidiaries or affiliates. The information in the Supplement regarding the Bloomberg Commodity IndexSM components has been derived solely from publicly available documents. None of Bloomberg, UBS AG, UBS Securities or any of their subsidiaries or affiliates has made any due diligence inquiries with respect to the Bloomberg Commodity IndexSM components in connection with the Fund. None of Bloomberg, UBS AG, UBS Securities or any of their subsidiaries or affiliates.

affiliates makes any representation that these publicly available documents or any other publicly available information regarding the Bloomberg Commodity IndexSM components, including without limitation a description of factors that affect the prices of such components, are accurate or complete.

NONE OF BLOOMBERG, UBS AG, UBS SECURITIES OR ANY OF THEIR SUBSIDIARIES OR AFFILIATES GUARANTEES THE ACCURACY AND/OR THE COMPLETENESS OF THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO AND NONE OF BLOOMBERG, UBS AG, UBS SECURITIES OR ANY OF THEIR SUBSIDIARIES OR AFFILIATES SHALL HAVE ANY LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS THEREIN. NONE OF BLOOMBERG, UBS AG, UBS SECURITIES OR ANY OF THEIR SUBSIDIARIES OR AFFILIATES MAKES ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE OBTAINED BY THE FUND OWNERS OF SHARES IN THE FUND OR ANY OTHER PERSON OR ENTITY FROM THE USE OF THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO. NONE OF BLOOMBERG, UBS AG, UBS SECURITIES OR ANY OF THEIR SUBSIDIARIES OR AFFILIATES MAKES ANY EXPRESS OR IMPLIED WARRANTIES AND EXPRESSLY DISCLAIMS ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE WITH RESPECT TO THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO. WITHOUT LIMITING ANY OF THE FOREGOING, TO THE MAXIMUM EXTENT ALLOWED BY LAW, BLOOMBERG, ITS LICENSORS (INCLUDING UBS), AND ITS AND THEIR RESPECTIVE EMPLOYEES, CONTRACTORS, AGENTS, SUPPLIERS AND VENDORS SHALL HAVE NO LIABILITY OR RESPONSIBILITY WHATSOEVER FOR ANY INJURY OR DAMAGES-WHETHER DIRECT, INDIRECT, CONSEQUENTIAL, INCIDENTAL, PUNITIVE OR OTHERWISE-ARISING IN CONNECTION WITH THE FUND OR BLOOMBERG COMMODITY INDEX OR ANY DATA OR VALUES RELATING THERETO—WHETHER ARISING FROM THEIR NEGLIGENCE OR OTHERWISE, EVEN IF NOTIFIED OF THE POSSIBILITY THEREOF. THERE ARE NO THIRD PARTY BENEFICIARIES OF ANY AGREEMENTS OR ARRANGEMENTS AMONG BLOOMBERG, UBS SECURITIES AND THE FUND OTHER THAN UBS AG.

UBS S&P 500 SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the UBS S&P 500 SF UCITS ETF (the "Fund"), a subfund of UBS (IrI) Fund Solutions PIc (the "Company") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "Central Bank").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions Plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being S&P 500 Net Total Return Index).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association), and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

(i) The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anticorruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is 0% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

In order to target a reduction in tracking difference and/or target an improved performance of the Fund, the Investment Manager may invest in swaps which reference a gross total return version of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be

made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of Financial Derivative Instruments. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 51% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained. **Disruption Events**

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner;

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- VII. adjust the terms of the Swaps; and/or
- VIII. delay payment to the Fund under the terms of the Swaps; and/or
- IX. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant

Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, EUR acc, CHF acc, GBP acc, HKD acc, SGD acc, JPY acc, SEK acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).
Dealing Day	Any Business Day.
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket size.
Website	www.ubs.com/etf - information on Fund Assets and the Fund's iNAV is set out on the Website.

 * Non-Base Currency Share Classes performance may differ from that of the Base Currency Share Classes due to currency hedging costs.

Description of the Shares

Share Class *	**"USD acc"	**"EUR acc"	**"CHF acc"	**"GBP acc"	**"HKD acc"	**"SGD acc"	**"JPY acc"
Initial Issue Price	N/A.	EUR 1/100 the closing level of the Reference Index on the launch date.	CHF 1/100 the closing level of the Reference Index on the launch date.	GBP 1/100 of the closing level of the Reference Index on the launch date	HKD 1/100 of the closing level of the Reference Index on the launch date	SGD 1/100 of the closing level of the Reference Index on the launch date	JPY 1/100 of the closing level of the Reference Index on the launch date

Share	**"SEK	**hEUR	**hCHF	**hGBP	**hHKD	**hSGD	**hJPY	**hSEK
Class *	acc"	acc"	acc"	acc"	acc	acc	acc	acc
Initial Issue Price	SEK 1/100 of the closing level of the Reference Index on the launch date	EUR 100	CHF 100	GBP 100	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000

Share Class *	**hPLN	**hNOK	***USD	***hEUR	***hCHF	***hGBP
	acc	acc	UKdis	UKdis	UKdis	UKdis
Initial Issue Price	PLN 1,000	NOK 1,000	USD 100	EUR 100	CHF 100	GBP 100

Share Class *	***hHKD	***hSGD	***hJPY	***hSEK	***hPLN	***hNOK
	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue Price	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000	PLN 1,000	NOK 1,000

* Non-Base Currency share classes will have a currency hedged exposure to the Reference Index.

**Together "acc" Share Class.

*** Together the "dis" Share Class.

Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Details on the Reference Index can be found at <u>www.standardandpoors.com</u>.

Initial Offer Period

The USD acc Share Class has launched and Shares in this Share Class are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the EUR acc Share Class, CHF acc Share Class, GBP acc Share Class, HKD acc Share Class, SGD acc Share Class, JPY acc Share Class, SEK acc Share Class, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc, USD UKdis, hEUR UKdis, hCHF UKdishCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat-Fees	0.19% per annum	0.19% per annum

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index is an equity index of securities which focuses on the large cap segment of the US market. Securities listed on national stock exchanges in the US are eligible for inclusion. The Reference Index represents companies that are available to investors worldwide and provides a representation of 10 sectors within 500 constituents.

The Reference Index is rebalanced on a quarterly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is sponsored by Standard & Poor's Financial Services LLC. More details on the Reference Index, including its underlying components, can be found at <u>www.spindices.com</u>.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.

- (h) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (i) The Fund is exposed to the investment risk arising from the Reference Index underlying the Swaps. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (j) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" "or" "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (k) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (I) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (m) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF

UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

Standard & Poor's® and S&P® are registered trademarks of Standard & Poor's Financial Services LLC ("S&P Finance") and have been licensed for use by the Manager. The Fund is not sponsored, endorsed, sold or promoted by S&P Finance or its Affiliates, and S&P Finance and its Affiliates make no representation, warranty or condition regarding the advisability of buying, selling or holding shares in the Fund.

THIS FUND IS NOT SPONSORED, ENDORSED, SOLD OR PROMOTED BY STANDARD & POOR'S AND ITS AFFILIATES ("S&P"). S&P MAKES NO REPRESENTATION, CONDITION OR WARRANTY, EXPRESS OR IMPLIED, TO THE OWNERS OF THE FUND OR ANY MEMBER OF THE PUBLIC REGARDING THE ADVISABILITY OF INVESTING IN SECURITIES GENERALLY OR IN THE FUND PARTICULARLY OR THE ABILITY OF THE S&P 500 TO TRACK THE PERFORMANCE OF CERTAIN FINANCIAL MARKETS AND/OR SECTIONS THEREOF AND/OR OF GROUPS OF ASSETS OR ASSET CLASSES. S&P'S ONLY RELATIONSHIP TO THE MANAGER IS THE LICENSING OF CERTAIN TRADEMARKS AND TRADE NAMES AND OF THE S&P 500 WHICH IS DETERMINED, COMPOSED AND CALCULATED BY S&P WITHOUT REGARD TO THE MANAGER OR THE FUND. S&P HAS NO OBLIGATION TO TAKE THE NEEDS OF THE MANAGER OR THE OWNERS OF THE FUND INTO CONSIDERATION IN DETERMINING, COMPOSING OR CALCULATING THE S&P 500. S&P IS NOT RESPONSIBLE FOR AND HAS NOT PARTICIPATED IN THE DETERMINATION OF THE PRICES AND AMOUNT OF THE FUND OR THE TIMING OF THE ISSUANCE OR SALE OF THE FUND OR IN THE DETERMINATION OR CALCULATION OF THE EQUATION BY WHICH THE FUND SHARES ARE TO BE CONVERTED INTO CASH. S&P HAS NO OBLIGATION OR LIABILITY IN CONNECTION WITH THE ADMINISTRATION, MARKETING, OR TRADING OF THE FUND.

S&P DOES NOT GUARANTEE THE ACCURACY AND/OR THE COMPLETENESS OF THE S&P 500 OR ANY DATA INCLUDED THEREIN AND S&P SHALL HAVE NO LIABILITY FOR ANY ERRORS, OMISSIONS, OR INTERRUPTIONS THEREIN. S&P MAKES NO WARRANTY, CONDITION OR REPRESENTATION, EXPRESS OR IMPLIED, AS TO RESULTS TO BE OBTAINED BY THE MANAGER, OWNERS OF THE FUND, OR ANY OTHER PERSON OR ENTITY FROM THE USE OF THE S&P 500 OR ANY DATA INCLUDED THEREIN. S&P MAKES NO EXPRESS OR IMPLIED WARRANTIES, REPRESENTATIONS OR CONDITIONS, AND EXPRESSLY DISCLAIMS ALL WARRANTIES OR CONDITIONS OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE AND ANY OTHER EXPRESS OR IMPLIED WARRANTY OR CONDITION WITH RESPECT TO THE S&P 500 OR ANY DATA INCLUDED THEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL S&P HAVE ANY LIABILITY FOR ANY SPECIAL, PUNITIVE, INDIRECT, OR CONSEQUENTIAL DAMAGES (INCLUDING LOST PROFITS) RESULTING FROM THE USE OF THE S&P 500 OR ANY DATA INCLUDED THEREIN, EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

THE DIRECTORS OF THE COMPANY, THE MANAGER AND THE INVESTMENT MANAGER TOGETHER THE "RESPONSIBLE PARTIES" DO NOT GUARANTEE THE ACCURACY AND/OR THE COMPLETENESS OF ANY DESCRIPTION RELATING TO THE REFERENCE INDEX OR ANY DATA INCLUDED THEREIN AND THE RESPONSIBLE PARTIES SHALL HAVE NO LIABILITY FOR ANY ERRORS, OMISSIONS, OR INTERRUPTIONS HEREIN. THE APPROVED RESPONSIBLE PARTIES MAKES NO WARRANTY, EXPRESS OR IMPLIED, AS TO THE FUND, ANY SHAREHOLDER IN THE FUND, OR TO ANY OTHER PERSON OR ENTITY IN RESPECT OF THE REFERENCE INDEX DESCRIBED HEREIN.

THE INDEX PROVIDER MAKES NO EXPRESS OR IMPLIED WARRANTIES, AND EXPRESSLY DISCLAIMS ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE WITH RESPECT TO THE REFERENCE INDEX OR STRATEGY OR ANY DATA INCLUDED HEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL THE INDEX PROVIDER HAVE ANY LIABILITY FOR ANY SPECIAL, PUNITIVE, INDIRECT, OR CONSEQUENTIAL DAMAGES OR FOR ANY LOST PROFITS, EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

UBS MSCI USA SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS MSCI USA SF UCITS ETF** (the **"Fund**"), a sub-fund of UBS (Irl) Fund Solutions Plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions Plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any

registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being MSCI USA Index Net Total Return).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the "**Fund Assets**" for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index**" and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is 0% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

b) An explanation for the divergence between the anticipated and realised tracking error for the year;

c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and

d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

In order to target a reduction in tracking difference and/or target an improved performance of the Fund, the Investment Manager may invest in swaps which reference a gross total return version of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit

risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a **"Counterparty"**). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and

95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of Financial Derivative Instruments. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 51% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
 - (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, EUR acc, CHF acc, GBP acc, HKD acc, SGD acc, JPY acc, SEK acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges"**) for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges. Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), (B) banks are open for business in New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager). The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.

Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket size.
Website	www.ubs.com/etf - information on Fund Assets and the Fund's iNAV is set out on the Website.

* Non-Base Currency Share Classes performance may differ from that of the Base Currency Share Classes due to currency hedging costs.

Description of the Shares

Share Classes

Share	**USD	**EUR	**CHF	**GBP	**HKD	**SGD	**JPY	**SEK
Class*	acc	acc	acc	acc	acc	acc	acc	acc
Initial Issue	N/A	EUR 1/100 of the closing level of the Reference Index on the Launch Date.	CHF 1/100 of the closing level of the Reference Index on the Launch Date	GBP 1/100 of the closing level of the Reference Index on the Launch Date	HKD 1/100 of the closing level of the Reference Index on the Launch Date	SGD 1/100 of the closing level of the Reference Index on the Launch Date	JPY 1/100 of the closing level of the Reference Index on the Launch Date	SEK 1/100 of the closing level of the Reference Index on the Launch Date

Share	**hEUR	**hCHF	**hGBP	**hHKD	**hSGD	**hJPY	**hSEK	**hPLN
Class*	acc							
Initial Issue	EUR 100	CHF 100	GBP 100	HKD 100	SGD 100	JPY 100	SEK 100	PLN 1,000

Share	**hNOK	***USD	***hEUR	***hCHF	***hGBP	***hHKD	***hJPY
Class*	acc	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue	NOK 1,000	USD 100	EUR 100	CHF 100	GBP 100	HKD 1,000	JPY 1,000

Share	***hSEK	***hPLN	***hNOK	***hSGD	
Class*	UKdis	UKdis	UKdis	UKdis	
Initial Issue	SEK 1,000	PLN 1,000	NOK 1,000	SGD 100	

* Non-Base Currency share classes will have a currency hedged exposure to the Reference Index

** Together "acc" Share Class.

*** Together the "**dis**" Share Class.

Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Information regarding the Initial Issue Price is available from the Administrator. Details on the Reference Index can be found at www.mscibarra.com

Initial Offer Period

The USD acc Share Class has launched and Shares in this Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the EUR acc Share Class, CHF acc Share Class, GBP acc Share Class, HKD acc Share Class, SGD acc Share Class, JPY acc Share Class, SEK acc Share Class, hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the hHKD acc Share Class, the hSGD acc Share Class, the hJPY acc Share Class, the hSEK acc Share Class, USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund).

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat-Fees	0.15% per annum	0.15% per annum

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index is a free float adjusted market capitalization index that is designed to measure large and mid cap US equity market performance. The Reference Index is member of the MSCI international equity index series and represents the US equity portion of the global benchmark MSCI ACWI (All Country World Index) Index.

The Reference Index is rebalanced on a quarterly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is sponsored by MSCI Limited. More details on the Reference Index, including its underlying components, can be found at <u>www.msci.com</u>.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of

currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.

- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (i) The Fund is exposed to the investment risk arising from the Reference Index underlying the Swaps. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (j) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (k) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (I) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (m) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment

Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

THIS FINANCIAL PRODUCT IS NOT SPONSORED, ENDORSED, SOLD OR PROMOTED BY MSCI LIMITED ("MSCI"), ANY AFFILIATE OF MSCI OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX. THE MSCI INDEXES ARE THE EXCLUSIVE PROPERTY OF MSCI. MSCI AND THE MSCI INDEX NAMES ARE SERVICE MARK(S) OF MSCI OR ITS AFFILIATES AND HAVE BEEN LICENSED FOR USE FOR CERTAIN PURPOSES BY LICENSEE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, TO THE OWNERS OF THIS FINANCIAL PRODUCT OR ANY MEMBER OF THE PUBLIC REGARDING THE ADVISABILITY OF INVESTING IN FINANCIAL SECURITIES GENERALLY OR IN THIS FINANCIAL PRODUCT PARTICULARLY OR THE ABILITY OF ANY MSCI INDEX TO TRACK CORRESPONDING STOCK MARKET PERFORMANCE. MSCI OR ITS AFFILIATES ARE THE LICENSORS OF CERTAIN TRADEMARKS, SERVICE MARKS AND TRADE NAMES AND OF THE MSCI INDEXES WHICH ARE DETERMINED, COMPOSED AND CALCULATED BY MSCI WITHOUT REGARD TO THIS FINANCIAL PRODUCT OR THE ISSUER OR OWNER OF THIS FINANCIAL PRODUCT. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION TO TAKE THE NEEDS OF THE ISSUERS OR OWNERS OF THIS FINANCIAL PRODUCT INTO CONSIDERATION IN DETERMINING, COMPOSING OR CALCULATING THE MSCI INDEXES. NEITHER MSCI, ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX IS RESPONSIBLE FOR OR HAS PARTICIPATED IN THE DETERMINATION OF THE TIMING OF, PRICES AT, OR QUANTITIES OF THIS FINANCIAL PRODUCT TO BE ISSUED OR IN THE DETERMINATION OR CALCULATION OF THE EQUATION BY WHICH THIS FINANCIAL PRODUCT IS REDEEMABLE FOR CASH. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, THE MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION OR LIABILITY TO THE OWNERS OF THIS FINANCIAL PRODUCT IN CONNECTION WITH THE ADMINISTRATION, MARKETING OR OFFERING OF THIS FINANCIAL PRODUCT.

ALTHOUGH MSCI SHALL OBTAIN INFORMATION FOR INCLUSION IN OR FOR USE IN THE CALCULATION OF THE MSCI INDEXES FROM SOURCES WHICH MSCI CONSIDERS RELIABLE, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX WARRANTS OR GUARANTEES THE ORIGINALITY, ACCURACY AND/OR THE COMPLETENESS OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE **OBTAINED BY LICENSEE, LICENSEE'S CUSTOMERS OR COUNTERPARTIES, ISSUERS OF** THE FINANCIAL SECURITIES, OWNERS OF THE FINANCIAL SECURITIES, OR ANY OTHER PERSON OR ENTITY, FROM THE USE OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN IN CONNECTION WITH THE RIGHTS LICENSED HEREUNDER OR FOR ANY OTHER USE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX SHALL HAVE ANY LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS OF OR IN CONNECTION WITH ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. FURTHER, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY EXPRESS OR IMPLIED WARRANTIES OF ANY KIND, AND MSCI, ANY OF ITS AFFILIATES AND ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX HEREBY EXPRESSLY DISCLAIM ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, WITH RESPECT TO ANY MSCI INDEX AND ANY DATA INCLUDED THEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL MSCI, ANY OF ITS AFFILIATES OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, SPECIAL, PUNITIVE, CONSEQUENTIAL OR ANY OTHER DAMAGES (INCLUDING LOST PROFITS) EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

No purchaser, seller or holder of this security, or any other person or entity, should use or refer to any MSCI trade name, trademark or service mark to sponsor, endorse, market or promote this product without first contacting MSCI to determine whether MSCI's permission is required. Under no circumstances may any person or entity claim any affiliation with MSCI without the prior written permission of MSCI.

UBS MSCI EM SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS MSCI EM SF UCITS ETF** (the "**Fund**"), a subfund of UBS (IrI) Fund Solutions plc (the "**Company**") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "**Central Bank**").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

(i) **"EM**" (ii) means emerging markets.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being MSCI Emerging Markets Net Total Return Index).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

(iii) The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anticorruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index"** and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.03% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

b) An explanation for the divergence between the anticipated and realised tracking error for the year;

c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and

d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

In order to target a reduction in tracking difference and/or target an improved performance of the Fund, the Investment Manager may invest in swaps which reference a gross total return version of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be

made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of Financial Derivative Instruments. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 51% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
- (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
- (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers),

(i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

	1
Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), and (B) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).
	The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00pm. (Dublin time) 1 Business Day prior to the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket size.
Website	www.ubs.com/etf - information on Fund Assets and the Fund's iNAV is set out on the Website.

Details on the Reference Index can be found at <u>www.mscibarra.com</u>

Description of the Shares

The Fund offers the following Share Classes:

Share Class*	**hEUR acc	**hCHF acc	**hGBP acc	**hHKD acc	**hSGD acc	**hJPY acc	**hSEK acc
Initial Issue	EUR 100	CHF 100	GBP 100	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000

Share	**hPLN	**hNOK	***USD	***hEUR	***hCHF	***hGBP	***hHKD
Class*	acc	acc	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue	PLN 1,000	NOK 1,000	USD 100	EUR 100	CHF 100	GBP 100	HKD 1,000

Share	***hSGD	***hJPY	***hSEK	***hPLN	***hNOK	USD
Class*	UKdis	UKdis	UKdis	UKdis	UKdis	acc
Initial Issue	SGD 100	JPY 1,000	SEK 1,000	PLN 1,000	NOK 1,000	N/A

* Non-Base Currency share classes will have a currency hedged exposure to the Reference Index

** Together "acc" Share Class.

*** Together the "dis" Share Class

Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Initial Offer Period

The USD acc Share Class has launched and Shares in this Share Class are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the hHKD acc Share Class, the hSGD acc Share Class, the hJPY acc Share Class, the hSEK acc Share Class, USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund).

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat-Fees	0.205% per annum	0.205% per annum

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- d) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- e) Any costs associated with foreign currency hedging.
- f) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed **"Fees and Expenses"** should be read in conjunction with the sections headed **"Fees and Expenses"** and **"Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets"** in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is the MSCI Emerging Markets Net Total Return Index ("**the Reference Index**"). The Reference Index is described below but only represents an extract of information available and neither the Directors, the Manager, UBS AG and the Investment Manager have taken any responsibility for the accuracy or completeness of such information.

The Reference Index is a free float-adjusted market capitalization index that is designed to measure equity market performance of emerging markets. As of July 2014 the MSCI Emerging Markets Index consisted of the following 23 emerging market country indices: Brazil, Chile, China, Colombia, Czech Republic, Egypt, Greece, Hungary, India, Indonesia, Korea, Malaysia, Mexico, Peru, Philippines, Poland, Qatar, Russia, South Africa, Taiwan, Thailand, Turkey and United Arab Emirates.

The Reference Index is rebalanced on a quarterly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is sponsored by MSCI Limited. More details on the Reference Index, including its underlying components, can be found at <u>www.msci.com</u>.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.

- (h) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (i) The Fund is exposed to the investment risk arising from the Reference Index underlying the Swaps. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (j) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchangebased" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (k) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (I) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (m) The Reference Index is comprised of emerging market country indices including India. The Securities and Exchange Board of India ("SEBI") imposes certain obligations under the FII (Foreign Institutional Investor) Regulations on investors seeking to invest either directly and or indirectly in securities listed or proposed to be listed in India ("Indian Securities"). In the case of the Fund, such obligations would apply where the Indian listed stock component of the Reference Index exceeds certain prescribed limits which are set by SEBI. Currently, the Indian listed stock component in the Reference Index remains below the prescribed limits. However, should the SEBI limit requirements change, this may have an impact on the Fund as an indirect investor in Indian Securities by referencing the Reference Index and consequently the Fund may need to be closed to further subscriptions.
- (n) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment

decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

THIS FUND IS NOT SPONSORED, ENDORSED, SOLD OR PROMOTED BY MSCI LIMITED. ("MSCI"), ANY AFFILIATE OF MSCI OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX. THE MSCI INDEXES ARE THE EXCLUSIVE PROPERTY OF MSCI. MSCI AND THE MSCI INDEX NAMES ARE SERVICE MARK(S) OF MSCI OR ITS AFFILIATES AND HAVE BEEN LICENSED FOR USE FOR CERTAIN PURPOSES BY LICENSEE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, TO THE OWNERS OF THIS FUND OR ANY MEMBER OF THE PUBLIC REGARDING THE ADVISABILITY OF INVESTING IN FINANCIAL SECURITIES GENERALLY OR IN THIS FUND PARTICULARLY OR THE ABILITY OF ANY MSCI INDEX TO TRACK CORRESPONDING STOCK MARKET PERFORMANCE. MSCI OR ITS AFFILIATES ARE THE LICENSORS OF CERTAIN TRADEMARKS, SERVICE MARKS AND TRADE NAMES AND OF THE MSCI INDEXES WHICH ARE DETERMINED, COMPOSED AND CALCULATED BY MSCI WITHOUT REGARD TO THIS FUND OR THE ISSUER OR OWNER OF THIS FUND. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION TO TAKE THE NEEDS OF THE ISSUERS OR OWNERS OF THIS FUND INTO CONSIDERATION IN DETERMINING, COMPOSING OR CALCULATING THE MSCI INDEXES. NEITHER MSCI, ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX IS RESPONSIBLE FOR OR HAS PARTICIPATED IN THE DETERMINATION OF THE TIMING OF, PRICES AT, OR QUANTITIES OF THIS FUND TO BE ISSUED OR IN THE DETERMINATION OR CALCULATION OF THE EQUATION BY WHICH THIS FUND IS REDEEMABLE FOR CASH. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, THE MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION OR LIABILITY TO THE OWNERS OF THIS FUND IN CONNECTION WITH THE ADMINISTRATION, MARKETING OR OFFERING OF THIS FUND.

ALTHOUGH MSCI SHALL OBTAIN INFORMATION FOR INCLUSION IN OR FOR USE IN THE CALCULATION OF THE MSCI INDEXES FROM SOURCES WHICH MSCI CONSIDERS RELIABLE, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX WARRANTS OR GUARANTEES THE ORIGINALITY, ACCURACY AND/OR THE COMPLETENESS OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE OBTAINED BY LICENSEE, LICENSEE'S CUSTOMERS OR COUNTERPARTIES, ISSUERS OF THE FINANCIAL SECURITIES, OWNERS OF THE FINANCIAL SECURITIES, OR ANY OTHER PERSON OR ENTITY, FROM THE USE OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN IN CONNECTION WITH THE RIGHTS LICENSED HEREUNDER OR FOR ANY OTHER USE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX SHALL HAVE ANY LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS OF OR IN CONNECTION WITH ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. FURTHER, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY EXPRESS OR IMPLIED WARRANTIES OF ANY KIND, AND MSCI, ANY OF ITS AFFILIATES AND ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX HEREBY EXPRESSLY DISCLAIM ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, WITH RESPECT TO ANY MSCI INDEX AND ANY DATA INCLUDED THEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL MSCI, ANY OF ITS AFFILIATES OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, SPECIAL, PUNITIVE, CONSEQUENTIAL OR ANY OTHER DAMAGES (INCLUDING LOST PROFITS) EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

No purchaser, seller or holder of this security, or any other person or entity, should use or refer to any MSCI trade name, trademark or service mark to sponsor, endorse, market or promote this product without first contacting MSCI to determine whether MSCI's permission is required. Under no circumstances may any person or entity claim any affiliation with MSCI without the prior written permission of MSCI.

UBS MSCI AC ASIA EX JAPAN SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS MSCI AC Asia ex Japan SF UCITS ETF** (the **"Fund"**), a sub-fund of UBS (IrI) Fund Solutions plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being MSCI AC Asia ex Japan Net Total Return Index).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

(i) The Fund is passively managed.

(ii) The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-

corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.03% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

b) An explanation for the divergence between the anticipated and realised tracking error for the year;

c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and

d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

In order to target a reduction in tracking difference and/or target an improved performance of the Fund, the Investment Manager may invest in swaps which reference a gross total return version of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of Financial Derivative Instruments. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
- (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
- (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by

such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The Fund will not be leveraged. The market risk of the Fund associated with the use of financial derivative instruments will be measured using the commitment approach in accordance with the requirements of the Central Bank.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc and the USD acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

_	
Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), and (B) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).
	The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00p.m. (Dublin time) 1 Business Day prior to the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10.30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to 3 (three) Business Days after the relevant Dealing Day and in any event shall not exceed 10 (ten) Business Days from the relevant Dealing Day.
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.
Minimum Holding	1 Share
Website	www.ubs.com/etf - information on Fund Assets and the Fund's iNAV is set out on the Website.

Details on the Reference Index can be found at <u>www.mscibarra.com</u>.

Description of the Shares

Share Classes

Share	USD	**hEUR	**hCHF	**hGBP	**hHKD	**hSGD	**hJPY	**hSEK
Class*	acc	acc	acc	acc	acc	acc	acc	acc
Initial Issue Price	N/A	EUR 100	CHF 100	GBP 100	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000

Share	**hPLN	**hNOK	***USD	***hEUR	***hCHF	***hGBP	***hHKD	***hSGD
Class*	acc	acc	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue Price	PLN 1000	NOK 1000	USD 100	EUR 100	CHF 100	GBP 100	HKD 1,000	SGD 100

Share	***hJPY	***hSEK	***hPLN	***hNOK
Class*	UKdis	UKdis	UKdis	UKdis
Initial Issue Price	JPY 1,000	SEK 1,000	PLN 1,000	NOK 1,000

* Non-Base Currency share classes will have a currency hedged exposure to the Reference Index

** Together "acc" Share Class.

*** Together the "dis" Share Class

Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Initial Offer Period

The USD acc Shares are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the hHKD acc Share Class, the hSGD acc Share Class, the hJPY acc Share Class, the hSEK acc Share Class, USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund).

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%

Primary Market Transaction Costs***	Up to 1%	Up to 1%	
-------------------------------------	----------	----------	--

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"	
Flat-Fees	0.233% per annum	0.233% per annum	

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index captures large and mid cap representation across 2 of 3 developed market countries (excluding Japan) (as at July 2014 these include Hong Kong and Singapore) and 8 emerging markets countries in Asia (as at July 2014 such markets include China, India, Indonesia, Malaysia, the Philippines, South Korea, Taiwan and Thailand). With 616 constituents, the Reference Index covers approximately 84% of the free float-adjusted market capitalisation in each country.

The Reference Index is rebalanced on a quarterly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollars).

Index Provider and website

The Reference Index is sponsored by MSCI Limited. More details on the Reference Index, including its underlying components, can be found at <u>www.msci.com</u>.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or

unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.

- (i) The Fund is exposed to the investment risk arising from the Reference Index underlying the Swaps. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (j) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchangebased" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (k) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (I) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (m) The Reference Index is comprised of developed market and emerging market country indices including India. The Securities and Exchange Board of India ("SEBI") imposes certain obligations under the FII (Foreign Institutional Investor) Regulations on investors seeking to invest either directly and or indirectly in securities listed or proposed to be listed in India ("Indian Securities"). In the case of the Fund, such obligations would apply where the Indian listed stock component of the Reference Index exceeds certain prescribed limits which are set by SEBI. Currently, the Indian listed stock component in the Reference Index remains below the prescribed limits. However, should the SEBI limit requirements change, this may have an impact on the Fund as an indirect investor in Indian Securities by referencing the Reference Index and consequently the Fund may need to be closed to further subscriptions.
- (n) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment

decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

THIS FUND IS NOT SPONSORED, ENDORSED, SOLD OR PROMOTED BY MSCI Limited. ("MSCI"), ANY AFFILIATE OF MSCI OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX. THE MSCI INDEXES ARE THE EXCLUSIVE PROPERTY OF MSCI. MSCI AND THE MSCI INDEX NAMES ARE SERVICE MARK(S) OF MSCI OR ITS AFFILIATES AND HAVE BEEN LICENSED FOR USE FOR CERTAIN PURPOSES BY LICENSEE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, TO THE OWNERS OF THIS FUND OR ANY MEMBER OF THE PUBLIC REGARDING THE ADVISABILITY OF INVESTING IN FINANCIAL SECURITIES GENERALLY OR IN THIS FUND PARTICULARLY OR THE ABILITY OF ANY MSCI INDEX TO TRACK CORRESPONDING STOCK MARKET PERFORMANCE. MSCI OR ITS AFFILIATES ARE THE LICENSORS OF CERTAIN TRADEMARKS, SERVICE MARKS AND TRADE NAMES AND OF THE MSCI INDEXES WHICH ARE DETERMINED, COMPOSED AND CALCULATED BY MSCI WITHOUT REGARD TO THIS FUND OR THE ISSUER OR OWNER OF THIS FUND. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION TO TAKE THE NEEDS OF THE ISSUERS OR OWNERS OF THIS FUND INTO CONSIDERATION IN DETERMINING, COMPOSING OR CALCULATING THE MSCI INDEXES. NEITHER MSCI, ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX IS RESPONSIBLE FOR OR HAS PARTICIPATED IN THE DETERMINATION OF THE TIMING OF, PRICES AT, OR QUANTITIES OF THIS FUND TO BE ISSUED OR IN THE DETERMINATION OR CALCULATION OF THE EQUATION BY WHICH THIS FUND IS REDEEMABLE FOR CASH. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, THE MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION OR LIABILITY TO THE OWNERS OF THIS FUND IN CONNECTION WITH THE ADMINISTRATION, MARKETING OR OFFERING OF THIS FUND.

ALTHOUGH MSCI SHALL OBTAIN INFORMATION FOR INCLUSION IN OR FOR USE IN THE CALCULATION OF THE MSCI INDEXES FROM SOURCES WHICH MSCI CONSIDERS RELIABLE, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX WARRANTS OR GUARANTEES THE ORIGINALITY, ACCURACY AND/OR THE COMPLETENESS OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE OBTAINED BY LICENSEE, LICENSEE'S CUSTOMERS OR COUNTERPARTIES, ISSUERS OF THE FINANCIAL SECURITIES, OWNERS OF THE FINANCIAL SECURITIES, OR ANY OTHER PERSON OR ENTITY, FROM THE USE OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN IN CONNECTION WITH THE RIGHTS LICENSED HEREUNDER OR FOR ANY OTHER USE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX SHALL HAVE ANY LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS OF OR IN CONNECTION WITH ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. FURTHER, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY EXPRESS OR IMPLIED WARRANTIES OF ANY KIND, AND MSCI, ANY OF ITS AFFILIATES AND ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX HEREBY EXPRESSLY DISCLAIM ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, WITH RESPECT TO ANY MSCI INDEX AND ANY DATA INCLUDED THEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL MSCI, ANY OF ITS AFFILIATES OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, SPECIAL, PUNITIVE, CONSEQUENTIAL OR ANY OTHER DAMAGES (INCLUDING LOST PROFITS) EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

No purchaser, seller or holder of this security, or any other person or entity, should use or refer to any MSCI trade name, trademark or service mark to sponsor, endorse, market or promote this product without first contacting MSCI to determine whether MSCI's permission is required. Under no circumstances may any person or entity claim any affiliation with MSCI without the prior written permission of MSCI.

UBS MSCI ACWI SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS MSCI ACWI SF UCITS ETF** (the **"Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes, it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any

registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being MSCI ACWI Net Total Return Index) which captures the large and mid cap representation of equities across developed markets and emerging markets countries.

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include fully funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index / Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include common stock and preferred stock), e.g. equity-linked notes (debt instruments based on the return of a single stock or basket of stocks) or global depositary receipts; fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, short-term money market deposits, short-term commercial paper, floating rate notes, medium term notes, securities issued or guaranteed by any OECD government, its agencies and shall be in accordance with the requirements of the Central Bank) and convertible bonds and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS Group AG group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus.

All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies at (i) and (ii) above.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currencies of countries which make up the Reference Index. Currency hedging of a share class will be limited to the developed market currencies underlying the Reference Index which typically account for 80-90% of the Reference Index. For example, the (hedged to USD) A- acc share class will hedge the developed market currency exposures inherent in the Reference Index against the US dollar currency. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error in normal conditions after fees and transaction costs is 0.02% per annum.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed

and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

In order to target a reduction in tracking difference and/or target an improved performance of the Fund, the Investment Manager may invest in swaps which reference a gross total return version of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the "**ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of Financial Derivative Instruments. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits) put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

(i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and

(ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- IV. adjust the terms of the Swaps; and/or
- V. delay payment to the Fund under the terms of the Swaps; and/or
- VI. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may

temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The Fund will not be leveraged. The market risk of the Fund associated with the use of financial derivative instruments will be measured using the commitment approach in accordance with the requirements of the Central Bank.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading "**Borrowing** and Lending Powers", the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hUSD acc, hEUR acc, hCHF acc, hJPY acc, hGBP acc, hPLN acc, hNOK acc, hSEK acc, hSGD acc, hHKD acc, EUR acc, CHF acc, GBP acc and Class JPY acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the hUSD UKdis, hEUR UKdis, hGBP UKdis, hGBP dis, hJPY UKdis, hCHF UKdis, hPLN UKdis, hNOK UKdis, USD UKdis, EUR UKdis, CHF UKdis, GBP UKdis, hSEK UKdis, hSGD UKdis, hHKD UKdis and JPY UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or

realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the declaration of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges on or about the launch date of the relevant Share Class. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended
Base Currency	US Dollar ("USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), and (B) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).
	The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00pm. (Dublin time) 1 Business Day prior to the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share
Website	www.ubs.com/etf - information on Fund Assets and the Fund's iNAV is set out on the Website.

Description of the Shares

Share Classes

Share Class***	*USD acc	*hUSD acc	*hEUR acc	*hCHF acc	*hGBP acc	**hUSD UKdis
Initial Issue Price	N/A	N/A	N/A	N/A	GBP 100	N/A
Minimum Subscription Amount****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000
Minimum Redemption Amount*****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000

Share Class***	**hEUR UKdis	**hGBP UKdis	**hCHF UKdis	**hGBP dis	*hJPY acc	*hJPY UKdis
Initial Issue Price	N/A	N/A	N/A	GBP 100	N/A	JPY 1,000
Minimum Subscription Amount****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000
Minimum Redemption Amount*****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000

Share Class***	*EUR acc	*CHF acc	*GBP acc	*JPY acc	**USD UKdis	**EUR UKdis
Initial Issue Price	EUR 100	CHF 100	GBP 100	JPY 1,000	USD 100	EUR 100
Minimum Subscription Amount****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000
Minimum Redemption Amount*****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000

Share Class***	**CHF UKdis	**GBP UKdis	**JPY UKdis	*hPLN acc	*hNOK acc	*hSEK acc
Initial Issue Price	CHF 100	GBP 100	JPY	PLN	NOK	SEK
			1,000	1,000	1,000	1,000
Minimum	USD	USD	USD	USD	USD	USD
Subscription Amount****	100,000	100,000	100,000	100,000	100,000	100,000
Minimum	USD	USD	USD	USD	USD	USD
Redemption Amount*****	100,000	100,000	100,000	100,000	100,000	100,000

Share Class***	*hSGD acc	*hHKD acc	**hPLN UKdis	**hNOK UKdis	**hSEK UKdis	**hSGD UKdis
Initial Issue Price	SGD 100	HKD 1,000	PLN 1,000	NOK 1,000	SEK 1,000	SGD 100
Minimum Subscription Amount****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000
Minimum Redemption Amount*****	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000	USD 100,000

Share	**hHKD	
Class***	UKdis	
Initial Issue	HKD	
Price	1,000	
Minimum Subscription Amount****	USD 100,000	
Minimum Redemption Amount*****	USD 100,000	

* Together the "acc" Share Class

** Together the "**dis**" Share Class

*** Each hedged Share Class will have a currency hedged exposure through the relevant Currency Index. Hedged Share Classes performance may differ from that of the Base Currency Share Classes due to, inter alia, currency hedging and currency hedging costs. **** The Minimum Subscription Amount may also be its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size and the Supplement will be updated accordingly.

***** The Minimum Redemption Amount may also be its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket size and the Supplement will be updated accordingly.

Information regarding the Initial Issue Price is available from the Administrator. Details on the Reference Index can be found at <u>www.msci.com</u>.

Initial Offer Period

The hCHF acc Share Class, the hCHF UKdis Share Class, the hEUR acc Share Class, the hEUR UKdis Share Class, the hGBP UKdis Share Class, the hUSD acc Share Class, the hUSD UKdis Share Class, the USD acc Share Class, the USD UKdis Share Class and the hJPY acc Share Class and the hSGD acc Share Class have launched and Shares in these Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hGBP acc Share Class, the hGBP dis Share Class, the hJPY UKdis Share Class, the EUR acc Share Class, the CHF acc Share Class, the GBP acc Share Class, the JPY acc Share Class, the EUR UKdis Share Class, the CHF UKdis Share Class, the GBP UKdis Share Class, the JPY UKdis Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the hSEK acc Share Class, the hHKD acc Share Class, the hPLN UKdis Share Class, the hNOK UKdis Share Class, the hSEK UKdis Share Class, the hSGD UKdis Share Class and the hHKD UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund).

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"	
Flat-Fees 0.21% per annum		0.21% per annum	

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index is a free float-adjusted market capitalization-weighted index that captures the large and mid-cap representation of equities across 23 developed markets and 23 emerging markets countries, with a current aggregate total of 2,464 constituents. The components of the Reference Index are primarily composed of consumer discretionary (being businesses that tend to be the most sensitive to economic cycles. Its manufacturing segment includes automotive, household durable goods, leisure equipment and textiles & apparel. The services segment includes hotels, restaurants and other leisure facilities, media production and services, and consumer retailing and services), financials and information technology companies. For the purposes of the Reference Index, developed market countries include: Australia, Austria, Belgium, Canada, Denmark, Finland, France, Germany, Hong Kong, Ireland, Israel, Italy, Japan, Netherlands, New Zealand, Norway, Portugal, Singapore, Spain, Sweden, Switzerland, the UK and the US. Emerging market countries include: Brazil, Chile, China, Colombia, Czech Republic, Egypt, Greece, Hungary, India, Indonesia, Korea, Malaysia, Mexico, Peru, Philippines, Poland, Russia, Qatar, South Africa, Taiwan, Thailand, Turkey and United Arab Emirates. The Reference Index is based on the MSCI Global Investable Market Indexes (GIMI) Methodology, further information on which is set out on the Index Provider's website detailed below.

The Reference Index is calculated in USD. The Reference Index has different Currency Indices, including Currency Indices denominated in Euro, GBP, JPY and CHF. Such Currency Indices employ forward FX contracts to provide the relevant currency return. The Currency Indices are designed to represent the returns of the Reference Index while reducing currency risk but not the underlying equity market risk. By employing such a strategy, the Currency Indices seek to reduce the risk of downward currency fluctuations although potential currency gains may also be sacrificed. The Reference Index value in respect of the Currency Indices is determined in accordance with the methodology index and further information available is at http://www.msci.com/eqb/methodology/meth_docs/MSCI_May14_GIMIMethod.pdf

The Reference Index is rebalanced on a quarterly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is sponsored by MSCI Limited.

- 1. General information on the Reference Index (a variant of the MSCI ACWI Index) may be found at:
- <u>http://www.msci.com/resources/factsheets/index_fact_sheet/msci-acwi-net.pdf</u> 2. Reference Index Methodology:
- http://www.msci.com/eqb/methodology/meth_docs/MSCI_May14_GIMIMethod.pdf 3. Reference Index Calculation Methodology:
- <u>http://www.msci.com/eqb/methodology/meth_docs/MSCI_Apr14_IndexCalcMethodology.pdf</u> 4. Reference Index Constituents:
- <u>http://www.msci.com/products/indexes/licensing/constituents.html</u> Select "ACWI" from the drop down menu.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences in

currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cashflows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.

- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due, inter alia, to currency hedging costs.
- (i) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (j) The Fund is exposed to the investment risk arising from the Reference Index underlying the Swaps. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (k) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (I) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index licence, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (m) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including, without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (n) The Reference Index is comprised of developed market and emerging market country indices including India. The Securities and Exchange Board of India ("SEBI") imposes

certain obligations under the FII (Foreign Institutional Investor) Regulations on investors seeking to invest either directly and or indirectly in securities listed or proposed to be listed in India ("Indian Securities"). In the case of the Fund, such obligations would apply where the Indian listed stock component of the Reference Index exceeds certain prescribed limits which are set by SEBI. Currently, the Indian listed stock component in the Reference Index remains below the prescribed limits. However, should the SEBI limit requirements change, this may have an impact on the Fund as an indirect investor in Indian Securities by reference Index and consequently the Fund may need to be closed to further subscriptions.

(o) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

THIS FUND IS NOT SPONSORED, ENDORSED, SOLD OR PROMOTED BY MSCI LIMITED. ("MSCI"), ANY AFFILIATE OF MSCI OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX. THE MSCI INDEXES ARE THE EXCLUSIVE PROPERTY OF MSCI. MSCI AND THE MSCI INDEX NAMES ARE SERVICE MARK(S) OF **MSCI OR ITS AFFILIATES AND HAVE BEEN LICENSED FOR USE FOR CERTAIN PURPOSES** BY LICENSEE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, TO THE OWNERS OF THIS FUND OR ANY MEMBER OF THE PUBLIC REGARDING THE ADVISABILITY OF INVESTING IN FINANCIAL SECURITIES GENERALLY OR IN THIS FUND PARTICULARLY OR THE ABILITY OF ANY MSCI INDEX TO TRACK CORRESPONDING STOCK MARKET PERFORMANCE. MSCI OR ITS AFFILIATES ARE THE LICENSORS OF CERTAIN TRADEMARKS, SERVICE MARKS AND TRADE NAMES AND OF THE MSCI INDEXES WHICH ARE DETERMINED, COMPOSED AND CALCULATED BY MSCI WITHOUT REGARD TO THIS FUND OR THE ISSUER OR OWNER OF THIS FUND. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION TO TAKE THE NEEDS OF THE ISSUERS OR OWNERS OF THIS FUND INTO CONSIDERATION IN DETERMINING, COMPOSING OR CALCULATING THE MSCI INDEXES. NEITHER MSCI, ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX IS RESPONSIBLE FOR OR HAS PARTICIPATED IN THE DETERMINATION OF THE TIMING OF, PRICES AT, OR QUANTITIES OF THIS FUND TO BE ISSUED OR IN THE DETERMINATION OR CALCULATION OF THE EQUATION BY WHICH THIS FUND IS REDEEMABLE FOR CASH. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, THE MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION OR LIABILITY TO THE OWNERS OF THIS FUND IN CONNECTION WITH THE ADMINISTRATION, MARKETING OR OFFERING OF THIS FUND.

ALTHOUGH MSCI SHALL OBTAIN INFORMATION FOR INCLUSION IN OR FOR USE IN THE CALCULATION OF THE MSCI INDEXES FROM SOURCES WHICH MSCI CONSIDERS RELIABLE, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX WARRANTS OR GUARANTEES THE ORIGINALITY, ACCURACY AND/OR THE COMPLETENESS OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE **OBTAINED BY LICENSEE, LICENSEE'S CUSTOMERS OR COUNTERPARTIES, ISSUERS OF** THE FINANCIAL SECURITIES, OWNERS OF THE FINANCIAL SECURITIES, OR ANY OTHER PERSON OR ENTITY, FROM THE USE OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN IN CONNECTION WITH THE RIGHTS LICENSED HEREUNDER OR FOR ANY OTHER USE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX SHALL HAVE ANY LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS OF OR IN CONNECTION WITH ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. FURTHER, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY EXPRESS OR IMPLIED WARRANTIES OF ANY KIND, AND MSCI, ANY OF ITS AFFILIATES AND ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX HEREBY EXPRESSLY DISCLAIM ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE. WITH RESPECT TO ANY MSCI INDEX AND ANY DATA INCLUDED THEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL MSCI, ANY OF ITS AFFILIATES OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, SPECIAL, PUNITIVE, CONSEQUENTIAL OR ANY OTHER DAMAGES (INCLUDING LOST PROFITS) EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

No purchaser, seller or holder of this security, or any other person or entity, should use or refer to any MSCI trade name, trademark or service mark to sponsor, endorse, market or promote this product without first contacting MSCI to determine whether MSCI's permission is required. Under no circumstances may any person or entity claim any affiliation with MSCI without the prior written permission of MSCI.

UBS CMCI EX-AGRICULTURE SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS CMCI EX-AGRICULTURE SF UCITS ETF** (the **"Fund"**), a sub-fund of UBS (IrI) Fund Solutions plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being UBS CMCI Ex-Agriculture Ex-Livestock Capped Index Total Return).

The Fund will return the Reference Index or Currency Index performance less fees and expenses on each Dealing Day.

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index / Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the

nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anticorruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index"** and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.04% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be

made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDI. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's maximum exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

(i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and

For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus. German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by

such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The Fund will not be leveraged. The market risk of the Fund associated with the use of FDIs will be measured using the commitment approach in accordance with the requirements of the Central Bank.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers**", the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund (**"ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Turne	Onen anded	
Туре	Open-ended	
Base Currency	US Dollar (" USD ")	
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), (B) banks are open for business in London and New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.	
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnig subject always to the Directors' discretion to temporarily suspend the determination of Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund accordance with the provisions of the Prospectus and the Articles.	
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).	
	The dealing calendar for the Fund is available from the Manager.	
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.	
Minimum Fund Size	USD 100 million	
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.	
Settlement Date	Up to three Business Days after the relevant Dealing Day.	
Minimum Holding	1 Share	
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.	
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.	
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.	

Description of the Shares

Share Classes

Share	USD acc	**hEUR	**h CHF	**hGBP	**hHKD	**hSGD	**hJPY	**hSEK
Class*		acc	acc	acc	acc	acc	acc	acc
Initial Issue Price	N/A	N/A	N/A	N/A	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000

Share	**hPLN	**hNOK	***USD	***hEUR	***hCHF	***hGBP	***hHKD	***hSGD
Class*	acc	acc	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue Price	PLN 1000	NOK 1000	USD 100	EUR 100	CHF 100	GBP 100	HKD 1,000	SGD 100

Share	***hJPY	***hSEK	***hPLN	***hNOK
Class*	UKdis	UKdis	UKdis	UKdis
Initial Issue Price	JPY 1,000	SEK 1,000	PLN 1,000	NOK 1,000

**Together the "acc" Share Classes.

*** Together the "dis" Share Class.

Each hedged Share Class will have a currency hedged exposure through the relevant Currency Index. Hedged Share Classes performance may differ from that of the Base Currency Share Classes due to, inter alia, currency hedging and currency hedging costs.

Initial Offer Period

The hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class and the USD acc Share Class have launched and Shares in these Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hHKD acc, Share Class, the hSGD) acc Share Class, the hJPY acc Share Class, the hSEK acc Share Class, hPLN acc Share Class, the hNOK acc Share Class, the USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"	
Flat-Fees	0.34% per annum	0.34% per annum	

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed **"Fees and Expenses"** should be read in conjunction with the sections headed **"Fees and Expenses"** and **"Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets"** in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index was developed to offer investors a commodity index that recognised the various maturities of commodity futures contracts that are available in commodity markets but excludes exposure to two commodity categories of agriculture and livestock. Initially, commodity indices have been limited to those offering an index based on short term commodity futures contracts, whereas the Reference Index enables the construction of commodity exposures across maturities and across three commodity categories of energy, industrial metals and precious metals. The Reference Index is made up of 14 components across three commodity categories.

For an index providing exposure to commodities such as the Reference Index, where categories of components making up the relevant index are highly correlated, such highly correlated categories must be considered as sub-categories of the same commodity (in other words, components that are highly correlated must be considered as representing one component of the relevant index for the purposes of calculating the Fund's exposures). The Reference Index includes one component with a weighting of more than 20% and up to 35% (taking into account high correlation between certain components of the Reference Index such that they are considered together to constitute one component). Pursuant to the Regulations, the Fund is permitted to gain exposure to one component that represents more than 20% and up to 35% of the Reference Index where justified by exceptional market conditions.

Whilst the Reference Index is designed to provide broad-based exposure to commodities as an asset class across three main groups (energy, industrial metals and precious metals) the oil and gasoline components within the energy group are highly correlated, and on a combined basis, they have a typical weighting of more than 20% and up to 35% of the Reference Index. Their high weighting in the Reference Index is partly due to the economic significance of the energy group and of these components as measured by their relative share of global consumption. In order to properly reflect the economic significance of the oil and gasoline components within the energy group and within the commodities sector as a whole, it is necessary for the Reference Index, and consequently the Fund, to have a typical weighting of more than 20% and up to 35% to these highly correlated components.

The Reference Index is calculated in USD, and includes a notional exposure to overnight interest rates. The Reference Index has different Currency Indices, including Currency Indices denominated in Euro, GBP, CHF, HKD, SGD, JPY and SEK. Such Currency Indices employ forward FX contracts to provide the relevant currency return. The Currency Indices are designed to represent the returns of the Reference Index while reducing currency risk but not the underlying commodity market risk. By employing such a strategy, the Currency Indices seek to reduce the risk of downward currency fluctuations although potential currency gains may also be sacrificed.

The Reference Index is rebalanced on a monthly basis back to its target weights.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is governed by MerQube, Inc (the "Index Administrator") via its internal processes. The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index. More details on the Reference Index, including its underlying components, can be found at https://www.ubs.com/global/en/investment-bank/cmci/ubs-strategies.html.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of Share Class denomination. Consequently, returns in

respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due, inter alia, to currency hedging costs.

- (i) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (j) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in commodity markets.
- (k) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchangebased" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (I) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (m) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (n) The Fund, through its use of the Total Return Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (o) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

UBS Index Disclaimer

No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the Index Administrator. The Index is subject to the terms set out in the Index Manual, as such Index Manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to UBS.

CMCI is a trademark of UBS AG.

Index Disclaimer for MerQube

All information provided by MerQube in this Prospectus is impersonal and not tailored to the needs of any person, entity or group of persons or entities. MerQube receives remuneration in connection with the licensing of its indices to third parties and the provision of customised calculation services.

Past performance of an index is not an indication or guarantee of future results.

It is not possible to invest directly in an index. Exposure to an asset class represented by an index is only possible through investable instruments based on that index. MerQube does not sponsor, endorse, sell, promote or manage investment funds or other investment vehicles offered by third parties whose objective is to achieve an investment return based on the performance of an index. MerQube makes no representation that investment products based on the Index will accurately track the performance of the Index or provide positive investment returns. MerQube, Inc. is not an investment adviser and MerQube makes no representation as to the advisability of investing in any such mutual fund or other investment vehicle. Any decision to invest in such mutual fund or other investment vehicle should not be made in reliance on the statements made in this document. Prospective investors are advised to make an investment in such a fund or other investment vehicle only after careful consideration of the risks associated with an investment in such funds, which are set out in detail in a prospectus or similar document prepared by or on behalf of the issuer of the fund or other investment product or vehicle.

The closing prices of the indices are calculated by MerQube on the basis of the closing prices of the individual components of the index as determined by the respective main stock exchange. MerQube receives the closing prices from one of its third-party providers and verifies them by comparing them with the prices of another provider. The providers receive the closing prices from the primary exchanges.

MerQube and its data suppliers (collectively, the "MerQube Parties") do not warrant the accuracy, completeness, timeliness or availability of this Prospectus. The MerQube Parties are not responsible for any errors or omissions, regardless of cause, in the results obtained from the use of the Content. THE MERQUBE PARTIES DISCLAIM ALL WARRANTIES, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE, FREE OF BUGS, SOFTWARE ERRORS OR DEFECTS, THAT THE OPERATION OF THE CONTENT WILL NOT BE UN-INTERRUPTED, OR THAT THE CONTENT WILL WORK WITH ANY SOFTWARE OR HARDWARE CONFIGURATION. In no event shall the MerQube Parties be liable for any direct, incidental, exemplary, compensatory, punitive, special or consequential damages, costs, expenses, attorneys' fees or losses (including, without limitation, lost revenues or lost profits and opportunity costs) in connection with the use of the Content, even if advised of the possibility of such damages.

Publication of performance

Past performance is not an indicator or guarantee of future results. All information presented prior to the launch date has been retrospectively tested. The back-tested performance is not an actual performance, but a hypothetical one. Backtest calculations are based on the same methodology in effect at the time of the official launch of the index(es). However, back-tested data may reflect the application of the index methodology in retrospect and historical calculations may change due to revisions in the economic data used for index calculation. Full details of the index methodology can be found at www.merqube.com. It is not possible to invest directly in an index.

The first value date is the first day for which a calculated value (either live or backtested) is available for a particular index. The inception date is the date on which the values of a particular index are first considered live: Index values provided for any date or time period prior to the launch date of the index are considered backtested. MerQube defines the Launch Date as the date on which the values of an index are known to have been made available to the public.

The index returns shown do not represent the results of actual securities trading. MerQube, Inc. does not manage actual assets. The index returns do not reflect the payment of any sales charges or fees that an investor may be required to pay for the purchase of the securities underlying the index(es) or for mutual funds designed to track the performance of the index(es). The imposition of these fees and costs would result in the actual and retrospectively tested performance of the fund/investment vehicle being lower than the performance of the index that the vehicle/fund tracks.

UBS BBG Commodity CMCI SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS BBG Commodity CMCI SF UCITS ETF** (the "**Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the "**Company**") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "**Central Bank**").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

a) IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being UBS BCOM Constant Maturity Commodity Index Total Return).

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index/ Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

- a) The Fund will take long positions only and the Fund will invest 100% of its net assets in long positions.
- b) The Fund is passively managed.

c) The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-

bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.04% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the "**Approved Counterparty**") pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This

may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

(i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and

(ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The Fund will not be leveraged. The market risk of the Fund associated with the use of FDIs will be measured using the commitment approach in accordance with the requirements of the Central Bank.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application has been and will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges"**) for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and

institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

a) GENERAL INFORMATION RELATING TO THE FUND

-	
Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), (B) banks are open for business in London and New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the
	Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager). The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	USD 100 million.
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share.
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.

Description of the Shares

Share Classes †			
"acc" Share Classes	Initial Issue Price	"dis" Share Classes	Initial Issue Price
*USD acc	N/A	**USD UKdis	USD 100
*hEUR acc	N/A	**hEUR UKdis	EUR 100
*h CHF acc	N/A	**hCHF UKdis	CHF 100
*hGBP acc	N/A	**hGBP UKdis	GBP 100
*hHKD acc	HKD 1,000	**hHKD UKdis	HKD 1,000
*hSGD acc	SGD 100	**hSGD UKdis	SGD 100
*hJPY acc	JPY 1,000	**hJPY UKdis	JPY 1,000
*hSEK acc	SEK 1,000	**hSEK UKdis	SEK 1,000
*hPLN acc	PLN 1,000	**hPLN UKdis	PLN 1,000
*hNOK acc	NOK 1,000	**hNOK UKdis	NOK 1,000

* Together the "acc" Share Classes

** Together the "dis" Share Class

† Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Initial Offer Period

The hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class and the USD acc Share Class have launched and Shares in these Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hSEK acc Share Class, the hJPY acc Share Class, hSGD acc Share Class, hHKD acc Share Class, the hPLN acc Share Class, the hNOK acc Share Class, USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class "acc"		"dis"	
Flat-Fees	0.34% per annum	0.34% per annum	

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information

The Reference Index was developed to offer investors a commodity index that recognised the various maturities of commodity futures contracts that are available in commodity markets. Previously, commodity indices have been limited to those offering an index based on short term commodity futures contracts, whereas the Reference Index enables the construction of commodity exposures across maturities and across commodity categories of energy, industrial metals, precious metals, agriculture and livestock. The Reference Index is made up of 22 components across commodity categories.

For an index providing exposure to commodities such as the Reference Index, where categories of components making up the relevant index are highly correlated, such highly correlated categories must be considered as sub-categories of the same commodity (in other words, components that are highly correlated must be considered as representing one component of the relevant index for the purposes of calculating the Fund's exposures). The Reference Index includes one component with a weighting of more than 20% and up to 35% (taking into account high correlation between certain components of the Reference Index such that they are considered together to constitute one component). Pursuant to the Regulations, the Fund is permitted to gain exposure to one component that represents more than 20% and up to 35% of the Reference Index where justified by exceptional market conditions.

Whilst the Reference Index is designed to provide broad-based exposure to commodities as an asset class across five main groups (agriculture, energy, industrial metals, precious metals and livestock), the oil and gasoline components within the energy group are highly correlated, and on a combined basis, they have a typical weighting of more than 20% and up to 35% of the Reference Index. Their high weighting in the Reference Index is partly due to the economic significance of the energy group and of these components as measured by their relative share of global production and futures liquidity. In order to properly reflect the economic significance of the oil and gasoline components within the energy group and within the commodities sector as a whole, it is necessary for the Reference Index, and consequently the Fund, to have a typical weighting of more than 20% and up to 35% to these highly correlated components.

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index is calculated in USD, and includes a notional exposure to overnight interest rates. The Reference Index has different Currency Indices, including Currency Indices denominated in Euro, GBP, CHF, HKD, SGD, JPY and SEK. Such Currency Indices employ forward FX contracts to provide the relevant currency return. The Currency Indices are designed to represent the returns of the Reference Index while reducing currency risk but not the underlying commodity market risk. By employing such a strategy, the Currency Indices seek to reduce the risk of downward currency fluctuations although potential currency gains may also be sacrificed.

The Reference Index is rebalanced on a monthly basis back to its target weights. Details of the target weights are available at https://www.ubs.com/global/en/investment-bank/cmci/ubs-strategies.html.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is governed by MerQube, Inc (the "Index Administrator") via its internal processes. The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index. More details on the Reference Index, including its

underlying components and its calculation methodology can be found at <u>https://www.ubs.com/global/en/investment-bank/cmci/ubs-strategies.html</u>.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into a separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.

- (i) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (j) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in commodity markets.
- (k) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchangetraded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (I) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (m) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (n) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (o) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

Bloomberg Disclaimer

"Bloomberg®" and "Bloomberg Commodities IndexSM" are service marks of Bloomberg Finance L.P. and its affiliates, including Bloomberg Index Services Limited ("BISL"), the administrator of the indices (collectively, "Bloomberg") and have been licensed for use for certain purposes by UBS AG.

The Fund is not sponsored, endorsed, sold or promoted by Bloomberg. Bloomberg does not make any representation or warranty, express or implied, to the owners of or counterparties to the Fund or any member of the public regarding the advisability of investing in securities or commodities generally or in the Fund particularly. The only relationship of Bloomberg to UBS AG is the licensing of certain trademarks, trade names and service marks and of the Bloomberg Commodities IndexSM, which is determined, composed and calculated by BISL without regard to UBS AG or the Fund. Bloomberg has no obligation to take the needs of UBS AG or the owners of the Fund into consideration in determining, composing or calculating Bloomberg Commodities IndexSM. Bloomberg is not responsible for and has not participated in the determination of the timing of, prices at, or quantities of the Fund to be issued or in the determination or calculation of the equation by which the Fund are to be converted into cash. Bloomberg shall not have any obligation or liability, including, without limitation, to Fund customers, in connection with the administration, marketing or trading of the Fund.

The Supplement relates only to the Fund and does not relate to the exchange-traded physical commodities underlying any of the Bloomberg Commodity IndexSM components. Purchasers of the Fund should not conclude that the inclusion of a futures contract in the Bloomberg Commodity IndexSM is any form of investment recommendation of the futures contract or the underlying exchange-traded physical commodity by Bloomberg. The information in the Supplement regarding the Bloomberg Commodity IndexSM components has been derived solely from publicly available documents. Bloomberg has not made any due diligence inquiries with respect to the Bloomberg Commodity IndexSM components in connection with the Fund. Bloomberg makes no representation that these publicly available documents or any other publicly available information regarding the Bloomberg Commodity IndexSM components, including without limitation a description of factors that affect the prices of such components, are accurate or complete.

BLOOMBERG DOES NOT GUARANTEE THE ACCURACY AND/OR THE COMPLETENESS OF THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO AND SHALL HAVE NO LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS THEREIN. BLOOMBERG DOES NOT MAKE ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE OBTAINED BY UBS AG, OWNERS OF THE FUND OR ANY OTHER PERSON OR ENTITY FROM THE USE OF THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO. BLOOMBERG DOES NOT MAKE ANY EXPRESS OR IMPLIED WARRANTIES AND EXPRESSLY DISCLAIMS ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE WITH RESPECT TO THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO. WITHOUT LIMITING ANY OF THE FOREGOING, TO THE MAXIMUM EXTENT ALLOWED BY LAW, BLOOMBERG, ITS LICENSORS, AND ITS AND THEIR RESPECTIVE EMPLOYEES, CONTRACTORS, AGENTS, SUPPLIERS, AND VENDORS SHALL HAVE NO LIABILITY OR RESPONSIBILITY WHATSOEVER FOR ANY INJURY OR DAMAGES—WHETHER DIRECT, INDIRECT, CONSEQUENTIAL, INCIDENTAL, PUNITIVE OR OTHERWISE—ARISING IN CONNECTION WITH THE FUND OR THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA OR VALUES RELATING THERETO—WHETHER ARISING FROM THEIR NEGLIGENCE OR OTHERWISE, EVEN IF NOTIFIED OF THE POSSIBILITY THEREOF.

BLOOMBERG and BLOOMBERG INDICES are trademarks or service marks of Bloomberg Finance L.P. Bloomberg Finance L.P. and its affiliates, including Bloomberg Index Services Limited, the administrator of BLOOMBERG INDICES (collectively, "Bloomberg") or Bloomberg's licensors own all proprietary right in the BLOOMBERG INDICES. Bloomberg does not guarantee the timeliness, accuracy or completeness of any data or information relating to BLOOMBERG INDICES. Bloomberg makes no warranty, express or implied, as to the BLOOMBERG INDICES or any data or values relating thereto or results to be obtained therefrom, and expressly disclaims all warranties of merchantability and fitness for a particular purpose with respect thereto. It is not possible to invest directly in an index. Back-tested performance is not actual performance. Past performance is not an indication of future results. To the maximum extent allowed by law, Bloomberg, its licensors, and its and their respective employees, contractors, agents, suppliers and vendors shall have no liability or responsibility whatsoever for any injury or damages - whether direct, indirect, consequential, incidental, punitive or otherwise - arising in connection with BLOOMBERG INDICES or any data or values relating thereto - whether arising from their negligence or otherwise. This document constitutes the provision of factual information, rather than financial product advice. Nothing in the BLOOMBERG INDICES shall constitute or be construed as an offering of financial instruments or as investment advice or investment recommendations (i.e., recommendations as to whether or not to "buy", "sell", "hold", or to enter or not to enter into any other transaction involving any specific interest or interests) by Bloomberg or its affiliates or a recommendation as to an investment or other strategy by Bloomberg or its affiliates. Data and other information available via the BLOOMBERG INDICES should not be considered as information sufficient upon which to base an investment decision. All information provided by the BLOOMBERG INDICES is impersonal and not tailored to the needs of any person, entity or group of persons. Bloomberg and its affiliates do not express an opinion on the future or expected value of any security or other interest and do not explicitly or implicitly recommend or suggest an investment strategy of any kind. Customers should consider obtaining independent advice before making any financial decisions. © 2022 Bloomberg Finance L.P. All rights reserved. This document and its contents may not be forwarded or redistributed without the prior consent of Bloomberg.

This Supplement has not been reviewed by Bloomberg Finance. This Supplement should be read in conjunction with the prospectus in respect of the relevant product.

UBS Index Disclaimer

No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the Index Administrator. The Index is subject to the terms set out in the Index Manual, as such Index Manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner. The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to UBS.

CMCI is a trademark of UBS AG.

Index Disclaimer for MerQube

All information provided by MerQube in this Prospectus is impersonal and not tailored to the needs of any person, entity or group of persons or entities. MerQube receives remuneration in connection with the licensing of its indices to third parties and the provision of customised calculation services.

Past performance of an index is not an indication or guarantee of future results.

It is not possible to invest directly in an index. Exposure to an asset class represented by an index is only possible through investable instruments based on that index. MerQube does not sponsor, endorse, sell, promote or manage investment funds or other investment vehicles offered by third parties whose objective is to achieve an investment return based on the performance of an index. MerQube makes no representation that investment products based on the Index will accurately track the performance of the Index or provide positive investment returns. MerQube, Inc. is not an investment adviser and MerQube makes no representation as to the advisability of investing in any such mutual fund or other investment vehicle. Any decision to invest in such mutual fund or other investment vehicle should not be made in reliance on the statements made in this document. Prospective investors are advised to make an investment in such a fund or other investment vehicle only after careful consideration of the risks associated with an investment in such funds, which are set out in detail in a prospectus or similar document prepared by or on behalf of the issuer of the fund or other investment product or vehicle.

The closing prices of the indices are calculated by MerQube on the basis of the closing prices of the individual components of the index as determined by the respective main stock exchange. MerQube receives the closing prices from one of its third-party providers and verifies them by comparing them with the prices of another provider. The providers receive the closing prices from the primary exchanges.

MerQube and its data suppliers (collectively, the "MerQube Parties") do not warrant the accuracy, completeness, timeliness or availability of this Prospectus. The MerQube Parties are not responsible for any errors or omissions, regardless of cause, in the results obtained from the use of the Content. THE MERQUBE PARTIES DISCLAIM ALL WARRANTIES, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE, FREE OF BUGS, SOFTWARE ERRORS OR DEFECTS, THAT THE OPERATION OF THE CONTENT WILL NOT BE UN-INTERRUPTED, OR THAT THE CONTENT WILL WORK WITH ANY SOFTWARE OR HARDWARE CONFIGURATION. In no event shall the MerQube Parties be liable for any direct, incidental, exemplary, compensatory, punitive, special or consequential damages, costs, expenses, attorneys' fees or losses (including, without limitation, lost revenues or lost profits and opportunity costs) in connection with the use of the Content, even if advised of the possibility of such damages.

Publication of performance

Past performance is not an indicator or guarantee of future results. All information presented prior to the launch date has been retrospectively tested. The back-tested performance is not an actual performance, but a hypothetical one. Backtest calculations are based on the same methodology in effect at the time of the official launch of the index(es). However, back-tested data may reflect the application of the index methodology in retrospect and historical calculations may change due to revisions in the economic data used for index calculation. Full details of the index methodology can be found at www.merqube.com. It is not possible to invest directly in an index.

The first value date is the first day for which a calculated value (either live or backtested) is available for a particular index. The inception date is the date on which the values of a particular index are first considered live: Index values provided for any date or time period prior to the launch date of the index are considered backtested. MerQube defines the Launch Date as the date on which the values of an index are known to have been made available to the public.

The index returns shown do not represent the results of actual securities trading. MerQube, Inc. does not manage actual assets. The index returns do not reflect the payment of any sales charges or fees that an investor may be required to pay for the purchase of the securities underlying the index(es) or for mutual funds designed to track the performance of the index(es). The imposition of these fees and costs would result in the actual and retrospectively tested performance of the fund/investment vehicle being lower than the performance of the index that the vehicle/fund tracks

UBS CMCI Commodity Carry SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS CMCI Commodity Carry SF UCITS ETF** (the **"Fund"**), a sub-fund of UBS (IrI) Fund Solutions plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KID, and may not be distributed unless accompanied by the Prospectus and the KID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a medium to high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being UBS CM-BCOM Outperformance Strategy Index ex-Precious Metals 2.5 Leveraged Net of Cost Total Return) less fees and expenses on each Dealing Day.

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index/ Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company.
- (iii) Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such transferable securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus.

All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

- (i) The Fund will take long positions only and the Fund will invest 100% of its net assets in long positions.
- (ii) The Fund may not invest more than 10% of its net assets in collective investment schemes.
- (iii) The Fund is passively managed.

(iv) The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

(v) The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

(vi) This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anticorruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.02% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and

d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the "**Approved Counterparty**") pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index and a cash payment which will equate to the Flat Fee (as described below).

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow (for example, Euribor):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances, the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances, the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically unfunded swaps do not require any upfront payment of cash.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;

(iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The total expected level of leverage of the Fund as a result of its investments shall not exceed 100% of its Net Asset Value, as calculated using the commitment approach.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

The Reference Index to which the Fund will be exposed will employ leverage approximately 2.5 times the value of the Reference Index, as further detailed in the Section entitled "General Description of the Reference Index" below.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended	
Base Currency	US Dollar (" USD ")	
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and 26 December), (B) banks are open for business in London and New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.	
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.	
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).	
	The dealing calendar for the Fund is available from the Manager.	
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.	
Minimum Fund Size	USD 100 million.	
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.	
Settlement Date	Up to three Business Days after the relevant Dealing Day.	
Minimum Holding	1 Share.	
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.	
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.	
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.	

Initial Offer Period

The hEUR acc Share Class, the hCHF acc Share Class, and the (USD) A- acc Share Class have launched and Shares in these Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period for the h GBP acc Share Class, the hHKD acc Share Class, the hSGD acc Share Class, the hJPY acc Share Class, the hSEK acc Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, h SEK UKdis Share Class, hPLN UKdis Share Class and hNOK UKdis Share Class has remained open during the period from 9 am on 13 May 2025 (Irish time) and has closed at 5 pm on 12 November 2025 (Irish time).

	Share Classes				
" Acc" Share Classes	" dis" Share Classes	Initial Issue Price			
*USD acc	**USD UKdis	USD 100			
*hEUR acc	**hEUR UKdis	EUR 100			
*hCHF acc	**hCHF UKdis	CHF 100			
*hGBP acc	**hGBP UKdis	GBP 100			
*hHKD acc	**hHKD UKdis	HKD 1,000			
*hSGD acc	**hSGD UKdis	SGD 100			
*hJPY acc	**hJPY UKdis	JPY 1,000			
*hSEK acc	**hSEK UKdis	SEK 1,000			
*hPLN acc	**hPLN UKdis	PLN 1,000			
*hNOK acc	**hNOK UKdis	NOK 1,000			

Description of the Shares

* Together the "acc" Share Class

** Together the "dis" Share Class

‡ Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%

Primary Market Transaction Costs***	Up to 1%	Up to 1%
--	----------	----------

**The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat Fees	0.34% per annum	0.34% per annum

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are estimated to be in the region of $\in 60,000.00$ but may seek to recover such establishment and approval costs if there is money available to it out of the Flat Fee. Any such expenses recovered from the Flat Fee will be charged over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will (in order to deliver a return which includes the Flat Fee per the agreed terms of the Swap) take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The UBS CM-BCOM Outperformance Strategy Index Ex-Precious Metals 2.5x Leveraged Net of Cost Total Return (the "Reference Index") represents a long-short exposure to a broad base of futures on commodities. The Reference Index takes a leveraged exposure of approximately 2.5 times the value of the UBS CM-BCOM Outperformance Strategy Index ex-Precious Metals Excess Return (the "COSIXPER Index").

The COSIXPER Index takes a long position in the UBS BCOM Constant Maturity ex-Precious Metals Excess Return Index (the "Long Underlying") and a short position in the Bloomberg Commodity ex-Precious Metals Index (the "Short Underlying") and accrues for a per annum cost and adds an interest rate return. Each of the Long Underlying and Short Underlying are highly liquid and diversified benchmarks for commodity investments. Each commodity component included in the Long Underlying is also included in the Short Underlying – however, the maturity of the exposures and frequency of rolling the exposures differs between the Long Underlying and Short Underlying.

The exposure through the futures to the underlying commodities, as represented in the Long Underlying and Short Underlying is rebalanced on a monthly basis on the Index Rebalancing Date, which is last Index Business Day of each month. The Reference Index rebalances the exposure to have a notional allocation of 250% to the COSIXPER Index on the Index Rebalancing Date, which is the last Index Business Day of each month.

On each Index Rebalancing Date, the weight allocated to the Long Underlying will be equal but opposite to the weight allocated to the Short Underlying. The Long Underlying rebalances monthly to the effective weights of the Short Underlying over a rolling period. Consequently on the Index Rebalance Date, the sum of the weights of each underlying component in the Long Underlying and Short Underlying on Rebalancing Date will be close to zero. For the purposes of calculating the exposure to any commodity component in accordance with Regulation 71(1)(a) of the UCITS Regulations within the COSIXPER Index, the exposure of a commodity component included in the Long Underlying will be netted against the exposure of that same commodity component included in the Short Underlying.

The aim of the Reference Index is to extract a leveraged premium from the difference in the methodologies between the Long Underlying and the Short Underlying.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is governed by MerQube, Inc (the "Index Administrator"). The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index.

More details on the Reference Index, including its underlying components can be found at <u>https://www.ubs.com/global/en/investment-bank/cmci/ubs-strategies.html</u>. Index Methodology is available separately on request from the Manager, free of charge, and without restriction.

The Index performance will include an index replication cost of 0.85% per annum. This cost is based on operational clearing; management (including the costs of creating a leveraged exposure); execution of commodities futures; ensuring the closing level for all tracked commodities; and futures margining requirements.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) The Reference Index provides an exposure to the difference in performance of two commodity indices. The Reference Index utilises leverage, meaning that its value may rise or fall more than the difference in performance of the two commodity indices to which it has exposure over the short, medium and/or long term.
- (g) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.

- (h) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (i) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (j) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (k) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in commodity markets.
- (I) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (m) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (n) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (o) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (p) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks are not considered as part of the index selection process and the Investment

Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

Bloomberg Index Disclaimer

"Bloomberg®" and "Bloomberg Commodities IndexSM" are service marks of Bloomberg Finance L.P. and its affiliates, including Bloomberg Index Services Limited ("BISL"), the administrator of the indices (collectively, "Bloomberg") and have been licensed for use for certain purposes by UBS AG.

The Fund is not sponsored, endorsed, sold or promoted by Bloomberg. Bloomberg does not make any representation or warranty, express or implied, to the owners of or counterparties to the Fund or any member of the public regarding the advisability of investing in securities or commodities generally or in the Fund particularly. The only relationship of Bloomberg to UBS AG is the licensing of certain trademarks, trade names and service marks and of the Bloomberg Commodities IndexSM, which is determined, composed and calculated by BISL without regard to UBS AG or the Fund. Bloomberg has no obligation to take the needs of UBS AG or the owners of the Fund into consideration in determining, composing or calculating Bloomberg Commodities IndexSM. Bloomberg is not responsible for and has not participated in the determination of the timing of, prices at, or quantities of the Fund to be issued or in the determination or calculation of the equation by which the Fund are to be converted into cash. Bloomberg shall not have any obligation or liability, including, without limitation, to Fund customers, in connection with the administration, marketing or trading of the Fund.

The Supplement relates only to the Fund and does not relate to the exchange-traded physical commodities underlying any of the Bloomberg Commodity IndexSM components. Purchasers of the Fund should not conclude that the inclusion of a futures contract in the Bloomberg Commodity IndexSM is any form of investment recommendation of the futures contract or the underlying exchange-traded physical commodity by Bloomberg. The information in the Supplement regarding the Bloomberg Commodity IndexSM components has been derived solely from publicly available documents. Bloomberg has not made any due diligence inquiries with respect to the Bloomberg Commodity IndexSM components in connection with the Fund. Bloomberg makes no representation that these publicly available documents or any other publicly available

information regarding the Bloomberg Commodity IndexSM components, including without limitation a description of factors that affect the prices of such components, are accurate or complete.

BLOOMBERG DOES NOT GUARANTEE THE ACCURACY AND/OR THE COMPLETENESS OF THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO AND SHALL HAVE NO LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS THEREIN. BLOOMBERG DOES NOT MAKE ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE OBTAINED BY UBS AG, OWNERS OF THE FUND OR ANY OTHER PERSON OR ENTITY FROM THE USE OF THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO. BLOOMBERG DOES NOT MAKE ANY EXPRESS OR IMPLIED WARRANTIES AND EXPRESSLY DISCLAIMS ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE WITH RESPECT TO THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA RELATED THERETO. WITHOUT LIMITING ANY OF THE FOREGOING, TO THE MAXIMUM EXTENT ALLOWED BY LAW, BLOOMBERG, ITS LICENSORS, AND ITS AND THEIR RESPECTIVE EMPLOYEES, CONTRACTORS, AGENTS, SUPPLIERS, AND VENDORS SHALL HAVE NO LIABILITY OR RESPONSIBILITY WHATSOEVER FOR ANY INJURY OR DAMAGES-WHETHER DIRECT, INDIRECT, CONSEQUENTIAL, INCIDENTAL, PUNITIVE OR OTHERWISE-ARISING IN CONNECTION WITH THE FUND OR THE BLOOMBERG COMMODITY INDEXSM OR ANY DATA OR VALUES RELATING THERETO-WHETHER ARISING FROM THEIR NEGLIGENCE OR OTHERWISE, EVEN IF NOTIFIED OF THE POSSIBILITY THEREOF.

BLOOMBERG and BLOOMBERG INDICES are trademarks or service marks of Bloomberg Finance L.P. Bloomberg Finance L.P. and its affiliates, including Bloomberg Index Services Limited, the administrator of BLOOMBERG INDICES (collectively, "Bloomberg") or Bloomberg's licensors own all proprietary right in the BLOOMBERG INDICES. Bloomberg does not guarantee the timeliness, accuracy or completeness of any data or information relating to BLOOMBERG INDICES. Bloomberg makes no warranty, express or implied, as to the BLOOMBERG INDICES or any data or values relating thereto or results to be obtained therefrom, and expressly disclaims all warranties of merchantability and fitness for a particular purpose with respect thereto. It is not possible to invest directly in an index. Back-tested performance is not actual performance. Past performance is not an indication of future results. To the maximum extent allowed by law, Bloomberg, its licensors, and its and their respective employees, contractors, agents, suppliers and vendors shall have no liability or responsibility whatsoever for any injury or damages - whether direct, indirect, consequential, incidental, punitive or otherwise - arising in connection with BLOOMBERG INDICES or any data or values relating thereto - whether arising from their negligence or otherwise. This document constitutes the provision of factual information, rather than financial product advice. Nothing in the BLOOMBERG INDICES shall constitute or be construed as an offering of financial instruments or as investment advice or investment recommendations (i.e., recommendations as to whether or not to "buy", "sell", "hold", or to enter or not to enter into any other transaction involving any specific interest or interests) by Bloomberg or its affiliates or a recommendation as to an investment or other strategy by Bloomberg or its affiliates. Data and other information available via the BLOOMBERG INDICES should not be considered as information sufficient upon which to base an investment decision. All information provided by the BLOOMBERG INDICES is impersonal and not tailored to the needs of any person, entity or group of persons. Bloomberg and its affiliates do not express an opinion on the future or expected value of any security or other interest and do not explicitly or implicitly recommend or suggest an investment strategy of any kind. Customers should consider obtaining independent advice before making any financial decisions. © 2022 Bloomberg Finance L.P. All rights reserved. This document and its contents may not be forwarded or redistributed without the prior consent of Bloomberg.

This Supplement has not been reviewed by Bloomberg Finance. This Supplement should be read in conjunction with the prospectus in respect of the relevant product.

UBS Index Disclaimer

No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or

change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the Index Administrator. The Index is subject to the terms set out in the Index Manual, as such Index Manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to UBS.

CMCI is a trademark of UBS AG.

Index Disclaimer for MerQube

All information provided by MerQube in this Prospectus is impersonal and not tailored to the needs of any person, entity or group of persons or entities. MerQube receives remuneration in connection with the licensing of its indices to third parties and the provision of customised calculation services.

Past performance of an index is not an indication or guarantee of future results.

It is not possible to invest directly in an index. Exposure to an asset class represented by an index is only possible through investable instruments based on that index. MerQube does not sponsor, endorse, sell, promote or manage investment funds or other investment vehicles offered by third parties whose objective is to achieve an investment return based on the performance of an index. MerQube makes no representation that investment products based on the Index will accurately track the performance of the Index or provide positive investment returns. MerQube, Inc. is not an investment adviser and MerQube makes no representation as to the advisability of investing in any such mutual fund or other investment vehicle. Any decision to invest in such mutual fund or other investment vehicle should not be made in reliance on the statements made in this document. Prospective investors are advised to make an investment in such a fund or other investment vehicle only after careful consideration of the risks associated with an investment in such funds, which are set out in detail in a prospectus or similar document prepared by or on behalf of the issuer of the fund or other investment product or vehicle.

The closing prices of the indices are calculated by MerQube on the basis of the closing prices of the individual components of the index as determined by the respective main stock exchange. MerQube receives the closing prices from one of its third-party providers and verifies them by comparing them with the prices of another provider. The providers receive the closing prices from the primary exchanges.

MerQube and its data suppliers (collectively, the "MerQube Parties") do not warrant the accuracy, completeness, timeliness or availability of this Prospectus. The MerQube Parties are not responsible for any errors or omissions, regardless of cause, in the results obtained from the use of the Content. THE MERQUBE PARTIES DISCLAIM ALL WARRANTIES, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE, FREE OF BUGS, SOFTWARE ERRORS OR DEFECTS, THAT THE OPERATION OF THE CONTENT WILL NOT BE UN-INTERRUPTED, OR THAT THE CONTENT WILL WORK WITH ANY SOFTWARE OR

HARDWARE CONFIGURATION. In no event shall the MerQube Parties be liable for any direct, indirect, incidental, exemplary, compensatory, punitive, special or consequential damages, costs, expenses, attorneys' fees or losses (including, without limitation, lost revenues or lost profits and opportunity costs) in connection with the use of the Content, even if advised of the possibility of such damages.

Publication of performance

Past performance is not an indicator or guarantee of future results. All information presented prior to the launch date has been retrospectively tested. The back-tested performance is not an actual performance, but a hypothetical one. Backtest calculations are based on the same methodology in effect at the time of the official launch of the index(es). However, back-tested data may reflect the application of the index methodology in retrospect and historical calculations may change due to revisions in the economic data used for index calculation. Full details of the index methodology can be found at www.merqube.com. It is not possible to invest directly in an index.

The first value date is the first day for which a calculated value (either live or backtested) is available for a particular index. The inception date is the date on which the values of a particular index are first considered live: Index values provided for any date or time period prior to the launch date of the index are considered backtested. MerQube defines the Launch Date as the date on which the values of an index are known to have been made available to the public.

The index returns shown do not represent the results of actual securities trading. MerQube, Inc. does not manage actual assets. The index returns do not reflect the payment of any sales charges or fees that an investor may be required to pay for the purchase of the securities underlying the index(es) or for mutual funds designed to track the performance of the index(es). The imposition of these fees and costs would result in the actual and retrospectively tested performance of the fund/investment vehicle being lower than the performance of the index that the vehicle/fund tracks.

UBS MSCI China A SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS MSCI China A SF UCITS ETF** (the "**Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the "**Company**") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "**Central Bank**").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID, and may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes, it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

An investment in the sub-fund should not constitute a substantial proportion of an investment portfolio and may not be appropriate for all investors.

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being MSCI China A Net Total Return Index (USD)) less fees and expenses on each Dealing Day.

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index/ Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equityrelated securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include, U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company.

Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such transferable securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the "**Fund Assets**" for the purposes of the Prospectus.

All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

(i) The Fund will take long positions only and the Fund will invest 100% of its net assets in long positions.

(ii) The Fund is passively managed.

(iii) The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anticorruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the securities which are underlying the Reference Index. Such Currency Indices will not implement a currency hedge against the currency of the Reference index itself. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.09% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and

d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index and a cash payment which will equate to the Flat Fee (as described below).

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow (for example, Euribor)):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances, the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances, the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically unfunded swaps do not require any upfront payment of cash.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

In order to target a reduction in tracking difference and/or target an improved performance of the Fund, the Investment Manager may invest in swaps which reference a gross total return version of the Reference Index.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer

collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to

the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include, U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under **"Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or

- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The Fund will not be leveraged. The market risk of the Fund associated with the use of FDIs will be measured using the commitment approach in accordance with the requirements of the Central Bank.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund.

The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hPLN acc, hNOK acc, hHKD acc, hSGD acc, hUSD acc, hJPY acc, hSEK acc, hEUR acc, hGBP acc and hCHF acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hPLN UKdis, *hNOK UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hUSD UKdis, hEUR UKdis hCHF UKdis and hGBP UKdis in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended	
Base Currency	US Dollar (" USD ")	
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), and (B) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.	
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.	
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).	
	The dealing calendar for the Fund is available from the Manager.	
Dealing Deadline	2.00 p.m. (Dublin time) one Business Day prior to the relevant Dealing Day.	
Minimum Fund Size	Fund USD 100 million.	
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.	
Settlement Date	Up to three Business Days after the relevant Dealing Day.	
Minimum Holding	1 Share.	
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.	
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.	
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.	

Initial Offer Period

The USD acc Share Class has launched and Shares in this Share Class are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The hCHF acc Share Class, the hEUR acc Share Class, the hGBP acc Share Class, the hPLN acc Share Class, the hNOK acc Share Class, the hHKD acc Share Class, the hSGD acc Share Class, the hJPY acc Share Class, the hSEK acc Share Class, hUSD acc Share Class, USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hUSD UKdis Share Class, hPLN UKdis Share Class, hNOK UKdis Share Class will open at 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Description of the Shares

Share Classes ‡			
"acc" Share Classes	"dis" Share Classes	Initial Issue Price	
*USD acc	**USD UKdis	USD 100	
*hPLN acc	**hPLN UKdis	PLN 1,000	
*hNOK acc	**hNOK UKdis	NOK 1,000	
*hHKD acc	**hHKD UKdis	HKD 1,000	
*hSGD acc	**hSGD UKdis	SGD 100	
*hJPY acc	**hJPY UKdis	JPY 1,000	
*hSEK acc	**hSEK UKdis	SEK 1,000	
*hUSD acc	**hUSD UKdis	USD 100	
*hEUR acc	**hEUR UKdis	EUR 100	
*hCHF acc	**hCHF UKdis	CHF 100	
*hGBP acc	**hGBP UKdis	GBP 100	

*Together the "acc" Share Class

**Together the "dis" Share Class

‡ Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

**The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"	
Flat Fees	0.295% per annum	0.295% per annum	

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are estimated to be in the region of $\in 60,000.00$ but may seek to recover such establishment and approval costs if there is money available to it out of the Flat Fee. Any such expenses recovered from the Flat Fee will be charged over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will (in order to deliver a return which includes the Flat Fee per the agreed terms of the Swap) take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed **"Fees and Expenses"** should be read in conjunction with the sections headed **"Fees and Expenses"** and **"Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets"** in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is the MSCI China A Net Total Return Index (USD) ("the Reference Index"). The Reference Index is further described below but this only represents an extract of information available and the Directors, the Manager, UBS AG, MSCI. (UBS AG and MSCI together, and/or any such other successor sponsor(s) to the Reference Index, the "Index Provider") and the Investment Manager have taken all reasonable measures to ensure the accuracy and completeness of such information.

The Reference Index is derived from the MSCI China A International Index but covers only those securities that are accessible through "Stock Connect". The Index is a free float-adjusted market capitalization-weighted index that captures large and mid-cap representation across China securities listed on the Shanghai and Shenzhen exchanges. The index is designed for international investors and is calculated using China A Stock Connect listings based on the offshore RMB exchange rate (CNH).

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index rebalances on a quarterly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

The Reference Index is calculated in USD (Dollar).

- (i) Index Provider and website
- (ii) The Reference Index is sponsored by MSCI Ltd.

1. General information on the Reference Index (a variant of the MSCI China A International Index) may be found at:

https://www.msci.com/documents/10199/35cd04e7-a7f2-41de-9316-e92db9f17cbc

2. Reference Index Methodology:

https://www.msci.com/eqb/methodology/meth_docs/MSCI_Aug19_GIMIMethod.pdf

3. Reference Index Calculation Methodology:

https://www.msci.com/eqb/methodology/meth_docs/MSCI_July2018_IndexCalcMethodology.pdf

4. Reference Index Constituents:

http://www.msci.com/products/indexes/licensing/constituents.html

Select "China A" from the drop down menu.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) Swap price volatility risk: The Fund will enter into FDIs with the swap counterparty to get the relevant exposure to the Reference Index. The Swap prices and transaction costs linked to the Reference Index exposure are volatile and may be subject to significant fluctuations over time, typically as a result of the underlying market dynamics of China A Shares. Investors should therefore note that swap prices at the time of subscription may vary significantly thereafter and such movement can have either a positive or negative impact on the overall performance of the Fund. Investors should also note that Swap prices may also be impacted by any substantial subscription and redemption activity at the level of the Fund.
- (b) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (c) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (d) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (e) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (f) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (g) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate

the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.

- (h) The Company on behalf of the Fund will enter into a separate Derivative Contract in respect of each Class. Each Derivative Contract will be on similar terms except that each Derivative Contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (i) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (j) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (k) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (I) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (m) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (n) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (o) The Fund's performance may be adversely affected by variations in the exchange rates between the base currency of the Fund and the currencies to which the Fund is exposed.
- (p) Investment in the People's Republic of China ("PRC") and PRC Economic Risks: The economy in the PRC, which has been in a state of transition from a planned economy to a more market orientated economy, differs from the economies of most developed countries and investing in the PRC may be subject to greater risk of loss than investments in developed markets, due, among other factors, to greater market volatility, lower trading volume, political and economic instability, greater risk of market shut down, greater control of foreign exchange and greater limitations on foreign investment policy than those typically found in a developed market. There may be substantial government intervention in the PRC economy, including restrictions on investment in companies or

industries deemed to be sensitive to relevant national interests. The PRC government and regulators may also intervene in the financial markets, such as by the imposition of trading restrictions, which may affect the trading of Chinese securities and ultimately may impact on the performance of the Fund. The Chinese companies in which the Reference Index invests may be held to lower disclosure, corporate governance, accounting and reporting standards than companies in more developed markets. In addition, some of the securities held with the Reference Index may be subject to higher transaction and other costs, foreign ownership limits, the imposition of withholding or other taxes, or may have liquidity issues which make such securities more difficult to sell at reasonable prices. These factors ultimately may have an unpredictable impact on the Fund's investments and increase the volatility. Furthermore, market interventions may have a negative impact on market sentiment which may in turn affect the performance of the Reference Index and therefore the performance of the Fund.

The PRC economy has experienced significant and rapid growth in the past 20 years. However, such growth may or may not continue, and may not apply evenly across different geographic locations and sectors of the PRC economy. Economic growth has also been accompanied by periods of high inflation. The PRC government has implemented various measures from time to time to control inflation and restrain the rate of economic growth of the PRC economy. Furthermore, the PRC government has carried out economic reforms to achieve decentralisation and utilisation of market forces to develop the economy of the PRC. These reforms have resulted in significant economic growth and social progress. There can, however, be no assurance that the PRC government will continue to pursue such economic policies or, if it does, that those policies will continue to be successful. Any such adjustment and modification of those economic policies may have an adverse impact on the securities markets in China and therefore ultimately on the performance of the Fund.

These factors may increase the volatility of the Fund and hence the risk of loss to the value of your investment.

- (q) The Fund's PRC Political Risks: Any political changes, social instability and adverse diplomatic developments which may take place in or in relation to the PRC could result in the imposition of additional governmental restrictions including expropriation of assets, confiscatory taxes or nationalisation of some or all of the constituents of the Reference Index. Investors should also note that any change in the policies of the government and relevant authorities of the PRC may adversely impact the securities markets in the PRC.
- (r) Legal System of the PRC: The PRC legal system is based on written statutes and their interpretation by the Supreme People's Court. Prior court decisions may be cited for reference but do not have precedent value. Since 1979, the PRC government has been developing a comprehensive system of commercial law, and progress has been made in introducing laws and regulations dealing with economic matters such as foreign investment, corporate organisation and governance, commerce, taxation and trade. However, because of the limited volume of published cases and judicial interpretation and their non-binding nature, the interpretation and enforcement of these regulations involves significant uncertainties.
- (s) Potential market volatility risk: Investors should note that the Shanghai Stock Exchange and Shenzhen Stock Exchange on which China A Shares are traded are undergoing development. Market volatility may result in significant fluctuation in the prices of securities traded on such markets, which could therefore ultimately impact upon the Net Asset Value of the Fund.
- (t) Accounting and Reporting Standards: PRC companies are required to follow PRC accounting standards and practices which follow international accounting standards to a certain extent. However, the accounting, auditing and financial reporting standards and practices applicable to PRC companies may be less rigorous, and there may be significant differences between financial statements prepared in accordance with the PRC accounting standards and practice and those prepared in accordance with international accounting standards. For example, there are differences in the valuation methods of properties and assets and in the requirements for disclosure of information to investors.
- (u) Taxation in the PRC: A 10% PRC withholding tax is applicable to the payment of dividends and interest to foreign investors by PRC listed companies. From a technical perspective, withholding tax is also legally applicable to capital gains realised by foreign investors on the disposal of China A Shares under the Corporate Income Tax. However, before the release of its Caishui ("Circular") 79,

the PRC tax authorities had not been reported to have sought in practice to collect such withholding tax on capital gains realised by foreign investors on the disposal of China A Shares.

Circulars 79 and 81 were released by the PRC tax authorities on 14 November 2014 in anticipation of the launch of the Stock Connect and in response to market concerns on the capital gains tax treatment of trading activity under this new programme and its comparability to the existing QFII/ RQFII regime. Under Circulars 79 and 81, capital gains realised on or after 17 November 2014 by foreign investors including QFIIs and RQFIIs from the disposal of China A Shares, including PRC "land-rich" companies (i.e. companies that have derived more than 50% of their value from immovable property located in the PRC at any time in the 3 years prior to the disposal), are temporarily exempt from Chinese taxation. The duration of this period of temporary exemption is not stated in either Circular 79 or 81 and is subject to termination by the PRC tax authorities with or without notice.

To the extent that the Fund uses swaps to achieve its investment objective, such instruments may include a provision whereby the Fund undertakes to keep the relevant counterparty fully or partially indemnified for any prospective or retrospective tax liability that it may suffer either directly or indirectly as a result of the relevant counterparty's direct or indirect holding of, interest in, or dealing in China A Shares for the purpose of hedging such swaps and/or other FDI.

If the relevant Calculation Agent determines, in its sole and absolute discretion, that the risk of the relevant counterparty suffering from such tax liability becomes material, it may at any time elect to make a downward adjustment to the valuation of the to reflect the expected amount of such tax liability. Any such adjustment of the valuation of the swap will remain in effect until such tax position of the relevant counterparty can be ascertained, at which point the calculation agent may make further upward or downward adjustments of the valuation of the swap accordingly. As a result, the attention of investors is drawn to the fact that i) the Net Asset Value of the Fund may be adversely impacted by any such adjustments to the valuation of the swap.

(v) Other risks associated with the PRC: Risks of trading liquidity and other disruptions

China A Shares are traded on the Shanghai and/or Shenzhen Stock Exchanges.

Liquidity for China securities will be impacted by any temporary or permanent suspensions of particular securities imposed from time to time by the Shanghai and/or Shenzhen Stock Exchanges or as a result of any regulatory or governmental intervention in relation to particular investments or the markets. Any such suspension may impact on the Net Asset Value of the Fund and may increase the tracking error of the Fund and may expose the Fund to losses.

If a significant portion of the constituents of the Reference Index are restricted or suspended, the Fund may, in the sole discretion of the Directors, suspend the determination of the Net Asset Value and the issue and redemption of Shares of the Fund in accordance with the section entitled "Suspension of Calculation of Net Asset Value" on page 51 of the Prospectus. In addition, where the last the closing or bid prices, as applicable, for suspended securities do not, in the opinion of the Manager, reflect their fair value or if prices are unavailable, the value will be calculated with care and in good faith by the Directors, or a competent person or firm appointed by the Directors and approved for that purpose by the Depositary, on the basis of the probable realisation value for such securities in accordance with the section titled "Valuation of the Assets" on page 48 to 50 of the Prospectus. Any temporary suspension of the issue and redemption of Shares in the Fund may cause the Shares of the Fund to trade at a significant premium or discount to the Net Asset Value on any stock exchange on which they are admitted for trading.

- (w) Investment Quota: Changes to foreign investment regulation in the PRC may be made at any time by the State Administration of Foreign Exchange (SAFE) and China Securities Regulatory Commission (CSRC), and such changes may have a detrimental impact on the ability of the Fund to achieve its Investment Objective. Any restriction in the amount of investment quota made available by the Chinese authorities may hinder the ability of the Swap Counterparty to increase the size of the OTC Swap transactions. The Board of Directors may elect to close the Fund for further subscriptions if no alternative swap counterparty can be appointed in respect of the Fund.
- (x) **Local Currency risk**: The Chinese government may impose capital controls on the flow of currency in and out of China which could impact the value of Fund investments negatively.

(y) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

THIS FINANCIAL PRODUCT IS NOT SPONSORED, ENDORSED, SOLD OR PROMOTED BY MSCI LTD.. ("MSCI"), ANY AFFILIATE OF MSCI OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX. THE MSCI INDEXES ARE THE EXCLUSIVE PROPERTY OF MSCI. MSCI AND THE MSCI INDEX NAMES ARE SERVICE MARK(S) OF MSCI OR ITS AFFILIATES AND HAVE BEEN LICENSED FOR USE FOR CERTAIN PURPOSES BY LICENSEE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, TO THE OWNERS OF THIS FINANCIAL PRODUCT OR ANY MEMBER OF THE PUBLIC REGARDING THE ADVISABILITY OF INVESTING IN FINANCIAL SECURITIES GENERALLY OR IN THIS FINANCIAL PRODUCT PARTICULARLY OR THE ABILITY OF ANY MSCI INDEX TO TRACK CORRESPONDING STOCK MARKET PERFORMANCE. MSCI OR ITS AFFILIATES ARE THE LICENSORS OF CERTAIN TRADEMARKS, SERVICE MARKS AND TRADE NAMES AND OF THE MSCI INDEXES WHICH ARE DETERMINED, COMPOSED AND CALCULATED BY MSCI WITHOUT REGARD TO THIS FINANCIAL PRODUCT OR THE ISSUER OR OWNER OF THIS FINANCIAL PRODUCT. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION TO TAKE THE NEEDS OF THE ISSUERS OR OWNERS OF THIS FINANCIAL PRODUCT INTO CONSIDERATION IN DETERMINING, COMPOSING OR CALCULATING THE MSCI INDEXES. NEITHER MSCI, ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX IS RESPONSIBLE FOR OR HAS PARTICIPATED IN THE DETERMINATION OF THE TIMING OF, PRICES AT, OR QUANTITIES OF THIS FINANCIAL PRODUCT TO BE ISSUED OR IN THE DETERMINATION OR CALCULATION OF THE EQUATION BY WHICH THIS FINANCIAL PRODUCT IS REDEEMABLE FOR CASH. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, THE MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION OR LIABILITY TO THE OWNERS OF THIS FINANCIAL PRODUCT IN CONNECTION WITH THE ADMINISTRATION, MARKETING OR OFFERING OF THIS FINANCIAL PRODUCT.

ALTHOUGH MSCI SHALL OBTAIN INFORMATION FOR INCLUSION IN OR FOR USE IN THE CALCULATION OF THE MSCI INDEXES FROM SOURCES WHICH MSCI CONSIDERS RELIABLE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX WARRANTS OR GUARANTEES THE ORIGINALITY, ACCURACY AND/OR THE COMPLETENESS OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE OBTAINED BY LICENSEE, LICENSEE'S CUSTOMERS OR COUNTERPARTIES, ISSUERS OF THE FINANCIAL SECURITIES, OWNERS OF THE FINANCIAL SECURITIES, OR ANY OTHER PERSON OR ENTITY, FROM THE USE OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN IN CONNECTION WITH THE RIGHTS LICENSED HEREUNDER OR FOR ANY OTHER USE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX SHALL HAVE ANY LIABILITY FOR ANY ERRORS, OMISSIONS OR INTERRUPTIONS OF OR IN CONNECTION WITH ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. FURTHER, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY EXPRESS OR IMPLIED WARRANTIES OF ANY KIND, AND MSCI, ANY OF ITS AFFILIATES AND ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX HEREBY EXPRESSLY DISCLAIM ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, WITH RESPECT TO ANY MSCI INDEX AND ANY DATA INCLUDED THEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL MSCI, ANY OF ITS AFFILIATES OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, SPECIAL, PUNITIVE, CONSEQUENTIAL OR ANY OTHER DAMAGES (INCLUDING LOST PROFITS) EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

No purchaser, seller or holder of this security, or any other person or entity, should use or refer to any MSCI trade name, trademark or service mark to sponsor, endorse, market or promote this product without first contacting MSCI to determine whether MSCI's permission is required. Under no circumstances may any person or entity claim any affiliation with MSCI without the prior written permission of MSCI.

UBS Euro Equity Defensive Covered Call SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS Euro Equity Defensive Covered Call SF UCITS ETF** (the "Fund"), a sub-fund of UBS (Irl) Fund Solutions plc (the "Company") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "Central Bank").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID, and may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

An investment in the sub-fund should not constitute a substantial proportion of an investment portfolio and may not be appropriate for all investors.

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already

complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being the Euro Equity Defensive Covered Call index) less fees and expenses on each Dealing Day.

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index/ Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equityrelated securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company.

Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange the price performance and income received in relation to such transferable securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus.

All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

a) The Fund will take long positions only and the Fund will invest 100% of its net assets in long positions.

b) The Fund is passively managed (i.e. its objective is to track the Reference Index).

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

CVO/657072-000001/26467820v31

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index**" and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.02% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. This position level disclosure will include the exposure received through the Swaps. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty**") pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index and a cash payment which will equate to the Flat Fee (as described below).

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow like Euribor, etc):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically unfunded swaps do not require any upfront payment of cash.
- 3. In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating

interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 4 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or

which are deemed by the Fund to have an implied credit rating at least equal to such level (a **"Counterparty"**). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German

Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The total expected level of leverage of the Fund as a result of its investments shall not exceed 100% of its Net Asset Value, as calculated using the commitment approach.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund.

The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the EUR acc, hUSD acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the EUR UKdis, hUSD UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing

the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended	
Base Currency	Euro ("EUR")	
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin, (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day); and (B) banks are open for business in London and the Trans-European Automated Real-time Gross Settlement Express Transfer (TARGET) system is open; and (C) the Reference Index is calculated and published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.	
Dealing DayAny Business Day. However, some Business Days will not be Dealing where, for example, markets on which the Fund Assets are listed or to or markets relevant to the Reference Index are closed provided there least one Dealing Day per fortnight, subject always to the Direct discretion to temporarily suspend the determination of the Net Asset and the sale, conversion and/or redemption of Shares in the Fund accordance with the provisions of the Prospectus and the Articles.The Investment Manager produces dealing calendars which det advance the Dealing Days for each Fund. The dealing calendar market operator, regulator or exchange (as applid declares a relevant market closed for trading and/or settlement closure may be made with little or no notice to the Investment Manager.		
Dealing Deadline	2.00 pm (Dublin time) on the relevant Dealing Day.	
Minimum Fund Size	EUR 100 million.	
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.	
Settlement Date	Settlement Date Up to three Business Days after the relevant Dealing Day.	
Minimum Holding	1 Share.	
Minimum Subscription Amount	100,000 EUR or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.	

Minimum	100,000 EUR or its equivalent in another currency or such amount as may
Redemption	be determined by the Directors. Investors will be notified of any change to
Amount	the Minimum Redemption Amount/basket.
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.

Initial Offer Period

The EUR acc Share Class has launched and Shares in this Share Class are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the *hUSD acc Share Class, hCHF acc Share Class, hGBP acc Share Class, hHKD acc Share Class, hSGD acc Share Class, hJPY acc Share Class, hSEK acc Share Class, hPLN acc Share Class, hNOK acc Share Class, EUR UKdis Share Class, hUSD UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class and hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Share Classes ‡			
"acc" Share Classes	"dis" Share Classes	Initial Issue Price	
*EUR acc	**EUR UKdis	EUR 100	
*hUSD acc	**hUSD UKdis	USD 100	
*hCHF acc	**hCHF UKdis	CHF 100	
*hGBP acc	**hGBP UKdis	GBP 100	
*hHKD acc	**hHKD UKdis	HKD 1,000	
*hSGD acc	**hSGD UKdis	SGD 100	
*hJPY acc	**hJPY UKdis	JPY 1,000	
*hSEK acc	**hSEK UKdis	SEK 1,000	
*hPLN acc	**hPLN UKdis	PLN 1,000	
*hNOK acc	**hNOK UKdis	NOK 1,000	

Description of the Shares

* Together the "acc" Share Class

** Together the "**dis**" Share Class

‡ Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	" acc"	" dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	" асс"	" dis"
Flat Fee	0.26% per annum	0.26% per annum

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are estimated to be in the region of $\in 60,000.00$ but will seek to recover such establishment and approval costs if there is money available to it out of the Flat Fee. Any such expenses recovered from the Flat Fee may be charged over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will (in

order to deliver a return which includes the Flat Fee per the agreed terms of the Swap) take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Euro Equity Defensive Covered Call index (the "Reference Index") is governed by a published, rules-based methodology and is designed to reflect the performance of a notional call option writing strategy, which consists of selling listed index call options ("Calls") on the Euro Stoxx 50 Index (the "**Underlying Index**") and a long investment in the Underlying Index. The Calls are traded on Eurex. The Underlying Index is a blue chip index that measures the performance of 50 European company stocks with the objective of reflecting the market sector leaders in the Eurozone.

A call writing strategy typically provides exposure to the sale of call options on a reference asset, which, in the case of the Reference Index, is the Underlying Index.

A call option is a financial derivative contract between two parties used for the purposes of entering into a transaction whereby the seller of the option agrees, in exchange for an upfront payment (the "premium"), to sell an asset (the "reference asset") to the purchaser of the option at an agreed price (the "strike price") on a specific date in the future (the "exercise date").

The purpose of the strategy of the Reference Index is to sell call options in order to collect the premium from their sale, while seeking to reduce the potential loss that may be incurred if the call options are "exercised". This is achieved by seeking to sell call options with strike prices that are above the actual price of the reference asset at the exercise date, so that there is no value to the purchaser of the call option in exercising the option.

Calls are sold under the Reference Index methodology with a target strike of 104% of the value of the option on the day the Calls are sold and a target maturity of approximately 1 month.

By selling a Call, the seller receives a premium from the buyer, which increases the Reference Index's return if the option is not exercised and expires worthless. If, however, the value of the Underlying Index at option expiry increases above the Call's strike price, the option finishes "in-the-money" and the seller pays the buyer the difference between the strike price and the value of the Underlying Index. Any such loss will be met by reducing the size of the long investment in the Underlying Index and therefore will be reflected in the level of the Reference Index.

The seller (writer) of a call option assumes the risk of a rise in the market price of the reference asset above the strike price of the call option in exchange for the payment of a premium by the purchaser of the call option. Where the seller of the call option holds a long position in the Underlying Index, the strategy is said to be 'covered' and any losses suffered on the call options are expected to be approximately off-set by gains on the long investment in the Underlying Index.

The Reference Index rebalances on a weekly basis by including a new component which reflects the sale of a Call. Each call option position included in the Reference Index consists of one listed contract with a notional exposure of 25% of the Index and which will be due to expire approximately one month after inclusion. The option will be retained in the Reference Index until its relevant expiry, when it may or may not be exercised. Once expired, each Call will be replaced by a new Call on the same terms.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index is calculated in Euro.

Index Provider and website

The Reference Index is governed by Solactive AG ("Index Administrator"). The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index.

More details on the Reference Index, including its underlying components can be found at <u>https://www.solactive.com/indices/?se=1&index=DE000SL0AS69</u>. Index Methodology is available separately on request.

The Reference Index performance will include an index replication cost of 0.27% per annum. In addition to the index replication cost the Reference Index performance will also reflect transaction costs within the Reference Index.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) Covered Call Option Risk. Options like Calls may be subject to volatile swings in price influenced by changes in the value of the underlying instrument. The seller (writer) of a call option assumes the risk of a rise in the market price of the underlying above the strike price of the call option in exchange for the payment of a premium by the call option buyer. Where the seller of a call option holds either a long position in the underlying asset, or holds similar exposure to the underlying asset, the strategy is stated to be 'covered' and any losses suffered on the call options are expected to be approximately offset by gains on the long investment in the Underlying Index. The Underlying Index is made up of the fifty largest and most liquid stocks listed in the Eurozone. Any investor would therefore not benefit of any positive performance of the long position in the Underlying Index above the strike price of a call option.
- (b) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (c) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (d) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (e) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the

Reference Index which, in turn, is exposed to general market movements (negative as well as positive).

- (f) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (g) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (h) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (i) Classes of Shares denominated in currencies other than EUR are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (j) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles. For example, the calculation of the Reference Index relies on the assumption that it is possible to establish and close out positions in exchange-traded derivatives. There is no assurance that a liquid market on an exchange will continue for any particular exchange-traded derivative or at any particular time.
- (k) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (I) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (m) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including

adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.

- (n) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (o) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (p) Implied Volatility Risk. Through selling Calls, the Reference Index value increases due to the premium it receives. However, the Reference Index is also exposed to the value of each Call it has sold, until each Call is either exercised or expires. The value of each Call is partly based on the volatility assumption used by market participants to price such options (i.e. implied volatility). Higher implied volatility typically correlates with higher prices for options. Consequently, increases in the implied volatility of each Call will cause the value of such options to increase which will result in a corresponding decrease in the value of the Reference Index and thus a decrease the Fund's NAV. The Fund is therefore exposed to implied volatility risk before the Calls expire. The implied volatility of each Call sold may increase due to general market and economic conditions.
- (q) The return of the Fund depends, amongst other things, on the performance of the Reference Index. The Fund gains exposure to the Reference Index by investing in one or more Swaps. An investment in one or more Swaps does not give the Fund a direct interest in the Reference Index and the portfolio of Calls or any underlying asset linked to such portfolio of calls or any long investment in the Underlying Index.
- (r) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the index administrator. The Index is subject to the terms set out in the index manual, as such index manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to the Index Owner.

The financial instrument is not sponsored, promoted, sold or supported in any other manner by Solactive AG nor does Solactive AG offer any express or implicit guarantee or assurance either with regard to the results of using the Index and/or Index trade mark or the Index level at any time or in any other respect. The Index is calculated and published by Solactive AG. Solactive AG uses its best efforts to ensure that the Index is calculated correctly. Irrespective of its obligations towards the issuer of the financial instrument, Solactive AG has no obligation to point out errors in the Index to third parties including but not limited to investors and/or financial instrument constitutes a recommendation by Solactive AG to invest capital in said financial instrument nor does it in any way represent an assurance or opinion of Solactive AG with regard to any investment in this financial instrument. Solactive AG is registered as a benchmark administrator under the Regulation (EU) 2016/2011 (BMR). Solactive appears on the ESMA register of administrators.

EUROSTOXX DISCLAIMER

STOXX Ltd., Deutsche Börse Group and their licensors, research partners or data providers have no relationship to UBS AG, other than the licensing of the EURO STOXX 50® Index and the related trademarks for use in connection with the Fund.

STOXX, Deutsche Börse Group and their licensors, research partners or data providers do not:

sponsor, endorse, sell or promote the Fund.

recommend that any person invest in the Fund or any other securities.

have any responsibility or liability for or make any decisions about the timing, amount or pricing of the Fund.

have any responsibility or liability for the administration, management or marketing of the Fund.

consider the needs of the Fund or the owners of the Fund in determining, composing or calculating the relevant index or have any obligation to do so.

STOXX, Deutsche Börse Group and their licensors, research partners or data providers give no warranty, and exclude any liability (whether in negligence or otherwise), in connection with the Fund or their performance.

STOXX does not assume any contractual relationship with the purchasers of the Fund or any other third parties.

Specifically,

STOXX, Deutsche Börse Group and their licensors, research partners or data providers do not give any warranty, express or implied, and exclude any liability about:

The results to be obtained by the Fund, the owner of the Fund or any other person in connection with the use of the relevant index and the data included in the relevant index;

The accuracy, timeliness, and completeness of the relevant index and its data;

The merchantability and the fitness for a particular purpose or use of the relevant index and its data;

The performance of the Fund generally.

STOXX, Deutsche Börse Group and their licensors, research partners or data providers give no warranty and exclude any liability, for any errors, omissions or interruptions in the relevant index or its data;

Under no circumstances will STOXX, Deutsche Börse Group or their licensors, research partners or data providers be liable (whether in negligence or otherwise) for any lost profits or indirect, punitive, special or consequential damages or losses, arising as a result of such errors, omissions or interruptions in the relevant index or its data or generally in relation to the Fund, even in circumstances where STOXX, Deutsche Börse Group or their licensors, research partners or data providers are aware that such loss or damage may occur.

The licensing agreement between UBS AG and STOXX is solely for their benefit and not for the benefit of the owners of the Fund or any other third parties.

UBS Euro Equity Defensive Put Write SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS Euro Equity Defensive Put Write SF UCITS ETF** (the "**Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the "**Company**") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "**Central Bank**").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID, and may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a medium to high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any

registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being the Euro Equity Defensive Put Write index) less fees and expenses on each Dealing Day.

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index/ Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equityrelated securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company.
- (iii) Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange the price performance and income received in relation to such transferable securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus.

All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

- a) The Fund will take long positions only and the Fund will invest 100% of its net assets in long positions.
- b) The Fund is passively managed (i.e. its objective is to track the Reference Index).

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

CVO/657072-000001/26467820v31

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.02% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. This position level disclosure will include the exposure received through the Swaps. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty**") pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index and a cash payment which will equate to the Flat Fee (as described below).

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow like Euribor, etc.):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically unfunded swaps do not require any upfront payment of cash.
- 3. In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating

interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the "**ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 4 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or

which are deemed by the Fund to have an implied credit rating at least equal to such level (a **"Counterparty"**). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German

Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The total expected level of leverage of the Fund as a result of its investments shall not exceed 100% of its Net Asset Value, as calculated using the commitment approach.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund.

The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the EUR acc, hUSD acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the EUR UKdis, hUSD UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing

the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended	
Base Currency	Euro (" EUR ")	
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin, (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day); and (B) banks are open for business in London and the Trans-European Automated Real-time Gross Settlement Express Transfer (TARGET) system is open; and (C) the Reference Index is calculated and published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.	
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager). The dealing calendar for the Fund is available from the Manager.	
Dealing Deadline	2.00 pm (Dublin time) on the relevant Dealing Day.	
Minimum Fund Size	EUR 100 million.	
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.	
Settlement Date	Up to three Business Days after the relevant Dealing Day.	
Minimum Holding	1 Share.	
Minimum Subscription Amount	100,000 EUR or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.	

Minimum	100,000 EUR or its equivalent in another currency or such amount as may
Redemption	be determined by the Directors. Investors will be notified of any change to
Amount	the Minimum Redemption Amount/basket.
Website	<u>www.ubs.com/ETF</u> - information on Fund Assets and the Fund's iNAV is set out on the Website.

Initial Offer Period

The EUR acc Share Class has launched and Shares in this Share Class are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the *hUSD acc Share Class, hCHF acc Share Class, hGBP acc Share Class, hHKD acc Share Class, hSGD acc Share Class, hJPY acc Share Class, hSEK acc Share Class, hPLN acc Share Class, hNOK acc Share Class, EUR UKdis Share Class, hUSD UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class and hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Share Classes ‡		
"acc" Share Classes	"dis" Share Classes	Initial Issue Price
*EUR acc	**EUR UKdis	EUR 100
*hUSD acc	**hUSD UKdis	USD 100
*hCHF acc	**hCHF UKdis	CHF 100
*hGBP acc	**hGBP UKdis	GBP 100
*hHKD acc	**hHKD UKdis	HKD 1,000
*hSGD acc	**hSGD UKdis	SGD 100
*hJPY acc	**hJPY UKdis	JPY 1,000
*hSEK acc	**hSEK UKdis	SEK 1,000
*hPLN acc	**hPLN UKdis	PLN 1,000
*hNOK acc	**hNOK UKdis	NOK 1,000

Description of the Shares

* Together the "acc" Share Class

** Together the "**dis**" Share Class

‡ Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat Fee	0.26% per annum	0.26% per annum

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are estimated to be in the region of $\in 60,000.00$ but will seek to recover such establishment and approval costs if there is money available to it out of the Flat Fee. Any such expenses recovered from the Flat Fee may be charged over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will (in

order to deliver a return which includes the Flat Fee per the agreed terms of the Swap) take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Euro Equity Defensive Put Write index (the "Reference Index") is governed by a published, rules-based methodology and is designed to reflect the performance of a notional put option writing strategy, which consists of selling listed index put options ("Puts") on the Euro Stoxx 50 Index (the "**Underlying Index**") and a notional cash exposure linked to a three-month German Bubil rate. The Puts are traded on Eurex. The Underlying Index is a blue chip index that measures the performance of 50 European company stocks with the objective of reflecting the market sector leaders in the Eurozone.

A put writing strategy typically provides exposure to the sale of put options on a reference asset, which, in the case of the Reference Index, is the Underlying Index.

A put option is a financial derivative contract between two parties used for the purposes of entering into a transaction whereby the seller of the option agrees, in exchange for an upfront payment (the "premium"), to buy an asset (the "reference asset") from the purchaser of the option at an agreed price (the "strike price") on a specific date in the future (the "exercise date").

The purpose of the strategy of the Reference Index is to sell put options in order to collect the premium from their sale, while seeking to reduce the potential loss that may be incurred if the put options are "exercised". This is achieved by seeking to sell put options with strike prices that are below the actual price of the reference asset at the exercise date, so that there is no value to the purchaser of the put option in exercising the option.

Puts are sold under the Reference Index methodology with a target strike of 97% of the value of the option on the day the Puts are sold and a target maturity of approximately 1 month.

By selling a Put, the seller receives a premium from the buyer, which increases the Reference Index's return if the option is not exercised and expires worthless. If, however, the value of the Underlying Index at option expiry falls below the Put's strike price, the option finishes "in-the-money" and the seller pays the buyer the difference between the strike price and the value of the Underlying Index. Any such loss will be paid from the notional cash exposure.

If the value of the Underlying Index at option expiry increases beyond the Puts' strike price, the Reference Index returns would not be expected to increase beyond the premium received, plus any interest earned with respect to the notional cash exposure.

The premia received upon selling each Put will be reflected in the balance of the notional cash exposure and therefore the level of the Reference Index. Upon expiry of each Put, any settlement loss will be reflected in the balance of the notional cash exposure and therefore the level of the Reference Index.

The Reference Index rebalances on a weekly basis by including a new component which reflects the sale of a Put. Each put option position included in the Reference Index consists of one listed contract with a notional exposure of 25% of the Index and which will be due to expire approximately one month after inclusion. The option will be retained in the Reference Index until its relevant expiry, when it may or may not be exercised. Once expired, each Put will be replaced by a new Put on the same terms.

Through the Reference Index, the Fund will be exposed to the risk that the level of the Underlying Index falls. In the event that the Underlying Index performs negatively and its level drops below the

strike price (in excess of any premium received), the put options which have been sold will be exercised by the purchaser and this will generate a loss in the Reference Index.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index is calculated in Euro.

Index Provider and website

The Reference Index is governed by Solactive AG ("Index Administrator"). The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index.

More details on the Reference Index, including its underlying components can be found at <u>https://www.solactive.com/indices/?se=1&index=DE000SL0AS77</u>. Index Methodology is available separately on request.

The Reference Index performance will include an index replication cost of 0.27% per annum. In addition to the index replication cost, the Reference Index performance will also reflect transaction costs within the Reference Index.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) Put Option Selling Risk. Options like Puts may be subject to volatile swings in price influenced by changes in the value of the underlying instrument. Although the value of the Reference Index increases due to the premiums it collects on the back of the options being written, the reduction to the value of the Reference Index if options expire in-the-money may outweigh the gains to the value of the Reference Index from the receipt of such option premiums. The Underlying Index is made up of the 50 largest and most liquid stocks listed on in the Eurozone. The potential return of a sold Put position is limited to the amount of option premium it receives; however, the position can potentially lose up to the entire strike price of each option it sells.
- (b) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (c) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (d) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (e) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).

- (f) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (g) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (h) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (i) Classes of Shares denominated in currencies other than EUR are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (j) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles. For example, the calculation of the Reference Index relies on the assumption that it is possible to establish and close out positions in exchange-traded derivatives. There is no assurance that a liquid market on an exchange will continue for

any particular exchange-traded derivative or at any particular time.

- (k) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (I) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (m) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.

- (n) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (o) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (p) Implied Volatility Risk. Through selling Puts, the Reference Index value increases due to the premium it receives. However, the Reference Index is also exposed to the value of each Put it has sold, until each Put is either exercised or expires. The value of each Put is partly based on the volatility assumption used by market participants to price such options (i.e. implied volatility). Higher implied volatility typically correlates with higher prices for options. Consequently, increases in the implied volatility of each Put will cause the value of such options to increase which will result in a corresponding decrease in the value of the Reference Index and thus a decrease in the Fund's NAV. The Fund is therefore exposed to implied volatility risk before the Puts expire. The implied volatility of each Put sold may increase due to general market and economic conditions.
- (q) The return of the Fund depends, amongst other things, on the performance of the Reference Index. The Fund gains exposure to the Reference Index by investing in one or more Swaps. An investment in one or more Swaps does not give the Fund a direct interest in the Reference Index and the portfolio of Puts or any underlying asset linked to such portfolio of puts.
- (r) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the index administrator. The Index is subject to the terms set out in the index manual, as such index manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to the Index Owner.

The financial instrument is not sponsored, promoted, sold or supported in any other manner by Solactive AG nor does Solactive AG offer any express or implicit guarantee or assurance either with regard to the results of using the Index and/or Index trade mark or the Index level at any time or in any other respect. The Index is calculated and published by Solactive AG. Solactive AG uses its best efforts to ensure that the Index is calculated correctly. Irrespective of its obligations towards the issuer of the financial instrument, Solactive AG has no obligation to point out errors in the Index to third parties including but not limited to investors and/or financial intermediaries of the financial instrument. Neither publication of the Index by Solactive AG nor the licensing of the Index or Index trade mark for the purpose of use in connection with the financial instrument constitutes a recommendation by Solactive AG to invest capital in said financial instrument nor does it in any way represent an assurance or opinion of Solactive AG with regard to any investment in this financial instrument. Solactive AG is registered as a benchmark administrator under the Regulation (EU) 2016/2011 (BMR). Solactive appears on the ESMA register of administrators.

EUROSTOXX DISCLAIMER

STOXX Ltd., Deutsche Börse Group and their licensors, research partners or data providers have no relationship to UBS AG, other than the licensing of the EURO STOXX 50® Index and the related trademarks for use in connection with the Fund.

STOXX, Deutsche Börse Group and their licensors, research partners or data providers do not:

sponsor, endorse, sell or promote the Fund.

recommend that any person invest in the Fund or any other securities.

have any responsibility or liability for or make any decisions about the timing, amount or pricing of the Fund.

have any responsibility or liability for the administration, management or marketing of the Fund.

consider the needs of the Fund or the owners of the Fund in determining, composing or calculating the relevant index or have any obligation to do so.

STOXX, Deutsche Börse Group and their licensors, research partners or data providers give no warranty, and exclude any liability (whether in negligence or otherwise), in connection with the Fund or their performance.

STOXX does not assume any contractual relationship with the purchasers of the Fund or any other third parties.

Specifically,

STOXX, Deutsche Börse Group and their licensors, research partners or data providers do not give any warranty, express or implied, and exclude any liability about:

The results to be obtained by the Fund, the owner of the Fund or any other person in connection with the use of the relevant index and the data included in the relevant index;

The accuracy, timeliness, and completeness of the relevant index and its data;

The merchantability and the fitness for a particular purpose or use of the relevant index and its data;

The performance of the Fund generally.

STOXX, Deutsche Börse Group and their licensors, research partners or data providers give no warranty and exclude any liability, for any errors, omissions or interruptions in the relevant index or its data;

Under no circumstances will STOXX, Deutsche Börse Group or their licensors, research partners or data providers be liable (whether in negligence or otherwise) for any lost profits or indirect, punitive, special or consequential damages or losses, arising as a result of such errors, omissions or interruptions in the relevant index or its data or generally in relation to the Fund, even in circumstances where STOXX, Deutsche Börse Group or their licensors, research partners or data providers are aware that such loss or damage may occur.

The licensing agreement between UBS AG and STOXX is solely for their benefit and not for the benefit of the owners of the Fund or any other third parties.

UBS US Equity Defensive Covered Call SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS US Equity Defensive Covered Call SF UCITS ETF** (the "**Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the "**Company**") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "**Central Bank**").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KID, and may not be distributed unless accompanied by the Prospectus and the KID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Shareholders should note that distributions may be payable out of the capital of the SOFR +4% Share Classes. The payment of distributions out of capital may result in the erosion of capital notwithstanding the performance of the Fund. As a result, distributions may be achieved by foregoing the potential for future capital growth and this cycle may continue until all capital is depleted.

An investment in the sub-fund should not constitute a substantial proportion of an investment portfolio and may not be appropriate for all investors.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already

complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being the US Equity Defensive Covered Call index) less fees and expenses on each Dealing Day.

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index/ Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equityrelated securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company.
- (iii) Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange the price performance and income received in relation to such transferable securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus.

All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund will take long positions only and the Fund will invest 100% of its net assets in long positions.

The Fund is passively managed (i.e. its objective is to track the Reference Index).

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index**" and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.02% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. This position level disclosure will include the exposure received through the Swaps. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty**") pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index and a cash payment which will equate to the Flat Fee (as described below).

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow like Euribor, etc):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically unfunded swaps do not require any upfront payment of cash.
- 3. In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating

interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the "**ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 4 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or

which are deemed by the Fund to have an implied credit rating at least equal to such level (a **"Counterparty"**). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German

Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The total expected level of leverage of the Fund as a result of its investments shall not exceed 100% of its Net Asset Value, as calculated using the commitment approach.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund.

The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD dis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

The Directors may declare defined dividends attributable to the SOFR +4% Share Classes (as defined below).

SOFR +4% Share Classes may distribute an annual percentage of the Net Asset Value of which payment will be scheduled on every 30 June and 31 December.

SOFR +4% Share Classes will seek to pay out a dividend regardless of the performance of the Shares. As a result, the Net Asset Value of such Shares may fluctuate more than other Share Classes, the potential for future appreciation of such Net Asset Value of such Shares may be eroded as the dividend may be composed of capital attributable to the Shares. In this regard, distributions made during the life of the SOFR +4% Share Classes may be understood as a type of capital reimbursement, depending on performance of the Fund.

The Fund is not obliged to communicate an expected distribution rate per Share to Shareholders of SOFR +4% Share Classes and prospective investors, and investors should note that any such rate may vary with market conditions. There can be no guarantee that any rate will be achieved, and in the event that there is insufficient distributable income or gains in the Fund to meet a specific level, investors in the Fund may receive no distribution or a lower level distribution.

The rationale for providing for payment of distributions out of capital is to allow for the ability to consistently provide distributions to investors seeking an income-oriented investment. Distributions out of capital may have different tax implications to distributions of income and investors should seek advice in this regard.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended	
Base Currency	US Dollar (" USD ")	
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin, (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day); and (B) banks are open for business in London and New York; and (C) the Reference Index is calculated and published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.	
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager). The dealing calendar for the Fund is available from the Manager.	
Dealing Deadline	2:00 p.m. (Dublin time) on the relevant Dealing Day.	
Minimum Fund Size	USD 100 million.	
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.	
Settlement Date	Up to three Business Days after the relevant Dealing Day.	
Minimum Holding	1 Share.	
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.	
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.	

Website www.ubs.com/ETF - information on Fund Assets and the Fund's set out on the Website.	NAV is
---	--------

Initial Offer Period

The USD acc Share Class has launched and Shares in this Share Class is available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the *hEUR acc Share Class, hCHF acc Share Class, hGBP acc Share Class, hHKD acc Share Class, hSGD acc Share Class, hJPY acc Share Class, hSEK acc Share Class, hPLN acc Share Class, hNOK acc Share Class, USD dis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class and hNOK UKdis Share Class was open during the period from 9 am on 20 October 2022 (Irish time) and has closed at 5 pm on 19 April 2023 (Irish time).

The initial offer period of the USD) dis SOFR+4% will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Share Classes ‡		
" acc" Share Classes	" dis" Share Classes	Initial Issue Price
*USD acc	**USD UKdis	USD 100
	**USD dis SOFR+4%	USD 20
*hEUR acc	**hEUR UKdis	EUR 100
*hCHF acc	**hCHF UKdis	CHF 100
*hGBP acc	**hGBP UKdis	GBP 100
*hHKD acc	**hHKD UKdis	HKD 1,000
*hSGD acc	**hSGD UKdis	SGD 100
*hJPY acc	**hJPY UKdis	JPY 1,000
*hSEK acc	**hSEK UKdis	SEK 1,000
*hPLN acc	**hPLN UKdis	PLN 1,000
*hNOK acc	**hNOK UKdis	NOK 1,000

Description of the Shares

* Together the " **acc**" Share Class

** Together the " **dis**" Share Class

‡Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

dis Share classes with "SOFR +4%" in their name shall distribute a defined dividend twice a year. The defined distribution will amount to the Secured Overnight Financing Rate ("SOFR") +4% on a per annum basis.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	" acc"	" dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	" асс"	" dis"
Flat Fee	0.20% per annum	0.20% per annum

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are estimated to be in the region of $\in 60,000.00$ but will seek to recover such establishment and approval costs if there is money available to it out of the Flat Fee. Any such expenses recovered from the Flat Fee may be charged over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will (in

order to deliver a return which includes the Flat Fee per the agreed terms of the Swap) take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The US Equity Defensive Covered Call index (the "Reference Index") is governed by a published, rules-based methodology and is designed to reflect the performance of a notional call option writing strategy, which consists of selling listed index call options ("SPX Calls") on the S&P 500 Index (the "**Underlying Index**") and a long investment in the Underlying Index. The SPX Calls are traded on Chicago Board of Options Exchange ("CBOE").

A call writing strategy typically provides exposure to the sale of call options on a reference asset, which, in the case of the Reference Index, is the Underlying Index.

A call option is a financial derivative contract between two parties used for the purposes of entering into a transaction whereby the seller of the option agrees, in exchange for an upfront payment (the "premium"), to sell an asset (the "reference asset") to the purchaser of the option at an agreed price (the "strike price") on a specific date in the future (the "exercise date").

The purpose of the strategy of the Reference Index is to sell call options in order to collect the premium from their sale, while seeking to reduce the potential loss that may be incurred if the call options are "exercised". This is achieved by seeking to sell call options with strike prices that are above the actual price of the reference asset at the exercise date, so that there is no value to the purchaser of the call option in exercising the option.

SPX Calls are sold under the Reference Index methodology with a target strike of 104% of the value of the option on the day the SPX Calls are sold and a target maturity of approximately 1 month.

By selling a SPX Call, the seller receives a premium from the buyer, which increases the Reference Index's return if the option is not exercised and expires worthless. If, however, the value of the Underlying Index at option expiry increases above the SPX Call's strike price, the option finishes "in-the-money" and the seller pays the buyer the difference between the strike price and the value of the Underlying Index. Any such loss will be met by reducing the size of the long investment in the Underlying Index and therefore will be reflected in the level of the Reference Index.

The seller (writer) of a call option assumes the risk of a rise in the market price of the reference asset above the strike price of the call option in exchange for the payment of a premium by the purchaser of the call option. Where the seller of the call option holds a long position in the Underlying Index, the strategy is said to be 'covered' and any losses suffered on the call options are expected to be approximately off-set by gains on the long investment in the Underlying Index.

The Reference Index rebalances on a weekly basis by including a new component which reflects the sale of a SPX Call. Each call option position included in the Reference Index consists of one listed contract with a notional exposure of 25% of the Index and which will be due to expire approximately one month after inclusion. The option will be retained in the Reference Index until its relevant expiry, when it may or may not be exercised. Once expired, each SPX Call will be replaced by a new SPX Call on the same terms.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is governed by Solactive AG ("Index Administrator"). The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index.

More details on the Reference Index, including its underlying components can be found at <u>https://www.solactive.com/indices/?se=1&index=DE000SL0AS44</u>. Index Methodology is available separately on request.

The Reference Index performance will include an index replication cost of 0.10% per annum. In addition to the index replication cost the Reference Index performance will also reflect transaction costs within the Reference Index.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) Covered Call Option Risk. Options like SPX Calls may be subject to volatile swings in price influenced by changes in the value of the underlying instrument. The seller (writer) of a call option assumes the risk of a rise in the market price of the underlying above the strike price of the call option in exchange for the payment of a premium by the call option buyer. Where the seller of a call option holds either a long position in the underlying asset, or holds similar exposure to the underlying asset, the strategy is stated to be 'covered' and any losses suffered on the call options are expected to be approximately offset by gains on the long investment in the Underlying Index. The Underlying Index represents the top 500 largest and most liquid stocks listed in the United States. Any investor would therefore not benefit of any positive performance of the long position in the Underlying Index above the strike price of a call option.
- (b) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (c) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (d) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (e) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the

Reference Index which, in turn, is exposed to general market movements (negative as well as positive).

- (f) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (g) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (h) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (i) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (j) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles. For example, the calculation of the Reference Index relies on the assumption that it is possible to establish and close out positions in exchange-traded derivatives. There is no assurance that a liquid market on an exchange will continue for any particular exchange-traded derivative or at any particular time.
- (k) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (I) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (m) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including

adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.

- (n) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (o) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (p) Implied Volatility Risk. Through selling SPX Calls, the Reference Index value increases due to the premium it receives. However, the Reference Index is also exposed to the value of each SPX Call it has sold, until each SPX Call is either exercised or expires. The value of each SPX Call is partly based on the volatility assumption used by market participants to price such options (i.e. implied volatility). Higher implied volatility typically correlates with higher prices for options. Consequently, increases in the implied volatility of each SPX Call will cause the value of such options to increase which will result in a corresponding decrease in the value of the Reference Index and thus a decrease the Fund's NAV. The Fund is therefore exposed to implied volatility risk before the Calls expire. The implied volatility of each SPX Call sold may increase due to general market and economic conditions.
- (q) The return of the Fund depends, amongst other things, on the performance of the Reference Index. The Fund gains exposure to the Reference Index by investing in one or more Swaps. An investment in one or more Swaps does not give the Fund a direct interest in the Reference Index and the portfolio of SPX Calls or any underlying asset linked to such portfolio of calls or any long investment in the Underlying Index.
- (r) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

CVO/657072-000001/26467820v31

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the index administrator. The Index is subject to the terms set out in the index manual, as such index manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to the Index Owner.

The financial instrument is not sponsored, promoted, sold or supported in any other manner by Solactive AG nor does Solactive AG offer any express or implicit guarantee or assurance either with regard to the results of using the Index and/or Index trade mark or the Index level at any time or in any other respect. The Index is calculated and published by Solactive AG. Solactive AG uses its best efforts to ensure that the Index is calculated correctly. Irrespective of its obligations towards the issuer of the financial instrument, Solactive AG has no obligation to point out errors in the Index to third parties including but not limited to investors and/or financial intermediaries of the financial instrument. Neither publication of the Index by Solactive AG nor the licensing of the Index or Index trade mark for the purpose of use in connection with the financial instrument constitutes a recommendation by Solactive AG to invest capital in said financial instrument nor does it in any way represent an assurance or opinion of Solactive AG with regard to any investment in this financial instrument. Solactive AG is registered as a benchmark administrator under the Regulation (EU) 2016/2011 (BMR). Solactive appears on the ESMA register of administrators.

US Equity Defensive Covered Call Index (the "Custom Index") is the property of UBS AG London Branch, which has contracted with S&P Opco, LLC (a subsidiary of S&P Dow Jones Indices LLC) ("SPDJI") to license the use of S&P 500 Index in connection with the Custom Index. The S&P 500 Index is the property of SPDJI, its affiliates (collectively with SPDJI, "S&P Dow Jones Indices") and/or their third-party licensors. S&P Dow Jones Indices and its third-party licensors shall have no liability for any errors or omissions in the S&P 500 Index and the Custom Index is not sponsored, endorsed, sold or promoted by S&P Dow Jones Indices or its third-party licensors.

The US Equity Defensive Covered Call SF UCITS ETF based on the Custom Index are not sponsored, endorsed, sold or promoted by S&P Dow Jones Indices or its third-party licensors. Neither S&P Dow Jones Indices nor its third party licensors make any representation or warranty, express or implied, to the owners of the US Equity Defensive Covered Call SF UCITS ETF or any member of the public regarding the advisability of investing in securities generally or in the US Equity Defensive Covered Call SF UCITS ETF particularly or the ability of the Index or Custom Index to track general market performance. S&P Dow Jones Indices' only relationship to UBS AG London Branch with respect to the Custom Index is the licensing of the S&P 500 Index certain rights with respect to the CBOE Option Settlement Prices with respect thereto and certain trademarks, service marks and trade names of S&P Dow Jones Indices or its third-party licensors. S&P Dow Jones Indices and its third party licensors are not responsible for and have not participated in the determination of the prices and amount of the US Equity Defensive Covered Call SF UCITS ETF or the timing of the issuance or sale of the US Equity Defensive Covered Call SF UCITS ETF or in the determination or calculation of the equation by which the US Equity Defensive Covered Call SF UCITS ETF may convert into cash or other redemption mechanics. S&P Dow Jones Indices and its third-party licensors have no obligation or liability in connection with the administration, marketing or trading of the US Equity Defensive Covered Call SF UCITS ETF. S&P Dow Jones Indices LLC is not an investment advisor. Inclusion of a security within the Index is not a recommendation by S&P Dow Jones Indices or its third-party licensors to buy, sell, or hold such security, nor is it investment advice.

NEITHER S&P DOW JONES INDICES NOR ITS THIRD-PARTY LICENSORS GUARANTEES THE ADEQUACY, ACCURACY, TIMELINESS AND/OR THE COMPLETENESS OF THE INDEX OR CUSTOM INDEX, THE CBOE OPTION SETTLEMENT PRICES OR ANY DATA RELATED THERETO OR ANY COMMUNICATION WITH RESPECT THERETO, INCLUDING, ORAL, WRITTEN, OR ELECTRONIC COMMUNICATIONS. S&P DOW JONES INDICES AND ITS THIRD-PARTY LICENSORS SHALL NOT BE SUBJECT TO ANY DAMAGES OR LIABILITY FOR ANY ERRORS, OMISSIONS, OR DELAYS THEREIN. S&P DOW JONES INDICES AND ITS THIRD PARTY LICENSORS MAKE NO EXPRESS OR IMPLIED WARRANTIES, AND EXPRESSLY DISCLAIM ALL WARRANTIES, OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE OR AS TO RESULTS TO BE OBTAINED BY UBS AG LONDON BRANCH, OWNERS OF THE US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF, OR ANY OTHER PERSON OR ENTITY FROM THE USE OF THE INDEX, CBOE OPTION SETTLEMENT PRICES OR CUSTOM INDEX OR WITH RESPECT TO ANY DATA RELATED THERETO. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT WHATSOEVER SHALL S&P DOW JONES INDICES OR ITS THIRD PARTY LICENSORS BE LIABLE FOR ANY INDIRECT, SPECIAL, INCIDENTAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES, INCLUDING BUT NOT LIMITED TO, LOSS OF PROFITS, TRADING LOSSES, LOST TIME, OR GOODWILL, EVEN IF THEY HAVE BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES, WHETHER IN CONTRACT, TORT, STRICT LIABILITY, OR OTHERWISE.

The S&P and S&P 500 are trademarks of Standard & Poor's Financial Services LLC.

UBS US Equity Defensive Put Write SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS US Equity Defensive Put Write SF UCITS ETF** (the "**Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the "**Company**") an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the "**Central Bank**").

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID, and may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a medium to high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "Directors of the Company" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already

complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the return of the Reference Index (being the US Equity Defensive Put Write index) less fees and expenses on each Dealing Day.

Investment Policy

In order to achieve the investment objective of delivering the return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index/ Currency Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equityrelated securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company.
- (iii) Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange the price performance and income received in relation to such transferable securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets"** for the purposes of the Prospectus.

All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund will take long positions only and the Fund will invest 100% of its net assets in long positions.

The Fund is passively managed (i.e. its objective is to track the Reference Index).

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index**" and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.02% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

- b) An explanation for the divergence between the anticipated and realised tracking error for the year;
- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. This position level disclosure will include the exposure received through the Swaps. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty**") pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index and a cash payment which will equate to the Flat Fee (as described below).

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow like Euribor, etc.):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically unfunded swaps do not require any upfront payment of cash.
- 3. In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating

interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 4 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or

which are deemed by the Fund to have an implied credit rating at least equal to such level (a **"Counterparty"**). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German

Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The total expected level of leverage of the Fund as a result of its investments shall not exceed 100% of its Net Asset Value, as calculated using the commitment approach.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund.

The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges"**) for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing

the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended	
Base Currency		
Base Currency	US Dollar ("USD")	
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin, (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day); and (B) banks are open for business in London and New York; and (C) the Reference Index is calculated and published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.	
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager). The dealing calendar for the Fund is available from the Manager.	
Dealing Deadline	2:00 p.m (Dublin time) on the relevant Dealing Day.	
Minimum Fund Size	USD 100 million.	
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.	
Settlement Date	Up to three Business Days after the relevant Dealing Day.	
Minimum Holding	1 Share.	
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.	
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.	

Website www.ubs.com/ETF - information on Fund Assets and the Fund's i set out on the Website.	NAV is
---	--------

Initial Offer Period

The USD acc Share Class has launched and Shares in this Share Class is available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the *hEUR acc Share Class, hCHF acc Share Class, hGBP acc Share Class, hHKD acc Share Class, hSGD acc Share Class, hJPY acc Share Class, hSEK acc Share Class, hPLN acc Share Class, hNOK acc Share Class, USD UKdis Share Class, hEUR UKdis Share Class, hCHF UKdis Share Class, hGBP UKdis Share Class, hHKD UKdis Share Class, hSGD UKdishSGD UKdis Share Class, hJPY UKdis Share Class, hSEK UKdis Share Class, hPLN UKdis Share Class and hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Description of the Shares

Share Classes ‡				
"A-acc" Share Classes	"A-dis" Share Classes	Initial Issue Price		
*USD acc	**USD UKdis	USD 100		
*hEUR acc	**hEUR UKdis	EUR 100		
*hCHF acc	**hCHF UKdis	CHF 100		
*hGBP acc	**hGBP UKdis	GBP 100		
*hHKD acc	**hHKD UKdis	HKD 1,000		
*hSGD acc	**hSGD UKdishSGD UKdis	SGD 100		
*hJPY acc	**hJPY UKdis	JPY 1,000		
*hSEK acc	**hSEK UKdis	SEK 1,000		
*hPLN acc	**hPLN UKdis	PLN 1,000		
*hNOK acc	**hNOK UKdis	NOK 1,000		

* Together the " **acc**" Share Class

** Together the " dis" Share Class

‡ Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	" acc"	" dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	" acc"	" dis"
Flat Fee	0.29% per annum	0.29% per annum

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are estimated to be in the region of \notin 60,000.00 but will seek to recover such establishment and approval costs if there is money available to it out of the Flat Fee. Any such expenses recovered from the Flat Fee may be charged over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will (in

order to deliver a return which includes the Flat Fee per the agreed terms of the Swap) take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL INFORMATION RELATING TO THE FUND

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The US Equity Defensive Put Write index (the "Reference Index") is governed by a published, rulesbased methodology and is designed to reflect the performance of a notional put option writing strategy, which consists of selling listed index put options ("SPX Puts") on the S&P 500 Index (the "**Underlying Index**") and a notional cash exposure linked to a three-month US treasury bill rate. The SPX Puts are traded on Chicago Board of Options Exchange ("CBOE").

A put writing strategy typically provides exposure to the sale of put options on a reference asset, which, in the case of the Reference Index, is the Underlying Index.

A put option is a financial derivative contract between two parties used for the purposes of entering into a transaction whereby the seller of the option agrees, in exchange for an upfront payment (the "premium"), to buy an asset (the "reference asset") from the purchaser of the option at an agreed price (the "strike price") on a specific date in the future (the "exercise date").

The purpose of the strategy of the Reference Index is to sell put options in order to collect the premium from their sale, while seeking to reduce the potential loss that may be incurred if the put options are "exercised". This is achieved by seeking to sell put options with strike prices that are below the actual price of the reference asset at the exercise date, so that there is no value to the purchaser of the put option in exercising the option.

SPX Puts are sold under the Reference Index methodology with a target strike of 97% of the value of the option on the day the SPX Puts are sold and a target maturity of approximately 1 month.

By selling a SPX Put, the seller receives a premium from the buyer, which increases the Reference Index's return if the option is not exercised and expires worthless. If, however, the value of the Underlying Index at option expiry falls below the SPX Put's strike price, the option finishes "in-themoney" and the seller pays the buyer the difference between the strike price and the value of the Underlying Index. Any such loss will be paid from the notional cash exposure.

If the value of the Underlying Index at option expiry increases beyond the SPX Puts' strike price, the Reference Index returns would not be expected to increase beyond the premium received, plus any interest earned with respect to the notional cash exposure.

The premia received upon selling each SPX Put will be reflected in the balance of the notional cash exposure and therefore the level of the Reference Index. Upon expiry of each SPX Put, any settlement loss will be reflected in the balance of the notional cash exposure and therefore the level of the Reference Index.

The Reference Index rebalances on a weekly basis by including a new component which reflects the sale of a SPX Put. Each put option position included in the Reference Index consists of one listed contract with a notional exposure of 25% of the Index and which will be due to expire approximately one month after inclusion. The option will be retained in the Reference Index until its relevant expiry, when it may or may not be exercised. Once expired, each SPX Put will be replaced by a new SPX Put on the same terms.

Through the Reference Index, the Fund will be exposed to the risk that the level of the Underlying Index falls. In the event that the Underlying Index performs negatively and its level drops below the

strike price (in excess of any premium received), the put options which have been sold will be exercised by the purchaser and this will generate a loss in the Reference Index.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is governed by Solactive AG ("Index Administrator"). The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index.

More details on the Reference Index, including its underlying components can be found at https://www.solactive.com/indices/?se=1&index=DE000SL0AS51https://www.ubs.com/global/en/investment-bank/ib/bloomberg-cmci/ubs-strategies.html. Index Methodology is available separately on request.

The Reference Index performance will include an index replication cost of 0.27% per annum. In addition to the index replication cost, the Reference Index performance will also reflect transaction costs within the Reference Index.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) Put Option Selling Risk. Options like SPX Puts may be subject to volatile swings in price influenced by changes in the value of the underlying instrument. Although the value of the Reference Index increases due to the premiums it collects on the back of options being written, the reduction to the value of the Reference Index if options expire in-the-money may outweigh the gains to the value of the Reference Index from the receipt of such option premiums. The Underlying Index represents the top 500 largest and most liquid stocks listed in the United States. The potential return of a short SPX Put position is limited to the amount of option premium it receives; however, the position can potentially lose up to the entire strike price of each option it sells.
- (b) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (c) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (d) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (e) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).

- (f) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (g) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (h) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (i) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (j) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles. For example, the calculation of the Reference Index relies on the assumption that it is possible to establish and close out positions in exchange-traded derivatives. There is no assurance that a liquid market on an exchange will continue for any particular exchange-traded derivative or at any particular time.
- (k) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (I) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (m) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.

- (n) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (o) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (p) Implied Volatility Risk. Through selling SPX Puts, the Reference Index value increases due to the premium it receives. However, the Reference Index is also exposed to the value of each SPX Put it has sold, until each SPX Put is either exercised or expires. The value of each SPX Put is partly based on the volatility assumption used by market participants to price such options (i.e. implied volatility). Higher implied volatility typically correlates with higher prices for options. Consequently, increases in the implied volatility of each SPX Put will cause the value of such options to increase which will result in a corresponding decrease in the value of the Reference Index and thus a decrease in the Fund's NAV. The Fund is therefore exposed to implied volatility risk before the Puts expire. The implied volatility of each SPX Put sold may increase due to general market and economic conditions.
- (q) The return of the Fund depends, amongst other things, on the performance of the Reference Index. The Fund gains exposure to the Reference Index by investing in one or more Swaps. An investment in one or more Swaps does not give the Fund a direct interest in the Reference Index and the portfolio of SPX Puts or any underlying asset linked to such portfolio of puts.
- (r) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the index administrator. The Index is subject to the terms set out in the index manual, as such index manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to the Index Owner.

The financial instrument is not sponsored, promoted, sold or supported in any other manner by Solactive AG nor does Solactive AG offer any express or implicit guarantee or assurance either with regard to the results of using the Index and/or Index trade mark or the Index level at any time or in any other respect. The Index is calculated and published by Solactive AG. Solactive AG uses its best efforts to ensure that the Index is calculated correctly. Irrespective of its obligations towards the issuer of the financial instrument, Solactive AG has no obligation to point out errors in the Index to third parties including but not limited to investors and/or financial intermediaries of the financial instrument. Neither publication of the Index by Solactive AG nor the licensing of the Index or Index trade mark for the purpose of use in connection with the financial instrument constitutes a recommendation by Solactive AG to invest capital in said financial instrument nor does it in any way represent an assurance or opinion of Solactive AG with regard to any investment in this financial instrument. Solactive AG is registered as a benchmark administrator under the Regulation (EU) 2016/2011 (BMR). Solactive appears on the ESMA register of administrators.

US Equity Defensive Put Write Index (the "Custom Index") is the property of UBS AG London Branch, which has contracted with S&P Opco, LLC (a subsidiary of S&P Dow Jones Indices LLC) ("SPDJI") to license the use of S&P 500 Index in connection with the Custom Index. The S&P 500 Index is the property of SPDJI, its affiliates (collectively with SPDJI, "S&P Dow Jones Indices") and/or their third-party licensors. S&P Dow Jones Indices and its third-party licensors shall have no liability for any errors or omissions in the S&P 500 Index and the Custom Index is not sponsored, endorsed, sold or promoted by S&P Dow Jones Indices or its third-party licensors.

The UBS US Equity Defensive Put Write SF UCITS ETF based on the Custom Index are not sponsored, endorsed, sold or promoted by S&P Dow Jones Indices or its third-party licensors. Neither S&P Dow Jones Indices nor its third party licensors make any representation or warranty, express or implied, to the owners of the UBS US Equity Defensive Put Write SF UCITS ETF or any member of the public regarding the advisability of investing in securities generally or in the UBS US Equity Defensive Put Write SF UCITS ETF particularly or the ability of the Index or Custom Index to track general market performance. S&P Dow Jones Indices' only relationship to UBS AG London Branch with respect to the Custom Index is the licensing of the S&P 500 Index certain rights with respect to the CBOE Option Settlement Prices with respect thereto and certain trademarks, service marks and trade names of S&P Dow Jones Indices or its third-party licensors. S&P Dow Jones Indices and its third party licensors are not responsible for and have not participated in the determination of the prices and amount of the UBS US Equity Defensive Put Write SF UCITS ETF or the timing of the issuance or sale of the UBS US Equity Defensive Put Write SF UCITS ETF or in the determination or calculation of the equation by which the UBS US Equity Defensive Put Write SF UCITS ETF may convert into cash or other redemption mechanics. S&P Dow Jones Indices and its third-party licensors have no obligation or liability in connection with the administration, marketing or trading of the UBS US Equity Defensive Put Write SF UCITS ETF. S&P Dow Jones Indices LLC is not an investment advisor. Inclusion of a security within the Index is not a recommendation by S&P Dow Jones Indices or its third-party licensors to buy, sell, or hold such security, nor is it investment advice.

NEITHER S&P DOW JONES INDICES NOR ITS THIRD-PARTY LICENSORS GUARANTEES THE ADEQUACY, ACCURACY, TIMELINESS AND/OR THE COMPLETENESS OF THE INDEX OR CUSTOM INDEX, THE CBOE OPTION SETTLEMENT PRICES OR ANY DATA RELATED THERETO OR ANY COMMUNICATION WITH RESPECT THERETO, INCLUDING, ORAL, WRITTEN, OR ELECTRONIC COMMUNICATIONS. S&P DOW JONES INDICES AND ITS THIRD-PARTY LICENSORS SHALL NOT BE SUBJECT TO ANY DAMAGES OR LIABILITY FOR ANY ERRORS, OMISSIONS, OR DELAYS THEREIN. S&P DOW JONES INDICES AND ITS THIRD PARTY LICENSORS MAKE NO EXPRESS OR IMPLIED WARRANTIES, AND EXPRESSLY DISCLAIM ALL WARRANTIES, OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE OR AS TO RESULTS TO BE OBTAINED BY UBS AG LONDON BRANCH, OWNERS OF THE UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF, OR ANY OTHER PERSON OR ENTITY FROM THE USE OF THE INDEX, CBOE OPTION SETTLEMENT PRICES OR CUSTOM INDEX OR WITH RESPECT TO ANY DATA RELATED THERETO. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT WHATSOEVER SHALL S&P DOW JONES INDICES OR ITS THIRD PARTY LICENSORS BE LIABLE FOR ANY INDIRECT, SPECIAL, INCIDENTAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES, INCLUDING BUT NOT LIMITED TO, LOSS OF PROFITS, TRADING LOSSES, LOST TIME, OR GOODWILL, EVEN IF THEY HAVE BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES, WHETHER IN CONTRACT, TORT, STRICT LIABILITY, OR OTHERWISE.

The S&P and S&P 500 are trademarks of Standard & Poor's Financial Services LLC.

UBS CMCI Future Commodity SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS CMCI Future Commodity SF UCITS ETF** (the **"Fund"**), a sub-fund of UBS (IrI) Fund Solutions plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KID may not be distributed unless accompanied by the Prospectus and the KID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the

CVO/657072-000001/26467820v31

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being UBS CMCI Future Commodity Index).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes for the performance of the Approved to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the **"Fund Assets**" for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

(z) The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The Fund is passively managed. The Fund complies with Article 8 of SFDR by tracking the Reference Index which has a sustainability profile ecological-social score (the "ES Score") that is higher than the UBS Constant Maturity Commodity Index ES Score to achieve an improvement in the portfolio level ES Score over time.

Information related to environmental and/or social characteristics is available in an annex to this document (SFDR RTS Art. 14(2)).

The Investment Manager applies climate transition benchmarks exclusion criteria as defined in Article 12(1)(a) to (c) of the Commission Delegated Regulation (EU) 2020/1818 on the portfolio of transferable securities.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective.

The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a **"Currency Index"** and together the **"Currency Indices"**). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

See "General Description of the Reference Index" section for more information on the Currency Indices.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.04% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;

An explanation for the divergence between the anticipated and realised tracking error for the year;

The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and

An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the following types of Swaps:

- 1. Fully funded Swaps whereby the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index;
- 2. Unfunded Swaps typically do not require any upfront payment and the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. The Fund will make a payment to the counterparty which is linked to the return on some or all of the transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index.
- 3. In addition, the Fund may use unfunded total return swaps in order to achieve the Investment

Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) a swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed and (b), in turn the fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

Fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded swaps, the unfunded swaps and the total return swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement"**), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a "**Counterparty**"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund

will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets In securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 51% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of

the Reference Index; and/or

(b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

CVO/657072-000001/26467820v31

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the "**Relevant Stock Exchanges**") for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended
Base Currency	US Dollar (" USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public
	and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), (B) banks are open for business in London and New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles.
	The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager).
	The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share
Minimum Subscription Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket size.
Minimum Redemption Amount	100,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.

Description of the Shares

Share Classes

Share	"USD	**"hEUR	**"hCHF	**"hGBP	**"hHKD	**"hSGD	**"hJPY
Class*	acc"	acc"	acc"	acc"	acc"	acc"	acc"
Initial Issue	USD 100	EUR 100	CHF 100	GBP 100	HKD 1,000	SGD 100	JPY 1,000

Share Class*	**"hSEK acc"	**hPLN acc	**hNOK acc	***USD UKdis	***hEUR UKdis	***hCHF UKdishCHF UKdis	***hGBP UKdis
Initial Issue	SEK 1,000	PLN 1,000	NOK 1,000	USD 100	EUR 100	CHF 100	GBP 100

Share	***hHKD	***hSGD	***hJPY	***hSEK	***hPLN	***hNOK
Class*	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue	HKD 1,000	SGD 100	JPY 1,000	SEK 1,000	PLN 1,000	NOK 1,000

* Non-Base Currency share classes will have a currency hedged exposure to the Reference Index

**Together the "acc" Share Class

*** Together the "dis" Share Class

Each Share Class not denominated in the Base Currency will have a currency hedged exposure through the relevant Currency Index. Such Share Class's performance may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund):

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat-Fees	0.34% per annum	0.34% per annum

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are not expected to exceed $\in 60,000.00$. The Manager will seek to recover such establishment and approval costs out of a portion of the Flat Fee. Any such establishment expenses recovered from the Flat Fee may be charged by the Manager over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- d) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- e) Any costs associated with foreign currency hedging.
- f) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index offers investors a commodity index that recognises the various maturities of commodity futures contracts that are available in commodity markets. Some other commodity indices offer an index based on short term commodity futures contracts, whereas the Reference Index enables the construction of commodity exposures across maturities. The Reference Index is made up of components across commodity categories of energy, industrial metals, precious metals, agriculture and livestock.

The Reference Index is designed to reflect the future of the global economy with regards to the resulting changes to production methods for, and utilization of, commodities. The Reference Index aims to incorporate these changes into the UBS Constant Maturity Commodity Index (the "CMCI") methodology by targeting an improvement in a portfolio level ecological-social score (the "ES Score") over time. To calculate this ES Score, the Index Owner (as defined below) uses an external provider, Reinhard Friesenbichler Unternehmensberatung or "RFU", to assess the current and future sustainability prospects of commodities within the CMCI across the energy, agriculture and metals industries. The ES Score is a numerical value per commodity that is computed by combining a continuous variable that assesses the sustainability of commodities from a production, utilisation, ecological and social perspective (e.g., with regard to waste and emissions, end-user industry impact, human rights, health and safety and supply chain impact), with a categorical variable that reflects the sustainability outlook of that commodity from a future utilisation and production point of view.

The composition of the Reference Index is rebalanced to new commodity target weights on an annual basis and the allocation to each commodity component is determined via the aid of optimization. The optimization objective is to minimize deviations from the CMCI target weights subject to: (i) liquidity, (ii) diversification constraints; and (iii) achieving an overall improvement in the ES Score at the Reference Index level.

For an index providing exposure to commodities such as the Reference Index, where categories of components making up the relevant index are highly correlated, such highly correlated categories must be considered as subcategories of the same commodity (in other words, components that are highly correlated must be considered as representing one component of the relevant index for the purposes of calculating the Fund's exposures). The Reference Index currently includes one component with a weighting of more than 20% and up to 35% (taking into account high correlation between certain components of the Reference Index such that they are considered together to constitute one component). Pursuant to the Regulations, the Fund is permitted to gain exposure to one component that represents more than 20% and up to 35% of the Reference Index where justified by exceptional market conditions.

The Reference Index is calculated in USD, and includes a notional exposure to overnight interest rates. The Reference Index has different Currency Indices, including Currency Indices denominated in Euro, GBP, CHF, HKD, SGD, JPY and SEK. Such Currency Indices employ forward FX contracts to provide the relevant currency return. The Currency Indices are designed to represent the returns of the Reference Index while reducing currency risk but not the underlying commodity market risk. By employing such a strategy, the Currency Indices seek to reduce the risk of downward currency fluctuations although potential currency gains may also be sacrificed.

The Reference Index is rebalanced to the target weights on a monthly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be staticThe Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is governed by MerQube, Inc (the "Index Administrator") via its internal processes. The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index. More details on the Reference Index, including its underlying components, can be found at https://www.ubs.com/global/en/investment-bank/cmci.html

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) Classes of Shares denominated in currencies other than USD are exposed to the Currency Index of the Reference Index relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging costs.
- (i) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is

terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.

- (j) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in commodity markets.
- (k) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchangetraded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (I) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (m) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (n) The Fund, through its use of the Reference Index has exposure to interest rates. Such interest rates may be negative or positive, which will impact the return the Fund negatively or positively respectively.
- (o) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF

CVO/657072-000001/26467820v31

UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS CMCI COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

UBS Index Disclaimer

No legal relationship (whether in contract, tort, or otherwise) exists between any investor buying, selling, entering into or holding products linked to the Index (such products, the "Index Products" and such investors, the "Index Product Investors") and UBS AG, London Branch (in its capacity as owner of the Index the "Index Owner") and the Index Owner owes no duty (whether in contract, tort, or otherwise) to any Index Product Investor. No claims, actions or legal proceedings may therefore be brought against the Index Owner in any manner whatsoever by an Index Product Investor or any other person.

The Index Owner makes no representation, warranty or guarantee whatsoever, express or implied, either as to the results to be obtained as to the use of the Index or the figures or levels at which the Index stands at any particular day or otherwise. In addition, the Index Owner gives no assurance regarding any modification or change in any methodology used in calculating the Index and is under no obligation to continue the calculation, publication and dissemination of the Index.

The Index Owner does not warrant or represent or guarantee to any person the accuracy or completeness of the Index and its computation or any information related thereto and makes no warranty or representation or guarantee of any kind whatsoever relating to the Index. The process and basis of computation and compilation of the Index and the related formulae, constituent benchmarks and factors may at any time be changed or altered by the Index Administrator. The Index is subject to the terms set out in the Index Manual, as such Index Manual may be amended from time to time.

No responsibility or liability is accepted by the Index Owner (whether for negligence or otherwise) in respect of the use of and/or reference to the Index by us or any other person in connection with securities, or for any inaccuracies, omissions, mistakes or errors in the computation of the Index (and the Index Owner shall not be obliged to advise any person or any Index Product Investor of any error therein) or for any economic or other loss which may be directly or indirectly sustained by any Index Product Investor or any other persons dealing with securities as a result. Any Index Product Investor or other person dealing with securities does so, therefore, in full knowledge of this disclaimer and can place no reliance whatsoever on the Index Owner.

The Index Owner makes no representation, warranty or guarantee as to the accuracy, completeness or appropriateness of the methodology, nor does it accept any liability for the use of such information.

Nothing in this disclaimer shall exclude or limit liability to the extent such exclusion or limitation is not permitted by law or regulations to which the Index Owner is subject.

The mark and name of the Index is proprietary to UBS.

CMCI is a trademark of UBS AG.

Index Disclaimer for MerQube

MerQube, Inc. administers and calculates the UBS CMCI Future Commodity Index for UBS. MerQube, Inc. does not guarantee the accuracy, timeliness, or completeness of the calculations or any data or information relating to the Index. MerQube, Inc. makes no warranty, express or implied, as to the Index, or any values or data related thereto or results to be obtained therefrom, and expressly disclaims all warranties of merchantability and fitness

for a particular purpose with respect thereto. To the maximum extent permitted by law, MerQube, Inc., its affiliates, and all their respective employees, subcontractors, partners, agents, suppliers, and vendors (collectively, the "protected parties") shall have no liability, contingent or otherwise, for any injury or damages, whether caused by the negligence of a protected party or otherwise, arising in connection with the calculation or administration of the Index or any data or values included therein or in connection therewith and shall not be liable for any losses (including lost profits), punitive, incidental, or consequential damages.

Sustainable

investment means an investment in an economic activity that contributes to an environmental or social objective, provided that the investment does not significantly harm any environmental or social objective and that the investee companies follow good governance practices.

Pre-contractual disclosure for the financial products referred to in Article 8, paragraphs 1, 2 and 2a, of Regulation (EU) 2019/2088 and Article 6, first paragraph, of Regulation (EU) 2020/852 Product name: UBS CMCI Future Commodity SF UCITS ETF 54930002MV1UTBISOC34

🛛 No

ANNEX II

BS CMCI Future Commodity SF UCITS ETF 54930002MV1UTBISOC34 Environmental and/or social characteristics

Does this financial product have a sustainable investment objective?

•• 🗌 Yes

It will make a minimum of sustainable investments with an environmental objective: %

The EU Taxonomy is a classification system laid down in Regulation (EU) 2020/852, establishing a list of environmentally sustainable economic activities. That Regulation does not include a list of socially sustainable economic activities. Sustainable investments with an environmental objective might be aligned with the Taxonomy or not.

in economic activities that qualify as environmentally sustainable under the EU Taxonomy

in economic activities that do not qualify as environmentally sustainable under the EU Taxonomy

It will make a minimum of sustainable investments with a social objective:

____%

Lt promotes Environmental/Social (E/S) characteristics and while it does not have as its objective a sustainable investment, it will have a minimum proportion of __% of sustainable investments

> with an environmental objective in economic activities that qualify as environmentally sustainable under the EU Taxonomy

with an environmental objective in economic activities that do not qualify as environmentally sustainable under the EU Taxonomy

with a social objective

It promotes E/S characteristics, but will not make any sustainable investments

Sustainability indicators measure	What environmental and/or social characteristics are promoted by this financial product?
how the sustainable objectives of this financial product are attained.	 This financial product is passively managed and tracks an index ("Index/Reference Index"). The following characteristics are promoted by the financial product: tracks/tracking of a benchmark with a sustainability profile ecological-social score (the "ES Score") that is higher than the parent benchmark's ES Score to achieve an improvement in the portfolio level ES Score over time. The parent benchmark is the UBS Constant Maturity Commodity Index. The Reference Index designated for the purpose of attaining the characteristic pro-moted by the financial product is the UBS CMCI Future Commodity Index.
	What sustainability indicators are used to measure the attainment of each of the environmental or social characteristics promoted by this financial product?
	The Reference Index is constructed by incorporating ecological and social criteria into the UBS Constant Maturity Commodity Index (the "CMCI" - parent benchmark) to achieve an improvement in the portfolio level ecological-social score (the "ES Score") over time. To calculate this ES Score, the Index Owner uses an external provider, Reinhard Friesenbichler Unternehmensberatung or "RFU", to assess the current and future sustainability prospects of commodities within the CMCI across the energy, agriculture and metals industries.
	The ES score is a numerical value per commodity that is computed by combining a continuous variable that assesses the sustainability of commodities from a production, utilisation, ecological and social perspective (e.g., with regard to waste and emissions, end-user industry impact, human rights, health and safety and supply chain impact), with a categorical variable that reflects the sustainability outlook of that commodity from a future utilisation and production point of view. The composition of the Reference Index is rebalanced to new commodity target weights on an annual basis and the allocation to each commodity component is determined via the aid of optimization. The optimization objective is to minimize deviations from the CMCI target weights subject to: (i) liquidity, (ii) diversification constraints; and (iii) achieving an overall improvement in the ES Score at the Reference Index level.

Principal adverse impacts are the most significant negative impacts of investment decisions on sustainability factors	How do the sustainable investments that the financial product partially intends to make, not cause significant harm to any environmental or social sustainable investment objective?
relating to environmental, social	N/A
and employee matters, respect for human rights, anti- corruption and anti- bribery matters.	How have the indicators for adverse impacts on sustainability factors been taken into account?
	N/A
	How are the sustainable investments aligned with the OECD Guidelines for Multinational Enterprises and the UN Guiding Principles on Business and Human Rights?
	N/A
	The EU Taxonomy sets out a "do not significant harm" principle by which Taxonomy- aligned investments should not significantly harm EU Taxonomy objectives and is accompanied by specific EU criteria.

The "do no significant harm" principle applies only to those investments underlying the financial product that take into account the EU criteria for environmentally sustainable economic activities. The investments underlying the remaining portion of this financial product do not take into account the EU criteria for environmentally sustainable economic activities.

Any other sustainable investments must also not significantly harm any environmental or social objectives.

	Does this financial product consider principal adverse impacts on sustainability factors?
	⊠ No
	What investment strategy does this financial product follow?
The investment strategy guides investment decisions based on factors such as investment objectives and risk tolerance.	The investment objective of the financial product is to track the performance before fees and expenses of the Reference Index. The Reference Index offers investors a commodity index that recognises the various maturities of commodity futures contracts that are available in commodity markets. Some other commodity indices offer an index based on short term commodity futures contracts, whereas the Reference Index enables the construction of commodity

 exposures across maturities. The Reference Index is made up of components across commodity categories of energy, industrial metals, precious metals, agriculture and livestock. The Reference Index is constructed by incorporating ecological and social criteria into the UBS Constant Maturity Commodity Index (the "CMCI" - parent benchmark) to achieve an improvement in the portfolio level ecological-social score over time. Currently, the CMCI weights future tenors across the curve in accordance with a 75% liquidity and 25% equal weighting rule. For the CMCI Sustainability Transition, longer dated tenors are overweighted proportional to maturity for commodities with a higher weight than the CMCI benchmark – the goal is to increase allocation at the back end of the curve and enhance liquidity for producers.
What are the binding elements of the investment strategy used to select the investments to attain each of the environmental or social characteristics promoted by this financial product?
The investment objective of the financial product is to track the performance before fees and expenses of the Reference Index.
The Reference Index aims to incorporate ecological and social criteria that has been developed by RFU, specialists in sustainable and investment research, into the UBS Constant Maturity Commodity Index (the "CMCI" - parent benchmark) and gradually improve its portfolio level Ecological-Social (E-S) score each year by +0.245. This gradual improvement will happen until no feasible solution as part of the optimisation process can be found to increase the score.
Beyond the year-on-year improvement of the index E-S score by +0.245, the CMCI Sustainability Transition leverages CMCI's tenor diversification to enhance these characteristics.
For the Ecological-Social (E-S) score, a total rating for each commodity is calculated, based on deep analysis of contributory factors four key areas broken into two sub-groups (Social and Ecological, and Production and Utilization). Together, this results in a score on the scale from -10 to +10, transformed into a rating on the nine-point scale from C- to A+. The weightings with which the individual criteria and dimensions are included in the overall rating are dependent on the commodity group:
Social impact : 44%–53% weighting
Ecological impact: 47%–56% weighting
Production impact: 51%–60% weighting
Utilization: 40%–49% weighting
The E-S Score for the reference index is calculated annually (every January).

Good governance practices include sound management structures, employee relations, remuneration of staff	What is the committed minimum rate to reduce the scope of the investments considered prior to the application of that investment strategy?
and tax compliance.	This financial product does not have a committed minimum rate to reduce the scope of investments considered prior to the application o the investment strategy.
	What is the policy to assess good governance practices of the investee companies?
	N/A
	What is the asset allocation planned for this financial product?
	The minimum proportion of the investments used to meet the environmental and/or social characteristics promoted by the financial product is 90%. The financial product does not commit to a minimum proporation of sustainable investments.
Asset allocation describes the share of investments in specific assets.	□ #1A Sustainable 0% with E/S characteristics 90%
Taxonomy-aligned activities are expressed as a share of: - turnover reflecting the share of revenue from green activities	 □ Investments □ #2 Other 10% □ #1B Other E/S characteristics 100%
of investee companies - capital expenditure	#1 Aligned with E/S characteristics includes the investments of the financial product used to attain the environmental or social characteristics promoted
(CapEx) showing the green investments made by investee companies, e.g. for a transition to a green economy.	by the financial product. #2 Other includes the remaining investments of the financial product which are neither aligned with the environmental or social characteristics, nor are qualified as sustainable investments.
- operational expenditure (OpEx) reflecting green operational activities of investee	The category #1 Aligned with E/S characteristics covers: - The sub-category #1A Sustainable covers sustainable investments with environmental or social objectives.
companies.	- The sub-category #1B Other E/S characteristics covers investments aligned with the environmental or social characteristics that do not qualify as sustainable investments
	How does the use of derivatives attain the environmental or

How does the use of derivatives attain the environmental or social characteristics promoted by the financial product?

Image: construction of the structure of the			(such swaps may include fun such the Fund promotes the those of the Reference Index index of commodity futures w social criteria that has been of sustainable and investment r Commodity Index (the "CMC	the Reference Index through swaps ided, unfunded or total return swaps), as same environmental characteristics as x, which is to gain indirect exposure to an which aims to incorporate ecological and developed by RFU, specialists in esearch, into the UBS Constant Maturity CI" - parent benchmark) and gradually blogical-Social (E-S) score each year.
are sustainable investments with an env0% Does the financial product invest in fossil gas and/or nuclear energy related activities that comply with the EU Taxonomy from the EU Taxonomy, the criteria for fossil gas include imitations on emissions and switching to renewable power or low-carbon fuels by the end of 2035. For nuclear energy, the criteria include energy, the criteria include so the financial product of the financial product other than sovereign bonds. The two graphs below show in green the minimum percentage of investments that are aligned with the EU Taxonomy. As there is no appropriate methodology to determine in Taxonomy-alignment of sovereign bonds, the first graph shows the Taxonomy alignment of sovereign bonds. In the financial product other than sovereign bonds. 1. Taxonomy-aligned including sovereign bonds. 1. Taxonomy-aligned activities are and graph shows the Taxonomy alignment of investments of the financial product tothe investments of the financial product tothe investments of the financial product other than sovereign bonds. 1. Taxonomy-aligned including sovereign bonds* 1. Taxonomy-alig		****		
energy related activities that comply with the EU Taxonomy fill To comply with the EU Taxonomy, the criteria for fossil gas include limitations on emissions and switching to renewable power or low-carbon fuels by the end of 2035. For nuclear energy, the criteria include comprehensive safety and waste management rules. In fossil gas In nuclear energy The two graphs below show in green the minimum percentage of investments that are aligned with the EU Taxonomy. As there is no appropriate methodology to determine traxonomy-alignment of sovereign bonds*, the first graph shows the Taxonomy alignment in relation to all the investments of the financial product including sovereign bonds, we the second graph shows the Taxonomy alignment of investments including sovereign bonds* 1. Taxonomy-alignment of investments including sovereign bonds* 2. Taxonomy-alignment of investments including sovereign bonds* 1. Taxonomy-alignment of investments including sovereign bonds* 1. Taxonomy-alignment of investments including sovereign bonds* 1. Taxonomy-alignment of investments including sovereign bonds* 1. Taxonomy-alignment of investments including sovereign bonds* 1. Taxonomy-aligneed (no fossil gas & miclear] alternatives are not yet available and among others have greenhouse gas emission levels corresponding to the 1.00%				
EU Taxonomy, the criteria for fossil gas include limitations on emissions and switching to renewable power or low-carbon fuels by the end of 2035. For nuclear energy, the criteria include 2035. For nuclear energy, the criteria include comprehensive safety and waste management rules. In fossil gas In nuclear energy				
low-carbon fuels by the end of 2035. For nuclear energy, the criteria include comprehensive safety and waste management rules. Enabling activities directly enable other activities to make a substantial contribution to an environmental objective. Transitional activities are activities for which low-carbon alternatives are not yet available and among others have greenhouse gas emission levels corresponding to the	EU Taxonomy, the criteria for fossil gas include limitations on emissions and switching to renewable power or low-carbon fuels by the end of 2035. For nuclear energy , the criteria include comprehensive safety and waste management rules. Enabling activities directly enable other		🗆 In fossil gas 🛛 🗆 In nuclear	energy
contribution to an environmental objective. 1. Taxonomy-alignment of investments including sovereign bonds* 2. Taxonomy-alignment of investments excluding sovereign bonds* Transitional activities are activities for which low-carbon alternatives are not yet available and among others have greenhouse gas emission levels corresponding to the Taxonomy-aligned (no fossil gas & nuclear) Taxonomy-aligned (no fossil gas & nuclear) 100% 100% 100%			aligned with the EU Taxonomy. As the Taxonomy-alignment of sovereign bor in relation to all the investments of the the second graph shows the Taxonomy	re is no appropriate methodology to determine the ads*, the first graph shows the Taxonomy alignment e financial product including sovereign bonds, while y alignment only in relation to the investments of
Transitional activities are activities for which low-carbon alternatives are not yet available and among others have greenhouse gas emission levels corresponding to the	contribution to an environmental			
emission levels corresponding to the	Transitional activities are activities for which low-carbon alternatives are not yet available and among others have		(no fossil gas & nuclear) Non Taxonomy-	(no fossil gas & nuclear) Non Taxonomy-
	emission levels		100%	100%
best performance. * For the purpose of these graphs, 'sovereign bonds' consist of all sovereign exposure	corresponding to the best performance.		* For the purpose of these graphs, 'so	overeign bonds' consist of all sovereign exposures

¹ Fossil gas and/or nuclear related activities will only comply with the EU Taxonomy where they contribute to limiting climate change ("climate change mitigation") and do not significantly harm any EU Taxonomy objective – see explanatory note in the left-hand margin. The full criteria for fossil gas and nuclear energy economic activities that comply with the EU Taxonomy are laid down in Commission Delegated Regulation (EU) 2022/1214.

are environmentally sustainable investments that do not take into account the criteria		What is the minimum share of investments in transitional and enabling activities?
for environmentally sustainable economic activities under the EU Taxonomy.		N/A
		What is the minimum share of sustainable investments with an environmental objective that are not aligned with the EU Taxonomy?
		N/A
		What is the minimum share of socially sustainable investments?
		N/A
		What investments are included under "#2 Other", what is their purpose and are there any minimum environmental or social safeguards?
		Included in "#2 Other" are cash deposits, short-term money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes for the purpose of efficient portfolio and cash management.
		Investments included in "#2 Other" are not considered against minimum environmental or social safeguards.
Reference benchmarks are indexes to measure whether the financial product attains the sustainable investment objective.	?	Is a specific index designated as a reference benchmark to determine whether this financial product is aligned with the environmental and/or social characteristics that it promotes?
		The Reference Index is designated for the purpose of attaining the characteristics promoted by the financial product.
		How is the reference benchmark continuously aligned with each of the environmental or social characteristics promoted by the financial product?
		The Reference Index is constructed by incorporating ecological and social criteria into the UBS Constant Maturity Commodity Index (the "CMCI") to achieve an improvement in the portfolio level ecological- social score (the "ES Score") over time. To calculate this ES Score, the Index Owner (as defined below) uses an external provider, Reinhard Friesenbichler Unternehmensberatung or "RFU", to assess the current and future sustainability prospects of commodities within

www.ubs.com/etf
Where can I find more product specific information online? More product-specific information can be found on the website:
<u>https://merqube.com/files/9056055b-fa51-42f6-be8c-</u> ebc0d6ad9163/methodology/UBS%20CMCI%20Technical%20Docu ment.pdf
The Reference Index is governed by MerQube, Inc (the "Index Administrator") via its internal processes. The Index Administrator controls the creation and operation of the administrative process, including all stages and processes involved in the production and dissemination of the Reference Index. UBS AG ("Index Owner") is the sole owner of the intellectual property rights with respect to the Reference Index. More details on the Reference Index, including its underlying components, can be found at:
Where can the methodology used for the calculation of the designated index be found?
The Reference Index is constructed by incorporating ecological and social criteria into the UBS Constant Maturity Commodity Index (the relevant broad market index) to achieve an improvement in the portfolio level ecological-social score over time.
How does the designated index differ from a relevant broad market index?
The Fund synthetically replicates the performance of the Reference Index by receiving the return of the Reference Index through Swaps.
How is the alignment of the investment strategy with the methodology of the index ensured on a continuous basis?
The composition of the Reference Index is rebalanced to new commodity target weights on an annual basis and the allocation to each commodity component is determined via the aid of optimization. The optimization objective is to minimize deviations from the CMCI target weights subject to: (i) liquidity, (ii) diversification constraints; and (iii) achieving an overall improvement in the ES Score at the Reference Index level.
the CMCI across the energy, agriculture and metals industries. The ES score is a numerical value per commodity that is computed by combining a continuous variable that assesses the sustainability of commodities from a production, utilisation, ecological and social perspective (e.g., with regard to waste and emissions, end-user industry impact, human rights, health and safety and supply chain impact), with a categorical variable that reflects the sustainability outlook of that commodity from a future utilisation and production point of view.

UBS MSCI USA SF INDEX FUND

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS MSCI USA SF INDEX FUND** (the **"Fund**"), a sub-fund of UBS (Irl) Fund Solutions plc (the **"Company"**) an umbrella type open-ended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KIID, and may not be distributed unless accompanied by the Prospectus and the KIID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest principally on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions plc

An umbrella fund with segregated liability between the sub-funds

Dated 12 May

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking capital appreciation and who are prepared to accept a high level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. Your attention is drawn to Section entitled "Share Dealing" hereof which provides general information about offers of shares in the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company.

You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KIID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KIID and the latest annual report and audited accounts of the Company and the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then

latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being the MSCI USA Index Net Total Return).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- (i) swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association, and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts – Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts – Swaps" below); and
- (ii) a portfolio of transferable securities which will include (without limitation) equity and equity-related securities (which may include common stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such transferable securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the "**Fund Assets**" for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed (i.e. its objective is to track the Reference Index).

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment strategy and the nature of underlying investments.

Principal adverse impacts (the "PAI") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to the Reference Index in addition to referencing one month rolling FX forward contracts. The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to FX forwards contracts is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

The anticipated level of tracking error after fees and transaction costs is 0.10% per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;

- c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and
- d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Use of FDI Contracts – Swaps

The Fund will enter into Swaps with a member of the UBS AG Group (being the "**Approved Counterparty**") pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index and a cash payment which will equate to the Flat Fee (as described below).

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow like Euribor, etc):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically unfunded swaps do not require any upfront payment of cash.
- 3. In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the components of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets

In order to target a reduction in tracking difference and/or target an improved performance of the Fund, the Investment Manager may invest in swaps which reference a gross total return version of the Reference Index.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap

agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the **"ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 4 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a **"Counterparty"**). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets

that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus. As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act. Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Leverage

The total expected level of leverage of the Fund as a result of its investments shall not exceed 100% of its Net Asset Value, as calculated using the commitment approach.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund.

The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, hEUR acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc, hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the USD UKdis, hEUR UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hSGD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

The Directors do not intend to list or trade the Shares of the Fund on any Regulated Stock Exchanges.

GENERAL INFORMATION RELATING TO THE FUND

Turne	Onen ended
Туре	Open-ended
Base Currency	US Dollar ("USD ")
Business Day	A day (other than a Saturday or Sunday) on which (i) (A) banks are open for business in Dublin (inclusive of all Irish public and/or bank holidays other than the following – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day), (B) banks are open for business in New York; and (C) the Reference Index is published by the Index Provider; or (ii) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager). The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	1.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	USD 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share
Minimum Initial	10,000,000 USD or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Initial Investment Amount.

Investment Amount	
Minimum Additional Investment Amount	1 Share or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Additional Investment Amount.
Minimum Redemption Amount	1 Share or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount.
Website	www.ubs.com/funds - information on the Fund's NAV is set out on the Website.

Initial Offer Period

The USD acc Share Class, the USD UKdis Share Class, the hEUR acc Share Class, the hEUR UKdis Share Class, the hCHF acc Share Class, the hCHF UKdis Share Class, the hGBP acc Share Class and the hGBP UKdis Share Class have launched and Shares in these Share Classes are available on each Dealing Day at the then prevailing Net Asset Value per Share.

The initial offer period of the hHKD acc Share Class, the hHKD UKdis Share Class, the hSGD acc Share Class, the hSGD UKdis Share Class, the hJPY acc Share Class, the hJPY UKdis Share Class, the hSEK acc Share Class, the hSEK UKdis Share Class, the hPLN acc Share Class, the hPLN UKdis Share Class, the hNOK acc Share Class and the hNOK UKdis Share Class will remain open during the period from 9 am on 13 May 2025 (Irish time) and will close at 5 pm on 12 November 2025 (Irish time).

Share Classes ‡			
"acc" Share Classes	"dis" Share Classes	Initial Issue Price	
*USD acc	**USD UKdis	N/A	
*hEUR acc	**hEUR UKdis	N/A	
*hCHF acc	**hCHF UKdis	N/A	
*hGBP acc	**hGBP UKdis	N/A	
*hHKD acc	**hHKD UKdis	HKD 1,000	
*hSGD acc	**hSGD UKdis	SGD 100	
*hJPY acc	**hJPY UKdis	JPY 1,000	
*hSEK acc	**hSEK UKdis	SEK 1,000	
*hPLN acc	**hPLN UKdis	PLN 1,000	
*hNOK acc	**hNOK UKdis	NOK 1,000	

Description of the Shares

* Together the "**acc**" Share Class.

** Together the "dis" Share Class.

‡ Each Share Class not denominated in the Base Currency will have a currency hedged exposure to the Reference Index. The performance of such Share Class may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund, and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund).

Share Class	"acc"	"dis"
Subscription Fee*	Up to 5%	Up to 5%
Exchange Charge	Up to 3%	Up to 3%
Redemption Fee**	Up to 3%	Up to 3%

*The Subscription Fee which may be incurred will be allocated as follows; up to 2% may be charged by the Company and payable solely to the Approved Counterparty; the balance of up to 3% is payable to the distributors.

**The Redemption Fee which may be incurred will be allocated as follows; up to 1% may be charged by the Company and payable solely to the Approved Counterparty; the balance of up to 2% is payable to the distributors.

Separate to and distinct from the charges set out above, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat Fee	0.12% per annum	0.12% per annum

The Manager shall discharge the fees associated with the establishment and approval of the Fund which are estimated to be in the region of \in 60,000 but will seek to recover such establishment and approval costs if there is money available to it out of the Flat Fee. Any such expenses recovered from the Flat Fee may be charged over a period of up to five years.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will (in order to deliver a return which includes the Flat Fee per the agreed terms of the Swap) take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.

c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Redemption Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index is described below but this only represents an extract of information available and the Directors, the Manager, UBS AG and the Investment Manager have taken all reasonable efforts to ensure the accuracy and completeness of such information.

The Reference Index is a free float adjusted market capitalization index that is designed to measure large and mid-cap US equity market performance. The Reference Index is member of the MSCI international equity index series and represents the US equity portion of the global benchmark MSCI ACWI (All Country World Index) Index.

The Reference Index is rebalanced on a quarterly basis.

The rebalancing frequency of the Reference Index has no direct impact on the transaction costs associated with the Fund itself as any rebalancing within the Reference Index is not expected to require any higher frequency of position turnover in the Fund than would otherwise be the case were the Reference Index to be static.

Where the Investment Manager becomes aware that the weighting of any particular component in the Reference Index exceeds the permitted investment restrictions, the Investment Manager will seek to amend the Fund's investment exposure to ensure the Fund operates within the permitted investment restrictions, while keeping the best interests of the Shareholders in consideration.

The Reference Index is calculated in USD (Dollar).

Index Provider and website

The Reference Index is sponsored by MSCI Limited.

More details on the Reference Index, including its underlying components, can be found at <u>www.msci.com</u>.

Subscription for Shares

General

Applications for initial subscriptions will be accepted at the Initial Issue Price plus Subscription Fee (if applicable). Subsequent subscriptions will be accepted at a price corresponding to the Net Asset Value per Share as determined as at the Valuation Point for the relevant Dealing Day, plus Subscription Fee (if applicable).

Applications for Shares

Applications for Shares may be made through the Administrator or through a duly appointed distributor for onward transmission to the Administrator. Applications received by the Administrator or duly appointed distributor prior to the Dealing Deadline for any Dealing Day will be processed on that Dealing Day. As set out in the Prospectus, any applications received after the Dealing Deadline for a particular Dealing Day will be processed on the following Dealing Day unless the Directors in their absolute discretion, in exceptional circumstances, otherwise determine to accept one or more applications received after the Dealing Deadline for processing on that Dealing Day provided that such application(s) have been received prior to the Valuation Point for the particular Dealing Day.

Initial applications should be made using an Application Form obtained from the Administrator which may be submitted by electronic means or by fax with the original form to follow as soon as possible and signed. All initial applications shall be subject to transmission to the Administrator of such other papers in such format (such as documentation relating to money laundering prevention checks) as may be required by the Directors or their delegate. Shares will not be issued, and no subscription monies will be accepted, until such time as the Administrator has received and is satisfied with all the information and original documentation required to verify the identity of the investor. In the case of subsequent applications submitted by electronic means or by fax, it shall not be necessary for the Administrator to subsequently receive the original Application Form provided that the Directors are satisfied that the appropriate controls and procedures are in place to comply with applicable anti-money laundering legislation and to ensure that any risk of fraud associated with the processing of transactions based on such means are adequately mitigated.

Amendments to a Shareholder's registration details and payment instructions will only be made following receipt of written or electronic instructions and appropriate original documentation from the relevant Shareholder.

Any applications submitted by electronic means must be in a form and method agreed by the Manager and the Administrator.

Applications will be irrevocable unless the Manager, or a delegate, otherwise agrees.

The Application Form contains certain conditions regarding the application procedure for Shares in the Company and certain indemnities in favour of the Company, the Manager, the Fund, the Administrator, the Depositary and the other Shareholders for any loss suffered by them as a result of certain applicants acquiring or holding Shares.

Form of Shares and Confirmation of Ownership

Confirmation of each purchase of Shares will normally be sent to Shareholders within two (2) Business Days of the purchase being made. Shares shall be issued in registered form only and

title to Shares will be evidenced by written confirmation of entry of the investor's name on the Company's register of Shareholders and no certificates will be issued.

Method of Payment and Subscriptions/ Redemptions Account

Subscription payments net of all bank charges should be paid by SWIFT or electronic transfer to the Subscriptions/ Redemptions Account as specified in the Application Form. Other methods of payment are subject to the prior approval of the Manager or its delegate.

No interest will be paid in respect of payments received in circumstances where the application is received in advance of a Dealing Day or held over until a subsequent Dealing Day.

Where the subscription monies are received into the Subscriptions/Redemptions Account from an investor in advance of Shares being issued (as will be the case in the context of the Fund which operates on a cleared funds basis), such subscription monies will be the property of the Fund and accordingly an investor will be treated as a general unsecured creditor of the Company during the period between receipt of subscription monies into the Subscriptions/ Redemptions Account and the issue of Shares.

Currency of Payment

Subscription monies are payable in the denominated currency of the Share Class.

Timing of Payment

Payment in respect of subscription must be received in cleared funds into the Subscriptions/Redemptions Account on or before the Settlement Date as outlined above – ("General Information Relating to the Fund").

If payment in full in respect of the issue of Shares has not been received by the relevant time on the relevant Settlement Date, or in the event of non-clearance of funds, the allotment of Shares made in respect of such application may, at the discretion of the Directors, be cancelled, or, alternatively, the Directors may treat the application as an application for such number of Shares as may be purchased with such payment on the Dealing Day next following receipt of payment in full of cleared funds. In such cases the Company may charge the applicant for any resulting bank charges or market losses incurred by the Fund.

Anti-Dilution Levy

The Directors reserve the right to impose an Anti-Dilution Levy on a transaction basis in the case of net subscriptions as a percentage adjustment (to be communicated to the Administrator) on the value of the relevant subscription calculated for the purposes of determining a subscription price to cover dealing costs and to preserve the value of the underlying assets of the Fund where they consider such a provision to be in the best interests of the Fund. Such amount will be added to the price at which Shares will be issued in the case of net subscription requests. Any such sum will be paid into the account of the Fund.

Redemption of Shares

General

Shareholders may redeem their Shares on a Dealing Day at the Redemption Price which shall be the Net Asset Value per Share, less Redemption Fee, (as disclosed herein) and any applicable duties and charges (save during any period when the calculation of the Net Asset Value is suspended). Please see the section in the Prospectus entitled "Suspension of Calculation of Net Asset Value" for further information in this regard.

Redemption Requests

Requests for the redemption of Shares should be made to the Administrator on behalf of the Company or through a duly appointed distributor for onward transmission to the Administrator and may be submitted in original form, by electronic means or by fax with the original to follow promptly and must be signed and should include such information as may be specified from time to time by the Directors or their delegate. Requests for repurchase received prior to the Dealing Deadline for any Dealing Day will be processed on that Dealing Day. As set out in the Prospectus, any requests for repurchase received after the Dealing Deadline for a Dealing Day will be processed on the next Dealing Day unless the Directors in their absolute discretion in exceptional circumstances, otherwise determine to accept one or more applications received after the Dealing Day provided that such request(s) have been received prior to the Valuation Point for the particular Dealing Day.

The Minimum Redemption Amount (if any) may vary according to the Class of Share.

In the event of a Shareholder requesting a repurchase which would, if carried out, leave the Shareholder holding Shares having a Net Asset Value less than the Minimum Holding, the Company may, if it thinks fit, repurchase the whole of the Shareholder's holding.

Method of Payment

The amount due on repurchase of Shares will be paid by electronic transfer only to the relevant Shareholder's account of record on the initial Application Form by the Settlement Date.

In no event shall redemption proceeds be paid until such papers as may be required by the Directors have been received from the investor and all of the necessary anti-money laundering checks have been carried out, verified and received in original form.

Amendments to a Shareholder's registration details and payment instructions will only be made following receipt of written instructions and appropriate original documentation from the relevant Shareholder.

Currency of Payment

Shareholders will normally be repaid in the denominated currency of the relevant Class.

Timing of Payment and Subscriptions/Redemptions Account

Investors should note that although the Directors intend to pay all redemption proceeds by the relevant Settlement Date, such payment may be delayed if the Directors consider such delay to be in the best interests of the Shareholders. In all instances the payment of the redemption proceeds shall be within 10 Business Days of the relevant Dealing Deadline.

Investors should note that any redemption proceeds being paid out by the Fund and held for any time in the Subscriptions/ Redemptions Account shall remain an asset of the Fund until such time as the proceeds are released to the investor. This would include, for example, cases where redemption proceeds are temporarily withheld pending the receipt of any outstanding identity verification documents as may be required by the Company or the Administrator – enhancing the need to address these issues promptly so that the proceeds may be released. It should also be noted that the investor shall have ceased being considered a Shareholder and instead will rank as a general unsecured creditor of the Company.

Withdrawal of Redemption Requests

Requests for redemption may not be withdrawn save with the written consent of the Directors or their delegate.

Anti-Dilution Levy

The Directors reserve the right to impose an Anti-Dilution Levy in the case of net repurchases on a transaction basis as a percentage adjustment (to be communicated to the Administrator) on the value of the relevant repurchase calculated for the purposes of determining a Redemption Price to cover dealing costs and to preserve value of the underlying assets of the Fund where they consider such a provision to be in the best interests of a Fund. Such amount will be deducted from the price at which Shares will be repurchased in the case of net repurchase requests. Any such sum will be paid into the account of the Fund.

Fractions

Fractions of Shares will be issued where any part of the subscription monies for Shares represents less than the subscription price for one Share, provided however, that fractions shall not be less than 0.0001 of a Share. Subscription monies representing less than 0.0001 of a Share will be retained by the Company in order to defray administration costs.

For further information on "Restrictions on Subscriptions", "Minimum Subscription Amount, Minimum Initial Investment Amount, Minimum Additional Investment Amount and Minimum Holding", "Subscriptions/Redemptions Account", "Data Protection" and "Restrictions on Redemptions" please refer to paragraphs 9.2.3, 9.2.4, 9.2.5, 9.2.6, and 9.3.2 of the Prospectus.

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Unlike the other sub-funds of the Company, it is not anticipated that there will be an active secondary market for the Shares of the Fund, and it is not expected that such a market will develop. Subject to certain conditions outlined herein, including when repurchases or the registration of transfers of Shares are suspended, Shareholders will, however, be able to realise their investment in a Fund by redeeming their Shares or by a transfer to an investor who is an eligible transferee.
- (c) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.
- (d) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (e) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).

- (f) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (g) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (h) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (i) Classes of Shares denominated in currencies other than USD have an exposure to the Reference Index in addition to FX forward contracts relating to the currency of the Share Class denomination. Consequently, returns in respect of a Base Currency Share Class may differ from those in respect of Non-Base Currency Share Classes due to currency hedging and currency hedging costs.
- (j) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (k) The Fund is exposed to the investment risk arising from the Reference Index underlying the ISDA Master Agreement. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (I) Some of the markets in which the Fund may effect derivative transactions are "over-the-counter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.
- (m) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (n) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could

change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.

(o) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. Accordingly, the Investment Manager will evaluate and integrate sustainability risks where relevant throughout the investment process. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS S&P 500 SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

THIS FINANCIAL PRODUCT IS NOT SPONSORED, ENDORSED, SOLD OR PROMOTED BY MSCI LIMITED ("MSCI"), ANY AFFILIATE OF MSCI OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX. THE MSCI INDEXES ARE THE EXCLUSIVE PROPERTY OF MSCI. MSCI AND THE MSCI INDEX NAMES ARE SERVICE MARK(S) OF MSCI OR ITS AFFILIATES AND HAVE BEEN LICENSED FOR USE FOR CERTAIN PURPOSES BY LICENSEE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, TO THE OWNERS OF THIS FINANCIAL PRODUCT OR ANY MEMBER OF THE PUBLIC REGARDING THE ADVISABILITY OF INVESTING IN FINANCIAL SECURITIES GENERALLY OR IN THIS FINANCIAL PRODUCT PARTICULARLY OR THE ABILITY OF ANY MSCI INDEX TO TRACK CORRESPONDING STOCK MARKET PERFORMANCE. MSCI OR ITS AFFILIATES ARE THE LICENSORS OF CERTAIN TRADEMARKS, SERVICE MARKS AND TRADE NAMES AND OF THE MSCI INDEXES WHICH ARE DETERMINED, COMPOSED AND CALCULATED BY MSCI WITHOUT REGARD TO THIS FINANCIAL PRODUCT OR THE ISSUER OR OWNER OF THIS FINANCIAL PRODUCT. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAS ANY **OBLIGATION TO TAKE THE NEEDS OF THE ISSUERS OR OWNERS OF THIS FINANCIAL** PRODUCT INTO CONSIDERATION IN DETERMINING, COMPOSING OR CALCULATING THE MSCI INDEXES. NEITHER MSCI, ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX IS RESPONSIBLE FOR OR HAS PARTICIPATED IN THE DETERMINATION OF THE TIMING OF, PRICES AT, OR QUANTITIES OF THIS FINANCIAL PRODUCT TO BE ISSUED OR IN THE DETERMINATION OR CALCULATION OF THE EQUATION BY WHICH THIS FINANCIAL PRODUCT IS REDEEMABLE FOR CASH. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, THE MAKING OR COMPILING ANY MSCI INDEX HAS ANY OBLIGATION OR LIABILITY TO THE OWNERS OF THIS FINANCIAL PRODUCT IN CONNECTION WITH THE ADMINISTRATION, MARKETING OR OFFERING OF THIS FINANCIAL PRODUCT.

ALTHOUGH MSCI SHALL OBTAIN INFORMATION FOR INCLUSION IN OR FOR USE IN THE CALCULATION OF THE MSCI INDEXES FROM SOURCES WHICH MSCI CONSIDERS RELIABLE, NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX WARRANTS OR GUARANTEES THE ORIGINALITY, ACCURACY AND/OR THE COMPLETENESS OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX MAKES ANY WARRANTY, EXPRESS OR IMPLIED, AS TO RESULTS TO BE **OBTAINED BY LICENSEE, LICENSEE'S CUSTOMERS OR COUNTERPARTIES, ISSUERS OF** THE FINANCIAL SECURITIES, OWNERS OF THE FINANCIAL SECURITIES, OR ANY OTHER PERSON OR ENTITY, FROM THE USE OF ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN IN CONNECTION WITH THE RIGHTS LICENSED HEREUNDER OR FOR ANY OTHER USE. NEITHER MSCI, ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN. OR RELATED TO. MAKING OR COMPILING ANY MSCI INDEX SHALL HAVE ANY LIABILITY FOR ANY ERRORS. OMISSIONS OR INTERRUPTIONS OF OR IN CONNECTION WITH ANY MSCI INDEX OR ANY DATA INCLUDED THEREIN. FURTHER. NEITHER MSCI. ANY OF ITS AFFILIATES NOR ANY OTHER PARTY INVOLVED IN. OR RELATED TO. MAKING OR COMPILING ANY MSCI INDEX MAKES ANY EXPRESS OR IMPLIED WARRANTIES OF ANY KIND. AND MSCI. ANY OF ITS AFFILIATES AND ANY OTHER PARTY INVOLVED IN. OR RELATED TO MAKING OR COMPILING ANY MSCI INDEX HEREBY EXPRESSLY DISCLAIM ALL WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE. WITH RESPECT TO ANY MSCI INDEX AND ANY DATA INCLUDED THEREIN. WITHOUT LIMITING ANY OF THE FOREGOING, IN NO EVENT SHALL MSCI, ANY OF ITS AFFILIATES OR ANY OTHER PARTY INVOLVED IN, OR RELATED TO, MAKING OR COMPILING ANY MSCI INDEX HAVE ANY LIABILITY FOR ANY DIRECT, INDIRECT, SPECIAL, PUNITIVE, CONSEQUENTIAL OR ANY OTHER DAMAGES (INCLUDING LOST PROFITS) EVEN IF NOTIFIED OF THE POSSIBILITY OF SUCH DAMAGES.

No purchaser, seller or holder of this security, or any other person or entity, should use or refer to any MSCI trade name, trademark or service mark to sponsor, endorse, market or promote this product without first contacting MSCI to determine whether MSCI's permission is required. Under no circumstances may any person or entity claim any affiliation with MSCI without the prior written permission of MSCI.

UBS EUR Overnight Rate SF UCITS ETF

Supplement to the Prospectus

This Supplement contains information in relation to the **UBS EUR Overnight Rate SF UCITS ETF** (the **"Fund"**), a sub-fund of UBS (Irl) Fund Solutions Plc (the **"Company"**) an umbrella type openended investment company with variable capital, governed by the laws of Ireland and authorised by the Central Bank of Ireland (the **"Central Bank"**).

This Supplement forms part of, and must be read in conjunction with, the Prospectus of the Company dated 12 May 2025, as may be amended, supplemented or modified from time to time, (the "Prospectus") and the KID, and may not be distributed unless accompanied by the Prospectus and the KID (other than to prior recipients of the Prospectus).

THIS DOCUMENT IS IMPORTANT. BEFORE YOU PURCHASE ANY OF THE SHARES REPRESENTING INTERESTS IN THE FUND DESCRIBED IN THIS SUPPLEMENT YOU SHOULD ENSURE THAT YOU FULLY UNDERSTAND THE NATURE OF SUCH AN INVESTMENT, THE RISKS INVOLVED AND YOUR OWN PERSONAL CIRCUMSTANCES. IF YOU ARE IN ANY DOUBT ABOUT THE CONTENTS OF THIS SUPPLEMENT YOU SHOULD TAKE ADVICE FROM AN APPROPRIATELY QUALIFIED ADVISOR.

Capitalised terms used in this Supplement will have the meanings given to them in the Definitions section below or in the Prospectus.

In addition to investing in transferable securities and/or units of collective investment schemes it is the intention of the Company to invest on behalf of the Fund in financial derivative instruments ("FDIs") for investment and efficient portfolio management purposes, where applicable.

Certain risks attached to investments in FDIs are set out in the Prospectus under the section headed "Risk Factors".

You should inform yourself as to (a) the possible tax consequences, (b) the legal and regulatory requirements, (c) any foreign exchange restrictions or exchange control requirements and (d) any other requisite governmental or other consents or formalities which you might encounter under the laws of the country of your citizenship, residence or domicile and which might be relevant to your purchase, holding or disposal of the Shares.

UBS (Irl) Fund Solutions Plc

An umbrella fund with segregated liability between the sub-funds

Dated 14 May 2025

IMPORTANT INFORMATION

Suitability of Investment

The Shares are not principal protected. The value of the Shares may go up or down and you may not get back the amount you have invested. See the section headed "Risk Factors" of the Prospectus and the section headed "Other Information – Risk Factors" of this Supplement for a discussion of certain risks that should be considered by you.

An investment in the Shares is only suitable for you if you (either alone or with the help of an appropriate financial or other advisor) are able to assess the merits and risks of such an investment and have sufficient resources to be able to bear any losses that may result from such an investment. The contents of this document are not intended to contain and should not be regarded as containing advice relating to legal, taxation, investment or any other matters.

Profile of a Typical Investor

The Fund is suitable for investors seeking to generate a return linked to money market rates and who are prepared to accept a low level of volatility. Investment in the Fund should be viewed as a medium to long term investment.

Responsibility

The Directors (whose names appear under the heading "**Directors of the Company**" of the Prospectus) accept responsibility for the information contained in the Prospectus and this Supplement. To the best of the knowledge and belief of the Directors (who have taken all reasonable care to ensure that such is the case) the information contained in this Supplement when read together with the Prospectus (as complemented, modified or supplemented by this Supplement) is in accordance with the facts as at the date of this Supplement and does not omit anything likely to affect the import of such information.

General

This Supplement sets out information in relation to the Shares and the Fund. You must also refer to the Prospectus which is separate to this document and describes the Company and provides general information about offers of shares in the Company. You should not take any action in respect of the Shares unless you have received a copy of the Prospectus and the KID. Should there be any inconsistency between the contents of the Prospectus and this Supplement, the contents of this Supplement will, to the extent of any such inconsistency, prevail. This Supplement and the Prospectus should both be carefully read in their entirety before any investment decision with respect to Shares is made.

As of the date of this Supplement, the Company does not have any loan capital (including term loans) outstanding or created but unissued and no outstanding mortgages, charges or other borrowings or indebtedness in the nature of borrowings, including bank overdrafts and liabilities under acceptances or acceptance credits, hire purchase or finance lease commitments, guarantees or other contingent liabilities.

Distribution of this Supplement and Selling Restrictions

Distribution of this Supplement is not authorised unless accompanied by a copy of the Prospectus, the KID and the latest annual report and audited accounts of the Company and

the Fund (other than to prior recipients of the Prospectus) and if published after such report, a copy of the then latest semi-annual report and unaudited accounts. The distribution of this Supplement and the offering or purchase of the Shares may be restricted in certain jurisdictions. If you receive a copy of this Supplement and/or the Prospectus you may not treat such document(s) as constituting an offer, invitation or solicitation to you to subscribe for any Shares unless, in the relevant jurisdiction, such an offer, invitation or solicitation could lawfully be made to you without compliance with any registration or other legal requirement other than those with which the Company has already complied. If you wish to apply for the opportunity to purchase any Shares it is your duty to inform yourself of, and to observe, all applicable laws and regulations of any relevant jurisdiction. In particular, you should inform yourself as to the legal requirements of so applying, and any applicable exchange control regulations and taxes in the countries of your respective citizenship, residence or domicile.

Definitions

Words and expressions defined in the Prospectus will, unless otherwise defined in this Supplement, have the same meaning when used in this Supplement.

TERMS OF THE SHARES REPRESENTING INTERESTS IN THE FUND

Investment Objective

The investment objective of the Fund is to deliver the net total return of the Reference Index (being Solactive €STR Daily Total Return Index).

Investment Policy

In order to achieve the investment objective of delivering the net total return of the Reference Index, the Company on behalf of the Fund intends to invest all or substantially all of the net proceeds of any issue of Shares in the manner set out below:

- swaps (each swap being an agreement between the Company on behalf of the Fund and an Approved Counterparty pursuant to a master agreement in accordance with the requirements of the International Swaps and Derivatives Association), and such swaps may include funded, unfunded or total return (as further described below under "Use of FDI Contracts Swaps") (the "Swaps"). The purpose of the Swaps shall be to gain indirect exposure to the Reference Index (as described in further detail under the heading "Use of FDI Contracts Swaps" below); and
- (ii) a portfolio of transferable securities which will include equity and equity-related securities (which may include stock and preferred stock); fixed income securities such as government and corporate bonds (such bonds may be rated or unrated and may be fixed or floating rate), money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), convertible bonds and commercial paper and/or units of other open-ended collective investment schemes including other Funds of the Company. Any of the foregoing may be issued or arranged by a member of the UBS AG Group. Where the Fund invests in such transferable securities and/or units of collective investment schemes, the Fund will enter into unfunded or total return swaps to exchange, the price performance and income received in relation to such securities and/or units of collective investment schemes for the performance of the Reference Index. In this instance, the Fund will have neither market nor credit exposure to the particular transferable securities and/or units of collective investment schemes referred to above, except in the event of a default on the part of the Approved Counterparty to the Swaps.

The Swaps, transferable securities and/or units of collective investment schemes, as listed at (i) and (ii) above, and any ancillary cash (either for investment or efficient portfolio management purposes) shall constitute the "**Fund Assets**" for the purposes of the Prospectus. All instruments shall be listed and/or traded on the exchanges and markets set out in Appendix II of the Prospectus.

The Fund is passively managed.

The Fund may, with due regard to the best interest of its Shareholders, decide from time to time to switch partially or totally from one of the above described policies to the other.

The investments underlying this financial product do not take into account the EU criteria for environmentally sustainable economic activities (Taxonomy Regulation Art. 7).

This Fund complies with Article 6 of SFDR.

The Fund considers principal adverse impacts on sustainability factors due to its investment

strategy and the nature of underlying investments.

Principal adverse impacts (the "**PAI**") are the most significant negative impacts of investment decisions on sustainability factors relating to environmental, social and employee matters, respect for human rights, anti-corruption, and anti-bribery matters. The Investment Manager integrates PAI indicators in its Index selection process.

At present, the following PAI indicator is considered as part of the basket of securities selection process:

1.14 "Exposure to controversial weapons (anti-personnel mines, cluster munitions, chemical weapons and biological weapons)":

By selecting the basket of securities, the Investment Manager ensures that the Fund will not directly hold companies involved in cluster munitions, anti-personnel mines or chemical and biological weapons, nor will it hold companies in breach of the Treaty on the Non-Proliferation of Nuclear Weapons. A company is considered to be involved in controversial weapons if the company is involved in development, production, storage, maintenance or transport of controversial weapons, or is a majority shareholder (>50% ownership stake) of such a company.

The annual report of the Company will include information on the principal adverse impacts on sustainability factors.

Further information relevant to the Fund's investment policy is contained in the main part of the Prospectus under "Investment Objectives and Policies" and under "Investment Restrictions".

Index Tracking Strategy

The Fund gains exposure to the Reference Index by investing in one or more Swaps. It is not intended that the Fund will invest directly in the components of the Reference Index in order to meet the Fund's investment objective. The Fund (and the investors in the Fund) will be exposed to the full performance of the Reference Index.

The hedged Share Classes intend to reduce currency risk by investing in Swaps with exposure to a currency hedged version of the Reference Index (each a "**Currency Index**" and together the "**Currency Indices**"). The aim of the currency hedge is to limit (though not eliminate) the FX exposure of the Class of Shares versus the currency of the Reference Index. The purpose of gaining exposure to a Currency Index is to limit, at a Share Class level, the profit or loss generated from foreign exchange exposure. This is achieved by the Currency Indices methodology referencing one month rolling forward FX contracts, as opposed to the Fund itself.

The Fund may have counterparty risk as a result of the investment in the swaps – further information is provided in the section below headed "Use of FDI Contracts – Swaps" and in section 6 "Risk Factors" of the Prospectus.

The anticipated level of tracking error after fees and transaction costs is up to 0.02 % per annum in normal market conditions.

The annual report of the Fund, which can be found on the Website, will include the following information:

- a) The size of the realised tracking error for the year, being the volatility of the difference between the return of the Fund and the return of the Reference Index;
- b) An explanation for the divergence between the anticipated and realised tracking error for the year;

c) The size of the annual tracking difference for the year, being the difference between the performance of the Fund and the Reference Index; and

d) An explanation for the annual tracking difference, including the impact of transaction costs, fees and expenses on the performance of the Fund.

Portfolio Transparency Policy

The Fund's transparency policy regarding the composition of its portfolio, and any collateral received in relation to the Swaps, is to provide position level disclosure on a daily basis. The information is published on the Website.

The assets in which the Fund will invest in order to achieve its investment objective are set out in the "Investment Policy" section above. Details on the composition of the Fund Assets will be set out on the Website.

Use of FDI Contracts – Swaps

The Fund may enter into Swaps with a member of the UBS AG Group or any other eligible entity (being the **"Approved Counterparty"**) pursuant to which the Fund will be entitled to receive from the Approved Counterparty the performance of the components of the Reference Index.

Swap agreements are two-party contracts whereby two parties agree to exchange the returns (or differentials in rates of return) earned or realised on particular agreed investments or instruments.

The Fund may enter into the types of Swaps detailed below. Such Swaps shall be in the form of index swaps (whereby one party exchanges the return on an index of any kind (in this instance it is the Reference Index) in return for an agreed amount from the other party) or equity swaps (whereby one party exchanges the return of an equity index and/or basket of equity securities while the other exchanges an amount based on a fixed-income cash flow (for example, Euribor):

- 1. Fully funded Swaps in the form of index swaps. In such circumstances, the Fund will pay upfront all or substantially all (some cash may be retained to pay expenses and fees) of the net proceeds of any issue of Shares to the counterparty in return for the performance of the components of the Reference Index.
- 2. Unfunded Swaps in the form of index swaps and/or equity swaps. In such circumstances, the Fund will invest all or substantially all of the net proceeds of any issue of Shares in a portfolio of transferable securities and/or units of collective investment schemes as listed at (ii) above. Through an index swap, the Fund will make a payment to the counterparty which is linked to the return on some or all of these transferable securities and units of collective investment schemes held by the Fund in return for receiving the performance of the components of the Reference Index. Typically, unfunded swaps do not require any upfront payment of cash.

In addition, the Fund may use unfunded total return swaps in order to achieve the Investment Objective whereby the Fund makes or receives periodical payments to or from the counterparty based on a set rate agreed between the parties in return for receiving or paying the performance of an underlying asset. An example of such total return swaps would include the Fund entering into (a) an equity total return swap whereby the Fund swaps out the performance of the transferable securities and/or units in collective investment schemes held by the Fund as listed at (ii) above in return for a fixed or floating interest rate agreed between the parties and (b), in turn via an index total return swap the Fund swaps out the fixed or floating interest rate received by it in return for receiving the performance of the Reference Index.

The Fund will limit its exposure to Swaps that are fully funded Swaps to 10% of its net assets.

As detailed above, fully funded Swaps are swap agreements pursuant to which a Fund transfers a cash amount in full consideration of the swap value to the counterparty. In return the Fund will be entitled to receive the performance of the relevant investment strategy under the terms of the swap agreement. In respect of each of the fully funded Swaps, the unfunded Swaps and the total return Swaps, the counterparty will transfer collateral to the Fund in accordance with the UCITS rules to mitigate credit risk to the counterparty arising from entering into the swap agreement. Fully funded Swaps are used to enhance the liquidity of the Fund.

The Approved Counterparty to the Swaps and the Company on behalf of the Fund have entered into a 2002 International Swaps and Derivatives Association Master Agreement (including any supporting agreements, annexes or schedules thereto) (the "**ISDA Master Agreement**"), and will enter into a confirmation, which will supplement, form part of and be subject to, the ISDA Master Agreement, for each Swap transaction. Such confirmations may be entered into before or after the relevant transaction and may be in electronic form.

The Swaps will at all times be valued in accordance with the provisions of the Prospectus. The valuation of the Swaps will reflect the relative movements in the performance of the Reference Index and the Fund's holding of transferable securities and/or other Fund Assets as may be referenced under the Swaps. Depending on the value of the Swaps, the Fund will have to make a payment to the Approved Counterparty or will receive such a payment. Where the Fund has to make a payment to the Approved Counterparty, this payment will be made from the proceeds and, as the case may be, the disposal of some or all of the transferable securities and/or other Fund Assets in which the Fund has invested.

As set out in Section 5 of the Prospectus, the Company will ensure that the counterparty risk exposure under the Swaps never exceeds the limits required by the Regulations and the Central Bank. Accordingly, the Company will reduce such counterparty exposure by causing the Approved Counterparty, where necessary, to provide appropriate Collateral to the Company on behalf of the relevant Fund (or as otherwise permitted by the Central Bank) under the terms of ISDA Master Agreement, in accordance with the Investment Restrictions. Alternatively, the Company may reduce its risk exposure to the Approved Counterparty by causing the Approved Counterparty to reset the Swaps or vice versa. This may result in a corresponding payment from the Fund to the Approved Counterparty.

The Swaps may be terminated by either party at any time or on the occurrence of certain events with respect to either the Fund or the Approved Counterparty including, but not limited to, an event of default (such as a failure to pay, breach of agreement or bankruptcy) or a termination event (which is not due to the fault of either party, for example, illegality or a change in the tax or accounting laws), in which case the non-defaulting party or the unaffected party respectively is entitled to calculate the close-out value of the Swaps and it will do so in line with the industry standard requirements set out in the ISDA Master Agreement. The Fund may then enter into new Swaps (unless the Directors resolve that it is inadvisable to do so) or, if the Directors determine that there is no reasonable way to achieve the investment objective, the Fund may be terminated in accordance with the provisions of the Prospectus.

Securities Lending and Repurchase Transactions

The Company may (for the purposes of efficient portfolio management only) enter into stock lending arrangements and/or Repurchase Transactions with financial institutions (including with a member of the UBS AG Group as the stock borrower or counterparty to the Repurchase Transaction) with a minimum credit rating at the time the relevant transaction as permitted by the Regulations, or which are deemed by the Fund to have an implied credit rating at least equal to such level (a

"Counterparty"). Any such stock lending arrangement and Repurchase Transaction will be subject to the conditions, limits and requirements of the Central Bank and the provisions of the Prospectus. In these transactions, Collateral will move between the Company and the relevant Counterparty in order to mitigate any counterparty risk.

There is no limit on the proportion of assets that may be subject to such Securities Financing Transactions and Total Return Swaps and therefore the maximum proportion of the Fund's assets that can be subject to Securities Financing Transactions or Total Return Swaps can be as much as 100%, i.e. all of the assets of the Fund. The expected proportion of the Fund's total assets that can be subject to each type of Securities Financing Transaction should be in the range of 0% and 95%. In any case the most recent semi-annual and annual accounts of the Fund will express the amount of the Fund's assets subject to Securities Financing Transactions and Total Return Swaps.

Investors should also note the information contained in the "Securities Lending and Repurchase Transactions" and "Collateral Policy" sections in the main body of the Prospectus.

Risk Management

The Company on behalf of the Fund has filed with the Central Bank its risk management policy which enables it to accurately measure, monitor and manage the various risks associated with the use of FDIs. The Company will, on request, provide supplementary information to Shareholders relating to the risk management methods employed, including the quantitative limits that are applied and any recent developments in the risk and yield characteristics of the main categories of investments.

The Company will use the commitment approach for the purposes of calculating global exposure for the Fund. The Fund's total exposure to the Reference Index will be limited to 100% of Net Asset Value.

Efficient Portfolio Management

The Company may also, for the purposes of efficient portfolio and cash management and not for investment purposes, invest in cash deposits, money market instruments (which shall include U.S. treasury bills, cash deposits, and short-term money market deposits), put and/or call options and units of UCITS regulated money market collective investment schemes.

Further information on efficient portfolio management is contained in the main body of the prospectus under the heading "Use of Financial Derivative Instruments and Efficient Portfolio Management".

Investment Restrictions

Investors in particular must note that the general investment restrictions set out under "**Investment Restrictions**" in the Prospectus apply to the Fund. In addition, the following investment restrictions shall apply to the Fund:

- (i) The Fund shall not invest in a fund of the Company which itself holds shares in any other fund of the Company; and
- (ii) For the avoidance of doubt, in accordance with Regulation 68(2)(a) of the UCITS Regulations, the Fund is permitted to invest no more than 10% of its net assets in securities which are not "permitted securities", being those securities which are detailed in section 5.1 of the Prospectus.

German Investment Tax Act Disclosure

It is the intention of the Company to seek German tax reporting status for all Share Classes of the Fund. On this basis, in addition to the investment restrictions set out in the Prospectus, the Fund will also adhere to the restriction that at least 80% of the Net Asset Value of the Fund shall be invested in equity participations (the "Equity Participation Ratio"). More details on the Equity Participation Ratio can be found in the Prospectus.

As the Fund will continuously invest more than 50% of their respective assets in equity investments (as defined in Section 2(8) of the German Investment Tax Act and related guidelines), it will qualify as an "Equity fund" within the meaning of Section 2(6) of the German Investment Tax Act for the partial exemption pursuant to Section 20(1) of the German Investment Tax Act.

Investors should refer to their tax advisors in relation to the implications of German tax reporting status being obtained.

Disruption Events

The following Disruption Events shall apply to the Fund in addition to those set out in the Prospectus:

- (i) the Index Provider makes an adjustment or modification to the calculation of the Reference Index;
- (ii) the Approved Counterparty is:
 - (a) prevented, hindered or impaired from conducting one or more of its activity/activities in a commercially reasonable manner in connection with the establishment, maintenance, adjustment or termination of any transaction(s) in order to fully hedge its exposure in respect of the Reference Index; and/or
 - (b) prevented, hindered or impaired from performing one or more of its obligation(s) in a commercially reasonable manner,

under the terms of the relevant Swaps, with the consequence that the Approved Counterparty or the Calculation Agent (as applicable) elects to:

- I. adjust the terms of the Swaps; and/or
- II. delay payment to the Fund under the terms of the Swaps; and/or
- III. terminate the Swaps;
- (iii) the Index Provider fails or delays in calculating, announcing and/or publishing the Reference Index level.

Consequences of Disruption Events

As further described in the Prospectus and as set out above, upon the occurrence of a Disruption Event including an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers), (i) the Approved Counterparty (where acting as the relevant Calculation Agent or otherwise) may either (a) terminate one or more of the relevant Swaps or (b) make adjustments to the terms or valuation of the relevant Swaps and other derivative contracts to account for such event and the Net Asset Value may be affected by such adjustment; (ii) the Directors may temporarily suspend the calculation of the Net Asset Value and any subscription, redemption and exchange of Shares; and/or (iii) the Directors may, in certain circumstances, terminate the Fund.

Borrowings

In accordance with the general provisions set out in the Prospectus under the heading **"Borrowing and Lending Powers"**, the Company on behalf of the Fund may borrow up to 10% of the Net Asset Value of the Fund on a temporary basis. Such borrowings may only be used for short term liquidity purposes to cover the redemption of Shares.

Dividend Policy

The Fund does not intend to issue dividends in respect of the USD acc, EUR acc, CHF acc, GBP acc, HKD acc, SGD acc, JPY acc, SEK acc, hUSD acc, hCHF acc, hGBP acc, hHKD acc, hSGD acc, hJPY acc, hSEK acc, hPLN acc and hNOK acc Share Classes. However, the Directors may, in their absolute discretion, determine to declare dividends in respect of such Share Classes and Shareholders will be notified accordingly. Any dividends payable by the Fund will be effected in accordance with the Dividend Policy stated in the Prospectus.

The Directors may declare dividends attributable to the EUR UKdis, hUSD UKdis, hCHF UKdis, hGBP UKdis, hHKD UKdis, hJPY UKdis, hSEK UKdis, hPLN UKdis, hNOK UKdis, hSGD UKdis, and EUR dis Share Classes in respect of any six or twelve month period ending on 31 December and/or 30 June. Such dividends may be declared out of:

- (i) net income (being the accumulated revenue consisting of all revenue accrued including interest and dividends) less expenses; or
- (ii) realised and unrealised capital gains on the disposal/valuation of investments less realised and unrealised capital losses of the Fund.

Any such dividends will be paid within four calendar months after declaration. Shareholders will be notified in advance of the payment of any such dividend and the period for which it will be determined.

The Directors reserve the right to increase or decrease the frequency of dividend payments, if any, at their discretion for income distribution Shares. In the event of a change of policy full details will be disclosed in an updated Supplement and Shareholders will be notified in advance.

Any dividend unclaimed after six years from the date when it first became payable shall be forfeited automatically, without the necessity for any declaration or other action by the Company.

Further information is set out in the "Dividend Policy" section of the Prospectus.

The dividend payments may be as low as zero. There is no guarantee that any dividend will be declared. For the avoidance of doubt, the payment of a dividend (if any) will reduce the Net Asset Value of the relevant Share Class accordingly.

Limited Recourse

A Shareholder will solely be entitled to look to the Fund Assets in respect of all payments in respect of its Shares. If the realised net assets of the Fund are insufficient to pay any amounts payable in respect of the Shares, the Shareholder will have no further right of payment in respect of such Shares nor any claim against or recourse to any of the assets of any other Fund or any other asset of the Company.

Listing

Application will be made to such exchanges as the Directors may determine from time to time (the **"Relevant Stock Exchanges"**) for the listing of the Shares issued and available to be issued and to be admitted to listing on the official list and trading on each of the Relevant Stock Exchanges. This Supplement and the Prospectus together comprise listing particulars for the purposes of listing

the Shares on the official list and trading on the main market of each of the Relevant Stock Exchanges.

Exchange Traded Fund

The Fund is an Exchange Traded Fund ("**ETF**"). The Shares of this Fund (where they are listed on the Relevant Stock Exchanges) are fully transferable among investors. It is envisaged that Shares will be bought and sold by public and institutional investors in the secondary market in the same way as the ordinary shares of a listed trading company.

The Directors of the Company may create new Share Classes from time to time, provided that the creation of any such new Share Classes is notified in advance to and cleared by the Central Bank. A separate pool of assets will not be maintained for each of the Share Classes.

GENERAL INFORMATION RELATING TO THE FUND

Туре	Open-ended
Base Currency	Euro (" Euro ")
Business Day	 A day (other than a Saturday or Sunday) on which: (A) banks are open for business in Dublin (including all Irish public and/or bank holidays except – New Year's Day, Good Friday, Easter Monday, Christmas Day and St Stephen's Day) and (B) the Reference Index is published by the Index Provider; or i) such other day or days as the Directors may from time to time determine and notify in advance to Shareholders. For the avoidance of doubt, there shall be at least two Dealing Days per month at regular intervals.
Dealing Day	 Any Business Day. However, some Business Days will not be Dealing Days where, for example, markets on which the Fund Assets are listed or traded or markets relevant to the Reference Index are closed provided there is at least one Dealing Day per fortnight, subject always to the Directors' discretion to temporarily suspend the determination of the Net Asset Value and the sale, conversion and/or redemption of Shares in the Fund in accordance with the provisions of the Prospectus and the Articles. The Investment Manager produces dealing calendars which detail in advance the Dealing Days for each Fund. The dealing calendar may be amended from time to time by the Investment Manager where, for example, the relevant market operator, regulator or exchange (as applicable) declares a relevant market closed for trading and/or settlement (such closure may be made with little or no notice to the Investment Manager). The dealing calendar for the Fund is available from the Manager.
Dealing Deadline	2.00 p.m. (Dublin time) on the relevant Dealing Day.
Minimum Fund Size	EUR 100 million
Valuation Point	10:30 pm (Dublin time) on the relevant Dealing Day.
Settlement Date	Up to three Business Days after the relevant Dealing Day.
Minimum Holding	1 Share

Minimum Subscription Amount	100,000 EUR or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Subscription Amount/basket.
Minimum Redemption Amount	100,000 EUR or its equivalent in another currency or such amount as may be determined by the Directors. Investors will be notified of any change to the Minimum Redemption Amount/basket.
Website	www.ubs.com/ETF - information on Fund Assets and the Fund's iNAV is set out on the Website.

* Non-Base Currency Share Classes performance may differ from that of the Base Currency Share Classes due to currency hedging costs.

Description of the Shares Share Classes

Share	**EUR	**USD	**CHF	**GBP	**HKD	**SGD	**JPY	**SEK
Class*	acc	acc	acc	acc	acc	acc	acc	acc
Initial Issue	EUR 5	USD 5	CHF 5	GBP 5	HKD 50	SGD 5	JPY 500	SEK 50

Share	**hUSD	**hCHF	**hGBP	**hHKD	**hSGD	**hJPY	**hSEK acc	**hPLN
Class*	acc	acc	acc	acc	acc	acc		acc
Initial Issue	USD 5	CHF 5	GBP 5	HKD 50	SGD 5	JPY 500	SEK 50	PLN 50

Share	**hNOK	***hEUR	***hUSD	***hCHF	***hGBP	***hHKD	***hJPY
Class*	acc	UKdis	UKdis	UKdis	UKdis	UKdis	UKdis
Initial Issue	NOK 50	EUR 5	USD 5	CHF 5	GBP 5	HKD 50	JPY 500

Share	***hSEK	***hPLN	***hNOK	***hSGD	***EUR dis
Class*	UKdis	UKdis	UKdis	UKdis	
Initial Issue	SEK 50	PLN 50	NOK 50	SGD 5	EUR 5

* Non-Base Currency share classes will have a currency hedged exposure to the Reference Index.

** Together the "**acc**" Share Class.

*** Together the "dis" Share Class.

Each Share Class not denominated in the Base Currency will have a currency hedged exposure

through the relevant Currency Index. The performance of such Share Classes may differ from that of the Base Currency Share Classes due to currency hedging and currency hedging costs.

Information regarding the Initial Issue Price is available from the Administrator. Details on the Reference Index can be found at <u>https://www.solactive.com/.</u>

Initial Offer Period

The initial offer period of the EUR acc Share Class will remain open during the period from 9 am on 15 May 2025 (Irish time) and will close at 5 pm on 13 November 2025 (Irish time). The initial offer period for all remaining Share Classes is closed as of the date of this Supplement.

Fees and Expenses

The following fees will be incurred on each Share by Shareholders (and will not be incurred by the Company on behalf the Fund and accordingly will not affect the Net Asset Value of the relevant Share Class of the Fund).

Share Class	"acc"	"dis"
Exchange Charge	Up to 3%	Up to 3%
Primary Market Transaction Costs***	Up to 1%	Up to 1%

***The Manager may waive all or part of the Primary Market Transaction Costs in the event that the Shares subscribed or redeemed by a Shareholder correspond at least to the Minimum Redemption Amount/basket or a multiple thereof or for any other reason at the Manager's sole discretion.

Separate to and distinct from the Exchange Charge and Primary Market Transaction Costs, which shall be deducted from the investment amount received from an investor or deducted from the redemption proceeds payable to the investor or used to purchase the Shares of the New Class (as appropriate), the following fees and expenses will be incurred by the Company on behalf of the Fund and will affect the Net Asset Value of the relevant Share Class of the Fund:

Share Class	"acc"	"dis"
Flat-Fees	0.10% per annum	0.10% per annum

The Promoter shall discharge the fees associated with the establishment and approval of the Fund.

Swap charges

In determining the price payable by the Fund for the Swaps, the Approved Counterparty will take into account the following factors:

- a) Any transaction fees and costs and charges that arise relating to trading the Swaps which may include any brokerage costs, index licence fees, relevant collateral costs and other costs of hedging the Swap agreements. Such costs and charges will be levied at normal commercial rates.
- b) Any costs associated with foreign currency hedging.
- c) All external fees and costs relating to the Fund or the Company, as further described in the Flat Fee section of the Prospectus.

This section headed "Fees and Expenses" should be read in conjunction with the sections headed "Fees and Expenses" and "Issue and Repurchase Price/Calculation of Net Asset Value/Valuation of Assets" in the Prospectus.

GENERAL DESCRIPTION OF THE REFERENCE INDEX

The Reference Index reflects the return of a daily compounded Euro short term rate (€STR).

€STR is an overnight reference rate calculated by the European Central Bank and reflects the wholesale euro unsecured overnight borrowing costs of euro area banks. The rate is based on transactions conducted and settled on the previous day which are deemed to be executed at arm's length and thereby reflect market rates in an unbiased way. Further details on the calculation methodology of €STR can be found on:

https://www.ecb.europa.eu/stats/financial_markets_and_interest_rates/euro_short-term_rate/html/index.en.html

The compounding is done with an actual over 360 day count convention.

The Reference Index is not rebalanced.

The Reference Index is calculated in EUR (Euro).

Index Provider and website

The Reference Index is governed by Solactive AG. More details on the Reference Index, including its underlying components can be found at https://www.solactive.com/

OTHER INFORMATION

Risk Factors

Certain risks relating to the Shares are set out under the heading "**Risk Factors**" in the Prospectus. In addition, Shareholders must also note that:

- (a) The sub-funds of the Company are segregated as a matter of Irish law and as such, in Ireland, the assets of one sub-fund will not be available to satisfy the liabilities of another sub-fund. However, it should be noted that the Company is a single legal entity which may operate or have assets held on its behalf or be subject to claims in other jurisdictions which may not necessarily recognise such segregation. There can be no guarantee that the courts of any jurisdiction outside Ireland will respect the limitations on liability as set out above.
- (b) Upon the occurrence of a Disruption Event or an Index Disruption and Adjustment Event (and without limitation to the Directors' personal powers as further described in the Prospectus); (i) adjustments may be made to account for any such event (including adjustment to the Reference Index, the calculation of the Reference Index level and/or changes to the terms of the derivative contracts) which may have a significant impact on the Net Asset Value of the Fund; (ii) the Directors may temporarily suspend the calculation

of the Net Asset Value and any subscription, redemption and exchange of Shares and payment of redemption proceeds in accordance with the provisions of the Prospectus under the section "Suspension of Calculation of Net Asset Value"; and/or (iii) the Directors may, in certain circumstances as set out in the Prospectus, terminate the Fund.

- (c) The return payable under the Swaps with a counterparty is subject to the credit risk of the counterparty. In addition, the counterparty will generally act as the calculation agent under the Swaps (the "Calculation Agent") and perform those duties agreed in the ISDA Master Agreement and confirmation for the relevant Swaps. Shareholders should note that not only will they be exposed to the credit risk of the counterparty but also potential conflicts of interest in the performance of the function of Calculation Agent by the counterparty. The counterparty will use its reasonable endeavours to resolve any such conflicts of interest fairly (having regard to its respective obligations and duties) and to ensure that the interests of the Company and the Shareholders are not unfairly prejudiced. The Directors believe that the counterparty will be suitable and competent to act as Calculation Agent. Any valuations provided by the counterparty in its role as Calculation Agent will be verified at least weekly by a party independent of the counterparty as sourced by the Administrator and approved by the Depositary.
- (d) The value of investments and the income from them, and therefore the value of and income from the Shares can go down as well as up and an investor may not get back the amount invested. The Fund's exposure is linked to the performance of the components of the Reference Index which, in turn, is exposed to general market movements (negative as well as positive).
- (e) There can be no assurance that the Reference Index will be successful at producing positive returns consistently or at all. The Index Provider makes no representation or warranty, express or implied, that the Reference Index will produce positive returns at any time.
- (f) Additional Classes of Shares may be created at any time without the consent of the then existing Shareholders in accordance with the Central Bank's requirements. Each Class of Shares issued in respect of the Fund will perform differently as a result of differences of currency and fees (as applicable). The Company on behalf of the Fund will enter into Swaps that are designed to generate the cash flows payable in respect of the Shares of the relevant Classes. There is no legal segregation of assets and liabilities between Classes and there is no separate portfolio of assets held for each Class.
- (g) The Company on behalf of the Fund will enter into separate derivative contract(s) in respect of each Class. Each derivative contract will be on similar terms except that each derivative contract will provide a return in the currency of denomination of each Class of Shares to which it relates.
- (h) The Fund's use of the Reference Index is licensed by the Index Provider. If at any time, the licence is terminated or the Reference Index otherwise becomes unavailable, unreliable, inaccurate or unrepresentative, the Fund Directors may exercise their discretion to terminate the Fund in accordance with the terms of the Articles.
- (i) The Fund is exposed to the investment risk arising from the Reference Index underlying the Swaps. The Fund is therefore exposed to the risks inherent to investments in equity markets.
- (j) Some of the markets in which the Fund may effect derivative transactions are "over-thecounter" or "interdealer" markets, which may be illiquid and are sometimes subject to larger spreads than exchange-traded derivative transactions. The participants in such markets are typically not subject to credit evaluation and regulatory oversight, which would be the

case with members of "exchange-based" markets. This exposes the Fund to the risk that a counterparty will not settle a transaction in accordance with its terms and conditions because of a credit or liquidity problem with the counterparty. Delays in settlement may also result from disputes over the terms of the contract (whether or not bona fide) since such markets may lack the established rules and procedures for swift settlement of disputes among market participants found in "exchange-based" markets. These factors may cause the Fund to suffer a loss due to adverse market movements while replacement transactions are executed or otherwise.

- (k) Whilst the Company has the right to use and reference the Reference Index in connection with the Fund in accordance with the terms of the Reference Index license, in the event that the licence is terminated the Fund may be terminated or if any other Index Disruption and Adjustment Event occurs, adjustments may be made to the terms of the Swaps after negotiation with each Approved Counterparty to account for any such event including adjustment to the Reference Index or the calculation of the Reference Index level which may have a significant impact on the Net Asset Value of the Fund.
- (I) Subject to certain pre-defined parameters, it is possible that the methodology used to calculate the Reference Index or the formulae underlying the Reference Index could change and such change may result in a decrease in the performance of the Reference Index. As such, aspects of the Reference Index could change in the future, including without limitation, the methodology and third party data sources. Any changes may be made without regard to the interests of a holder of any product linked to the Reference Index. Additionally, the Reference Index was created by the Index Provider, who has the right to permanently cancel the Reference Index at any time. Such cancellation may have a material adverse effect on any linked investments or transactions.
- (m) Sustainability risk within the meaning of Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector are environmental, social and governance events or conditions whose occurrence could have an actual or potential principal adverse impact on the value of the Fund's investment. Consistent with the requirements applicable to the Manager under the UCITS Regulations, the Manager is required to take sustainability risks into account in the ongoing monitoring of investments. However, sustainability risks can affect all known types of risk (for example, market risk, liquidity risk, counterparty risk and operational risk), and as a factor, contribute to the materiality of these risk types. This Fund is passively managed, tracking the Reference Index. Sustainability risks are not considered as part of the index selection process and the Investment Manager aims to track the index in line with the limits set out in the investment policy of the Fund. Therefore, sustainability risks are not the primary consideration for investment decisions and the Investment Manager does not expect that the assessment of likely impacts of sustainability risks will materially impact the expected risk or return characteristics of the Fund. Given the investment objective and policy of the Fund, the Manager does not consider the adverse impact of investment decisions on sustainability factors as it believes focusing on the selection of investment opportunities for the Fund to be a more efficient use of its resources.
- (n) The Fund relies on a floating rate, the Euro short-term rate (€STR). Alterations or reforms in the determination or oversight of the €STR might negatively affect the market, value of any securities, or payments connected to the €STR. Moreover, any alternative to the €STR and any pricing adjustments mandated by a regulator, counterparties, or other entities may negatively impact the Fund's performance and/or Net Asset Value. In addition, changes in interest rates may positively or negatively affect the Fund's performance, including a potential negative return net of fees and expenses.

Investors should also refer to the Prospectus for additional disclosure of risks.

Miscellaneous

The Company has the following Funds established as at the date of this Supplement.

UBS S&P 500 SF UCITS ETF UBS BBG COMMODITY INDEX SF UCITS ETF UBS CMCI COMPOSITE SF UCITS ETF UBS MSCI USA SF UCITS ETF UBS MSCI EM SF UCITS ETF UBS MSCI AC ASIA EX JAPAN SF UCITS ETF UBS MSCI ACWI SF UCITS ETF UBS CMCI EX-AGRICULTURE SF UCITS ETF UBS BBG COMMODITY CMCI SF UCITS ETF UBS CMCI COMMODITY CARRY SF UCITS ETF UBS MSCI CHINA A SF UCITS ETF UBS EURO EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS EURO EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS US EQUITY DEFENSIVE COVERED CALL SF UCITS ETF UBS US EQUITY DEFENSIVE PUT WRITE SF UCITS ETF UBS MSCI USA SF INDEX FUND UBS CMCI COMMODITY CARRY EX-AGRICULTURE SF UCITS ETF UBS CMCI FUTURE COMMODITY SF UCITS ETF UBS EUR OVERNIGHT RATE SF UCITS ETF UBS USD OVERNIGHT RATE SF UCITS ETF UBS GBP OVERNIGHT RATE SF UCITS ETF UBS S&P 500 EQUAL WEIGHT SF UCITS ETF

Disclaimers

THE FINANCIAL INSTRUMENT IS NOT SPONSORED, PROMOTED, SOLD OR SUPPORTED IN ANY OTHER MANNER BY SOLACTIVE AG NOR DOES SOLACTIVE AG OFFER ANY EXPRESS OR IMPLICIT GUARANTEE OR ASSURANCE EITHER WITH REGARD TO THE RESULTS OF USING THE INDEX AND/OR INDEX TRADE MARK OR THE INDEX LEVEL AT ANY TIME OR IN ANY OTHER RESPECT. THE INDEX IS CALCULATED AND PUBLISHED BY SOLACTIVE AG. SOLACTIVE AG USES ITS BEST EFFORTS TO ENSURE THAT THE INDEX IS CALCULATED CORRECTLY. **IRRESPECTIVE OF ITS OBLIGATIONS TOWARDS THE ISSUER OF THE FINANCIAL INSTRUMENT,** SOLACTIVE AG HAS NO OBLIGATION TO POINT OUT ERRORS IN THE INDEX TO THIRD PARTIES INCLUDING BUT NOT LIMITED TO INVESTORS AND/OR FINANCIAL INTERMEDIARIES OF THE FINANCIAL INSTRUMENT. NEITHER PUBLICATION OF THE INDEX BY SOLACTIVE AG NOR THE LICENSING OF THE INDEX OR INDEX TRADE MARK FOR THE PURPOSE OF USE IN CONNECTION WITH THE FINANCIAL INSTRUMENT CONSTITUTES A RECOMMENDATION BY SOLACTIVE AG TO INVEST CAPITAL IN SAID FINANCIAL INSTRUMENT NOR DOES IT IN ANY WAY REPRESENT AN ASSURANCE OR OPINION OF SOLACTIVE AG WITH REGARD TO ANY INVESTMENT IN THIS FINANCIAL INSTRUMENT. SOLACTIVE AG IS REGISTERED AS A BENCHMARK ADMINISTRATOR UNDER THE REGULATION (EU) 2016/2011 (BMR). SOLACTIVE APPEARS ON THE ESMA REGISTER OF ADMINISTRATORS.

INFORMATION FOR INVESTORS IN SWITZERLAND DATED 04 JUNE 2025

This Country Supplement forms part of, and should be read in conjunction with the Prospectus for UBS (Irl) Fund Solutions plc (the "Company") dated 04 June 2025 and with its Supplements in respect of the Funds dated as stated below, and is for use in Switzerland only.

Words and expressions defined in the Prospectus shall, unless the context otherwise requires, have the same meaning when used herein.

- UBS CMCI Composite SF UCITS ETF (dated 12 May 2025)
- UBS BBG Commodity Index SF UCITS ETF (dated 12 May 2025)
- UBS S&P 500 SF UCITS ETF (dated 12 May 2025)
- UBS MSCI USA SF UCITS ETF (dated 12 May 2025)
- UBS MSCI EM SF UCITS ETF (dated 12 May 2025)
- UBS MSCI AC Asia ex Japan SF UCITS ETF (dated 12 May 2025)
- UBS MSCI ACWI SF UCITS ETF (dated 12 May 2025)
- UBS CMCI Ex-Agriculture SF UCITS ETF (dated 12 May 2025)
- UBS BBG Commodity CMCI SF UCITS ETF (dated 12 May 2025)
- UBS CMCI Commodity Carry SF UCITS ETF (dated 12 May 2025)
- UBS MSCI China A SF UCITS ETF (dated 12 May 2025)
- UBS Euro Equity Defensive Covered Call SF UCITS ETF (dated 12 May 2025)
- UBS Euro Equity Defensive Put Write SF UCITS ETF (dated 12 May 2025)
- UBS US Equity Defensive Covered Call SF UCITS ETF (dated 12 May 2025)
- UBS US Equity Defensive Put Write SF UCITS ETF (dated 12 May 2025)
- UBS CMCI Future Commodity SF UCITS ETF (dated 12 May 2025)
- UBS S&P 500 Equal Weight SF UCITS ETF (dated 12 May 2025)
- UBS MSCI USA SF Index Fund (dated 12 May 2025)
- UBS EUR Overnight Rate SF UCITS ETF (dated 14 May 2025)

Dated 04 June 2025

1. Representative

The representative in Switzerland is UBS Fund Management (Switzerland) AG, Aeschenplatz 6, 4052 Basel.

2. Paying agent

The paying agent in Switzerland is UBS Switzerland AG, Bahnhofstrasse 45, 8001 Zurich and its branches in Switzerland.

3. Location where the relevant documents may be obtained

The Prospectus, Key Information Documents, the Articles as well as the annual and semi-annual reports may be obtained free of charge from the representative as well as under www.ubs.com/etf or www.ubs.com/fonds.

4. Publications

- 1. Publications concerning the foreign collective investment scheme are made in Switzerland on the website of Swiss Fund Data AG at www.swissfunddata.ch.
- 2. Each time Shares are issued or redeemed, the issue and the redemption prices or the Net Asset Value together with a reference stating «excluding commissions» must be published for all Share Classes at www.ubs.com/fonds and on the website of Swiss Fund Data AG at www.swissfunddata.ch. Prices must be published at least twice per month (on the first and third Monday or the subsequent banking day).

5. Payment of retrocessions and rebates

- 1. The Management Company and its agents may pay retrocessions as remuneration for distribution activity in respect of fund units in or from Switzerland. This remuneration may be deemed payment for the following services in particular:
 - Promoting and rendering the distribution of fund units;
 - Training client advisers and salespersons;
 - Organization of and participation in road shows, events and shows of all kinds in connection with the distribution of fund units;
 - Contacting potential investors;
 - Central relationship management and servicing of existing client relationships;
 - Responding to specific requests from investors regarding the investment product or the provider;
 - Produce and issue marketing and legal fund documents;
 - Provide administrative services of all kinds in connection with the distribution of fund units;
 - Brokering and processing subscription and redemption of fund units;
 - Subscribe units as a nominee for clients on behalf of the provider;

- Appointment and monitoring of sub-distributors;
- Performing due diligence delegated by the Provider in areas such as ascertaining client needs and distribution restrictions;
- Instruction of the external auditors to verify compliance (by the fund provider) with the legal and self-regulatory obligations of the Distributor, such as in particular the SFAMA Guidelines for the Distribution of Collective Investment Schemes;
- Operation and maintenance of electronic distribution platforms;
- Central reporting for fund providers and distribution distributers;
- Provide administrative services of all kinds including fulfilment of the due diligence obligations in combating money laundering and the financing of terrorism in connection with the distribution of fund units.

Retrocessions are not deemed to be rebates even if they are ultimately passed on, in full or in part, to the investors.

The recipients of the retrocessions must ensure transparent disclosure and inform investors, unsolicited and free of charge, about the amount of remuneration they may receive for distribution.

On request, the recipients of retrocessions must disclose the amounts they actually receive for distributing the collective investment schemes of the investors concerned.

2. In the case of distribution activity in Switzerland, the Company, the Management Company and its agents may, upon request, pay rebates directly to investors. The purpose of rebates is to reduce the fees or costs incurred by the investor in question. The objective criteria for the granting of rebates defined by the Company or the Management Company apply.

6. Place of performance and jurisdiction

In respect of the Shares distributed in and from Switzerland, the place of performance and jurisdiction is the registered office of the representative.

7. Information for the purposes of the listing on SIX Swiss Exchange (ETFs)

7.1 Security Number / ISIN

Name of Fund	Class	Security Number	ISIN-Code
UBS CMCI Composite S	F UCITS USD acc	11926603	IE00B53H0131
	EUR acc CHF acc	11925970 11926029	IE00B58HMN42 IE00B58FQX63
	GBP acc HKD acc*) SGD acc*) JPY acc*) SEK acc*)	11926049	IE00B50XJX92
UBS BBG Commodity	USD acc	11926045	IE00B5B3W843 495

CMCI SF UCITS ETF	EUR acc CHF acc GBP acc HKD acc*) SGD acc*) JPY acc*) SEK acc*)	11925964 11926022 11926038	IE00B58ZM503 IE00B598DX38 IE00B5993T22
UBS S&P 500 SF UCITS ETF	USD acc EUR acc*) CHF acc*) GBP acc*) HKD acc*) SGD acc*) JPY acc*) SEK acc*)	12126271	IE00B4JY5R22
UBS MSCI USA SF UCITS ETF	USD acc EUR acc*) CHF acc*) GBP acc*) HKD acc*) SGD acc*) JPY acc*) SEK acc*)	12126257	IE00B3SC9K16
UBS MSCI EM SF UCITS ETF	USD acc	12894611	IE00B3Z3FS74
UBS MSCI AC Asia Ex J SF UCITS ETF	apan USD acc	18824468	E00B7WK2W23
UBS MSCI ACWI SF UC ETF	ITS USD acc USD Ukdis hUSD acc hEUR acc hCHF acc hCHF UKdis hGBP acc*) hUSD UKdis hEUR UKdis hGBP UKdis hGBP dis*) hJPY acc hSGD acc	28650087 47127930 28650238 28650240 28650241 30033438 29267293 29267355 29267361 34176211 55559336	IE00BYM11H29 IE00BJXFZ989 IE00BYM11J43 IE00BYM11K57 IE00BYM11L64 IE00BZ2GTT26 IE00BYVDRC61 IE00BYVDRC61 IE00BYVDRF92 IE00BD495N16 IE00BD495N16 IE00BMTD0S12
UBS CMCI Ex-Agricultur SF UCITS ETF	e USD acc hEUR acc hCHF acc hGBP acc	30034512 32132319 32132316 40539875	IE00BZ2GV965 IE00BYT5CV85 IE00BYT5CW92 IE00BYT5CX00

UBS BBG Commodity CMCI SF UCITS ETF	USD acc hEUR acc hCHF acc hGBP acc hHKD acc*) hSGD acc*) hJPY acc*) hSEK acc*)	36621183 36621185 36621184 40538348	IE00BYYLVH00 IE00BYYLVJ24 IE00BYYLVK39 IE00BF0V4615
UBS CMCI Commodity (SF UCITS ETF	Carry USD acc	50164024	IE00BKFB6L02
SF UCITS ETF	USD Ukdis*) hEUR acc**)	54094592	IE00BMC5DV85
	hEUR UKdis*) hCHF acc hCHF UKdis*) hGBP acc*) hGBP UKdis*) hHKD acc*) hHKD UKdis*) hSGD acc*) hSGD UKdis*) hJPY acc*) hJPY UKdis*) hSEK acc*) hSEK UKdis*) hPLN acc*) hPLN UKdis*) hNOK acc*) hNOK UKdis*)	54094558	IE00BMC5DW92
UBS MSCI China A SF UCITS ETF	USD acc USD Ukdis*) hPLN acc*) hPLN UKdis*) hNOK acc*) hNOK UKdis*) hHKD acc*) hHKD UKdis*) hSGD acc*) hSGD UKdis*) hJPY UKdis*) hSEK acc*) hSEK uKdis*) hUSD acc*) hUSD UKdis*) hEUR acc*) hEUR uKdis*) hEUR uKdis*) hEUR UKdis*) hCHF acc*) hCHF UKdis*) hCHF UKdis*)	50164051	IE00BKB6K94

	hGBP UKdis*)		
UBS Euro Equity Defens Call SF UCITS ETF	sive Covered EUR acc IE00BLDGHQ61 EUR UKdis*) hUSD acc*) hUSD UKdis*) hCHF acc*) hCHF UKdis*) hGBP acc*) hGBP UKdis*) hHKD acc*) hHKD UKdis*) hSGD acc*) hSGD UKdis*) hJPY acc*) hJPY UKdis*) hSEK acc*) hSEK UKdis*) hNCK acc*) hNOK acc*) hNOK UKdis*)	54'125'745	
UBS Euro Equity Defens Write SF UCITS ETF	sive Put EUR acc IE00BLDGHT92 EUR UKdis*) hUSD acc*) hUSD UKdis*) hCHF acc *) hCHF UKdis*) hGBP acc *) hGBP UKdis*) hHKD UKdis*) hSGD acc*) hSGD UKdis*) hJPY acc *) hJPY UKdis*) hSEK acc*) hSEK UKdis*) hPLN acc*) hPLN UKdis*) hNOK uKdis*)	54'125'731	
UBS US Equity Defensiv Call SF UCITS ETF	e Covered USD acc USD Ukdis*)	54'125'686	IE00BLDGHX39
	USD dis SOFR+4% hEUR acc*)	140'138'562	IE000LEVKOK2
	,		498

	hEUR UKdis*) hCHF acc*) hCHF UKdis*) hGBP acc*) hGBP UKdis*) hHKD acc*) hHKD UKdis*) hSGD acc*) hSGD UKdis*) hJPY acc*) hJPY UKdis*) hSEK acc*) hSEK UKdis*) hPLN acc*) hPLN UKdis*) hNOK acc*) hNOK UKdis*)		
UBS US Equity Defensiv SF UCITS ETF	e Put Write USD acc USD Ukdis*) hEUR acc*) hEUR UKdis*) hCHF acc*) hCHF UKdis*) hGBP acc*) hGBP UKdis*) hHKD UKdis*) hSGD acc*) hSGD UKdis*) hJPY acc*) hJPY UKdis*) hSEK acc*) hSEK uKdis*) hPLN acc*) hPLN UKdis*) hNOK acc*) hNOK UKdis*)	54'120'341	IE00BLDGHF56
UBS CMCI Future Comn UCITS ETF	nodity SF USD acc USD Ukdis*) hEUR acc*) hEUR UKdis*) hCHF acc*) hCHF UKdis*) hGBP acc*) hGBP UKdis*) hHKD acc*) hHKD UKdis*) hSGD acc*)	125'663'096	IE000WJCYGB4

	hJPY acc*) hJPY UKdis*) hSEK acc*) hSEK UKdis*) hPLN acc*) hPLN UKdis*) hNOK acc*) hNOK UKdis*)		
UBS S&P 500 Equal We SF UCITS ETF	ight USD acc EUR acc *) CHF acc *) GBP acc *) HKD acc *) SGD acc *) JPY acc *) SEK acc *) hEUR acc *) hCHF acc *) hCHF acc *) hGBP acc *) hKD acc *) hSGD acc *) hSGD acc *) hSGD acc *) hSGD acc *) hSGD acc *) hSEK acc *) hPLN acc *) hNOK acc *) USD Ukdis *) hEUR UKdis *) hCHF UKdis *) hGBP UKdis *) hSGD UKdis *) hSGD UKdis *) hSEK UKdis *) hSEK UKdis *) hNOK UKdis *)		IE0008GBXCA4
	hCHF dis *) hGBP dis *) hHKD dis *) hSGD dis *) hJPY dis *) hSEK dis *) hPLN dis *) hNOK dis *)		
UBS EUR Overnight Rate SF UCITS ETF	EUR acc	141'785'507	IE0006BDZN36
	USD acc *) CHF acc *) GBP acc *) HKD acc *)	141 100 001	IL UUUUBDZINGU
			500

*) not listed on SIX Swiss Exchange as of the date of this Prospectus **) share class has been launched but is offered in Switzerland to qualified investors only

Trade Currency 7.2

Name of Fund	Class	Trade Currency
UBS CMCI Composite SF UCITS ETF	USD acc EUR acc CHF acc GBP acc HKD acc SGD acc JPY acc SEK acc	USD EUR CHF GBP HKD SGD JPY SEK
UBS BBG Commodity CMCI SF UCITS ETF	USD acc EUR acc CHF acc GBP acc HKD acc SGD acc JPY acc SEK acc	USD EUR CHF GBP HKD SGD JPY SEK
UBS S&P 500 SF UCITS ETF	USD acc EUR acc CHF acc GBP acc	USD EUR CHF GBP

	HKD acc SGD acc JPY acc SEK acc	HKD SGD JPY SEK
UBS MSCI USA SF UCITS ETF	USD acc EUR acc CHF acc GBP acc HKD acc SGD acc JPY acc SEK acc	USD EUR CHF GBP HKD SGD JPY SEK
UBS MSCI EM SF UCITS ETF	USD acc	USD
UBS MSCI AC Asia Ex Japan SF UCITS ETF	USD acc	USD
UBS MSCI ACWI SF UCITS ETF	USD acc USD Ukdis hUSD acc hEUR acc hCHF acc hCHF UKdis hGBP acc hUSD UKdis) hEUR UKdis hGBP UKdis hGBP dis hJPY acc hSGD acc	USD USD EUR CHF CHF USD EUR GBP GBP JPY SGD
UBS CMCI Ex-Agriculture SF UCITS ETF	USD acc hEUR acc hCHF acc hGBP acc	USD EUR CHF GBP
UBS BBG Commodity CMCI SF UCITS ETF	USD acc hEUR acc hCHF acc hGBP acc hHKD acc hSGD acc hJPY acc hSEK acc	USD EUR CHF GBP HKD SGD JPY SEK
UBS CMCI Commodity Carry SF UCITS ETF	USD acc USD Ukdis hEUR acc hEUR UKdis	USD

	hCHF acc hCHF UKdis hGBP acc hGBP UKdis hHKD acc hHKD UKdis hSGD acc hSGD UKdis hJPY acc hJPY UKdis hSEK acc hSEK UKdis hPLN acc hPLN UKdis hNOK acc hNOK UKdis	CHF
UBS MSCI China A SF UCITS ETF	USD acc USD Ukdis hPLN acc hPLN UKdis hNOK acc hNOK UKdis hHKD acc hHKD UKdis hSGD acc hSGD UKdis hJPY acc hJPY UKdis hSEK acc hSEK UKdis hUSD acc hUSD UKdis hEUR acc hEUR UKdis hEUR acc hEUR UKdis hCHF acc hCHF UKdis hGBP acc hGBP UKdis	USD
UBS Euro Equity Defensive Covered Call SF UCITS ETF	EUR acc	EUR
	EUR UKdis USD dis SOFR+4% hUSD acc hUSD UKdis hCHF acc hCHF UKdis hGBP acc hGBP UKdis hHKD acc hHKD UKdis hSGD acc hSGD UKdis	EUR

	hJPY acc hJPY UKdis hSEK acc hSEK UKdis hPLN acc hPLN UKdis hNOK acc hNOK UKdis	
UBS Euro Equity Defensive Put Write SF UCITS ETF	EUR acc EUR UKdis hUSD acc hUSD UKdis hCHF acc hCHF UKdis hGBP acc hGBP UKdis hHKD acc hHKD UKdis hSGD acc hSGD UKdis hJPY acc hJPY UKdis hSEK acc hSEK UKdis hPLN acc hPLN UKdis hNOK acc hNOK UKdis	EUR
UBS US Equity Defensive Covered Call SF UCITS ETF	USD acc USD Ukdis hEUR acc hEUR UKdis hCHF acc hCHF UKdis hGBP acc hGBP UKdis hHKD acc hHKD UKdis hSGD acc hSGD UKdis hJPY acc hJPY UKdis hSEK acc hSEK UKdis hPLN acc hPLN UKdis hNOK acc	USD

hNOK UKdis

UBS US Equity Defensive Put Write SF UCITS ETF	USD acc USD Ukdis hEUR acc hEUR UKdis hCHF acc hCHF UKdis hGBP acc hGBP UKdis hHKD acc hHKD UKdis hSGD acc hSGD UKdis hJPY acc hJPY UKdis hSEK acc hSEK UKdis hPLN acc hPLN UKdis hNOK acc hNOK UKdis	USD	
UBS CMCI Future Commodity SF UCITS ETF	USD acc USD Ukdis hEUR acc hEUR UKdis hCHF acc hCHF UKdis hGBP acc hGBP UKdis hHKD acc hHKD UKdis hSGD acc hSGD UKdis hJPY acc hJPY UKdis hSEK acc hSEK UKdis hPLN acc hPLN UKdis hNOK acc hNOK UKdis		USD EUR EUR GBP HKD SGD JPY SEK PLN NOK
UBS S&P 500 Equal Weight SF UCITS ETF	USD acc EUR acc CHF acc GBP acc HKD acc SGD acc JPY acc SEK acc		USD

	hEUR acc hCHF acc hGBP acc hHKD acc hSGD acc hJPY acc hSEK acc hPLN acc hNOK acc USD Ukdis hEUR UKdis hGBP UKdis hGBP UKdis hJPY UKdis hSGD UKdis hSGD UKdis hSEK UKdis hNOK UKdis (USD) A-dis hEUR dis hCHF dis hGBP dis hHKD dis hSGD dis hJPY dis hSEK dis hDLN dis hNOK dis	USE
UBS EUR Overnight Rate SF UCITS ETF	EUR acc USD acc CHF acc GBP acc HKD acc SGD acc JPY acc SEK acc hUSD acc hCHF acc hGBP acc hHKD acc hSGD acc hJPY acc hSEK acc hDLN acc hDLN acc hNOK acc hEUR UKdis hUSD UKdis hCHF UKdis hGBP UKdis hGBP UKdis hJPY UKdis hJPY UKdis hSEK UKdis	EUF

SD

JR

hPLN UKdis hNOK UKdis hSGD UKdis EUR dis

7.3 Clearing Agent

SIX SIS AG

7.4 Form of Securities

Registered Shares represented by a Global Share Certificate.

7.5 Information on the Net Asset Value (NAV) performance of the past three years

Not available yet.

7.6 **Responsibility Clause**

The Company bears all responsibilities for the content of this Prospectus pursuant to Scheme A Point 4 of the Listing Rules of SIX Swiss Exchange. To the best of the knowledge of the Manager, all statements made in this Prospectus are true and do not omit any material information.

7.7 Declaration on annual and semi-annual reports

The latest annual and semi-annual reports constitute an integral part of this listing prospectus.

7.8 Market Marker

The listing of the Shares on SIX Swiss Exchange and their market on the SIX Swiss Exchange serves the purpose to provide the investors, in addition to the possibility to subscribe and redeem shares directly with the Company, a way to purchase and sell shares in a liquid and regulated secondary market, ie. on a stock exchange.

The following institutions have been appointed as market makers:

- UBS AG, Zurich

A market maker is responsible for maintaining a market for the traded shares and, in this connection, for entering bid and ask prices for fund units in the trading system.

The Swiss Financial Market Supervisory Authority requires that a Market Maker makes sure that the spread between (i) the indicative Net Asset Value of a Share and (ii) the price at which investors can buy and sell Shares is reduced to a reasonable level.

Pursuant to the agreement concluded between SIX Swiss Exchange and the market makers, the latter shall undertake to maintain a market for Shares within a certain framework and under normal conditions, and, within this context, enter bid and ask prices for fund units in the SIX trading system, which as a rule, and under normal conditions, shall not exceed a spread of 2% (1% above and below the indicative net asset value).

7.9 Declaration on primary / secondary listing

For the relevant Funds the primary listing took place on the following stock exchanges:

Information on Unlisted Funds

Shares of the following Sub-Funds are not listed on a stock exchange:

UBS MSCI USA SF Index Fund

25.040RS